Universiteit Gent



2002



The impact of colour in the store environment. An environmental psychology approach.



DOCTORAL DISSERTATION

submitted to the Faculty of Economics and Business Administration in candidacy for the degree of Doctor in Applied Economic Sciences

The Impact of Colour in the Store Environment

The Impact of Colour in the Store Environment

Preface and Acknowledgements

"Time is God's way of making sure everything doesn't happen at once..."

Anonymous

However, there are times that everything has to happen at once. At such times you lock yourself out, you cut yourself loose from the world around you, but paradoxically, at the same time you also come to appreciate the people around you the most. The last six months of this research endeavour have been such a time for me. Studying environmental psychology, I acknowledge the importance of my environment, which has been very supportive to me during the tedious process of accomplishing this dissertation. Therefore, I would like to thank everyone around me, my friends and colleagues, my neighbours, my family and especially my parents and Gunter, my partner, for being there, for their good cheer, for their understanding and moral support, for their love and care...

This is an achievement that could not have been possible without the support of a number of persons. Therefore, I would like to thank the people who contributed to the realization of this research endeavour.

In the first place, my thanks go out to Prof. Dr. Rudy Moenaert, without whom I never would have even considered writing a dissertation. I'm also very grateful to Prof. Dr. Rosette S'Jegers for her never-ending faith in me, during this, at times, cumbersome process. Nevertheless, it is Prof. Dr. Maggie Geuens, my promotor, who deserves most of the credit, because without her I would never have been able to bring this to a good ending. I would like to thank Maggie for her continuous support and encouragements. Her remarks and suggestions were always stimulating and constructive. She was so much more than an advisor to me. She was a like-minded source of inspiration and a true friend.

I am also extremely grateful to Wim Goris of BAT NV for developing, in Computer Aided Design (Arkey), the experimental store I used for this research. He did a tremendous job.

Furthermore, I wish to express my gratitude to Nathalie Heusschen, Stefanie Molenaers and the students of the 2000-2001 Market Research Practicum of Prof. Dr. Steve Muylle (i.e. Christophe Jans, Ellen Mombers, Jeroen Van der Weeën, Kim Vanrutten and Evelyne Verdeyen) for their greatly appreciated assistance in the data collection.

I also would like to thank Ann Morel, for her assistance as a marketing assistant, taking some work off my shoulders and Bert Saenen, for providing me, besides moral support, with technical assistance whenever needed.

Furthermore, I would like to thank my father, for reading parts of this work in order to correct my English and my mother for her helping hand in putting together the references section. Because these must have been painstaking jobs for both of them, their help is all the more appreciated.

A last word of thanks goes out to all the respondents who were willing to participate in this study, because I would be nowhere without them...

Finally, I am indebted to the VUB Research Counsel for funding this research project.

Malaika Brengman Brussels, August 30th, 2002

Table of Contents

Chapter 1

Introduction to the Research Problem

1.1.	Introduction	.4
1.2.	Research Justification	.5
1.2	2.1. Importance	.5
1.2	2.2. Shortcomings of Existing Research	.6
1.3.	Research Objective	.9
1.4.	Methodological and Empirical Contributions1	10
1.5.	Structure1	11

Chapter 2

Store Atmospherics

2.1. Introduction	16
2.2. Significance of the Store Environment	18
2.3. An Environmental Psychology Approach to Store Atmospherics	22
2.3.1. The Mehrabian-Russell Environmental Psychology Model	22
2.3.2. Application of the Model to Store Atmospherics	25
2.4. The Stimulus: Characterizing the Store Environment	27
2.4.1. A General Characterization of the Store Environment	27
2.4.1.1. The Information Rate of the Environment	27
2.4.1.2. The Behavioural Perspective Model	28
2.4.2. Towards a Classification of Store-environmental Characteristics	31

2.5. The C	rganism: Emotional Responses to Shopping Environments	35
2.5.1.	A Discrete Emotions Perspective	36
2.5.2.	A Dimensional Emotion Perspective	37
2.5.3.	Emotional Responses to the Shopping Experience	42
2.6. The R	esponse: Approach-Avoidance Behaviours in Shopping Environments	50
2.6.1.	Approach-Avoidance Behaviours: General Conceptualisation	50
2.6.2.	Approach-Avoidance Behaviours in a Retail Setting	53
2.7. Emoti	onal Determinants of Approach-Avoidance Responses to Shopping Environments	67
2.7.1.	The Effects of Pleasure on Approach or Avoidance Behaviours	68
2.7.2.	The Effects of Arousal on Approach or Avoidance Behaviours	70
2.7.3.	The Effects of Dominance on Approach or Avoidance Behaviours	74
2.8. Atmos	spheric Effects on Shopping Behaviour: A Review of Empirical Evidence	77

Colour Effects

3.1. Introduction			
3.2. Colou	r Perception	87	
3.2.1.	The Trichromatic Theory of Colour Vision		
3.2.2.	Opponent Colour Processing	90	
3.2.3.	The Two-Stage Model of Colour Vision		
3.2.4.	Colour Blindness	94	
3.3. The T	hree Dimensions of Colour	97	
3.3.1.	Hue		
3.3.2.	Saturation	99	
3.3.3.	Lightness	101	
3.3.4.	The Munsell Colour System		
3.3.5.	Conclusion	112	
3.4. Cogni	tive Responses to Colour		

3.5. Affect	3.5. Affective Responses to Colour		
3.5.1.	Introc	luction	116
3.5.2.	Meth	odological Caveats	117
3.5	5.2.1.	Colour stimuli specifications	117
3.5	5.2.2.	Colour responses and their assessment	119
3.5.3.	Colou	ır-evoked Pleasure	124
3.5	5.3.1.	Effects of Hue	
3.5	5.3.2.	Effects of Saturation	140
3.5	5.3.3.	Effects of Value	145
3.5	5.3.4.	Overview	153
3.5.4.	Colou	ır-evoked Arousal	156
3.5	5.4.1.	Effects of Hue	
3.5	5.4.2.	Effects of Saturation	176
3.5	5.4.3.	Effects of Value	
3.5	5.4.4.	Overview	
3.5.5.	Colou	ır-evoked Dominance	
3.5	5.5.1.	Effects of Hue	
3.5	5.5.2.	Effects of Saturation	196
3.5	5.5.3.	Effects of Value	
3.5	5.5.4.	Overview	
3.6. Behav	vioural	Responses to Colour	211
3.6.1.	The A	Application of "Colour Theory" to Real World Situations	211
3.6.2.	Empi	rical Evidence of Colour-Behaviour Effects	214
3.6	5.2.1.	Task Performance	
3.6	5.2.2.	Time Perceptions and Time Spending	219
3.6	5.2.3.	Miscellaneous Behaviours	

Effects of Colour in the Store Environment: Review of Empirical Results, Research Model and Hypotheses

4.1. In	troduction	227
4.2. R	esearch Objectives	228
4.3. Tl	e Need to Study Colour in Context	231

4.4	. The Ir	nfluence of Colour in The Store Environment: Empirical Evidence	235
	4.4.1.	Empirical Evidence provided by Bellizzi, Crowley and Hasty (1983) : "The Effects of Colour in Store Design"	236
	4.4.2.	Empirical Evidence provided by Middlestadt (1990) : "The Effect of Background and Ambient Colour on Product Attitudes and	239 Beliefs"
	4.4.3.	Empirical Evidence provided by Bellizzi and Hite (1992) : "Environmental Colour, Consumer Feelings and Purchase Likelihood"	242
	4.4.4.	Empirical Evidence provided by Crowley (1993) : "The Two-Dimensional Impact of Colour on Shopping"	246
	4.4.5.	Overview of Empirical Evidence	249
4.5	. The Ir	nfluence of Colour in the Store Environment: Research Model	255
4.6	. The Ir	nfluence of Colour in the Store Environment: Hypotheses	257
	4.6.1.	Store Colour-evoked Pleasure	257
	4.6 4.6 4.6	5.1.1.Store Colour Hue5.1.2.Store Colour Brightness5.1.3.Store Colour Saturation	
	4.6.2.	Store Colour-evoked Arousal	
	4.6 4.6 4.6	5.2.1.Store Colour Hue5.2.2.Store Colour Brightness5.2.3.Store Colour Saturation	
	4.6.3.	Store-Colour-evoked Dominance	
	4.6 4.6 4.6	5.3.1. Store Colour Hue5.3.2. Store Colour Brightness5.3.3. Store Colour Saturation	
	4.6.4.	The Impact of Store Colour-evoked Emotions on Approach-Avoidance Behaviours	
4.7	. The N	Anderating Role of Demographic Characteristics	
	4.7.1.	Age	
	4.7.2.	Gender	274

Research Methodology

5.1.	Introduction	.284
5.2.	Experimental Design	.286
5.3.	Stimuli	.287
5.4.	Subjects & Procedures	.296
5.5.	Measures	.299

Chapter 6

Empirical Results

6.1. Introduction		
6.2. Screening for	colour blindness or colour vision deficiency	
6.3. Data screening	g	
6.4. Respondents.		
6.5. Construct vali	idation	
6.5.1. Constru	ct validation of the PAD-emotion scale	
6.5.1.1.	Exploratory Factor Analysis	
6.5.1.2.	Uni-Dimensionality	
6.5.1.3.	Within-Method Convergent Validity	
6.5.1.4.	Reliability	
6.5.1.5.	Discussion	
6.5.2. Constr	ruct validation of the approach-avoidance scale	
6.5.2.1.	Exploratory Factor Analysis	
6.5.2.2.	Uni-Dimensionality	
6.5.2.3.	Within-Method Convergent Validity	
6.5.2.4.	Reliability	
6.5.2.5.	Discussion	

6.6	. Store-	Colour	-Evoked Emotions	339
	6.6.1.	Introd	luction 'Mood' as a Covariate	339
	6.6.2.	Hypot	theses Testing: Multivariate Analysis of Variance (MANCOVA)	340
	6.6.3.	Store-	Colour-Evoked Pleasure	344
	6.6	5.3.1.	Store Colour Hue	344
	6.6	5.3.2.	Store Colour Brightness	346
	6.6	5.3.3.	Store Saturation	348
	6.6.4.	Store-	Colour-Evoked Arousal	350
	6.6	5.4.1.	Tension and Excitement: A Two-Dimensional View of Arousal	350
	6.6	5.4.2.	Store Colour Hue	350
	6.6	5.4.3.	Store Colour Brightness	355
	6.6	5.4.4.	Store Colour Saturation	359
	6.6.5.	Store-	Colour Evoked Dominance	366
	6.6.6.	Mode	rating Effects	367
	6.6	5.6.1.	Testing for the Moderating Effect of Gender	367
	6.6	5.6.2.	Testing the Moderating Effect of Age	369
6.7	. Store-	Colour	-Evoked Emotions as Determinants of Approach/Avoidance Behaviours.	371
	6.7.1.	Hypot	theses Testing: Structural Equation Modeling	371
	6.7.2.	Pleasu	are as a Determinant of Approach/Avoidance Behaviours	372
	6.7.3.	Arous	al as a Determinant of Approach/Avoidance Behaviours	374
	6.7.4.	Overv	view	376
	6.7.5.	Mode	rating Effects	377
	67	7.5.1	Testing for the Moderating Effect of Gender	
	67	7.5.2	Testing for the Moderating Effect of Age	
	5.7			
6.8	. Testir	ng Medi	ation	384

Discussion, Limitations and Implications

7.1.	Intr	oduction	.396	
7.2. Overview of the Results				
7.3.	Imp	olications	.402	
7.3	.1.	Theoretical Implications	402	
7.3	.2.	Managerial Implications	.404	
7.3	.3.	Consumer Implications	405	

7.4.	Limitations	406
7.5.	Directions for Further Research	408
Referen	ices	
Append	lix	445

List of Tables

Chapter 1

Introduction to the Research Problem

Chapter 2

Store Atmospherics

Table 2-1:	Operant Classification of Consumer Behaviour	.29
Table 2-2:	The Behavioural Response Model: Contingency Category Matrix	.30
Table 2-3:	Elements of the Store Environment	.33
Table 2-4:	A Comprehensive Classification of Store Environmental Characteristics	.34
Table 2-5:	Eight PAD-Emotion Categories	.41
Table 2-6:	Donovan and Rossiter's (1982) Approach/Avoidance Scale	.54
Table 2-7:	Studies Measuring Approach/Avoidance Behaviour	.57

Chapter 3

Colour Effects

Table 3-1:	Mean Colour Liking Ratings for Different Countries	132
Table 3-2:	The Pleasure-Eliciting Properties of Colours: Effects of Hue	134
Table 3-3:	The Pleasure-Eliciting Properties of Colours: Effects of Saturation	143
Table 3-4:	The Pleasure-Eliciting Properties of Colours: Effects of Brightness	149
Table 3-5:	The Arousal-Eliciting Properties of Colours: Effects of Hue	169
Table 3-6:	The Arousal-Eliciting Properties of Colours: Effects of Saturation	179
Table 3-7:	The Arousal-Eliciting Properties of Colours: Effects of Brightness	186
Table 3-8:	The Dominance-Eliciting Properties of Colours: Effects of Hue	195
Table 3-9:	The Dominance-Eliciting Properties of Colours: Effects of Saturation	198
Table 3-10:	The Dominance-Eliciting Properties of Colours: Effects of Brightness	

Effects of Colour in the Store Environment: Review of Empirical Results, Research Model and Hypotheses

Table 4-1:	Empirical Evidence Provided by Bellizzi, Crowley and Hasty (1983)	.251
Table 4-2:	Empirical Evidence Provided by Middlestadt (1990)	.252
Table 4-3:	Empirical Evidence Provided by Bellizzi and Hite (1992)	.253
Table 4-4:	Empirical Evidence Provided by Crowley (1993)	.254

Chapter 5

Research Methodology

Table 5-1:	Respondents According to Age and Gender	296
Table 5-2:	Demographic and Socio-Economic Characteristics of Respondents	297
Table 5-3:	Mood Short Form	299
Table 5-4:	PAD Measurement Scale	300
Table 5-5:	Approach/Avoidance Measurement Scale	301

Chapter 6

Empirical Results

Table 6-1:	Colour Blindness among Respondents	311
Table 6-2:	Demographic Characteristics of Colour-Blind Respondents	312
Table 6-3:	The Impact of Gender on Colour Vision Deficiency	312
Table 6-4:	Respondents According to Age and Gender	315
Table 6-5:	Demographic and Socio-Economic Characteristics of Respondents	316
Table 6-6:	Number of Respondents in Each Condition	317
Table 6-7:	Principle Component Factor Loadings with Varimax Rotated Solution (PAD-Scale)	320
Table 6-8:	Goodness of Fit Measures for the Four Correlated Factor Structure in the PAD- Scale	328
Table 6-9:	Principle Component Factor Loadings with Varimax Rotated Solution (Approach-Avoidance-Scale)	334
Table 6-10:	Goodness of Fit Measures for the Two Correlated Factor Structure of the Approach-Avoidance Scale	336

Table 6-11:	Main and Interaction Effects of Store Colour Wavelength, Saturation and Value on Elicited Emotions	343
Table 6-12:	Average Pleasure Elicited by the Various Store-Colour-Hues	
Table 6-13:	Average Tension Elicited by the Various Store-Colour-Hues	.352
Table 6-14:	Average Excitement Elicited by the Various Store-Colour-Hues	.353
Table 6-15:	Average Tension Elicited by the Various Store-Colour-Hues among Males and Females	368
Table 6-16:	Average Excitement Elicited by the Various Store-Colour-Hues by Age-Group	.370
Table 6-17:	Construct Correlations	.372
Table 6-18:	Goodness of Fit Measures for the Hypothesized Model (Pleasure)	.373
Table 6-19:	Goodness of Fit Measures for the Hypothesized Model (Arousal)	.375
Table 6-20:	Store-Colour Evoked Emotions: Parameter Estimates	.376
Table 6-21:	Store-Colour-Evoked Emotions: Summary of Hypotheses and Results	.376
Table 6-22:	Goodness of Fit Measures for the Hypothesized Models for Male and Female Sub-Samples	379
Table 6-23:	Unstandardized Regression Weights for Gender	.379
Table 6-24:	Goodness of Fit Measures for the Hypothesized Models for the Different Age Sub- Samples	380
Table 6-25:	Unstandardized Regression Weights for Age-Groups	.382
Table 6-26:	Store-Colour-Evoked Emotions: Summary of Hypotheses and Results	.384
Table 6-27:	Effects of Store-Colour-Evoked Emotions on Approach/Avoidance Intentions: Summary of Hypotheses and Results	384

Discussion, Limitations and Implications

List of Figures

Chapter 1

Introduction to the Research Problem

Chapter 2

Store Atmospherics

Figure 2-1:	The Mehrabian-Russell Environmental Psychology Model	22
Figure 2-2:	The Causal Chain Connecting Atmosphere and Purchase Probability	25
Figure 2-3:	A Circumplex Structure of Emotion Concepts	37
Figure 2-4:	Russell and Mehrabian's Pleasure-Arousal Hypothesis	71

Chapter 3

Colour Effects

Figure 3-1:	The Electromagnetic Spectrum	87
Figure 3-2:	Three Levels of Structure in the Eye	88
Figure 3-3:	Cone Spectral Sensitivity Curves	89
Figure 3-4:	A Colour Circle Derived from Opponent Colour Processing	91
Figure 3-5:	Opponent Processing of Cone Outputs	93
Figure 3-6:	Illustration of Deficient Colour Vision	95
Figure 3-7:	Differences in Hue, with Saturation and Value Held Constant	97
Figure 3-8:	Wavelengths of Light	98
Figure 3-9:	The Visible Electromagnetic Spectrum	98
Figure 3-10:	Hue Denotes the Dominant or 'Average' Wavelenght of Light	99
Figure 3-11:	Differences in Saturation, with Hue and Value Held Constant1	00
Figure 3-12:	Saturation Denotes the Spectral Purity of Light	00

Figure 3-13:	Differences in Value, with Hue and Saturation Held Constant	101
Figure 3-14:	Lightness or Value Denotes the Total Quantity of Light	101
Figure 3-15:	10 Hues Represented in the Munsell Colour Circle	104
Figure 3-16:	Hue Designations in the Munsell Colour Circle	105
Figure 3-17:	Value and Chroma for a Red Violet Hue	106
Figure 3-18:	Value and Chroma Ranges for Yellow and Purple-Blue Demonstrate that the Natural Colour Space is Highly Irregular	107
Figure 3-19:	The Munsell System of Colour Notation	108
Figure 3-20:	Schematic Diagram of the Three Axes of the Munsell Colour System	109
Figure 3-21:	The Munsell Colour Space	110
Figure 3-22:	Traced Sketch by Munsell of his "Colour Tree"	111
Figure 3-23:	Mean Pleasure Level as a Function of Colour Wavelength	129
Figure 3-24:	Mean Arousal Level as a Function of Colour Wavelength	165
Figure 3-25:	Actual and Predicted Average Arousal Levels as Functions of Brightness of Achromatic Colours	182
Figure 3-26:	Actual and Predicted Average Arousal Levels as Functions of Colour Brightness	185
Figure 3-27:	Mean Dominance Level as a Function of Colour Wavelength	193
Figure 3-28:	Actual and Predicted Average Dominance Levels as Functions of Brightness of Achromatic Colours	201
Figure 3-29:	Actual and Predicted Average Dominance Levels as Functions of Colour Brightness	203

Effects of Colour in the Store Environment: Review of Empirical Results, Research Model and Hypotheses

Chapter 5

Research Methodology

Figure 5-1:	Selected Values and Chromas for Each Hue	289
Figure 5-2:	Value and Chroma Ranges for Yellow and Purple-Blue Demonstrate that the Natural Colour Space is Highly Irregular	290
Figure 5-3:	The Four Stimuli Selected to Represent the Munsell 5 Yellow-Red Hue Region	294
Figure 5-4:	The Four Stimuli Selected to Represent the Munsell 5 Green-Yellow Hue Region	295

Empirical Results

Figure 6-1:	Illustration of Pseudo-Isochromatic Plates from the Ishihara Colour Vision Test	310
Figure 6-2:	Scree Plot Arousal Construct	321
Figure 6-3:	Reversal Theory: Two Hypothetical Curves Relating Arousal to Hedonic Tone	322
Figure 6-4:	Matrix Scatterplot Revealing the Correlations between the Two Distinct Arousal Factors and Pleasure	324
Figure 6-5:	Scree Plot Dominance Construct	325
Figure 6-6:	Path Coefficients of the Four-Correlated Factor PAD Emotion Construct in our Sample	328
Figure 6-7:	Scree Plot Approach-Avoidance Construct	332
Figure 6-8:	Path Coefficients in the Two-Correlated-Factor Model Underlying the Approach- Advoidance Construct in our Sample	336
Figure 6-9:	Average Pleasure Elicited by the Various Store-Colour-Hues	345
Figure 6-10:	Average Pleasure Elicited by Light and Dark Colours	346
Figure 6-11:	Average Pleasure Elicited by Light and Dark Tints of the Various Store- Colour-Hues	347
Figure 6-12:	Average Pleasure Elicited by Saturated and Unsaturated Store-Colours	348
Figure 6-13:	Average Pleasure Elicited by Saturated and Unsaturated Tints of the Various Store-Colour-Hues	349
Figure 6-14:	Average Tension Elicited by the Various Store-Colour-Hues	351
Figure 6-15:	Average Excitement Elicited by the Various Store-Colour-Hues	352
Figure 6-16:	Store-Colour-Hue Evoked Tension and Excitement	354
Figure 6-17:	Average Tension Elicited by Dark and Light Store-Colours	355
Figure 6-18:	Average Tension Elicited by Light and Dark Tints of the Various Store- Colour-Hues	356
Figure 6-19:	Average Excitement Elicited by Dark and Light Store-Colours	357
Figure 6-20:	Average Excitement Elicited by Light and Dark Tints of the Various Store- Colour-Hues	358
Figure 6-21:	Average Tension Elicited by Saturated and Unsaturated Store-Colours	359
Figure 6-22:	Average Tension Elicited by Saturated and Unsaturated Tones of the Various Store-Colour-Hues	360
Figure 6-23:	Average Excitement Elicited by Saturated and Unsaturated Tones of the Various Store-Colour-Hues for Dark and Bright Tints	361
Figure 6-24:	Average Excitement Elicited by Light and Dark Tints of the Various Store- Colour-Hues for Unsaturated and Saturated Tones	362
Figure 6-25:	Comprehensive Illustration of the Complex 'Hue x Saturation x Value' Interactions	363

Figure 6-26:	Store-Colour-Saturation-Evoked Tension and Excitement (for Red and	
-	Blue Hues)	
Figure 6-27:	Store-Colour-Hue-Evoked Tension for Males and Females	
Figure 6-28:	Store-Colour-Evoked Excitement among Different Age-Groups	
Figure 6-29:	Store-Colour-Hue-Evoked Excitement among Different Age-Groups	
Figure 6-30:	Empirical Test of Hypothesized Model (Pleasure)	
Figure 6-31:	Empirical Test of Hypothesized Model (Arousal)	
Figure 6-32:	Path Coefficients for Males versus Females	
Figure 6-33:	Unstandardized Path Coefficients for Age-Groups	
Figure 6-34:	Path-Diagram of Mediational Model	
Figure 6-35:	Direct and Pleasure-Mediated Effects of Store-Colour-Appreciation on Approach/Avoidance Intentions	
Figure 6-36:	Direct and Tension-Mediated Effects of Store-Colour-Appreciation on Approach/Avoidance Intentions	
Figure 6-37:	Direct and Excitement-Mediated Effects of Store-Colour-Appreciation on Approach/Avoidance Intentions	

Discussion, Limitations and Implications

Introduction to the Research Problem



CHAPTER STRUCTURE

Chapter 1: Introduction to the Research Problem

- 1.1. Introduction
- 1.2. Research Justification
 - 1.2.1. Importance
 - 1.2.2. Shortcomings of Existing Research
- 1.3. Research Objective
- 1.4. Methodological and Empirical Contributions
- 1.5. Structure

1.1. INTRODUCTION

In this doctoral dissertation the impact of colour in the store environment is examined, applying an environmental psychology approach. This initial chapter is devoted to the introduction of this particular research problem. In section 1.2., a justification for this study is provided. First of all, the importance of the store environment in affecting in-store consumer behaviour is outlined in paragraph 1.2.1. Next, in paragraph 1.2.2., the shortcomings of existing research with regard to the effects of colour in the store environment are assessed, as well as the shortcomings with regard to the prevalent colour-affect research in general. Moreover, the need to study colour in context is emphasized. Subsequently, section 1.3. proposes the resulting research objectives and in section 1.4. the consequent methodological and empirical contributions of the current study are presented. Finally, in section 1.5., the structure of this study is depicted.

1.2.1. IMPORTANCE

The recent finding that more than two thirds of purchase decisions are made in the store (POPAI Europe, 1998) has evoked a renewed interest for "Place", a long neglected instrument in the marketing-mix. The 'place' where a product is bought or consumed is, according to Kotler (1973), one of the most significant features of the 'total product'. Three decades ago, Kotler (1973) introduced the term 'atmospherics' to describe "the effort to design buying environments to produce specific emotional effects in the buyer that enhance his(/her) purchase probability". He identified 'atmospherics' as a highly relevant marketing instrument for retailers and argued that 'spatial aesthetics' should be consciously used by marketing planners. Markin, Lilis and Narayana (1976) also acknowledged that the store environment can be an important element affecting and shaping in-store customer behaviour. Kotler (1973) contended that, as other marketing tools become neutralized in the competitive battle, 'atmospherics' would be likely to play a growing role in the unending search of firms for differential advantage. More recently, Bitner (1990) confirms that atmospheric planning can make the difference between a business success or failure. The effect of the store environment on retailer performance, has actually been demonstrated by Kumar and Karande (2000).

In fact, there has been a growing recognition among practitioners and marketing researchers that store interiors and exteriors can be designed to create specific feelings in shoppers, that can have an important cueing or reinforcing effect on purchase (Donovan and Rossiter, 1982; Grossbart et al., 1990; Donovan et al., 1994; Greenland and McGoldrick, 1994; Sherman, Mathur and Smith, 1997; McGoldrick and Pieros, 1998; Babin and Attaway, 2000; Babin and Babin, 2001; Dubé and Morin, 2001; Moye and Kincade, 2002; Sweeney and Wyber, 2002; Retail-Update 2002; Cahan, 2002). In the last two decades, there have been many studies concerning store atmospherics based upon environmental psychology and specifically on the Mehrabian and Russell (1974) Stimulus-Organism-Response model, which has first been applied to the study of store atmosphere by Donovan and Rossiter (1982). In this context it has been proposed that approach/avoidance behaviours of customers are largely determined by individual internal (cognitive, emotional and physiological) responses to the store

environment (Bitner, 1992). The accumulated empirical evidence, as reviewed by Turley and Milliman (2000), clearly shows that shoppers can be induced to behave in certain ways, based on the atmosphere created by retail management. Out of 28 articles examining the effect of the atmosphere on sales, cited in their review, 25 found some significant relationship between the environment and customer purchasing behaviour. The review presented by Turley and Milliman indicates that atmospheric variables influence a wide variety of consumer evaluations and behaviours. Turley and Milliman (2000, p209) conclude: "Although there may be some debate about whether the atmosphere can influence time spent in an environment, there is enough evidence to be able to clearly state that the atmosphere has an effect on consumer spending and that variations of atmospheric variables affect the amount of money people spend and the number of items they purchase". Consequently, the current study aims to add to this increasingly important stream of research.

1.2.2. SHORTCOMINGS OF EXISTING RESEARCH

Intuitively aware of the importance of the store atmosphere, store managers invest huge amounts redecorating their stores. In order for them to invest their money wisely, it is of the utmost importance first to develop a comprehensive understanding of how stimuli from the store environment affect the consumer. In this context Kotler (1973) first called the attention to the great need for further research into the subject of optimal atmosphere. One particular research question he found important to be investigated was how different atmospheric elements work and what messages are communicated by particular colours, sounds, odors and textures (Kotler, 1973). Although many atmospheric variables have been extensively examined, Turley and Milliman (2000) draw, in their suggestions for further research with regard to atmospheric effects on shopping behaviour, the attention to store interior colour, which has not received the attention it probably deserves. In fact, research on the impact of colour in the store environment appears to be scarce and methodologically flawed. Although it has been demonstrated that environmental colour can actually influence the emotions, beliefs, attitudes and behaviours of retail shoppers (e.g. Bellizzi et al., 1983; Middlestadt, 1990; Bellizzi and Hite, 1992; Crowley, 1993), it is still unclear how exactly colour affects approach behaviour. This is a critical shortcoming, especially since store-interior-colours, which could, in practice, be easily adapted at minimal costs, have been proven to be a potentially influential store-design element.

Nevertheless, studies on the subject appear to be flawed, because they failed to provide adequate specifications and controls of colour stimuli, a caveat noted to be common among colour research (Gelineau, 1981, Beach et al., 1988; Valdez and Mehrabian, 1994). Indeed, colour stimuli have not been specified according to a standard colour system, but only vague verbal descriptions have been reported. For exact specification, colours should be described with respect to three different colour characteristics, namely hue, saturation and brightness, where hue refers to the pigment contained in the colour; saturation (or chroma) identifies the richness or purity of the colour (with lower saturation colours containing more grey) and brightness (or value) signifies the depth of tone in the colour or its black-to-white quality. However, in previous studies hue effects have been tested, without controlling for saturation or brightness, this way confounding the effects of the three colour attributes. Moreover, only effects of colour-hue have been investigated in a retailing context, whereas effects of saturation and brightness have been ignored completely. Yet, these neglected colour dimensions may very well be the major determining factors in colour effects on human feelings and behaviour (Valdez and Mehrabian, 1994; Gorn et al., 1997). In fact, Bellizzi, Crowley and Hasty note in their 1983 paper that the fully saturated colours used in their study may be too strong for many design applications and that it still has to be determined whether less saturated versions of these colours would have similar effects. Up to now, however, this simple question still has not been investigated. Indeed the scarce research on the topic published thus far, focused on the effects of hue, without considering effects of brightness and saturation and worse, not even controlling for them. Also Bellizzi and Hite (1992) call for further research into the area in order to develop an understanding of how colour may affect consumers. Bellizzi and Hite also suggest to study the effects of 'other' colours, as up to now mainly red and blue seem to have been investigated. They also call for further research with respect to 'other' reds and blues, referring to reds and blues with different value and saturation levels, because studies thus far only focussed on fully saturated hues.

Most previous colour research is based on consumer evaluations of colour chips. However, due to several methodological caveats much of this general research on colour and affect has also been noted to be weak (Gelineau, 1981; Beach et al., 1988; Valdez, 1993). Valdez and Mehrabian (1994, p394) point out that "despite the substantial body of experimental work in this area, results have failed to provide a thorough and general characterisation of relationships between colour and affect". According to Valdez (1993), the methodological shortcomings in this regard can be grouped into two major categories. A first major area of

caveats concerns, as already mentioned, the failure of many colour-studies to provide adequate "specifications or controls of colour stimuli" (e.g., absence of controls for saturation and brightness, while investigating effects of hue). A second area of concern involves the failure to use sufficiently reliable, valid or comprehensive "measures of emotional responses" to colour stimuli. Furthermore, it is also still unclear whether findings with regard to affective responses to colour patches can be generalized to colours applied to the store interior. According to Norman and Scott (1952), judged preferences for colours presented as small chips reveal little about preferences for coloured objects or environments. Findings by Osgood et al. (1957) also suggest that colour acceptance is defined by the object with which it is associated. Arguing that colours should first be tested, applied to objects, Guilford and Smith (1959) caution their readers with regard to applying their findings to real world situations. Rather sceptical as to the effect of colour in the environment, Beach et al. (1988), point out that studying colours in isolation (e.g. colour boards or colour samples) provides us with little information, other than how a subject at that moment rates a given colour on a given scale. They also argue that colours need to be studied in context, especially in environmental contexts, and feel that even then caution must be taken with respect to the generalizability of the findings. On the other hand, Taft (1997) found few significant differences between affective ratings of colour-chips and objects of the same colour, suggesting that results of earlier colour-meaning research may be generalized and used for colour planning. However, his study did not include colours applied to the environment. Thus, as studies on colour preference and colour meaning still systematically rely on the use of colour chips, and the study of colours in environmental contexts is scarce, it remains a critical question whether preferences for and meanings of isolated colour-chips are generalizable to contextual colours, such as store colour. The current study aims to address the aforementioned caveats.

With regard to the "effects" of colours in the environment, actually, the literature seems to contain mostly anecdotal evidence, revealing a severe lack of systematic empirical research. This has, however, not restrained practitioners from applying specific colours in sports locker rooms, hospitals, jail cells, fast-food restaurants and other places, with the aim of influencing the emotional state of the environment's occupants.

1.3. RESEARCH OBJECTIVE

The study, presented here, aims to fill this gap in an extensive stream of research concerning the impact of store atmospherics on the consumer. As research on the impact of colour in the store environment is scarce and methodologically flawed, it is our challenge to assess the impact of this store design element more thoroughly. It is our purpose to examine the different effects of store-colour hue, brightness and saturation on the emotions elicited by the store and on consumers' subsequent approach-avoidance behavioral intentions towards the store. Special attention will be devoted to the elaboration of an appropriate methodology.

In the current research endeavour we will attempt to address the vital issues discussed in the previous paragraph. Two basic objectives are contemplated. First and foremost, a more accurate methodology will be proposed for assessing the impact of colour in the store environment on shopper behaviour. Applying an environmental psychology approach, the proposed study aims to assess (1) the direct effect of colour in the store environment on the emotions experienced in the store (notably pleasure, arousal and dominance) and (2) the direct and indirect effect of colour on consumers' appoach/avoidance behaviour through the store colour-evoked emotions. In particular, the specific effects of colour hue, saturation and brightness in the store design will be investigated. Moreover, the moderating impact of two demographic characteristics, age and gender, with regard to atmospheric responsiveness will be assessed.

Although Bellizzi et al. (1983) hoped that their initial study would serve as a catalyst to encourage additional research interest in the role of colour in retail store management, they did not have many academic adherents. With this research project we hope not only to gain a deeper understanding of the effects of colour in the store environment, but also to generate some revived academic interest into the subject.

From a more practical point of view, this research is intended to assist retailers in selecting and avoiding certain colours in order to produce desired consumer moods and in-store behaviours. We are convinced that the retailer as well as the consumer may profit from such research, as it could lead the way to better-designed store-interiors. Guided by our objective to investigate the specific effects of the three colour-attributes, a three-way (hue x value x saturation) factorial between-subjects experimental design is developed. In fact, the factorial design involves eight hues, by two saturation levels (unsaturated versus saturated), by two brightness levels (dark versus light). Based on the Munsell Colour System, accordingly, 32 specific colour-tones are selected by means of a tedious procedure. These colours are applied to an experimental store environment, which has especially been developed in Computer Aided Design to serve as a stimulus in this study. Thus manipulating store-interior colours allowed us to question respondents with regard to the emotions experienced while viewing a picture of the store-interior and to tap their approach/avoidance responses with respect to the store. A total of 874 respondents participated in this study. Respondents were matched as much as possible according to age and gender, so that the moderating effects of these demographic characteristics could be investigated.



Figure 1-1: Structure of the Study

As illustrated in figure 1-1, this dissertation is composed of two main parts: a theoretical framework and the empirical study. In the theoretical framework, a background and critical review is provided with regard to 'store atmospherics' (chapter 2) and 'colour effects' (chapter 3). Subsequently 'the effects of colour in the store environment' will be elaborated on in chapter 4. This chapter includes, besides a review of empirical results, our research objectives, the research model and the related hypotheses. In part 2, the empirical research of this study will be presented. Chapter 5 describes the research methodology and in chapter 6 the research results will be elaborated. Finally, in chapter 7, the main results of the study will be discussed and its limitations will be provided out. In conclusion, suggestions for several possible research avenues for the future will be provided and the implications of the findings will be discussed.
Part I Theoretical Framework

Chapter 2 Store Atmospherics



CHAPTER STRUCTURE

Chapter 2: Store Atmospherics

- 2.1. Introduction
- 2.2. Significance of the Store Environment
- 2.3. An Environmental Psychology Approach to Store Atmospherics
 - 2.3.1. The Mehrabian-Russell Environmental Psychology Model
 - 2.3.2. Application of the Model to Store Atmospherics
- 2.4. The Stimulus: Characterizing the Store Environment
 - 2.4.1. A General Characterization of the Store Environment
 - 2.4.1.1. The Information Rate of the Environment 2.4.1.2. The Behavioural Perspective Model
 - 2.4.2. Towards a Classification of Store-environmental Characteristics
- 2.5. The Organism: Emotional Responses to Shopping Environments
 - 2.5.1. A Discrete Emotions Perspective
 - 2.5.2. A Dimensional Emotion Perspective
 - 2.5.3. Emotional Responses to the Shopping Experience
- 2.6. The Response: Approach-Avoidance Behaviours in Shopping Environments
 - 2.6.1. Approach-Avoidance Behaviours : General Conceptualisation
 - 2.6.2. Approach-Avoidance Behaviours in a Retail Setting
- 2.7. Emotional Determinants of Approach-Avoidance Responses to Shopping Environments
 - 2.7.1. The Effects of Pleasure on Approach or Avoidance Behaviours
 - 2.7.2. The Effects of Arousal on Approach or Avoidance Behaviours
 - 2.7.3. The Effects of Dominance on Approach or Avoidance Behaviours
- 2.8. Atmospheric Effects on Shopping Behaviour: A Review of Empirical Evidence

2.1. INTRODUCTION

In this chapter a brief overview will be given of the store atmospherics literature. First of all, in section 2.2. the significance of the store environment will be established. Subsequently, in section 2.3. an environmental approach to store atmospherics will be elaborated. In this section the Environmental Psychology Model, proposed by Mehrabian and Russell (1974), will be introduced and the application of this model to store atmospherics will be demonstrated. Next, in section 2.4. the stimulus-part of the model will be extended to the store environment. Two general characterisations of the store environment will be proposed, followed by a classification of specific store environmental characteristics. In section 2.5. the organism-part of the model is elaborated, providing an overview of emotional responses from a discrete as well as from a dimensional perspective and extending these to the shopping experience. Section 2.6. centres around the response-part of the model, and focuses in particular on approach-avoidance responses to shopping environments are examined. Finally, in section 2.8. a brief review of empirical evidence regarding atmospheric effects on shopping behaviour is provided.

"Retail space, i.e. the proximate environment that surrounds the retail shopper, is never neutral."

> MARKIN, LILIS & NARAYANA 1976

Foxall (1993) comments on the "apparent placelessness" of the behaviour that many models of consumer choice seek to explain, hereby emphasizing the need to integrate environmental influences. Indeed, Kotler (1973) stresses that atmospheres should not be ignored, since they are a subtle factor present in every buying situation. Markin, Lilis and Narayana (1976) agree that the store environment itself should not be ignored as an important element affecting and shaping in-store customer behaviour and point out that "retail space, i.e. the proximate environment that surrounds the retail shopper, is never neutral." According to Philip Kotler (1973) the 'place' where a product is bought or consumed is one of the most significant features of the 'total product'. He argues that in some cases, the place and more specifically the 'atmosphere' of the place, is more influential than the product itself in the purchase decision and that the atmosphere may even be the primary product.

As 'atmosphere' refers to 'the quality of the surrounding space', Kotler (1973) introduces the term 'atmospherics' to describe "the conscious designing of space to create certain effects in buyers". He defines 'atmospherics' more specifically as "the effort to design buying environments to produce specific emotional effects in the buyer that enhance his purchase probability"

Markin, Lilis and Narayana (1976) refer to the retail store as "*a bundle of cues, messages and suggestions that communicate to shoppers.*" Kotler (1973) suggests that it is because atmospheres are a *'silent language'* in communication that they have been neglected for such a long time. But he argues that in the future *'spatial aesthetics'* should be consciously used by marketing planners as a real marketing tool. He identifies 'atmospherics' as a highly relevant marketing instrument for retailers. As other marketing tools become neutralized in the competitive battle, atmospherics – the conscious planning of atmospheres to contribute to the buyer's purchasing propensity – is likely to play a growing role in the unending search of firms for *differential advantage* (Kotler, 1973). Indeed, Kotler (1973) proposes that atmospherics becomes a more relevant marketing tool as the number of competitive outlets increases. Bitner (1990) confirms that atmospheric planning can make the difference between a business success or failure.

From that moment onwards there has been a growing recognition among practitioners and marketing researchers that store interiors and exteriors can be designed to create specific feelings in shoppers that can have an important cueing or reinforcing effect on purchase (Donovan and Rossiter, 1982; Grossbart et al., 1990; Donovan et al., 1994; Greenland and McGoldrick, 1994; Sherman, Mathur and Smith, 1997; McGoldrick and Pieros, 1998; Babin and Attaway, 2000; Babin and Babin, 2001; Dubé and Morin, 2001; Moye and Kincade, 2002; Sweeney and Wyber, 2002).

If consumers are influenced by stimuli experienced at the point of purchase, the practice of creating influential atmospheres should become an important marketing strategy for most exchange environments. In this context Kotler (1973) first called the attention to the great need for further research into the subject of optimal atmosphere. One particular research question he found important to be investigated was how different atmospheric elements work and what messages are communicated by particular colours, sounds, odors and textures (Kotler, 1973).

In the last two decades, there have been many studies concerning store atmospherics based upon environmental psychology. The accumulated empirical evidence, as reviewed by Turley and Milliman (2000), clearly shows that shoppers can be induced to behave in certain ways, based on the atmosphere created by retail management. Out of 28 articles examining the effect of the atmosphere on sales, cited in their review, 25 found some significant relationship between the environment and customer purchasing behaviour. The review presented by Turley and Milliman indicates that atmospheric variables influence a wide variety of consumer evaluations and behaviours. Moreover, these studies also show that these relationships occur across a number of different types of retail stores and situations. Even when consumers are not aware of particular facets of the retail atmosphere, these may be influencing their behaviour (Milliman, 1982; Gulas and Schewe, 1994). Turley and Milliman (2000, p209) conclude: "Although there may be some debate about whether the atmosphere can influence time spent in an environment, there is enough evidence to be able to clearly state that the atmosphere has an effect on consumer spending and that variations of atmospheric variables affect the amount of money people spend and the number of items they purchase".

The finding that more than two thirds of purchase decisions are made in the store (POPAI Europe, 1998) has evoked a renewed interest for a long neglected instrument in the marketing-mix: "*Place*". Aware of this fact, store managers invest huge amounts redecorating their stores. In order for them to invest their money wisely, it is of the utmost importance first to develop a comprehensive understanding of how stimuli from the store environment affect the consumer.

"Atmospherics refers to

the effort to design buying environments to produce specific emotional effects in the buyer

that enhance his purchase probability"

KOTLER

1973

2.3. AN ENVIRONMENTAL PSYCHOLOGY APPROACH TO STORE ATMOSPHERICS

It is interesting to note that until the 1960's, psychologists have largely ignored the effects of the physical setting in their attempts to predict and explain behaviour (Bitner, 1992). From that time on, the systematic study of environment-behaviour relationships by psychologists has produced a rapidly growing dicipline, known as "environmental psychology" (Darley and Gilbert, 1985; Holahan, 1986; Stokols and Altman, 1987). Environmental psychology can be defined as "*that segment of psychology concerned with the systematic accounting of the relationship between person and environment*" (Russell and Ward, 1982). The classic emotional approach of environmental psychology is based on the fundamental notion that emotions elicited by the environment determine human behaviour (Mehrabian, 1978, p14).

2.3.1. THE MEHRABIAN-RUSSELL ENVIRONMENTAL PSYCHOLOGY MODEL

Mehrabian and Russell (1974) explored an emotional theory of environmental psychology, proposing that a person's behaviour in an environment is influenced "by the emotions that are elicited from him/her by that environment" (Russell and Mehrabian, 1978). According to this theory, an environment evokes emotional reactions in a person, which cause this person to approach or avoid this environment (Mehrabian, 1978, p15).



Source: Mehrabian & Russell,1974

Based on the Stimulus-Organism-Response (S-O-R) Paradigm (Lewin, 1936; Kelly, 1955; Rotter, 1954), Mehrabian and Russell (1974) thus propose a model of environmental psychological behaviour (presented in figure 2-1), relating features of the environment (S) to

approach-avoidance behaviours within the environment (R), mediated by the individual's emotional states (O) induced by the environment. Or stated differently: various environmental stimuli (colours, music, shapes, etc.) engender primary emotional responses, which as intervening variables, determine reactions to that environment.

In environmental psychology, researchers have studied how to most efficiently describe the emotional experience in an attempt to provide better descriptors to assess emotional responses to places and experiences therein (Russell and Pratt, 1980). In accordance with the basic three dimensions of meaning ('Evaluation', Activity' and 'Potency') identified by Osgood, Suci, and Tannenbaum (1957), Mehrabian and Russell (1974) proposed that three basic emotional states 'Pleasure', 'Arousal' and 'Dominance' provide a general description of emotions generated by an environment. Pleasure (the emotional counterpart of 'evaluation') refers more specifically to the degree to which a person feels happy or satisfied in a place. Arousal (the emotional correlate of stimulus' activity) refers to the degree of stimulation caused by an atmosphere. And dominance (the converse of stimulus 'potency') refers to the degree to which a person feels in control in a situation.

The various stimuli of the environment can, according to Mehrabian and Russell (1974b), be characterised in terms of "*the information rate*" of the environment (i.e. the quantity of information drawn from, or perceived in, the environment per unit of time). That is, the more varied, complex, novel, surprising and lively the environment is, the higher its information rate will be. The information rate of the environment is assumed to be directly related to the degree of arousal it induces. A "high-load" environment (i.e. novel, surprising, crowded) is expected to make its occupant feel stimulated, excited and alert, whereas a "low-load" environment (i.e. simple and familiar) is expected to make a person calm, relaxed and sleepy.

The emotion-eliciting quality of an environment thus refers to the average pleasure, arousal and dominance engendered by that environment in a representative sample of subjects (Russell and Mehrabian, 1978).

The response to any environment can, according to Mehrabian and Russell (1974), be subsumed in two basic categories of "approach" and "avoidance" behaviours (Wundt, 1905). Approach or convergence means that an individual reacts positively to the environment (i.e. desire to stay, explore, work and affiliate), whereas avoidance is characterised by an aversion

to the environment (a desire not to affiliate in the environment and to leave the environment). These two forms of behaviour are assumed to be opposite.

Mehrabian and Russell (1974) posit that the emotional states (pleasure, arousal and dominance) engendered by the environment, mediate a person's approach-avoidance responses to the environment. More specifically, they hypothesize pleasure to be significantly related to approach-avoidance measures overall, and that arousal has an interactive (multiplicative drive-like) effect with pleasantness such that arousal would be positively related to approach behaviours in pleasant environments, but negatively related to approach behaviours in unpleasant environments. They also hypothesized that dominance would be positively related to approach behaviours.

Thus, according to this model, environmental stimuli are presumed to engender primary emotional reactions in an individual, which mediate approach-avoidance behaviours, such as the desire to affiliate with others in the setting, the desire to stay in, or escape from, the setting and the willingness to spend money and consume there (Mehrabian, 1979; Mehrabian and Riccioni, 1986; Mehrabian and de Wetter, 1987; Mehrabian and Russell, 1975; Russell and Mehrabian, 1976, 1978).

2.3.2. APPLICATION OF THE MODEL TO STORE ATMOSPHERICS

Introducing the concept of store atmospherics, Kotler (1973) gives a tentative explanation as to how the atmosphere of a place may affect purchase behaviour, based on a causal chain shown in figure 2-2.

- (1) A product is surrounded by a space, characterized by certain sensory qualities.
- (2) The consumer perceives certain qualities of this space. This perception is subject to selective attention, distortion, and retention.
- (3) The perceived qualities of the atmosphere can influence the consumer's information and affective state.
- (4) The consumer's modified information and affective state may increase his purchase probability of the product.



Figure 2-2 : The causal chain connecting atmosphere and purchase probability Source: Kotler, 1973

Kotler (1973) argues that the atmosphere can have an effect on purchase behaviour in at least three ways: it may serve as (1) *an attention-creating medium*, as (2) *a message creating medium* and as (3) *an affect creating medium*.

It is in particular this last aspect that was elaborated on by Donovan and Rossiter (1982), who introduced the theoretical model of Mehrabian and Russell (1974) to the study of store atmosphere by proposing the store environment as the environmental stimulus. Donovan and Rossiter found store-induced arousal, and in particular store-induced pleasure to be very

powerful determinants of in-store approach/avoidance reactions. Pleasure and arousal were found to influence consumers' stated (1) enjoyment of shopping in the store, (2) time spent browsing and exploring store offerings, (3) willingness to talk to sales personnel, (4) tendency to spend more money than originally planned and (5) likelihood of returning to the store.

As this "emotional" model of environmental psychology has proven to be a valuable framework for generating hypotheses concerning the store atmosphere, it has successfully been applied by many atmospheric researchers since (Darden and Babin, 1994; Van Kenhove and Desrumaux, 1997; Hui, Dube and Chebat, 1997; Sherman, Mathur and Smith, 1997).

In line with Kotler's (1973) more comprehensive conceptualisation of atmospherics, Bitner (1992) extended the original M-R model to service-settings by adding to her framework cognitive and physiological internal responses to the emotional ones originally proposed by Mehrabian and Russell. According to Russell and Pratt (1980) the meaning that persons attribute to an environment can indeed be divided into perceptual-cognitive meaning and affective meaning.

However, as the first level of response to an environment is generally accepted to be affective (Ittelson, 1973, p16), in this thesis the classical emotional environmental psychology model (Mehrabian and Russell, 1974; Donovan and Rossiter, 1982) will be applied, with an exclusive focus on emotional internal responses.

In the next paragraphs we will elaborate this S-O-R conceptualisation of store atmospherics a little further. We will give a more detailed presentation on the store environment (the Stimulus), on the internal emotional responses evoked by shopping environments (the Organism) and on approach-avoidance behaviours in a shopping context (the Response). Furthermore, we will review whether the proposed relations in the model are applicable in a shopping context.

2.4. THE STIMULUS: CHARACTERIZING THE STORE ENVIRONMENT

2.4.1. A GENERAL CHARACTERIZATION OF THE STORE ENVIRONMENT

2.4.1.1. The Information Rate of the Environment

Based on information theory, Mehrabian and Russell (1974) characterize the various stimuli of the environment in terms of "*the information rate*" or "*load*" of the environment, which they define as the quantity of information drawn from, or perceived in, the environment per unit of time, measured by its degree of "novelty" (i.e. newness, unexpectedness) and "complexity" (i.e. the number of elements or features and the extent of motion or change in the environment). That is, the more varied, complex, novel, surprising and lively the environment is, the higher its information rate will be. A direct link is conceived between an environment's information rate and its arousability.

Actually, the development of an adequate stimulus taxonomy for research in environmental psychology has been a challenge, because of the many stimuli involved in any environmental setting. For this reason Mehrabian and Russell (1974) aimed for a measure of environmental stimulation, applicable across various physical and social settings.

As Donovan and Rossiter (1982) introduced the theoretical model of Mehrabian and Russell (1974) to the study of store atmosphere, they introduced the store environment as stimulus. Nevertheless, they maintained the general (i.e. non-specific) stimulus taxonomy proposed by Mehrabian and Russell (1974).

Bitner (1992) makes a distinction between '*lean*' and '*elaborate*' environments, based on the physical complexity of the 'servicescape' (i.e. the physical surroundings of the place where a service is delivered or consumed). In this context, Bitner (1992) describes '*lean*' environments as very simple, with few elements, few spaces and few forms and '*elaborate*' environments as very complicated with many elements and many forms.

In an attempt to explain the affective assessment of outdoor environments, Kaplan (1987) extends the environmental information rate notion of Mehrabian and Russell and contends that preference for, or liking of, a particular environment can be predicted according to three environmental dimensions: its '*complexity*' (i.e. visual richness, ornamentation, information rate), '*coherence*' (i.e. order, clarity, unity) and '*mystery*'.

2.4.1.2. The Behavioural Perspective Model

According to Foxall and Greenley (1999) research on the impact of the environment on consumer behaviour has generated mixed results, because it has lacked a theory-based classification of consumer environments.

With the Behavioural Perspective Model (BPM), Foxall and Greenley (1999) propose a theory-based classification of consumer settings, according to which the emotions elicited by the environment, and thus behaviour, could be predicted more accurately. According to this model consumer behaviour is predictable from two dimensions of situational influence: (1) the scope of the consumer behaviour setting; and (2) the utilitarian and informational reinforcement signalled by the setting as primed by the consumer's learning history (Foxall, 1999).

According to Foxall (1999), consumer behaviour settings, which comprise the stimuli that make up the physical and social surroundings in which the consumer behaves, can differ in the degree to which they promote or hinder specific consumer activities and the extent to which they guide or compell a consumer to act in a certain way.

Thus, consumer behaviour settings can be classified according to their "*scope*", on a continuum from *closed* to *open*. In relatively closed settings, consumers are expected to conform to a certain behaviour pattern in the setting (i.e. purchase and consumption are managed largely by persons or entities other than the consumer). In contrast, in a relatively open consumer behaviour setting, strong physical, social and verbal pressures to conform to a particular behaviour are largely absent. In open settings, "the consumer enjoys a degree of discretion over the physical and social contingencies, or can determine his or her own rules for choosing among the products and brands offered" (Foxall, 1999, p151). Think for instance

about the optician, where you go and explain what kind of glasses you would like. The optician selects some pairs of glasses and you try them on and decide which ones you prefer. A lot of people feel obliged to select one of the proposed models and to order the glasses right away. Therefore, this kind of retail setting can be conceived of as a rather closed setting. On the other hand, there are the more self-service oriented optician chains (e.g. Pearle Vision) where you can walk in, take the glasses off the shelves yourself, try them on, and walk out, without buying, as you please. Assistance is available if you want some, but it is not forced on you. This kind of setting is a more open one. The behaviour in open settings is usually positively reinforced. A store can also have a rather closed setting if it has guided pathways that you have to go through in order to get where you want to be and where you cannot easily skip parts of the store because it doesn't carry any merchandize you need (e.g. IKEA). The "scope" or "openness" of the environment is conceived to be closely related to the dominance it elicits.

Furthermore, settings can, according to Foxall (1999) and Foxall and Greenley (1999), be classified according to the "*reinforcement*" they provide, which can be *utilitarian* and/or *informational*. Utilitarian reinforcement refers to the utility or economic satisfaction received by consumers as a result of purchase or consumption, whereas informational reinforcement refers to the feed-back received on one's performance as a consumer, or to the level of achievement or social status conferred by shopping in the store. These types of reinforcement are assumed to be independent of each other and can each be high or low, depending on the setting.

	Utilitarian Reinforcement		
Informational Reinforcement	High	Low	
High	ACCOMPLISHMENT	ACCUMULATION	
Low	HEDONISM	MAINTENANCE	
Table 2 Source	-1: Operant classification : Foxall, 1997, p195	of consumer behaviour	

This way four classes of settings can be identified, depending on the relative levels or utilitarian and informational reinforcement they provide, with four corresponding classes of consumer behaviours being accomplishment, hedonism, accumulation and maintenance as illustrated in table 2-1 (Foxall, 1992; 1997).

By subdividing these classes further according to the relative scope or openness of the setting, a matrix of 8 contingency categories (illustrated in table 2-2) is proposed to provide a framework for a systematic investigation of Mehrabian and Russell's (1974) approach to environmental psychology in the consumer behaviour field (Foxall, 1997; Foxall and Greenley, 1999). According to Foxall and Greenley (1999, p151) this matrix "functionally defines an exhaustive range of consumer situations based on environmental contingencies.

	Closed	Open
ACCOMPLISHMENT		
	Fulfilment	Status
high utilitarian;		Consumption
high informational		
HEDONISM		
	Inescapable	Popular
high utilitarian;	Entertainment /	Entertainment
low informational	Amelioration	
ACCUMULATION		
	Token-based	Collecting
low utilitarian;	consumption	
high informational		
MAINTENANCE		
	Mandatory	Routine
low utilitarian;	Consumption	Purchasing
low informational		

BEHAVIOUR SETTING SCOPE

Table 2-2:The behavioural response model: contingency category matrixSource:Foxall & Greenley, 1999

Foxall and Greenley (2000) demonstrate by means of discriminant analysis that pleasure, arousal and dominance elicited by an environment can be predicted by the structural features of consumer situations proposed by the BPM. Consumer behaviour settings characterised by a relatively high utilitarian reinforcement appear to elicit more pleasure. Relatively high informational reinforcing settings evoke more arousal. And reported dominance appears to be higher in relatively open settings.

2.4.2. TOWARDS A CLASSIFICATION OF STORE ENVIRONMENTAL CHARACTERISTICS

One of the limitations of the Mehrabian-Russell (1974) and the behavioural perspective models is the lack of a classification system of specific environmental features (Donovan and Rossiter, 1982; McGoldrick & Pieros, 1998).

As '*atmosphere*' is apprehended through the senses, the atmosphere of a particular set of surroundings can, according to Kotler (1973), be described in sensory terms, with the main sensory channels for atmosphere being sight, sound, scent and touch. Accordingly, colour, brightness, size and shapes constitute the main visual dimensions of an atmosphere; volume and pitch the main aural dimensions; scent and freshness the main olfactory dimensions and finally softness, smoothness and temperature the main tactile dimension (Kotler, 1973).

Baker (1986) developed a more practical framework of environmental factors. She developed a typology categorizing the elements of the store environment into three critical dimensions: ambient, design and social factors:

- (1) <u>Store ambient factors</u> are nonvisual, background conditions in the environment including elements such as temperature, lighting, noise, music and scent (e.g., Milliman, 1982, 1986; Ward and Russell, 1981; Wineman, 1982; Yalch and Spangenberg, 1990). Customers may notice ambient factors when they exceed an acceptable range, such as when the lighting becomes too bright or the music too loud.
- (2) <u>Store design factors</u> are store environmental elements that are more visual in nature. These elements may be functional and/or aesthetic in nature (Marans and Spreckelmeyer, 1982). Functional store elements include layout, comfort and privacy.

Aesthetic elements include factors such as architecture, colour, materials, style and cleanliness.

(3) <u>Store social factors</u> relate to other people present in the store (Baker et al., 1994). Russell and Snodgrass (1987) noted that the physical presence of another person is an important part of any environment. The "people" component of the environment includes both store employees and customers (e.g. Bitner, 1992). The number, type and behaviour of people is proposed to influence consumers' perceptions of stores.

The Baker (1986) store environment framework has been applied in experimental settings to examine the effects of specific environmental stimuli on emotional states (Baker and Levy, 1993), on price acceptability (Grewal & Baker, 1994) and on merchandise and service quality inferences and store image (Baker, Grewal and Parasuraman, 1994).

Bitner (1992) presents a typology of "*Servicescapes*" (i.e., the physical surroundings of the place where a service is delivered or consumed), which is heavily based on the Baker (1986) framework.

According to Berman and Evans (1995, 2001) "atmosphere" can be divided into several elements of the store environment: the exterior of the store, the general interior, store layout, and interior (point-of-purchase) displays. Berman and Evans (1995, 2001) do not consider the social aspect as part of the store atmosphere, however they do include elements of the store exterior. In this regard, Kotler (1973) also referred to architecture as the atmospherics of the exterior structure of buildings. Everett et al. (1994) point out that it is important to recognize that attitudes and behaviour are influenced by environmental attributes at several levels, from those closest to the consumer, to the external store features (Ward et al., 1992), through to the 'macro-environment', for example the town or city. Table 2-3 contains a detailed breakdown of the elements of the store environment, as presented by Berman and Evans (2001, p604).

Elements of the store environment

Exterior

- Storefront
- Marquee
- Entrances
- Display windowsHeight of building
- Size of building
- Visibility
- Uniqueness
- Surrounding stores
- Surrounding area
- Parking
- Tarking
- Congestion

General interior

- Flooring
- Colours
- Lighting
- Scents, sounds
- Fixtures
- Wall textures
- Temperature
- Width of aisles
- Dressing facilities
- Vertical transportation
- Dead areas
- Personnel
- Self-Service
- Merchandise
- Prices (levels and displays)

- Cash register placement

- Technology/modernization
- Cleanliness

Store layout

- Allocation of floor space for selling, merchandise, personnel, and customers
- Product groupings
- Traffic flow
- Space/merchandise category
- Department locations
- Arrangements within
- departments

Interior (point of purchase) displays

- Assortment
- Theme-setting
- Ensemble
- Racks and cases
- Cut cases and dump bins
- Posters, signs and cards
- Mobiles
- Electronic displays

Table 2-3:Elements of the store environmentSource:Berman and Evans, 2001, p604

Turley and Milliman (2000) complete the typology proposed by Berman and Evans (1995) by including a fifth category of human variables, which is in accordance with Baker's (1986) store social factor.

In an attempt to create some organized and logical structure to the study of atmospheric variables potentially influencing consumer behaviour, we further integrate the various typologies and propose a comprehensive classification of store environmental cues, comprising 7 categories: the macro environment, the store exterior, store ambient factors, interior aesthetic design factors, interior functional design factors or store lay-out, point-of-purchase displays and store social factors, including store personnel and customers. Table 2-4 contains a detailed breakdown of the elements of these categories.

A Comprehensive Classification of Store-Environmental Characteristics

The macro environment

- Surrounding stores
 - Shopping center the store is located in
 - Surrounding area / neighbourhood, town or city the store is located in
 - Parking
 - Congestion

The store exterior

- Architecture
- Height of building
- Size of building
- Storefront
- Marquee
- Entrances
- Display windows
- Visibility
- Uniqueness

Store ambient factors

- Nonvisual, background conditions in the environment
- Temperature
- Noise
- Music
- Scent
- Cleanliness

Interior aesthetic design factors

- Interior architecture
- Colour
- Lighting
- Materials
- Style

Interior functional design factors / store layout

- Allocation of floor space for selling, merchandise, personnel, and customers
 - Product groupings
 - Traffic flow
 - Space/merchandise category
 - Department locations
 - Arrangements within departments

Point-of-purchase displays

- Assortment
- Theme-setting
- Ensemble
- Shelving / Product placement
- Racks and cases
- Cut cases and dump bins
- Posters, signs and cards
- Mobiles
- Electronic displays

Store social factors

- Store personnel
- Number of salespersons
- Employee characteristics
- Employee uniforms
- Store customers
- Customer characteristics
- Crowding
- Privacy

Table 2-4: A comprehensive classification of store-environmental characteristics

Source: Based upon Kotler, 1973; Baker, 1986; Bitner, 1992; Everett et al., 1994; Turley and Milliman, 2000; Berman and Evans, 2001.

2.5. THE ORGANISM: EMOTIONAL RESPONSES TO SHOPPING ENVIRONMENTS

That the first level of response to any environment is affective (emotional) is widely accepted in psychology (Ittelson, 1973). Because positive feelings have been found to lead to more positive product evaluations and vice versa (Bagozzi, Gopinath & Nyer, 1999), marketing researchers have generally acknowledged that emotions play a prominent role in the marketing of products and services. The pervasive influence of emotional response has been recognized in various contexts, such as advertising, product consumption and shopping (Holbrook, Chestnut, Oliva and Greenleaf, 1984; Batra and Ray, 1986; Westbrook, 1987; Batra and Holbrook, 1990; Cohen, 1990). As emotions play a central mediating role in environmental psychology, we will take a closer look at the nature of store-evoked emotions.

In accordance with Lazarus (1991) and Oatley (1992), Bagozzi, Gopinath and Nyer (1999, p184) define emotion as "a mental state of readiness that arises from cognitive appraisals of events or thoughts, has a phenomenological tone, is accompanied by physiological processes, is often expressed physically (e.g., in gestures, posture, facial features), and may result in specific actions to affirm or cope with the emotion, depending on its nature and meaning for the person having it". Furthermore, referring to Valdez and Mehrabian (1994, p407), Mehrabian (1996) proposes the premise that: "judgments, preferences or attitudes require an emotional foundation and cannot operate in an emotional vacuum". According to this premise: "emotions are the developmental precursors of elementary cognitive judgements and constitute the foundation for the latter".

In contrast to "emotional traits", which refer to personal emotional characteristics or temperaments, which are stable over periods of years or even a lifetime, "emotional states" refer to transitory conditions of the organism, which may vary substantially and rapidly over the course of a day (Mehrabian, 1996). "Moods", although similar, differ by convention from transient emotions in that a mood is generally longer lasting (from a few hours up to a couple of days) and lower in intensity. Also, an emotion has a clear object or referent (e.g. the store environment), whereas moods are generally non-intentional and global or diffused (Frijda, 1993). Moreover, moods are not as directly coupled with action tendencies and explicit actions as emotions often are (Bagozzi, Gopinath & Nyer, 1999). Although emotions are generally stronger in intensity than moods, they can also vary in intensity (Sokolov and

Boucsein, 2000). In this respect, milder, less intense emotions, which are more common in a marketing context, are also often referred to as feelings or affective responses (Frijda, 1987).

Gardner (1985) (see also Isen et al., 1978; Westbrook, 1980) hypothesized that transient feelings may have special impact in retail or service encounters because they may be natural outcomes of the interpersonal and dyadic nature of the retail environment. Indeed, empirical research has demonstrated that shopping environments can evoke emotional responses in consumers and that such emotions, in turn, influence shopping behaviours and outcomes (Donovan and Rossiter, 1982; Darden and Babin, 1994; Hui, Dube and Chebat, 1997; Sherman, Mathur and Smith, 1997).

2.5.1. A DISCRETE EMOTIONS PERSPECTIVE

One approach to the study of emotions, termed "*the discrete emotions perspective*", proposes that emotions can be conceptualised as a set of discrete and phenomenologically distinct affective states (e.g., Izard, 1977; see Batra and Ray, 1986).

Plutchik (1980) contends that the eight emotion categories of anger, joy, sadness, acceptance, disgust, expectancy, surprise and fear are at the root of all human emotional responses. According to Havlena and Holbrook (1986), the Plutchik measure has been widely applied in consumer research with regard to the measurement of emotions.

Izard (1977) distinguishes in his differential emotions theory among 10 fundamental emotions, which have also become highly popular among consumer researchers due to the diversity of emotions encompassed: joy, sadness, interest, anger, guilt, shyness/shame, disgust, contempt, surprise and fear. Izard's measure also includes a number of negative emotions, which according to Machleit and Eroglu (2000) may be pertinent to the shopping experience (such as anger or disgust due to poor interactions with salespersons, guilt over purchase and so forth).

2.5.2. A DIMENSIONAL EMOTION PERSPECTIVE

The second approach to the study of emotions, termed "*the dimensional perspective*", has suggested that more basic understanding of the impact of emotions can be derived from reducing the various emotion types into a set of underlying dimensions.

Indeed, emotion categories appear to be related to each other in a systematic way (Russell and Lemay, 2000). This underlying structure of emotions has received some renewed interest as researchers attempt to represent the relations among the various discrete emotions (see Diener, 1999; Russell and Feldman Barrett, 1999). Both, factor analysis and multi-dimensional scaling of emotion-related expressions appear to converge systematically in a two- or three-dimensional representation (Russell and Lemay, 2000). Figure 2-3 represents such a circumplex structure of emotions.



Figure 2-3:A circumplex structure of emotion conceptsSource:Russell and Lemay, 2000

In this two-dimensional circumplex, emotions range along an 'evaluation' dimension, from unpleasant to pleasant and, according to the degree of stimulation, along an 'activation' dimension from un-aroused and inactive to active and aroused. Whereas Watson and Tellegen (1985) originally questioned the bipolarity of such dimensions, claiming negative and positive affect to be independent of each other (see also Westbrook, 1987 and Mano and Oliver, 1993), they confirmed more recently that evidence supports such bipolarity (Watson and Tellegen, 1999; Russell and Carroll, 1999). Also Thayer's (1989) two dimensional view of tense and energetic arousal can be integrated into this two-dimensional space when viewed as 45° rotations of one another (Yik, Russell and Feldman Barrett, 1999). The eight combinations of pleasantness and activation identified by Larsen and Diener (1992) also clearly fit this circumplex. Thus, the existence of two basic dimensions, representing respectively hedonic tone and activation, has received extensive theoretical and empirical support (Mano, 1997; Russell and Feldman Barrett, 1999; Yik, Russell and Feldman Barret, 1999). It is argued that various emotions can be plotted as points on the circumplex of these dimensions (Russell and Pratt, 1980; Russell and Lemay, 2000).

From an environmental psychology perspective, Mehrabian and Russell (1974) identified three underlying dimensions of emotions – 'Pleasure', 'Arousal' and 'Dominance' – in accordance with the basic three dimensions of meaning ('Evaluation', Activity' and 'Potency') identified by Osgood, Suci, and Tannenbaum (1957). According to Mehrabian (1972, 1996) the considerable generality of these semantic differential factors suggests that they represent the lowest common denominators of cognition and, thus, are associated strongly with affective responses. Mehrabian and Russell (1974) therefore proposed that the three basic emotional states of pleasure, arousal and dominance (abbreviated PAD) provide a general description of the emotions generated by an environment.

- <u>Pleasure-displeasure</u> (the emotional counterpart of 'evaluation') concerns the positive-negative quality of emotional states and refers more specifically to the degree to which a person feels happy or satisfied in a place.
- (2) <u>Arousal-non arousal</u> (the emotional correlate of stimulus 'activity) refers to a combination of physical activity and mental alertness and concerns the degree of stimulation caused by an atmosphere.

(3) <u>Dominance-submissiveness</u> (the converse of stimulus 'potency', with higher potency stimuli eliciting lower dominance responses) refers to the degree to which a person feels in control of a situation and feels to have influence over his/her surroundings and others. In other words, dominance concerns the degree to which he/she feels "unrestricted or free to act in a variety of ways" (Mehrabian and Russell, 1974a, p19).

Pleasure, arousal and dominance have been identified as basic dimensions of emotions in studies of verbal descriptions of emotion (Bush, 1973) and in studies of facial, postural and vocal expressions of emotion (e.g. Mehrabian, 1972; Mehrabian and Ksionzky, 1974) (Russell and Mehrabian, 1978). Pleasure, arousal and dominance accounted respectively for 27, 23, and 14 percent of variance in emotional responses to various everyday settings and show considerable independence¹ (Mehrabian and Russell, 1974a, p26). Furthermore, studies have identified physiological correlates of behavioural or verbally reported pleasure, arousal and dominance (cited in Russell and Mehrabian, 1976 & 1978: Olds, 1956; Heath, 1964 – for pleasure; Evans and Day, 1971; Thayer, 1967, 1970 – for arousal; and Pribram, 1962 – for dominance). Thus, according to Mehrabian and Russell (1974), these dimensions, rather than the types per se, drive subsequent consumer responses.

Russell and Mehrabian (1977) and Mehrabian (1995) provide evidence that these three nearly independent bipolar PAD dimensions are both necessary and sufficient to adequately define emotional states. Russell and Mehrabian (1977) demonstrate that most of the reliable variance in 42 verbal-report emotion scales, developed by other investigators, could be accounted for in terms of the PAD dimensions. Although Russell (1980) showed that the two dimensions, pleasure and arousal, could account for a substantial portion of the variance in self-reported affective states, he attributed part of the remaining unexplained variance to social orientation and control (as identified by Sjoberg and Svensson, 1976) and/or dominance (as identified by Mehrabian and Russell, 1974a). In a study with regard to emotional responses towards advertisements, Holbrook and Batra (1987) generated a 94-item scale, to measure 29 emotional indices. Factor analysis revealed three underlying emotion dimensions: pleasure, arousal and dominance. Results reached by Shaver et al. (1987; 2001), who used multidimensional analyses to study 135 emotion terms, also corroborated the PAD Emotion Model. Although they obtained two-dimensional (Evaluation and Intensity) and three-

¹ Absolute values of the intercorrelations among the three scales were .07, .03 and .18.

dimensional (Evaluation, Potency and Activity) solutions, they found the three dimensional representation of affect to be statistically justifiable and highly interpretable and clearly more informative than the two dimensional one. The three dimensional solution helped to differentiate between the separate basic-emotion categories, suggested by cluster analysis (Shaver et al., 1987, p1071; Shaver et al., 2001, p39-40).

Mehrabian (1995 – Study 3) demonstrates that the PAD scales, which accounted for 58% of the total variance (respective PAD cronbach alphas of .97, .89 and .80), exhibit near independence, with inter-correlations among the pleasure, arousal and dominance scales not exceeding .09 in absolute value. However, Mehrabian (1998, p6) notes: "the PAD scales have exhibited such near independence <u>only when tested against a broad-based and balanced sample of stimuli</u>, such as the sample of 80 situation descriptions given by Mehrabian, Wihardja and Ljunggren (1997). In contrast the PAD scales have tended to intercorrelate significantly when used to assess stimuli that represented limited realms of affective experience (e.g. Mehrabian and De Wetter, 1987; Valdez and Mehrabian, 1994). Difficulty in developing broad-based and emotionally-balanced samples of stimuli may thus help explain the temptation for other investigators to delete the dominance factor and to rely only on the pleasure and arousal factors, or rotations thereof, to describe emotions".

A more detailed presentation on the theoretical rationale and experimental foundations for the PAD Emotional State Model is given by Mehrabian (1995, 1996, 1997). Experimental data bearing on the validity of the PAD Emotion Scales were reviewed by Mehrabian (1995). Additional evidence of construct validity of the PAD Emotion Scales was obtained by Mehrabian and Blum (1997).

Thus, specific emotions can be visualized as points in a three-dimensional PAD-emotion space. 'Bored', for example, could be represented by its respective PAD-coordinates (-.65, -.62, -.33), representing a highly unpleasant, highly unaroused and moderately submissive state (Mehrabian, 1998).

The generality of the PAD Emotion Model is illustrated by dichotomizing each of the dimensions: pleasure (+P) versus displeasure (-P), arousal (+A) versus non-arousal (-A), and dominance (+D) versus submissiveness (-D). The resulting 2P x 2A x 2D emotion categories are illustrated in table 2-5 (Valdez and Mehrabian, 1994; Mehrabian, 1998) by the following

groups, derived from ratings of 240 emotional states on the PAD scales (Mehrabian, 1978; Russell and Mehrabian, 1977).

+	P +	A + D:	admired, bold, creative, powerful, vigorous
+	P +	A - D :	amazed, awed, fascinated, impressed, infatuated
+	Р-	A + D :	comfortable, leisurely, relaxed, satisfied, unperturbed
+	P -	A - D :	consoled, docile, protected, sleepy, tranquilized
-	P +	A + D :	antagonistic, belligerent, cruel, hateful, hostile
-	P +	A - D :	bewildered, distressed, humiliated, in pain, upset
-	Р-	A + D :	disdainful, indifferent, selfish-uninterested, uncaring, unconcerned
-	P -	A - D :	bored, depressed, dull, lonely, sad

Table 2-5:Eight PAD-emotion categoriesSource:Valdez and Mehrabian, 1994

The PAD-measure has been widely employed in marketing and consumer research for evaluating emotions in various contexts (Holbrook, Chestnut, Oliva and Greenleaf, 1984; Havlena and Holbrook, 1986; Holbrook and Batra, 1987; Olney, Holbrook and Batra, 1991).

Havlena and Holbrook (1986) compared the Plutchnik and the Mehrabian and Russell (M-R) schemes with respect to consumption experiences. Their results showed evidence in favor of the latter, concluding that the three PAD dimensions captured more information about the emotional character of the consumer experience than did Plutchnik's eight categories.

However, Havlena and Holbrook's study focused on emotions concerning the general consumption experience, not the specific shopping context.

2.5.3. EMOTIONAL RESPONSES TO THE SHOPPING EXPERIENCE

Retail environments have been shown to evoke emotional responses in shoppers who actually appear to perceive substantial differences in the affective qualities of different stores (Darden and Babin, 1994).

In line with the discrete emotions approach, Dawson, Bloch and Ridgway (1990) measured seven types of emotions in their study on the effects of emotions in the retail environment: relaxed, content, satisfied, happy, surprised, excited and rewarded. In other studies (e.g. Menon and Dubé, 2000; Klemz and Boshoff, 2001) only a subset of emotion-types is investigated. Comparing the environmental and emotional influences on the willingness-tobuy among small and large retailers, Klemz and Boshoff (2001) focused, for instance, on the store-induced emotions of empathy and assurance. Menon and Dubé (2000) focused on two negative emotions (anxiety and anger) and on two positive emotions (joy and delight), in order to empirically investigate their framework that proposes that greater satisfaction can be ensured when retail managers engineer effective salesperson responses to in-store customer emotions. d'Astous (2000) focused on a negative feeling that may be experienced in the retail setting: irritation.

Yoo, Park & MacInnis (1998) note that extant studies on emotions evoked in a retail context have used standard emotions lists, which were developed for the purpose of studying human emotions generally. While these lists are useful and potentially comprehensive, they may over-represent emotions that are not an integral part of the retail context (e.g., pity), and may under-represent emotions that are (e.g., frustration). It is generally preferred to tap emotions most relevant to the domain of inquiry (Smith and Ellsworth, 1985). Thus, Yoo, Park & MacInnis (1998) feel that developing a list of emotional experiences, tailored to the retail context may be desirable. They decided to focus on retail-specific emotional responses, because of the growing acknowledgement that specific contexts (e.g., the retail shopping context) may have specific effects on consumer outcomes like emotions (Park and Smith, 1989; Huffman and Houston, 1993; Ratneshwar and Shocler, 1991). Moreover, they were

concerned that general measures of human emotions may not adequately characterize the nature and range of emotional experiences encountered in the retail context (see also Smith and Ellsworth, 1985). For this reason 21 hours of ethnographic interviews were conducted, based on the responses of four informants. Through these ethnographic interviews, Yoo, Park & MacInnis (1998) identified emotions generated in the retail shopping environment that are not typically tapped by standard inventories of general human emotions. These retail-specific emotions can be classified into positive feelings (pleased, attractive, excited, contented, pride, satisfied) and negative feelings (ignored, anxious, nullified, displeased and angry) (Yoo, Park & MacInnis, 1998).

Several interesting observations can be made about these emotions. While Dawson et al. (1990) examined only positive emotions, the ethnographic interviews, conducted by Yoo, Park & MacInnis (1998), revealed roughly equivalent numbers of positive and negative emotions. This finding also differs from the items typically assessed using standard emotion typologies (see Holbrook and Batra, 1987; Edell and Burke, 1987), which typically have a lower proportion of negative to positive emotion items. Several retail-specific emotions identified through the interviews, do not show up on standard emotion typologies. A feeling of nullification, for example, is aroused when shoppers do not achieve what they want, such as when they plan to buy a product but cannot find it in the store. Notably, nullification reflects less frustration (which has a heavy anger component) than a feeling of deflation. Shoppers feel ignored when they are not treated well as shoppers. Pride is also an emotional response that has not been identified in previous work on store emotions. In this particular case, shoppers feel proud when they display their economic ability and social status through shopping (i.e. when they shop in a prestigious department store and purchase high-priced items). It is interesting to note that several of the emotions identified through the interviews are socially oriented. The extent to which shoppers feel ignored, proud or attractive, for example, may be driven by the social benefits or costs encountered in the retail environment. Nevertheless, the interviews were based on a very small sample of consumers and hence the items revealed from the interviews may not comprehensively represent the range of emotional experiences manifest in the shopping encounter. Yoo, Park and MacInnis (1998) subsequently asked 60 Korean students to indicate the extent to which they had experienced each of these 11 emotional responses during a specific shopping episode. Seven-point Likert scales were used as response scales (1 = not at all; 7 = very much). Only the pleasant versus unpleasant factor structure emerged, with two factors clearly representing respectively positive and negative emotions. Coefficient alphas for each factor were .89 for positive feelings and .92 for negative feelings. In a field study among 294 Korean consumers (Yoo, Park & MacInnis, 1998) the same factor structure emerged.

From a purely hedonic (pleasure-displeasure) perspective, Turley and Bolton (1999) developed a five-item atmospheric affect scale, with each item assessing the affective response on a different category of atmospheric elements. Testing and purification of the scale in a fast-food restaurant setting resulted in a final 4-item scale of atmospheric affect, however they note that future research will have to confirm whether the deleted item is redundant for other types of retail categories as well. The proposed items are: (1) "in general, the outside of this restaurant is pleasing to me", (2) "in general, the interior of this building is attractive", (3) "I did not like the arrangement and layout of this restaurant" (reverse scored), (4) "The signs and decorations used in this store are not pleasant" (reverse scored), (5) "In general, I felt comfortable with the people I encountered while eating today". Deleting item (3) resulted in a coefficient alpha score of .72.

Babin and Attaway (2000) operationalize atmospheric-based affect as two separate, negatively related dimensions: positive and negative affect (cfr. Mano and Oliver, 1993; Oliver, 1993; Bagozzi and Moore, 1994). They do not incorporate a separate arousal component, claiming that, while in general arousal can be either positive or negative, in a retail setting arousal indicators appear to take on rather consistent positive or negative meanings (Smith and Ellsworth, 1985). Babin and Attaway (2000) thus propose that, as 'excitement' in a retail setting represents a more pervasively positive reaction, there is no valid reason to include arousal as a separate dimension.

In their study concerning perceived control and the effects of crowding, Hui and Bateson (1991) only measure Mehrabian and Russell's (1974) pleasure-displeasure dimension, arguing that existing evidence reveals that this dimension tends to produce the most direct and strongest effect on approach-avoidance reactions. However, they did use Mehrabian and Russell's (1974) dominance scale combined with a scale of helplessness as indicator of perceived control. Moreover, they included 27 emotional terms in their questionnaire, 24 of which were adopted from Havlena and Holbrook (1986) to measure the eight basic emotion components identified by Plutchik (1980). One more component, 'comfort', operationalized by three items (calm, peaceful and relaxed), was added to this study. They argue that the 27

emotional terms, used in their study, form a perfect two-dimensional (pleasure-displeasure; arousing-unarousing) circumplex model, as proposed by Russell and Pratt (1980).

Donovan and Rossiter (1982) assessed shoppers' in-store emotional states using Mehrabian and Russell's (1974) PAD scale. However, they adapted the scale a little to the retail setting. Several of the original dominance items (e.g. in control – cared for, autonomous – guided, important – awed) were replaced by more appropriate retail-specific scales (restricted – free, crowded – overcrowded, important – insignificant). According to Donovan and Rossiter (1982) the dominance-submissiveness elicited by a store refers to the degree to which a person feels free to act in the store. Martineau (1958) specifically referred to customers feeling overwhelmed by counters and displays, which are built too high. According to Martineau (1958) customers commented *"They build up the display way over eye level so that things are staring at you and it bears down on you*", or "On entering that store the whole place gives you the feeling of crushing you." He also talks about the "wage-earner's wife" who "is not going to expose herself to the possibility of humiliation by shopping in the quality store, even if she has the money to buy something there". These feelings could very well be captured by the dominance-submissiveness dimension of emotions (Russell and Mehrabian, 1977).

Nevertheless, Donovan and Rossiter (1982) demonstrated that the dominance dimension failed to predict approach-avoidance behaviours in a retail setting. They found that dominance does not strongly affect in-store behaviour. Moreover, conceptualising affective responses to the environment, Russell and Pratt (1980) deleted the dominance dimension altogether, arguing that dominance requires a cognitive interpretation by the individual and is therefore not purely applicable in environments calling for affective response.

Thus for theoretical reasons (Russell & Pratt's, 1980 conceptualization) as well as for a lack of empirical support (e.g. Donovan & Rossiter, 1982; Greenland and McGoldrick, 1994), several researchers investigating atmospheric effects of the store environment using the Mehrabian-Russell model, have disregarded the dominance dimension and simply researched the pleasure and arousal dimensions (e.g. Donovan et al., 1994; Sherman, Mathur and Smith, 1997; Van Kenhove and Desrumaux, 1997). As a consequence, dominance has received little attention in atmospherics research.

Mehrabian and Russell's (1974) PAD measure, as adapted to the study of store atmospherics by Donovan and Rossiter (1982), was used to capture pleasure and arousal by Flicker and Speer (1990), Donovan et al. (1994), Van Kenhove and Desrumaux (1997)...

In their 1994 study, Donovan et al. observed that several of the six arousal items from the M-R measure may require further investigation for their applicability in retail store environment studies. Anecdotal feed-back from shoppers suggested that some had difficulty relating the items aroused-unaroused, jittery-dull, and frenzied-sluggish to feelings experienced while shopping. Respondent confusion may have led to their failure to observe an unambiguous arousal factor.

Also Van Kenhove and Desrumaux (1997) note, when conducting exploratory and maximum likelihood confirmatory factor analysis using LISREL, that several of the original items from the pleasure-arousal scale do not appear to be good indicators of the pleasure and arousal constructs. They argue (on p306) "that careful inspection of the items selected to test the pleasure/arousal approach/avoidance relationship is needed".

Gröppel-Klein (1998) argues that 'arousal' and 'pleasure' seem to converge in a retail setting as 'positive activation' appears to be implicitly registered in the arousal construct. Nevertheless, according to Gröppel-Klein (1998, p309) "an optimized store-interior presents interaction of activating and de-activating stimuli". Empirical studies by Bost (1987) and Gröppel (1991) reveal that consumers in a retail setting can also experience a pleasant state of low arousal (i.e. relaxation). Thus, arguing that also at the point-of-sale, too much arousal can be experienced as hectic and unpleasant, Gröppel-Klein (1998) contends that the emotional dimension of "relaxation" experienced in the shopping environment should not be neglected in further studies. In her study concerning the influence of dominance perceived at the point of sale, Gröppel-Klein (1998) uses a modified version of Mehrabian and Russell's (1974) PAD conceptualization, including the dimensions of 'desire/pleasure' (implicitly a state of positive activation, captured by the items 'joyful' and 'in a good mood'), 'relaxation' (a positive state of low arousal, tapped by the items 'relaxed' and 'not hectic') and 'dominance' (incorporating the items 'superior', 'secure' and 'free'). These three dimensions were found to impact price assessment, assessment of service and assortment and reactions towards the store.
By means of several general human emotion scales (Plutchik, 1980; Izard, 1977; Mehrabian and Russell, 1974), Machleit and Eroglu (2000) also evaluated the nature of emotional responses to the shopping experience. In accordance to Darden and Babin's (1994) call for broadening the range of stores examined in order to show the full spectrum of affective qualities possible across all types of retailers, Machleit and Eroglu (2000) provide a descriptive account of emotions consumers feel across a wide variety of store types. The results of their study indicate that the broad range of emotions felt in the shopping context vary considerably across retail environments. Machleit and Eroglu (2000) also empirically compare the three emotion measures most frequently used in consumer research to determine which best captures the various emotions shoppers experience in retail environments, specifically, Izard's (Izard, 1977) differential emotion scale, Plutchik's (Plutchik, 1980) basic emotion scale, and Mehrabian and Russell's (Mehrabian and Russell, 1974) tri-dimensional scale of pleasure, arousal, and dominance. Their results show that the two emotion-specific scales (Izard, 1977 and Plutchik, 1980) outperform the dimensional PAD scale (Mehrabian and Russell, 1974) in capturing the richness of customers' emotional experience of shopping. However, they do also suggest that across a wide variety of shopping episodes it is not a good idea to combine such discrete emotion types into summary positive and negative affect factors. They recommend that researchers should use caution in constructing such summary dimensions and suggest that confirmatory factor analysis can aid in uncovering an appropriate underlying factor structure. Machleit and Eroglu (2000) do recognize that the M-R measure also has its strengths because it includes an arousal component, which is not adequately represented in the other scales. They also praise the dominance dimension as another benefit of the M-R measure. They argue that this dimension may be particularly relevant "in studies where control over one's environment is at issue, such as retail crowding, waiting time and so forth". Nevertheless, in their study, Machleit and Eroglu (2000) found the M-R measure to account for variance in shopping satisfaction only in terms of a pleasure/displeasure response.

Machleit and Eroglu (2000) emphasize that it is of particular interest which tangible or intangible environmental qualities instigate which types of emotional responses. Yalch and Spangenberg (1988) found, for instance, that environmental music conditions did not significantly affect pleasure nor dominance. They found only arousal to be significantly affected by store music.

Despite of the ambiguity concerning the arousal construct in a retail setting, atmospheric researchers generally acknowledge its potential importance (Donovan et al., 1994; Gröppel-Klein, 1997, 1998; Van Kenhove and Desrumaux, 1997; Gröppel-Klein and Baun, 2001).

As store-evoked arousal had always been measured by means of verbal self-report scales, causing rather ambiguous results, Gröppel-Klein and Baun (2001) conducted an experimental pilot-study using electrodermal activity as arousal indicator. Thus investigating the role of customers' arousal for retail stores more profoundly, they conclude that verbal scales may not be an appropriate measure to capture arousal. An empirical investigation was conducted in the fruit and vegetable department of two Austrian grocery stores. The experimental store (with enhanced atmosphere) was found to evoke higher arousal (electrodermal activity - EDA) than the control store. The pleasing atmosphere of the experimental store also led to a longer duration of shopping in the fruit and vegetable section than in the control store. In order to determine the valence of the evoked emotions, a German version of the positive dimensions of Izard's (1994) Differential Emotion Scale was used, including the dimensions 'interest', 'surprise' and 'joy'. Physical arousal (EDA) appeared to be only significantly correlated with the item 'glad' of the 'joy' dimension. The experimental store was rated as more 'varied', with more 'irresistible products' that 'suddenly caught the shopper's eye', making him/her feel more 'astonished', 'happy' and 'amazed', than in the control store. Moreover, the intentions 'to return to the store' and to 'recommend the store to others' were higher in the experimental store, as well as the 'actual amount of money spent' in the store. Results show that recording EDA at the point-of-sale is a practicable way to measure arousal in a valid manner and, thus, can be used to validate verbal arousal scales. Nevertheless, as EDA cannot reveal the hedonic tone of an arousing situation, verbal control of perceived emotions using a standardized questionnaire remains necessary. Moreover, EDA recording has a major limitation with regard to the time needed to collect satisfying sample sizes ("not every consumer is willing to be connected to electrodes") and to screen the output manually (Gröppel-Klein and Baun, 2001).

As 'emotional states' refer to transitory conditions (Mehrabian, 1996), which can vary substantially and rapidly, atmospheric researchers agree that it is difficult to assess emotions induced by the store environment. Donovan and Rossiter (1982) argue that the emotions engendered by the store atmosphere are transient and difficult to recall and verbalize. As Dawson et al. (1990) pointed out in their study, assessment of feeling states within the store

environment may reflect feeling states brought *to* the environment rather than induced *by* the environment. No doubt, emotions measured in the store reflect some combination of these. More research should be undertaken with regard to the pleasure and arousal dimensions prior to entering the store, because some of the motives for spending may have been to relieve negative emotions (cf. Mano, 1999). Therefore, Donovan et al. (1994) suggest that future studies should measure pleasure and arousal dimensions prior to entering the store as well as some time after entering the store. Emotions actually have been measured at different episodes of the shopping experience: before entering the store (Swinyard, 1993), a few minutes after entering the store (Donovan et al., 1994; Van Kenhove and Desrumaux, 1997), after considerable time in the store (Dawson, Bloch and Ridgeway, 1990) or after the shopping trip (Yalch and Spangenberg, 1993).

2.6. THE RESPONSE: APPROACH-AVOIDANCE BEHAVIOURS IN SHOPPING ENVIRONMENTS

2.6.1. APPROACH-AVOIDANCE BEHAVIOURS : GENERAL CONCEPTUALIZATION

The response to any environment can according to Mehrabian and Russell (1974) be subsumed in the two basic categories of "approach" and "avoidance" behaviours (Wundt, 1905). Approach or convergence means that an individual reacts positively to the environment, whereas avoidance is characterized by an aversion to the environment. Mehrabian and Russell (1974) posit that the emotional states (pleasure, arousal and dominance) engendered by the environment mediate approach or avoidance behaviours, such as the desire to affiliate with others in the setting, the desire to stay in or escape from the setting, and the willingness to spend money and consume there (Mehrabian, 1979; Mehrabian and Riccioni, 1986; Mehrabian and de Wetter, 1987; Mehrabian and Russell, 1974; Russell and Mehrabian, 1976, 1978).

Approach-avoidance behaviours as reactions to an environment can be framed within the more general notion of "action tendencies", referring to "a readiness to engage in or disengage from interaction with some goal object", including "impulses of 'moving towards', 'moving away', and 'moving against'" (Frijda, Kuipers and ter Schure, 1989, p213). In this regard some theorists (e.g. LeDoux, 1996) also maintain that "action tendencies" in general are automatic "prewired" responses connected to emotions (Bagozzi, Gopinath and Nyer, 1999).

Approach-avoidance behaviour as response to a situation was conceptualized and measured by Mehrabian and Russell (1974) and Russell and Mehrabian (1976, 1978) as a generic single uni-dimensional bipolar construct, composed of intercorrelated factors of preference, exploration, work performance and affiliation. Approach behaviours, thus, are seen as positive responses to an environment such as a desire to stay in a facility and explore it, the desire to work in the situation and the desire to affiliate there (i.e. seek contact with others). Avoidance behaviours include not wanting to stay or to spend time looking around or exploring the place, not wanting to solve problems or work in the situation and not wanting to affiliate there with others. In fact Mehrabian and Russell (1974 - experiments 1, 2 and 3) found only 'affiliation' to constitute a different subfactor within the approach-avoidance construct. The desire to stay, work in and explore the situation were found to be highly intercorrelated and appeared to define a single 'preference' subfactor of approach/avoidance. Both approach/avoidance subfactors, 'affiliation' and 'preference', also appeared to be significantly correlated. Nevertheless, it was suggested to analyse the four approach-avoidance indices separately when the object of a study is to obtain detailed answers for the particular environmental influences on each of them (Mehrabian and Russell, 1974, p144). This was also the approach taken by the authors in subsequent analyses. Based on their empirical findings, Russell and Mehrabian (1976) also conclude 'approach toward the setting itself' to be a separate factor from 'interpersonal approach' or the 'desire to affiliate'. Thus, Mehrabian and Russell (1974) and Russell and Mehrabian (1976, 1978) empirically demonstrated approach/avoidance to constitute a single 'bipolar' construct, composed of two interrelated subfactors 'preference' and 'affiliation'.

In contrast, Foxall (1990, 1997, 2000) suggests that approach and avoidance do not constitute a single dimension of consumer behaviour. In his argumentation he refers to Alhadeff (1982) who pointed out that consumer behaviour in any given situation is the outcome of two opposing learning histories: "the strength of approach is a function of the individual's learning history with respect to prior approach behaviour and their consequences, while the strength of avoidance/escape is a function of his or her history with respect to prior avoidance/escape responses and their consequences. The strength of consumer behaviour is, therefore, a vector quantity, which cannot be represented psychometrically by a single continuum from escape/avoidance to approach." (cited from Foxall, 1997, p514).

In his (1997) study, Foxall measured approach/avoidance by six of Mehrabian and Russell's eight statements for these items (those on thinking out a difficult task and working in the situation were deemed inappropriate). Selected "*approach*" statements were: 'How much time would you like to spend in this situation?', 'Once in this situation, how much time would you enjoy exploring around?', 'To what extent is this a situation in which you would feel friendly and talkative to a stranger who happens to be near you?'. Approach responses were scored from 1 (minimal approach tendency) to 7 (maximal approach tendency). Selected "*avoidance*" items were: 'How much would you try to leave or get out of this situation?', 'Is this a situation in which you might try to avoid other people or avoid having to talk to them?'. Avoidance responses were scored from 1 (minimal tendency to avoid/escape from the setting)

to 7 (maximal tendency to avoid/escape). Foxall found, as expected, *approach* and *avoidance* to be separate dimensions, which are negatively related to each other (Pearson correlation coefficient = -0.64). Proposing that emotions of pleasure, arousal and dominance would each exert independent influences on approach and avoidance responses, Foxall (1997) concludes: *"The separate conceptualisation and measurement of approach and avoidance has led to the finding that these behaviours are uniquely related to emotional response patterns"* (Foxall, 1997, p520). In the following chapter we will focus more specifically on the emotional determinants of approach and avoidance behaviours.

Coroborative of these findings by Foxall (1997) on a small sample are the results from a subsequent empirical study on 561 consumers by Foxall and Greenley (2000). As in the previous study by Foxall (1997), the respondents rated eight consumer situations, carefully chosen according to the behavioural perspective model, providing data on emotional responses and approach and avoidance intentions for 4.488 consumer situations. Besides a score for approach and one for avoidance, in this study also a composite measure termed 'aminusa' was calculated (i.e. approach minus avoidance, defined as the mean difference between approach and avoidance scores). This was done for the purpose of comparison with Mehrabian and Russell's approach-avoidance measure. However, Foxall and Greenley (2000) found that 'aminusa' (comparable to the M-R bipolar approach/avoidance measure) appeared to follow a similar pattern as their simple approach measure, indicating that respondents' approach-avoidance, measured as a single bipolar variable, tends towards the approach pole. They propose that the equivalence of their results for approach and those for aminusa suggests that the continued employment of the 'aminusa' variable is redundant. They conclude: "Aminusa, essentially Mehrabian and Russell's bipolar approach-avoidance variable, adds nothing to the explanation that is not provided by approach. However, the approach/avoidance distinction appears useful since the avoidance variable does add information to the analysis and should be retained (Foxall and Greenley, 1999)" (Foxall and Greenley, 2000, p55).

In fact, the idea that behaviour consists fundamentally of two sorts of motivations or action tendencies: appetitive or approach and aversive or avoidance (withdrawal), which are managed by two separate self-regulatory systems, is commonly tied to the work of Miller and Dollard (Miller and Dollard, 1941; Miller, 1944; see also Lewin, 1935, p62) and has reemerged over the past two decades (Carver, Sutton and Schreier, 2000). Approach and

avoidance tendencies are presumed to be managed by different structures in the nervous system (e.g. Konorski, 1948; Miller, 1944; Schneirla, 1959).

2.6.2. APPROACH-AVOIDANCE BEHAVIOURS IN A RETAIL SETTING

In their literature review, Turley and Milliman (2000) note that over the last thirty years of research, atmospheric effects have been measured on a wide variety of different dependent variables. The most widely examined dependent variables in experimental studies of the retail atmosphere are sales, time in the environment and approach-avoidance behaviour (Turley and Milliman, 2000). Some studies have used multiple dependent variables.

Donovan and Rossiter (1982) adapted Mehrabian and Russell's (1974) approach/avoidance scale to measure respondents' intentions to behave in the store. Transposed to a retail environment, Donovan and Rossiter (1982) distinguish four aspects of approach/avoidance behaviours:

- 1- a desire to stay in (approach) or to get out of the store (avoidance)
- 2- a desire to explore or look around in *(approach)* the store environment or to avoid moving through or interacting with the environment *(avoidance)*
- *3-* a desire to communicate with others in the store (sales personnel, floor staff, other clients) (*approach*) or to avoid this and to ignore communication attempts from other people (*avoidance*)
- 4- the degree of enhancement *(approach)* or hindrance *(avoidance)* of shopping performance (e.g. spending money, repeat-purchasing) and satisfaction with shopping performances

The 8 resulting approach-avoidance items (see table 2-6) were expected to indicate four underlying dimensions: time spent (items 1 and 2), affiliation (items 3 and 4), affect (items 5, 6 and 7) and money spent (item 8).

Affect:

- (1) Do you like this store environment?
- (2) Would you enjoy shopping in this store?
- (3) Would you avoid ever having to return to this store? (*)

Time dimension:

- (4) How much time would you spend browsing in this store?
- (5) Would you want to avoid looking around or exploring this environment? (*)

Affiliation:

- (6) Is this a place where you might try to avoid other people, and avoid to talk with them? (*)
- (7) Is this a place in which you would feel friendly and talkative to a stranger who happens to be near you?

Spending Behaviour:

(8) Is this the sort of place where I might end up spending more money than you originally set out to spend?

Table 2-6: Donovan and Rossiter's (1982) approach/avoidance scale

An exploratory principle component factor analysis yielded only one factor, which appeared to capture approach-avoidance. The authors note that this uni-dimensional approach factor should be interpreted more in the Fishbein and Ajzen (1975) sense of "attitude towards the act" than of intentions per se, because the factor seems highly loaded with affect. In accordance with Russell and Mehrabian's findings that affiliation seems to constitute a separate factor, not only this one factor, but the first three factors of the solution were explored (note that the two additional factors had eigenvalues < 1; these three factors accounted together for 78.8% of the variance). It was found that the affiliation items and the time and money items were in fact somewhat different from the other approach-avoidance items. The authors justify focusing on the various subscales (see table 2-7 for more details), by noting that the uni-dimensionality of approach-avoidance seems to be due to its higher-order factorial position relative to several sub-factors. For example, while their spending intentions measure is significantly related to Mehrabian and Russell's approach-avoidance responses (individual correlations with spend ranged from .35 to .48, all p<.01), the shared variance is quite low.

Based on the Donovan and Rossiter 1982 study, Van Kenhove and Desrumaux (1997) expected the 8 approach-avoidance items to indicate four underlying dimensions: time spent,

affiliation, affect and money spent. Note that they did not include the item 'I would avoid ever having to return to this store' in the 'affect' sub-dimension, as Donovan and Rossiter did, but rather in the 'money spent' dimension (see table 2-7 for more details). Exploratory factor analysis revealed however only two dimensions, with two of the reversed-scored items loading on the second dimension, which could be supportive of the existence of a separate avoidance dimension. Yet, this has not been contemplated by the authors. Conducting a maximum likelihood confirmatory factor analysis using LISREL 8, Van Kenhove and Desrumaux (1997), found the fit for the presumed four-factor solution to be unacceptable. Actually, some items had to be dropped because of very low factor-loadings. After dropping several items, eventually a satisfactory three factor-solution was achieved, indicative of the 'time', 'affiliation' and 'affect' sub-dimensions. Note that for the affiliation dimension only one item was retained.

Bateson and Hui (1991) considered in their study only the 'desire to stay' and the 'desire to affiliate' dimensions, based on Mehrabian and Russell (1974) and treated these together as one global approach/avoidance construct.

Also capturing approach/avoidance intentions by means of the 8 items proposed by Donovan and Rossiter (1982), Bellizzi and Hite (1992) discussed effects for each of the items separately. They considered the fact that in their study no effects could be discerned for both 'affiliation' items as indicative of the existence of a separate 'affiliation' factor. Yet, an average composite score was calculated afterwards, treating approach/avoidance as a uni-dimensional construct ($\alpha = .86$).

Mattilla and Wirtz (2001) also treated the 8 approach-avoidance items proposed by Donovan and Rossiter, as a bipolar uni-dimensional scale ($\alpha = .78$).

Thus, in a retailing context, traditional approach–avoidance studies have always considered approach and avoidance to be the opposite poles in a general approach-avoidance construct with several correlated underlying dimensions, such as exploration (time spending), affect and affiliation (Mehrabian and Russell, 1974; Russell and Mehrabian, 1976, 1978; Donovan and Rossiter, 1982; Bellizzi and Hite, 1992; Van Kenhove and Desrumaux, 1997, Matilla and Wirtz, 2001).

In accordance with Donovan and Rossiter (1982), but focusing on actual behaviour, rather than intentions, Sherman and Smith (1986) and Sherman et al. (1997) included the following items in their study: (1) number of items purchased

- (2) amount of time spent in the store
- (3) actual amount of money spent in the store
- (4) whether the shopper liked the store environment

In an extension of their 1982 study, Donovan et al. (1994) also measure actual shopping behaviour, rather than intentions. In this study pre-measures of estimated spending and time in the store are compared to post-measures of actual spending and actual time spent in the store. It is argued that post-only measurement of these behavioural dependent variables may be subject to memory error and post-hoc attributions.

Babin and Darden (1995) investigated the '*level of resources expended while shopping*' and the '*perceived shopping value*'. 'Perceived shopping value' was operationalized by means of two dimensions capturing "*an assessment of the total worth of shopping activity in terms of utilitarian and hedonic shopping value*", where 'utilitarian value' refers to task related worth and 'hedonic value' implies worth found in the shopping experience itself, aside from any task-related motives. Thus, 15 items were used to assess perceived utilitarian and hedonic shopping value).

Babin and Attaway (2000) introduce another measure of behavioural consequences of variations in store atmospherics, namely '*customer share*', which is a multi-item scale that captures the extent of temporal and economic resources devoted by a customer to a given store, in proportion to those resources devoted to the category as a whole. Moreover, they hypothesize shopping value (which in itself can serve as a reward) to mediate the relationship between affect and purchase behaviour.

Table 2-7: Studies measuring approach/avoidance behaviour			
Authors	Items used to measure approach/avoidance	Findings	
Mehrabian and Russell			
Experiment 1: 103 University under-gradu- ates' reactions to 8 (of 40)	Experiment 1: <u>Preference for the situation:</u> - How much do you like this situation? - How much do you prefer this situation? - How much would you seek out or try to find this situation?	A generic concept of approach-avoidance behaviour as response to a situation was hypothesized, composed of intercorrelated factors of preference, exploration, work performance and affiliation.	
(i.e. 824 sets of situations)	 Once in the situation, how much would you explore it? Once in the situation, how much would you like to leave, or get out of it? (*) Desire to solve problems in the situation: 	Experiment 1: Exploratory principle component factor analysis with oblique rotation on eight approach-avoidance items revealed three factors as indicated: a 'general preference'	
	 To what extent is this situation a good opportunity to think out some difficult problem you have been assigned in class? If the problem is an easy one, would you use this situation as an opportunity to solve the problem. 	factor, a 'problem solving' factor and an 'affiliation' factor. The former factors were significantly correlated r=0.36, p<0.05). 'Affiliation' appeared to be significantly correlated with 'preference' (r=0.22, p<0.05), but not with	

'problem solving' (r=0.17, p>0.05).

- Desire to affiliate in the situation:
 To what extent is this a situation in which you would be friendly and talkative to a stranger who happens to be near you?
 In this situation, would you initiate a conversation just to be friendly?

8-point scales, scored from 0 'not at all' to 7 'extremely so'.

Mehrabian and Russell	
(1974, chapter 8)	Experiment 2 and 3:
	Approach-avoidance items were redesigned to include equal numbers of
Experiment 2:	positively and negatively worded items.
163 subjects' responses to 20	
(of 65) verbally described	Desire to stay in the situation: ('Preference')
situations (i.e. 3260 sets of	 How much time would you like to spend in this situation?
situations)	- How much would you try to leave or get out of this situation? (*)
	Desire to explore the situation: ('Preference')
Experiment 3:	- Once in this situation, how much would you enjoy exploring around?
214 subjects' reactions to 6	- How much would you like to avoid any looking around or exploration
(of 55) verbally described	of this situation? (*)
situations (i.e. 1284 sets of	Desire to work in the situation: ('Preference')
situations)	- To what extent is this situation a good opportunity to think out some
	difficult task you have been working on?
	- How much would you dislike having to work in this situation? (*)
	Desire to affiliate: ('Affiliation')
	- To what extent is this a situation in which you would feel friendly and
	talkative to a stranger who happens to be near you?
	- Is this a situation in which you might try to avoid other people, avoid
	having to talk to them? (*)
	8-point scales, scored from 0 'not at all' to 7 'extremely so'.

The first item, involving time, could be rated as follows: 0 'none', 1 'a few expected significantly intercorrelated. The desire to minutes', 2 'half an hour', 3 'one hour', 4 'a few hours', 5 'a day', 7 'many, affiliate however, was identified as a separate factor in all many days'.

Experiment 2:

Exploratory principle component factor analysis actually revealed two factors with eigenvalues exceeding unity, accounting for 61% of the total variance: the first three categories of approach-avoidance were found to be highly intercorrelated and define a single factor called "preference", which has a low but significant correlation with the "desire to affiliate factor" (r=0.16, p<0.05).

Experiment 3:

Similar to the previous findings from experiment 2 an exploratory principle component factor analysis with oblique rotation revealed again two factors with eigenwaarde exceeding unity, accounting for 61% of the total variance, with similar factor loadings.

The composite "preference" factor was however also significantly correlated with the "affiliation" factor (r=0.23, p<0.01)

Conclusion:

All the behaviours that were assumed to be part of a generic approach-avoidance reaction to situations were as three experiments. In experiment 1 "problem solving" also popped up as a separate factor.

Therefore it was suggested that "the four approachavoidance indexes (as used in experiments 2 and 3) should be analyzed separately when the object of a study is to obtain detailed answers for the particular environmental influences on each of these" (Mehrabian and Russell, 1974, p144). This was also the approach taken by the authors in subsequent analyses. Therefore composite scores were calculated for each index, after reversing signs where appropriate.

Russell and Mehrabian (1976)

44 undergraduate students' responses to settings represented by a sample of 60 photographic slides of environments, selected according to a Desire to explore: 2x3x2 factorial design, with 2 levels of pleasantness, three levels of arousing quality and two levels of dominanceeliciting quality, as rated by Desire to affiliate: another group of subjects

Desire to stay:

- How much time would you like to spend, once you are back in these desire to drink items. settings?
- How much do you like the way you are feeling at this time?
- How hard would you try to avoid ever having to return? (*) -

- How much would you enjoy going back to these settings and exploring intercorrelated -.71. The two desire to affiliate items around?
 - How much would you try to avoid going back to look around or to measures were thus simply summed, after reversing signs explore these settings? (*)

- To what extent is this a time you would feel friendly and talkative to a Desire-to-stay and desire-to-explore appeared to be a stranger who happens to be near you?
- Is this a time when you might try to avoid other people, avoid having to coefficient of .88. As anticipated desire-for-affiliation was talk to them? (*)

Desire to smoke:

- At this time, would you feel like smoking a cigarette? -
 - Would you refuse an offer of a cigarette to smoke? (*)

Desire to drink:

-

- Would you feel like having a drink (something alcoholic)? -
- Do you think that "It's just not the right time to have an alcoholic is a separate factor from "interpersonal approach", or drink"? (*)

measured on nine-point scales ranging from "not at all" to "extremely so", except for the desire to stay item, concerning amount of time, which was measured on a nine-point scale, ranging from none, a few minutes, half an hour, an hour, a few hours, a day, a few days, many days, many months.

In this discussion we disregard the desire to smoke and the

Desire to stay consisted of three measures. The negatively worded item correlated -.59 with each of the two positively worded items. The positively worded items correlated .72. The two desire to explore items intercorrelated -.54. Responses to items within the where appropriate.

single approach-avoidance factor, with a correlation positively, but only moderately, related to exploration (correlation coeff.=.36) and staying (correlation coeff.=.37).

Based on their empirical findings Russell and Mehrabian (1976) conclude that "approach toward the setting itself" the desire to affiliate.

Russell and Mehrabian (1978)

Study 1

200 university students' reactions to 18 photographic slides of different physical settings, selected according to a completely crossed factorial design with pleasantness, arousing quality and domi- Desire to affiliate nance-eliciting quality of the setting (rated by other subjects) as orthogonal factors: 3Px3Ax2D (90 of 320 settings were selected (5

Study 2

310 university undergraduates' reaction to 9 photographic slides of different physical settings, selected according to an equally spaced grid with pleasantness and arousing quality of the setting as orthogonal dimensions (rated by other subjects), dominance was held average : PxA (90 of 320 settings were selected); (i.e. 2790 observations)

Approach-avoidance

- How much time would you like to spend in this place? -
- How much do you like this place?
- How much would you want to explore around in this place?
- setting? (*)
- How hard would you try to avoid ever having to return to this place? (*) The two 'desire to affiliate' items correlated -.57.

- To what extent is this a time you would feel friendly and talkative to a stranger who happens to be near you?
- Is this a time when you might try to avoid other people, avoid having to 'desire to affiliate' were in turn found to correlate .50. talk to them? (*)

slides per cell) for the 18 Scored on an eight-point scale from "not at all" to "extremely so", except for Among the 5 approach-avoidance items, the obtained cells); (i.e. 3600 observations) the 'time' item, which was rated on a 9-point scale: none, a few minutes, half an inter-correlations ranged from .56 to .75. The two desire to hour, one hour, a few hours, a day, a few days, many days, many months.

Study 1:

Among the 5 approach-avoidance items, inter-correlations ranged from .60 to .79. Since the standard deviations were similar (ranging from 2.0 to 2.3), these items were simply How much would you want to avoid looking around or exploring this summed to form a composite 'approach-avoidance' score.

> (standard deviations of 1.9 and 2.2) were also summed to form a composite 'desire-to-affiliate' score.

> These composite measures 'approach-avoidance' and

Study 2:

affiliate items correlated -.58. Composite scores were used for each factor. The composite 'approach-avoidance' factor correlated .43 with the composite 'desire to affiliate' factor.

Donovan and Rossiter (1982)

Items to measure approach-avoidance intentions were adapted from Mehrabian An exploratory principle component factor analysis with *and Russell to fit retail shopping intentions* An exploratory principle component factor analysis with varimax rotation on the eight approach-avoidance and

30 graduate students' responses to 2 to 3 different Affect:

actual retail environments

(i.e. 66 observations)

- (1) Do you like this store environment?
- (2) Would you enjoy shopping in this store?
- (3) Would you avoid ever having to return to this store? (*)

Time dimension:

- (4) How much time would you spend browsing in this store?
- (5) Would you want to avoid looking around or exploring this environment? (*)

Affiliation:

-

- (6) Is this a place where you might try to avoid other people, and avoid three factors of this solution were explored (these to talk with them? (*) accounted for 78.8% of the variance): it was found that the
- (7) Is this a place in which you would feel friendly and talkative to a affiliation items and the time and money items were in stranger who happens to be near you? fact somewhat different from the other approach-

Spending Behaviour: (single item)

- (8) Is this the sort of place where I might end up spending more money than you originally set out to spend?

An exploratory principle component factor analysis with varimax rotation on the eight approach-avoidance and spending items yielded only one factor with an eigenvalue greater than one, which appears to capture approachavoidance. The authors note that this uni-dimensional approach factor should be interpreted more in the Fishbein and Ajzen (1975) sense of "attitude towards the act" than of intentions per se, because the factor seems highly loaded with affect.

In accordance with Russell and Mehrabian's findings that affiliation seems to constitute a separate factor, the first three factors of this solution were explored (these accounted for 78.8% of the variance): it was found that the affiliation items and the time and money items were in fact somewhat different from the other approachavoidance items.

The following dependent approach-avoidance measures were calculated for further analyses:

- <u>APR-AVD</u>: average approach items (1-7) excluding the spend item (α =.88)
- <u>Spend:</u> single spend item (8)
- <u>Affect:</u> average of the three highest loading items on the first factor (like, enjoy, return) (α =.90)
- <u>Time:</u> average of items (4) and (5) (α =.67)
- <u>Affiliation</u>: average of items (6) and (7) (α =.72)

The authors justify focusing on the various subscales, by noting that the uni-dimensionality of approach-avoidance seems to be due to its higher-order factorial position relative to several subfactors. For example, while their spending intentions measure is significantly related to Mehrabian and Russell's approach-avoidance responses (individual correlations with spend ranged from .35 to .48, all p < .01), the shared variance is quite low.

Hui and Bateson (1991)

115 adult's reactions to a hypothetical person in more and less crowded bank and bar settings

Desire to stay Desire to affiliate (based on Mehrabian and Russell, 1974) Desire to stay and desire to affiliate were treated as one global approach/avoidance construct.

Bellizzi and Hite (1992)

experiment 2:

107 graduate students' reactions to projected furniture store interior drawings with manipulated background colours.

8 approach-avoidance items cfr. Donovan and Rossiter (1982) Effects were discussed separately on each of the 8 approach-avoidance items. Five of the eight items generated significant effects. Both affiliation items were among those that did not generate a significant effect, which may be indicative of a separate 'affiliation' factor

Afterwards, an average composite score was calculated $(\alpha = .86)$ for approach-avoidance, treating it as a unidimensional bipolar construct.

Babin and Darden

(1995)

Resource expenditures: (i.e. approach-avoidance behaviour)

shoppers intercepted the resource expenditure scale was adapted from approach-avoidance scales expended while shopping (based on approach-avoidance 130 while shopping at one of 10 commonly used in environmental psychology (Donovan and Rossiter, 1982; scales). stores (including a variety of Mehrabian and Russel, 1974) and included Likert items assessing

stores) in a major regional

mall

- time expenditure
- resource expenditure -
- social interactions

+ two behavioural items (normalized to a common metric) assessing

- amount of money spent -
- actual time spent shopping -

Babin an Darden (1995) investigated the level of resources

Confirmatory factor analysis demonstrated the unidimensionality of the resource expenditures scale, with construct reliability of .73 and 36 % of variance extracted, item reliabilities ranging from .19 to .59 and standardized factor loadings ranging from 0.44 to 0.77.

Van Kenhove and Desrumaux (1997)	
364	first-time-visitors'
store	reactions to one of s

in-

store reactions to one of seven <u>Af</u> large retail outlets in Belgium (two clothing stores, three furniture stores and three garden centers)

Time dir	nension:
-	(1) I like to spend much time browsing in this store
-	(2) I want to avoid looking around or exploring this store (*)
Affiliatio	<u>Dn:</u>
-	(3) This is a place where I try to avoid other people, and avoid to tall
	with them (*)
-	(4) This is a place in which I feel friendly and talkative to stor
	personnel who happens to be near me
Affect:	1 11
_	(5) I like this store environment
-	(6) I enjoy shopping in this store
Spend:	
	(7) I would avoid ever having to return to this store (*)
_	(8) This is the sort of place where I might end up spending more more

- (8) This is the sort of place where I might end up spending more money than I originally set out to spend

8 approach-avoidance items

cfr. Donovan and Rossiter (1982)

Measured on 5-point Likert scales

Van Kenhove and Desrumaux note that according to the Donovan & Rossiter (1982) study, the approach-avoidance items were expected to indicate four underlying dimensions: time spent, affiliation, affect and money k spent.

An exploratory factor analysis resulted in a two factor re solution with all items, except (3), (7), and (8) loading on the first factor. The loadings were relatively low for items (3) and (8) (below .40 on the second factor).

A maximum likelihood confirmatory factor analysis (LISREL 8) revealed that the fit for the four-factor solution, as presented by Donovan and Rossiter (1982) was not tenable, even after deleting items (3) and (8) with very low factor loadings (below .25). (χ^2 =11.44, df=5, p=0.043). Careful examination of the standardized residuals exceeding 2.58 resulted in dropping item 7. This resulted in a three-factor solution: time (items 1 and 2), affiliation (item 3) and affect (items 5 and 6). With a good fit (χ^2 =2.48, df=3, p=0.48, RMR=0.0083, AGFI=0.99). Reliability for the time dimension = .87, for the affect dimension = .97, for the affiliation dimension only one item was retained, with regard to spending all items were dropped.

Foxall (1997)

Exploratory study involving 27 students' responses to 8 verbally described consumer environments, selected by the systematic application of the behavioural perspective mo- Avoidance: del (BPM) (i.e. 216 consumer situations)

Approach:

- -How much time would you like to spend in this situation?
- Once in this situation, how much time would you enjoy exploring consumer behaviour. around?
- To what extent is this a situation in which you would feel friendly and related to each other (Pearson correlation coefficient = -_ talkative to a stranger who happens to be near you?

- How much would you try to leave or get out of this situation? -
- How much would you try to avoid any looking around or exploration in *emotional response patterns*" (Foxall, 1997, p520). _ the situation?
- Is this a situation in which you might try to avoid other people or avoid having to talk to them?

Approach responses were scored from 1 (minimal approach tendency) to 7 (maximal approach tendency); Avoidance responses were scored from 1 (minimal tendency to avoid/escape from the setting) to 7 (maximal tendency to avoid/escape).

Foxall (1990, 1997, 2000) suggests that approach and avoidance do not constitute a single dimension of

As expected approach and avoidance were negatively 0.64). However, "the separate conceptualisation and measurement of approach and avoidance has led to the finding that these behaviours are uniquely related to

Foxall and Greenley (2000)

561 consumers' reactions to 8 verbally described consumer situations, carefully chosen according to the behavioural perspective model (BPM), (i.e. 4.488 consumer situa- Avoidance: tions)

- Approach:
 - How much time would you like to spend in this situation? _
 - Once in this situation, how much time would you enjoy exploring calculated (i.e. approach minus avoidance, defined as the around?
 - To what extent is this a situation in which you would feel friendly and This was done for the purpose of comparison with talkative to a stranger who happens to be near you?

- How much would you try to leave or get out of this situation? -
- How much would you try to avoid any looking around or exploration in as approach, indicating that respondents' approach-the situation?
- Is this a situation in which you might try to avoid other people or avoid the approach pole. They propose that the equivalence of having to talk to them?

Approach responses were scored from 1 (minimal approach tendency) to 7 redundant. They state: "Aminusa, essentially Mehrabian (maximal approach tendency); Avoidance responses were scored from 1 (minimal tendency to avoid/escape from the setting) to 7 (maximal tendency to nothing to the explanation that is not provided by avoid/escape).

Besides a score for approach and one for avoidance, in this study also a composite measure termed 'aminusa' was mean difference between approach and avoidance scores). Mehrabian and Russell's approach-avoidance measure.

However, 'aminusa' appeared to follow a similar pattern avoidance, measured as a single variable, tends towards their results for approach and those for 'aminusa' suggests that the continued employment of both variables is and Russell's bipolar approach-avoidance variable, adds approach. However, the approach/avoidance distinction appears useful since the avoidance variable does add information to the analysis and should be retained (Foxall and Greenley, 1999)" (Foxall and Greenley, 2000, p55).

Matilla and Wirtz (2001)

8 approach-avoidance items 270 customers' responses to adapted from Donovan and Rossiter (1982) an actual store setting (a giftstore) in which music and odor were manipulated

Approach-avoidance was treated as a bipolar unidimensional scale (cronbach $\alpha = 0.78$)

"Emotions are fuels for drives, for all motion, every performance, and any behavioural act"

FONBERG

1986, p302

2.7. EMOTIONAL DETERMINANTS OF

Mehrabian and Russell (1974) posited that the emotional states (pleasure, arousal and dominance) engendered by the environment, mediate approach or avoidance behaviours, such as the desire to affiliate with others in the setting, the desire to stay in or escape from the setting, and the willingness to spend money and consume (Mehrabian, 1979; Mehrabian and Riccioni, 1986; Mehrabian and de Wetter, 1987; Mehrabian and Russell, 1975; Russell and Mehrabian, 1976, 1978). More specifically, they hypothesized that pleasure would be significantly related to approach-avoidance measures overall, and that arousal would have an interactive (multiplicative drive-like) effect with pleasantness such that arousal would be positively related to approach behaviours in pleasant environments, but negatively related to approach behaviours would be how in the set of approach behaviours.

The earliest research in the consumer field by Lutz and Kakkar (1975a,b) could however not confirm the hypotheses that pleasure, arousal and dominance relate predictably to consumer behaviour. They conclude (p148) that: "Despite the increase in explanatory power resulting from the use of pleasure, arousal and dominance variables, it is evident that *the situation in and of itself is not a powerful predictor of consumer behaviour*" (emphasis in original). Foxall and Greenley (1999) argue that the small and apparently arbitrarily selected scope of the consumer settings tested (i.e. verbal descriptions of one general and one snack situation), is responsible for their poor results.

Nevertheless, in a retail context, Donovan and Rossiter (1982) did find support for the contention that *pleasure* and *arousal* are related to respondents' intentions to remain in the setting and to spend money, but not for *dominance*. Over the various approach-avoidance measures, the proportion of variance accounted for by the PAD ratings ranged from a high of 50% for affect measures (liking, enjoyment, willingness to return) to a low for their single item 'spend' measure.

Based on the Behavioural Perspective Model, Foxall (1997) found pleasure, arousal and dominance to exert independent influences upon approach and avoidance behaviours, which were conceptualized not to be opposites in a uni-dimensional approach construct, but actually

two separate related constructs. For approach significant main effects could be found with regard to the three emotional states, pleasure, arousal and dominance, however for avoidance only a significant main effect could be revealed for pleasure. There was also no significant interaction. Based on these findings Foxall (1997) concludes that avoidance is explained by the lack of pleasantness in a consumer setting. The implication being that an inherently unpleasant consumer environment cannot be made more attractive by increasing dominance or arousal reactions. *"The separate conceptualisation and measurement of approach and avoidance has led to the finding that these behaviours are uniquely related to emotional response patterns"* (Foxall, 1997, p520). Nevertheless, these different effects have never been examined in a retailing context, as approach and avoidance have always been assumed to be the opposite poles in a uni-dimensional approach construct.

Next some empirical evidence will be reviewed on the specific roles of pleasure, arousal and dominance as antecedents of approach-avoidance intentions and behaviours.

2.7.1. THE EFFECTS OF PLEASURE ON APPROACH OR AVOIDANCE BEHAVIOURS

It has been generally acknowledged that consumers show a greater tendency for approach behaviours in the occurrence of positive emotions, aiming to share, maintain or increase the positive experience (see Bagozzi, Gopinath and Nyer, 1999). Likewise, positive emotions have been shown to increase customer satisfaction, whereas negative emotions tend to decrease satisfaction (Oliver, 1997).

In the past, valence has always been considered as the primary determinant of approach and avoidance behaviour (e.g. Campbell, 1963; Lewin, 1935; Osgood, 1953). Osgood (Osgood, 1953; Osgood et al., 1957) claimed that the 'evaluation' component of the semantic meaning of an object is of major importance as it serves as a guide for behaviour towards the object. According to the emotion-motivation model of Lang, Bradley and Cuthbert (1992), the mere presence of a stimulus object immediately activates, depending on its valence, either a positive approach or a negative avoidance motivational system.

A review of the effects of pleasure on various behaviours, indicates that pleasure induced by a setting increases the approach towards the setting (increased preference, liking, evaluation,

exploration, motivation at tasks and desire to affiliate and cooperate), even though the pleasure is not contingent of the approach (Russell and Mehrabian, 1976; e.g. Griffitt, 1970; Griffitt and Veitch, 1971; Janis, Kaye and Kirschner, 1965; Maslow and Mintz, 1956, Mehrabian and Russell, 1974, chapter 8; Razran, 1938, 1940). In Mehrabian and Russell's (1974) model, approach was therefore hypothesized to be a direct correlate of the situation's pleasure-eliciting quality.

In a retailing context, pleasant affective states have indeed been found to prolong shopping time (e.g. Donovan and Rossiter, 1982) and encourage spending (e.g., Baker, Levy and Grewal, 1992; Sherman, Mathur and Smith, 1997).

According to a study by Obermiller and Bitner (1984), respondents who viewed retail products in an emotionally pleasing environment evaluated the products more positively than subjects who viewed the same products in an unpleasing environment.

Moreover, it has been demonstrated that pleasurable shopping experiences induced by the store environment encourage consumers to spend more time in stores and to spend more money than intended (Donovan et al. 1994).

In the context of environmental crowding, Hui and Bateson (1991) found less perceived crowding and increased perceptions of personal control to be positively related to experienced pleasure, which in turn was found to exert a positive effect on approach/avoidance.

Sherman et al. (1997) found pleasure to have a positive influence on money spent in the store and store liking. On the other hand, pleasure was not found to affect the number of items purchased, nor the time spent in the store.

In their study regarding waiting attribution, Chebat, Gelinas-Chebat, Vaninski and Filiatrault (1995) found pleasure to have the strongest influence on approach-avoidance behaviour.

In contrast, Van Kenhove and Desrumaux (1997) could not conclude that pleasure is the major predictor of approach-avoidance intentions. They found arousal to play a more important role. Nevertheless, pleasure was found significantly related to a composite approach variable and to the time, affiliation and affect sub-dimensions.

Babin and Attaway (2000) find that "customer share" is directly influenced by a store's ability to create hedonic and utilitarian values (Babin, Darden and Griffin, 1994), which in turn, are both sensitive to the customer's positive and negative (i.e. evaluative) responses to the store environment.

2.7.2. THE EFFECTS OF AROUSAL ON APPROACH OR AVOIDANCE BEHAVIOURS

Arousal has for a long time been proposed to affect approach-avoidance behaviour (Berlyne, 1960, 1967; Dember and Earl, 1957; Fiske and Maddi, 1961, and Hebb, 1955). Actually, an inverted-U shaped relationship has been proposed, suggesting that extremes of arousal, whether high or low, produce avoidance, whereas moderate degrees of arousal generate approach. Likewise, in Mehrabian and Russell's 1974 model, approach to a situation was hypothesized to be an inverted-U-shaped function of the arousing quality of the situation.

Mehrabian and Russell (1974, ch 8) explored the independent and interactive effects of pleasure, arousal and dominance on approach-avoidance. Their findings were only partially supportive for the hypothesized inverted-U function. The inverted-U was found in the relationsip between arousal and desire to explore. It was also found for desire to work, but only in pleasant situations. For desire to affiliate, the inverted-U relation was found only in one study and again only in pleasant settings. Arousal failed to relate to desire to stay. Additional results helped clarify the inconsistent findings bearing on the inverted-U. These results showed an interaction between pleasure and arousal in determining approach: in pleasant settings, approach tended to vary directly with arousal, whereas in unpleasant settings approach tended to vary inversely with arousal. Based on this interaction effect and a reexamination of the previous literature Russell and Mehrabian (1976) suggested a modification of the inverted-U hypothesis. Figure 2-4 summarizes this new pleasure-arousal hypothesis, which incorporates three separate relationships: (1) pleasure is positively correlated with approach; (2) the relationship between arousal and approach describes an inverted-U function and (3) an interaction effect in which approach is generally positively related to arousal when pleasure is experienced, but generally negatively related with arousal when displeasure is experienced.



Figure 2-4, adopted from Russell and Mehrabian (1976) shows the combined effect of these three hypothesized relationships: The inverted U-function between approach-avoidance is represented by a parabolic curve. Increasing degrees of pleasure are represented by shifts of the same curve upwards and to the right. According to Russell and Mehrabian (1976) this set of curves indicates that pleasure determines the optimum level of arousal, in that the level of arousal producing maximum approach is a direct linear function of pleasure. In extremely pleasant settings, approach would be a monotonically increasing function of arousal, whereas in extremely unpleasant settings approach would be a monotonically decreasing function of arousal. The pleasure-arousal hypothesis proposed by Russell and Mehrabian (1976) thus

predicts that the inverted U only occurs in situations of intermediate pleasantness and thus explains Mehrabian and Russell's (1974) failure to find the inverted-U in most settings.

Situations labelled extremely high or low in arousal are typically more unpleasant than the moderately arousing situations (Mehrabian and Russell, 1974; Russell and Mehrabian, 1976). Therefore, the effects of pleasure should be controlled for when investigating the effects of arousal (Russell and Mehrabian, 1976). Results, provided by Russell and Mehrabian (1976) demonstrate that pleasure determines the desired level of arousal, a result not predicted by the simple inverted-U relation of arousal with approach.

Russell and Mehrabian (1976) speculate that approach behaviours as the desire to purchase increase with the pleasantness of the setting, and since arousal has a curvilinear relationship with approach behaviour, that desire is maximized in settings, which evoke an intermediate level of arousal. Thus, the model actually specifies a conditional interaction between pleasure and arousal in determining approach-avoidance: in pleasant environments, arousal should be positively correlated with approach behaviour (H1), whereas in unpleasant environments high arousal is expected to be negatively correlated with approach behaviour (H2).

In a retailing context, Donovan and Rossiter (1982) found arousal to be a significant predictor only of affiliation measures. They conclude (p50): "Arousal is not significantly related to approach-avoidance with the particular exception of affiliation". Nevertheless, when considering a pleasure-arousal interaction, Donovan and Rossiter (1982) found support for the first hypothesis, that for pleasant environments there is a positive correlation of arousal with approach intentions. As they were unable to collect information from enough customers who experienced the environment as unpleasant, they could not verify the second hypothesis.

In an extension of this original study, Donovan et al. (1994), were able to verify part of the second hypothesis, but failed to find support for the first one. They found arousal to account for spending less than intended in unpleasant store settings. In this study approach-avoidance was not measured by approach intentions, but by means of actual behaviour variables: 'extra time spent in the store' and 'unplanned spending'. Emotions experienced in the store were related to a comparison of pre-measures of estimated spending and time to be spent in the store and post-measures of actual spending and actual time spent in the store.

In a study focussing on music, Holbook and Gardner (1993) found clear evidence for the interaction effect.

Flicker and Speer (1990) could not reveal a significant relation between arousal and the response variables (avoidance - attraction; intended - actual behaviour), neither for the experimental renovated store, nor for the control store (i.e. the same store before renovations).

Sherman et al. (1997) on the other hand, found arousal to have a positive influence on money spent in the store, number of items purchased and time spent in the store. According to their study, arousal did not appear to have a significant impact on store liking.

Results from a maximum likelihood regression analysis (LISREL 8) by Van Kenhove and Desrumaux (1997) also clearly show the importance of the arousal scale. According to their results, arousal is significantly related to the three dependent approach-avoidance subconstructs (i.e. time, affect and affiliation). Because the sample by Van Kenhove and Desrumaux (1997) was composed of first-time visitors only, a substantial number of people included in the sample (41%) rated the store environment as unpleasant, in contrast to the Donovan and Rossiter (1982) study that included mainly pleasant scores. As pleasure and arousal appeared to be highly correlated (r = .78), the authors propose that both pleasure and arousal are necessary conditions for general approach behaviour intentions. In accordance with this hypothesis, in pleasant retail environments with high arousal, the approach intentions were found to be most positive. In unpleasant retail environments with low arousal, the approach intentions were found to be most negative. These results are, however, not in line with the bi-directional influence of pleasure and arousal in the M-R model. Reducing arousal in an unpleasant store environment is not found to increase approach, in this study. Therefore, support was found for the first, but not for the second hypothesis of the M-R interaction effect.

As findings regarding the role of arousal as an antecedent of approach-avoidance behaviour are ambiguous, even with regard to the hypothesized interaction effect, clearly more research is needed on this subject.

2.7.3. THE EFFECTS OF DOMINANCE ON APPROACH OR AVOIDANCE BEHAVIOURS

According to Donovan and Rossiter (1982) the dominance-submissiveness elicited by a store refers to the degree to which a person feels in control and feels free to act in the store. Control is widely accepted as a human driving force (White, 1959; Averill, 1973). Environmental psychologists, Proshansky, Ittelson and Rivlin (1974) have suggested that people tend to feel and behave more positively when they perceive to have more control in the environment. Schutz (1966) also suggested that a feeling of control is essential to having satisfactory interactions with other people. Although Mehrabian and Russell (1974) originally suggested a positive effect of feelings of dominance on approach behaviours, Russell and Mehrabian (1976) refrained from suggesting how feelings of dominance elicited by environments might affect consumer behaviour. In fact, conceptualising affective responses to the environment, Russell and Pratt (1980) deleted the dominance dimension altogether, arguing that dominance requires a cognitive interpretation by the individual and is therefore not purely applicable in environments calling for affective response.

In a retail setting, Donovan and Rossiter (1982) demonstrated that the dominance dimension failed to predict approach-avoidance behaviours. They found that dominance does not strongly affect in-store behaviour. Greenland and McGoldrick (1994) also report associations between *pleasure* and *arousal* and customers' ratings of traditional and modern bank designs, but not for *dominance*.

Hui and Bateson (1991) who operationalized perceived control combining the Mehrabian and Russel (1974) dominance scale and a helplessness scale, could not reveal a direct relationship from perceived control to approach-avoidance. Perceived control was found to be a positive determinant of pleasure, which in turn appeared to exert a positive effect on approach-avoidance.

Thus, for theoretical reasons (Russell & Pratt's, 1980 conceptualization) as well as for a lack of empirical support (e.g. Donovan & Rossiter, 1982; Greenland and McGoldrick, 1994), several researchers investigating atmospheric effects of the store environment using the Mehrabian-Russell model, have disregarded the dominance dimension and simply researched the pleasure and arousal dimensions (e.g. Donovan et al., 1994; Van Kenhove and Desrumaux, 1997). Therefore dominance has received little attention in atmospherics research.

Nevertheless, according to Foxall (1997) and Foxall and Greenley (1999) the failure to find a role for *dominance* might have been predictable given the ad hoc manner in which consumer settings have been selected for test and the single type of consumer setting used in each of these studies. Turley and Milliman (2000) agree that the findings with regard to dominance may be context specific and depending on the independent store environmental variables investigated.

With regard to music, for example, Yalch and Spangenberg (1988) found that environmental music conditions did not significantly affect pleasure nor dominance. Only arousal was significantly affected by store music. In contrast, with regard to retail crowding, the findings of Bateson and Hui (1987) suggest that dominance is positively correlated with pleasure and personal control and negatively correlated with crowding. In a subsequent crowding study by Hui and Bateson (1991), similar results were obtained.

Babin and Darden (1995) suggest that the lack of findings associated with dominance in previous studies involving retail store environments may also be due to the moderating impact of personality (Bitner, 1992) that has all too often been ignored. Indeed, consistent with previous research, Babin and Darden (1995) also find that feelings of dominance appear unimportant when considering their entire sample. However, integrating the moderating role of self-regulation tendency, they find that only state-oriented shoppers² appear to be negatively influenced by feelings of domination (with a path coefficient of -.49 as compared to a path coefficient of .02 for action-oriented shoppers). Based on these findings, Babin and Darden (1995, p61) emphasize that "*rather that dismissing dominance, as some studies have done, future theory development should include dominance as an important emotion influencing shopping behaviour*".

Investigating a variety of consumption contexts, selected according to the Behavioural Perspective Model, Foxall (1997) succeeded to demonstrate for the first time how all three independent variables – pleasure, arousal and dominance – impact consumers' approach

² Based on consumers' self-regulation tendency (Kuhl, 1986), Babin and Darden (1995) make a distinction between shoppers who are "action" oriented (who strongly hold on to original intentions) and people who are "state" oriented (who are more susceptible to social and emotional influences and therefore to competing, contextually derived action tendencies).

behaviour. In view of the failure of previous research to establish the role of dominance, the role of this factor was especially underscored.

Also Gröppel-Klein (1998) calls for the possible importance of the dominance factor with regard to its effect on price acceptibility.

2.8. ATMOSPERIC EFFECTS ON SHOPPING BEHAVIOUR: A REVIEW OF EMPIRICAL EVIDENCE

Turley and Milliman (2000) give an extensive overview of the empirical findings with regard to atmospheric effects on shopping behaviour. The accumulated empirical evidence, clearly shows that shoppers can be induced to behave in certain ways, based on the atmosphere created by retail management. Out of 28 articles examining the effect of the atmosphere on sales, cited in their review, 25 found some significant relationship between the environment and customer purchasing behaviour. The review presented by Turley and Milliman indicates that atmospheric variables influence a wide variety of consumer evaluations and behaviours. Turley and Milliman (2000, p209) conclude: "Although there may be some debate about whether the atmosphere can influence time spent in an environment, there is enough evidence to be able to clearly state that the atmosphere has an effect on consumer spending and that variations of atmospheric variables affect the amount of money people spend and the number of items they purchase".

In this regard, Chebat and Dubé (2000) point to the need for managers in the retailing industry to develop a better understanding of the interface between the resources they manipulate in atmospherics and the experience they want to create for the customer. Van Kenhove and Desrumaux (1997) also called for further research to determine what stimulus factors constitute pleasant and/or arousing environments. In this study, the specific effects of colour in the store environment will be further investigated, based on the environmental psychology model just presented. Before formulating the actual research objectives and the proposed hypotheses, a review on the literature concerning the effects of colour in general will be provided in the next chapter.

Complex decisions on the design and management of retail atmospherics are frequently made on the basis of a dearth of information that in other domains would be unimaginable.

> CHEBAT & DUBÉ 2000

Chapter 3

Colour Effects


CHAPTER STRUCTURE

Chapter 3: Colour Effects

- 3.1. Introduction
- 3.2. Colour Perception
 - 3.2.1. The Trichromatic Theory of Colour Vision
 - 3.2.2. Opponent Colour Processing
 - 3.2.3. The Two-Stage Model of Colour Vision
 - 3.2.4. Colour Blindness

3.3. The Three Dimensions of Colour

- 3.3.1. Hue
- 3.3.2. Saturation
- 3.3.3. Lightness
- 3.3.4. The Munsell Colour System
- 3.3.5. Conclusion
- 3.4. Cognitive Responses to Colour
- 3.5. Affective Responses to Colour
 - 3.5.1. Introduction
 - 3.5.2. Methodological Caveats
 - 3.5.2.1. Colour stimuli specifications
 - 3.5.2.2. Colour responses and their assessment
 - 3.5.3. Colour-evoked Pleasure
 - 3.5.3.1. Effects of Hue
 - 3.5.3.2. Effects of Saturation
 - 3.5.3.3. Effects of Value
 - 3.5.3.4. Overview
 - 3.5.4. Colour-evoked Arousal
 - 3.5.4.1. Effects of Hue
 - 3.5.4.2. Effects of Saturation
 - 3.5.4.3. Effects of Value
 - 3.5.4.4. Overview

3.5.5. Colour-evoked Dominance

- 3.5.5.1. Effects of Hue
- 3.5.5.2. Effects of Saturation
- 3.5.5.3. Effects of Value
- 3.5.5.4. Overview

3.6. Behavioural Responses to Colour

- 3.6.1. The Application of "Colour Theory" to Real World Situations
- 3.6.2. Empirical Evidence of Colour-Behaviour Effects

 - 3.6.2.1. Task Performance3.6.2.2. Time Perceptions and Time Spending3.6.2.3. Miscellaneous Behaviours

"The keenest of all our senses is the sense of sight"

CICERO

3.1. INTRODUCTION

Touching, tasting, smelling, hearing and seeing are the senses through which we perceive the environment surrounding us. However, our sight prevails, with 90 percent of what we know of the world coming to us through our vision (Swirnoff, 1989). An extremely important factor in visual perception is colour. Research suggests in fact that colour determines whether we see stimuli (Lindsay and Norman, 1973). According to Nassau (1983), colour is perceived by the eye and interpreted by the brain. What we see is the light that reflects off objects which enters the eye and registers these objects in our brains. Thus, the eye's retina absorbs the light and sends a signal, or sensation, to the brain, which makes us aware of a characteristic of light, namely colour (Herman Miller, 2001). In this respect, the phenomenon of colour is only a subjective visual sensation, produced by light and not a property of a given object (Wineman, 1979; Porter and Mikellides, 1976a).

Theories of colour have been devised for centuries. Nassau (1983) traces the historical roots of colour "from the philosophy of Plato where colour belonged unaccessibly in the realm of divine Nature, to the calculations of Newton, who discovered the spectrum and arbitrarily assigned to it seven fundamental colours, analogous to the seven notes of the musical scale" (cited in Beach et al., 1988, p3).

The use of colours as decoration in living (and working) environments has also received a good deal of attention (Beach et al., 1988; Birren, 1983; Wineman, 1979; Porter and Mikellides, 1976a, 1976b). In a critical review on the human factors of colour in environmental design, Beach et al. (1988) afirm that it was not until the late 1800's, with the advent of introspection, that colour was systematically investigated in a behavioural context. Nevertheless, the succeeding years have seen a plethora of research generated on the topic (Berlin and Kay, 1969; Kuller, 1981; Plack and Shick, 1974; Sucov, 1973; Hayward, 1972). We refer to Beach et al. (1988) for an extensive overview on this subject.

In this chapter we will focus more specifically on how colour is perceived and on the three distinct attributes of colour (hue, saturation and brightness). Moreover, we will give a detailed overview on human affective, cognitive and behavioural responses to colour. In the mean time, we will review the methodological concerns regarding colour research.

3.2. COLOUR PERCEPTION

Colour is our perceptual response to a very narrow span (less than 1%) of the total electromagnetic radiation emitted by the sun (Handprint Media, 2001). The wavelengths of the visible spectrum (light) range from about 380 to 750 nanometers. The invisible radiation at wavelengths shorter than 380 nanometers (higher energy) is called ultraviolet and includes x-rays and gamma rays; lower energy radiation at wavelengths longer than 750 nanometers is referred to as infrared, including heat, microwaves, television and radiowaves (Handprint Media, 2001). The electromagnetic spectrum is illustrated in figure 3-1.



Figure 3-1: The electromagnetic spectrum Source: Laura Funderburk, 2/03/2001

3.2.1. THE TRICHROMATIC THEORY OF COLOUR VISION

The National Bureau of Standards estimates that the human eye can distinguish over 10 million different colours (Herman Miller, 2001). However, according to *the trichromatic theory of colour vision*, our perception of these millions of colours depends on just three distinct colour receptors in the eye. This theory presents a framework universally used, both to specify colour stimuli and to define how the eye interprets colour from light (Handprint Media, 2001).

Without any knowledge of the nature and number of human photoreceptor cells, but based solely on his observation that all colours could be matched by a suitable mixture of three primary colours, Young proposed already in 1807 that there must be three fiber types in the human eye. Von Helmholtz (reprinted 1924) later formalized Young's "*trichomatric theory*" proposing hypothetical excitation curves for each of the three fiber types in order to make quantitative predictions of the ability to make discriminations on the basis of wavelength. The general idea of *trichromacy*, which thus was totally based on the results of psychophysical colour matching experiments, soon became accepted doctrine (Levine, 2000).



Source: Handprint Media, 2001 (http://www.handprint.com/HP/WCL/color1.html)

Later on, a firm physiological foundation was uncovered for the *trichromacy* of human vision. In fact, the eye appears to contain, between the retina and the sclera, two kinds of receptors: rods and cones, as illustrated in figure 3-2. While the rods convey shades of gray, the cones allow the brain to perceive colour hues. In fact there are three types of colour receptor cones, which are more sensitive to different wavelengths of light (according to the type of photopigment they contain): long wavelength or "red" (R) cones (which are actually most sensitive to "greenish yellow" wavelengths around 565 nm), middle wavelength or "green" (G) cones (most sensitive to 'green' wavelengths at 435 nm) (Brown and Wald, 1964; Marks et al., 1964, Merbs & Nathans, 1992; Neitz et al., 1995). The graphs in figure 3-3 below, demonstrate how each cone presents a curve of relative sensitivity across the entire spectrum.



Source: Smith and Pokorny, 1972, adopted from Handprint Media, 2001 (http://www.handprint.com/HP/WCL/color1.html)

³ Note that the cone spectral sensitivity curves in the graph above are displayed on a normal vertical scale, demonstrating clearly the low sensitivity at the extremes, making it obvious that each type of cone responds primarily to light with wavelengths close to their maximum sensitivity. In order to expand the detail of the graph at very low sensitivity, in the graph below the same curves are represented using a logarithmic or exponential scale on the vertical axis with each interval ten times larger than the previous one.

This peak response around the cone's most sensitive wavelength is the reason why colour vision can be modelled as a mixture of three "primary" colours of light, as discovered earlier from psychophysical colour matching experiments (Smith, Pokorny & Starr, 1976; Levine, 2000).

The large differences in the height and spread of the curves (the probability that a cone of each type will respond) arise in part because of the unequal proportions of R, G and B cones in the retina of the eye. About 64% of the colour receptor cells are long wavelength R cones, 32% are mid wavelength G cones and 4% short wavelength B cones (Handprint Media, 2001).

If a cone is sufficiently stimulated by light, anywhere within its range of sensitivity, it generates an invariable nerve impulse. Because this way information about the specific light wavelengths that stimulated the individual cones is lost, which is referred to as *"the principle of univariance"*⁴, the responses from all three types of cones must be combined to "triangulate" to the dominant wavelengths in the light (Abramov & Gordon, 1994; Levine, 2000).

3.2.2. OPPONENT COLOUR PROCESSING

While trichromatic theory accounts for many characteristics of human colour vision, there are a number of visual phenomena, which are not easily explained by this model (Abramov & Gordon, 1994; Levine, 2000). Data accumulated from experiments on colour afterimages, complementary colours and colour naming have given rise to an alternative model for the perception of colour, referred to as "*opponent process theory*", introduced by Ewald Hering in 1878.

Hering proposed that colour vision is mediated by three complex 'substances', also referred to as 'opponent processes', which account for the distinction between respectively red/green, blue/yellow and white/black. These 'substances' or 'processes' were postulated to be

⁴ The "principle of univariance" refers to the fact that "each visual pigment can only signal the rate at which it is effectively catching quanta; it cannot also signal the wavelength associated with the quanta caught" (Naka & Rushton, 1966, p538).

'opponent', which means that, for instance, the red/green 'substance' responds in opposite directions to red and green lights (i.e. it may be excited by red light and inhibited by green light).

Thus opponent processing induces four chromatic channels (four unique hues: red, green, yellow and blue) as well as two separate channels of luminosity information (white and black). These six fundamental colour sensations can combine to produce any colour giving way to a "natural colour system" (Hering, 1878, 1920).



Figure 3-4:A colour circle derived from opponent colour processingSource:From the original illustrations by Ewald Hering (1920)

Figure 3-4 (Hering, 1920) illustrates how the four unique hues create a colour circle. The two opponent colour pairs (red opposed to green and yellow opposed to blue) are placed opposite each other on the outer circle. These pairs define two opponent colour contrasts that never mix. A line through any part of this mixing ring indicates the relative proportions of two unique colours necessary to create the hue samples shown in the inner colour circle. Experimental evidence however demonstrates that the mixing of opponent colours is neither as symmetrical nor evenly spaced across the visible spectrum as the schematic representation by Hering implies (Handprint Media, 2001).

Although opponent process theory can explain a large variety of visual phenomena that cannot be explained by trichromatic theory (e.g. Hurvich & Jameson, 1957), it was criticized for many years due to a lack of physiological evidence (Levine, 2000). The basis of this criticism disappeared with the discovery of +R-G and -R+G (red-green opponency) and +B-Y and –B+Y (blue-yellow opponency) cells, which actually appear to be some sort of opponent processing cells (DeValois et al., 1966 and 1974; Wiesel and Hubel, 1966; Gouras, 1968; DeMonasterio et al., 1975; Dacey, 1996; Calkins et al., 1998).

3.2.3. THE TWO-STAGE PROCESS OF COLOUR VISION

Throughout the first half of the 20th century, the trichromatic theory (based on outputs from three types of cones) and Hering's opponent processing theory (including the four unique hues) were seen as competing explanations of colour vision. It was only after the middle of the century that scientists recognized that both were necessary to explain colour. In fact, the "opponent processing theory" does not contradict the "trichromatic theory", but actually extends it (Levine, 2000). This is referred to as the "two-stage model" of colour vision.

The trichromatic model describes how the combined outputs from three types of receptors can code any combination of light wavelengths. The opponent processing model explains then how those receptor codes are translated into two colour dimensions (with a luminosity dimension) that are the foundation of conscious colour perceptions, a process which also has been referred to as 'remapping' (Hurlbert, 1991). Thus, colour vision begins in a trio of receptor cells that respond preferentially to different wavelengths of light. These three types of cells are interconnected with other processing cells in the retina to produce at least six types of visual information, interpreted as colour by the mind. The translation from receptor responses to opponent codings happens in the retina of the eye: therefore the brain never "sees" the trichromatic outputs (Hurlbert, 1991, 1997; Levine, 2000; Handprint Media, 2001).

In the retina of the eye, colour information is actually synthesized before it is sent to the brain. Other cells in the retina transform the cone outputs into separate channels of colour information. Figure 3-5 shows the basic neural programming used to transform the raw R, G, and B cone responses through "opponent processing" (comparing responses of one cone against the others) into four new channels of chromatic 'colour' signals, and a channel of

achromatic (colourless) brightness or luminosity signal (Hurvich & Jameson, 1957; Hurlbert, 1991, 1997; Handprint media, 2001).



 Figure 3-5:
 Opponent processing of cone outputs

 Source:
 Handprint Media, 2001 (http://www.handprint.com/HP/WCL/color1.html)

This neural colour circuitry can be summarized as a few basic rules of colour vision (Handprint Media, 2001):

- "The combined stimulation of the R and G cones (right circle in figure...) is interpreted as the brightness or *lightness* of a colour." Luminosity (light/dark) is the dominant visual information recorded by the eye.
- "The relative proportion of stimulation received by the R cones in contrast to the G cones (central circle in figure...) creates the perception of *red* or *green*."
 (The B cones contribute slightly to clarify warm colour saturation)
- "If the R and G cones are stimulated approximately equally (and much more than the B cones), we see the colour *yellow*."
- "The relative proportion of stimulation received by the B cones, in contrast to the R and G cones combined (left circle in figure ...), creates the perception of *blue*."
- "If all three types of cones are stimulated approximately equally, we see no specific hue, but *white, grey* or *black*."

Thus, the opponent processes are formed by subtractive interactions among the three conetypes. This processing occurs in the retina, rather than in the brain, because these 'processed' colour impulses can be sent through the optic nerve with much less "noise" or error than the raw cone outputs. Finally, the optic nerves from both eyes lead to an area in each side of the brain, called the lateral geniculate nucleus (LGN). In that part of the brain colour, edge and spatial (depth) information from both eyes are processed once again. From the LGN, visual data travel to the visual cortex and from there radiate to many other areas of the brain involving language, body movement and long term memory.

'Optics' is the branch of science that analyses the mechanisms we use to perceive colour – the rods and cones of the eye's retina (Herman Miller, 2001). In this context, numerous experiments have been conducted with regard to colour perception and discrimination. It is, however, beyond the scope of this research project to expand on this issue. We intended to give an initial introduction with respect to colour vision in order to clarify some important colour related concepts that we will be referring to in the following paragraphs. For the interested reader, some critical assessments with regard to visual processing mechanisms are provided by Wright (1984), Hurvich (1981), Boynton (1979), Abramov & Gordon (1994), Hurlbert (1997) and Levine (2000).

Not everyone has the same colour discriminating abilities. In the next paragraph we will present a detailed discussion on colour blindness, a visual defect leading to the confusion of colours.

3.2.4. COLOUR BLINDNESS

Colour blindness is a visual defect resulting in the inability to distinguish colours (Medical Post, 1997; Neitz, 1998). There are many different kinds of colour defects (see Levine, 2000, p317-319). Completely colour-blind individuals can recognize only black, white, and shades of grey. This congenital condition of total colour blindness, in which all hues are perceived as variations of grey is called *"monochromatism"* or *"achromatopsia"*⁵ and is extremely rare

⁵ For a discussion on the difference between "achromatopsia" and "monochromacy", we refer to Levine, 2000, p319.

(Sewell, 1983). Most "monochromats" lack all three cone types (discussed earlier in paragraph 3.2.1.), leaving only the rod photoreceptors to mediate the visual process.

A much more common colour defect is "partial colour blindness" or "*dichromatism*". Indeed, most "colour-deficient" persons can see colours, but not in the same way as those with full colour vision (Rosenthal & Phillips, 1997; Wilson, 1998; Kaufman-Scarborough, 2000). In fact, one can be blind to one or more primary colours, leading to "colour confusion", which would be a more accurate term for this condition (Sewell, 1983). Most of those with defective colour vision are only partially colour-blind to red and green, i.e., they have a limited ability to distinguish reddish and greenish shades (illustrated in figure 3-6).



Figure 3-6: Illustration of deficient colour vision Source: Holland s., 2001

In fact they are missing one of the three cone pigments: "*protanopes*" lack the longwavelength-sensitive pigment, and have therefore a difficulty distinguishing red shades, while "*deuteranopes*" lack the medium-wavelength-sensitive pigment and have a difficulty distinguishing green shades (Nathans et al., 1986 a, b; Neitz et al., 1996). Those who are completely colour-blind to red and green, miss both cones and see both colours as a shade of yellow.

Colour vision deficiency, which depends on the light-sensitive pigment contained in the cone cells of the eye, is usually an inherited sex-linked characteristic (transmitted through, but recessive in, females) (Nathans et al., 1986 a, b; Neitz et al., 1996). Colour blindness can also be acquired as a result from certain degenerative diseases of the eyes. The colour vision of perfectly colour-normal persons can be affected over time by age, progressive illness (such as multiple sclerosis, leukemia, glaucoma, diabetes or Alzheimer's) or the use of certain

medications and can change subtly, either temporary or permanent (Braus, 1995; Wilson, 1998). Thus an individual who has had normal colour vision, may not realize a gradual decline in perceptual ability for some time.

The estimates concerning the actual number of persons suffering from colour vision deficiencies vary slightly. Buehler (1996) reports that 1 in 12 men (i.e. 8%) and 1 in 65 women (i.e. 2%) have a colour deficiency. Others suggest that 8 to 12 % of men and 0.4 to 0.5 % of women experience some difficulty in colour perception (Piantanida, 1988; Henthorne, 2000; Medical Post, 1997; Nash, 1994). Within the United States, estimates range from 19 million persons with congenital colour deficiencies (Corsino, 1985; Sewell, 1983) to approximately 41 million, including acquired colour deficiencies (Castleman, personal communication, cited in Kaufman-Scarborough, 2000). In contrast to partial colour blindness, which is obviously sex-linked, complete colour blindness, or *monochromatism*, affects men and women almost equally (Sewell, 1983).

Colour blindness is usually not related to visual acuity; and therefore, only significant when persons who suffer from it seek employment in occupations where colour recognition is important, such as airline pilots or railroad engineers, who must recognize red and green traffic signals (Wald, 1997) or in industries using colour coding such as for electrical components, chemicals and pharmaceuticals (Voke, 1998). Nevertheless, examining the impact of colour-deficient vision upon consumer emotions, enjoyment, efficiency and safetyrelated aspects of shopping, Kaufman-Scarborough (2000) reveal some prevalent problems and needs of colour deficient consumers that are infrequently addressed. Colour cues may not be processed and interpreted as intended and some information may be obscured, which may result in packaging design becoming illegible, or signing becoming indistinguishable. Kaufman-Scarborough (2000, p462) note that colour deficiencies are not typically addressed in marketing: "Studies of consumer reactions to retail, advertising, packaging and other informational cues traditionally assume that typical consumers have a relatively standard range of colour abilities (Edell & Staelin, 1983; Lee & Barnes, 1990; Meyers-Levy & Peracchio, 1995)". In order to accommodate for the colour blind, Kaufman-Scarborough (2000) recommends to test alternative colour combinations on packaging, store displays, store signs, advertising copy and websites by means of specific software programs developed to simulate how graphics are actually perceived by colour deficient consumers (Beard, 1995; Nakahara, 2000; Rigden, 2000; Vischeck, 2000; Wolfmaier, 1999).

Precise colour terminology is an invention of the 20th century, resulting from the scientific study of colour. By the mid 19th century, colour scientists concluded that three colour-making attributes serve extremely well to describe and predict conscious colour judgements. Although there are many slightly different colour description systems, most are based on these three colour-making attributes (Handprint Media, 2001). These colour systems posit that a given colour can be described according to the three dimensions: hue, saturation and lightness (Munsell, 1966; Thompson et al., 1992; Valdez and Mehrabian, 1994; Hoyer and MacInnis, 2001, p92-93). Because colour can be totally determined by these three attributes, these properties can be used to measure the stimulus intensity of colours (Hoyer and MacInnis, 2001, p96). 'Colorimetry', is the branch of science which measures the colour systems developed to precisely communicate colour.

3.3.1. HUE

Hue is the most familiar colour attribute (figure 3-7). It refers to the pigment contained in the colour, such as red, blue, green, yellow, ... Traditionally, statisticians (e.g., Armore, 1973) and many colour researchers alike have assumed that colour represents a nominal variable with no natural ordering (e.g., Jacobs and Suess, 1975; Bjerstedt, 1960; Sallis and Buckalew, 1984).



However, by using wavelength and the visible spectrum to map colours, the colour spectrum can be viewed as a continuum or ratio scale (Stevens, 1946; Crowley, 1993; Valdez and Mehrabian, 1994). Indeed, light (i.e. electromagnetic radiation emitted by the sun that

stimulates the eye), can be described as oscillations or waves. The distance required for one cycle in such a wave (from peak to peak) is referred to as "wavelength" (illustrated in figure 3-8). The wavelengths of light (the visible spectrum) range from about 380 to 750 nanometers. Waves increase in frequency (decrease in wavelength) as the radiation increases in energy.



 Figure 3-8:
 Wavelengths of light

 Source:
 Handprint Media, 2001 (http://www.handprint.com/HP/WCL/color1.html)

Within the spectrum, the spectral hues do not have clear boundaries, but appear to shade continuously from one hue to the next across colour bands of unequal width. Figure 3-9 shows the visible spectrum roughly as it appears in a brilliant rainbow, fading at either end into invisible ultraviolet (UV) or infrared (IR) radiation (Handprint Media, 2001).



Hue, the term most often used to describe colours, actually refers to the dominant wavelength of the colour. Indeed, as illustrated in figure 3-10, our perception of hue is usually determined by the average or dominant wavelengths in the light we see, regardless of the total range of wavelengths present in the stimulus (Handprint Media, 2001).



Figure 3-10: Hue denotes the dominant or "average" wavelength of light Source: Handprint Media, 2001 (http://www.handprint.com/HP/WCL/color2.html)

When viewed as wavelengths of visible light, colours can be ordered from short to long wavelengths as follows: violet, blue, green, yellow, orange, red. Researchers have tended to classify colours into two broad categories or colour hues: warm, long-wavelength colours, such as red, orange and yellow, and cool, short-wavelength colours, such as green, blue and violet.

Wavelength, however, is but one of three aspects of colour. Colour stimuli can be characterized completely in terms of hue (i.e., wavelength), saturation or chroma (i.e., purity or vividness) and lightness or value (i.e., black-to-white quality).

3.3.2. SATURATION

Saturation or "chroma" refers to the intensity, richness or purity of the colour. Regardless of how light or dark, saturated (high chroma) colours are deep and rich, almost glowing, with a greater proportion of pigment in them, whereas unsaturated (low chroma) colours are dull, containing more grey. A saturated colour looks very luminous or concentrated, just as if it came through a prism, while an unsaturated colour looks dull, gray, faded or diluted. The following definition, presented by Agoston (1979, p87), may be helpful for a better

understanding of chroma or saturation: "*The Munsell chroma of a colour sample is defined as the difference from a grey of the same lightness*". The example below (figure 3-11, adopted from Handprint Media, 2001) shows variations in saturation in a scarlet red of constant hue and lightness.



In technical terms, saturation is the chromatic intensity of a colour, judged in relation to its own brightness. As illustrated in figure 3-12, saturation represents the spectral purity of a colour. The less the variety of different wavelengths that make up the colour, the more saturated the colour.



Figure 3-12: Saturation denotes the spectral purity of light Source: Handprint Media, 2001 (http://www.handprint.com/HP/WCL/color2.html)

Single wavelength or mono-spectral lights are always maximally saturated. Unsaturated colours, on the other hand, are mixtures of many different wavelengths of light and do not have a narrow, precise location within the visible spectrum (i.e. they have a flat reflectance profile, comparable to the reflectance profile of grey).

As mentioned earlier, colour stimuli can be characterized completely in terms of hue (i.e., wavelength), saturation or chroma (i.e., purity or vividness) and lightness or value (i.e., black-to-white quality). This last, more familiar, colour-making attribute, which is independent of saturation, will be dealt with next.

3.3.3. LIGHTNESS

The third colour-making attribute, lightness or value, is a more familiar one and refers to the depth of tone of a colour or its black-to-white quality, i.e. the degree to which a colour varies along the white-black continuum (Farrell and Booth, 1984). More specifically, value reflects the degree of darkness or lightness of the colour, relative to a neutral scale, that extends from pure black to pure white. Low value colours have a "blackish" quality to them, as if the colour black was mixed in the pigment. High value colours have a "whitish" quality to them, as if the colour white was mixed into them. This makes high value colours pastel-like in appearance. The example below (adopted from Handprint Media, 2001) illustrates variations in tonal value for a constant dull (unsaturated) middle blue hue.



Figure 3-13: Differences in value, with hue and saturation held constant Source: Handprint Media, 2001 (http://www.handprint.com/HP/WCL/color2.html)

Value is determined by the total quantity of light reaching the eye from all parts of the spectrum: The diagram below (adopted from Handprint Media, 2001) clearly illustrates this: the greater the light, the higher the value (and thus the lighter the perceived colour).



Figure 3-14: Lightness or value denotes the total quantity of light Source: Handprint Media, 2001 (http://www.handprint.com/HP/WCL/color2.html)

In fact, value dominates our visual experience (Levine and Shefner, 2000). The mind relies on lightness and contrast information to infer the physical structure and surface qualities of the entire visual environment. That is why luminosity is handled as a separate channel of visual information from chromatic (colour) sensations (Handprint Media, 2001), as discussed earlier. To perform this critical function across the widest possible range of illumination, the eye and mind are continuously adjusting apparent values to accommodate the total visual context and to draw the maximum amount of contrast from available light. Our "objective" judgment of value suffers as a result.

Note that the lightness and the saturation of a colour are not related. For instance, a saturated pink could have a lot of lightness (a fluorescent pink) or a lot of darkness (a mauve). Although it is easy to confuse saturation (or chroma) and value (Melara et al., 1993; Melgosa et al., 2000), they really are distinct dimensions of colour, just as hue is. To eliminate any confusion, the Munsell Colour System will be elaborated on in the next paragraph, in order to give a more comprehensive view (both visually and conceptually) of how these three dimensions can be used to accurately describe colours.

3.3.4. THE MUNSELL COLOUR SYSTEM

When conducting colour research, one needs to employ a system by which colours can be quantified on the dimensions of hue, saturation (chroma) and lightness (value) (see Kelly and Judd, 1976). Available colour systems can precisely specify a colour through visual matching on these three dimensions. There are several colour calibration systems that scale colour along hue, chroma and value. The most widely used system in psychological research on colour is the Munsell System (Munsell, 1966) (Smets, 1982; Beach et al., 1988; Valdez and Mehrabian, 1994; Gorn et al., 1997).

"The Munsell colour-order system is a way of precisely specifying colours and showing the relationships among colours" (Munsell.com, 2002). The system was developed by the American painter Albert Henry Munsell between 1905 and 1916 and is based on the principle of 'perceived equidistance'. The concept of 'equality of differences' means that 'equal degrees of colour differences' have been envisioned with regard to each of the colour attributes: "equal hue steps, equal value steps and equal chroma steps" (Kuehni, 2002, p25). Using spinning colour-tops and a self-developed photometer, Munsell established numerical scales with visually uniform steps for each of these attributes. The interested reader is referred to Kuehni (2002) for a historical perspective on the early development of this colour system. In 1943, a subcommittee of the Colorimetric Committee of the Optical Society of America performed the first systematic investigation of the psychophysical properties of the Munsell System and formulated what has come to be known as the 'Munsell Renotation' (Indow, 1988). The Munsell System's ease of use and familiarity have given it near universal acceptance, making it one of the most widespread and most utilized and cited colour systems in the literature (Billmeyer, 1987). The Munsell Colour-Order System has been widely used in many fields of colour science, most notably as a model of uniformity for colorimetric spaces and has been recognized as a standard system of colour specification in standard Z138.2 of the American National Standards Institute, the JIS Z 8721 Japanese Industrial Standard for Colour, the German Standard Colour system DIN 6164 and several British national standards (Munsell.com, 2002).

Influenced by Rood's "Modern Chromatics" (1879), Munsell first introduced his coloursphere (A Colour Notation, 1905), placing five principle hues (red, yellow, green, blue and purple) at equal intervals around a circle and inserting – based on the principle of 'differential thresholds' or 'just noticeable differences' – five intermediate hues between them. This way hue can be classified into 10 equally spaced major hues: red, yellow-red, yellow, green-yellow, green, blue-green, blue, purple-blue, purple and purple-red (as illustrated in figure 3-15). The hues are arranged in such a way that opposing pairs are complementary or compensatory, which means that mixing them would result in an achromatic colour (Munsell.com, 2002; Colorsystem.com, 2002).



Figure 3-15:10 hues represented in the Munsell Colour CircleSource:The Munsell System of Colour Notation
(http://www.munsell.com/munsell1.htm)

The hue circle was then arbitrarily divided into 100 steps of equal visual change in hue, with the zero point at the beginning of the red sector (as illustrated in figure 3-16). This way hue may be identified by a number from 0 to 100, as shown in the outer circle. It is, however, more meaningful to identify hue by the hue-sector and a ten-scale step within this sector. This method of identification is demonstrated in the inner sector of the circle. The hue in the middle of the red sector is, for instance, called five red (5R). Note that the zero step is never used.



The value scale, which refers to the lightness of the colour, extends from 0 to 10. A value of 0 is used to symbolize pure black and a value of 10 is used for pure white. Black, white and the greytones between them are called neutral (N) or achromatic colours (i.e. they have no hue). In contrast, colours that have a hue are called chromatic colours. The value scale applies to chromatic as well as to neutral colours, with colours with low value being dark and colours with high value being light (Munsell.com, 2002).

The chroma scale, which refers to the saturation or purity of the colour, is actually the degree of departure (or perceptual distance) of a colour from the neutral colour of the same value and extends from 0 up. The scaling of chroma is intended to be visually uniform, with chroma 0 representing pure greytones or neutral colours and there being no arbitrary end to the scale. Colours of low chroma are unsaturated and weak (i.e. they appear to be mixed with grey), while those of high chroma are highly saturated, strong and vivid (Munsell.com, 2002).



Figure 3-17 (adopted from Handprint Media, 2001), represents an array of red violet colour samples across the complete range of chroma (saturation) and value. On the vertical axis value increases along equal steps of a gray-scale or value-scale, from black at the bottom to white at the top. On the horizontal axis, chroma changes in equal perceptual steps from achromatic (gray) at the left to maximum chroma at the right. The colour patches in each row of the diagram are of the same lightness, whereas those in each column have the same chroma. The overall shape of this array (bulging in the middle) illustrates that the range of chroma is much larger in the middle values than in the extremely light or dark values. Note that as value decreases, the amount of chroma, required for high saturation, also decreases. So the lines of equal saturation are not vertical, but diagonal.

⁶ To view displays like this of all the different Munsell hues, we refer to the Gretag Macbeth website (<u>http://standards.gretagmacbeth.com/cmc/index.htm</u>, last updated March 11, 2002), from which Munsell Conversion Software - V6.13 (Win 95/98/NT/2000/XP) Conversion of Munsell > XYZ > L*a*b* > RGB > CYMK > Color on Screen - can be downloaded for free.

To make matters even more complicated, different hues reach their maximum chroma at different values. For instance, purple-blue has the highest chroma at a very dark value, making its colour array bulge near the bottom, whereas yellow has its highest chroma at a very light value, making its array bulge at the top (Handprint Media, 2001). Indeed, yellow is brighter than red, which, in turn, is brighter than violet and so on...(Colorsystem.com, 2002). For the colours yellow and purple-blue, this contrast is illustrated in figure 3-18. This particularity makes it especially difficult to do colour research. For example, although Gorn et al. (1997) intended to include yellow in their investigations –as it is a primary colour– they excluded the hue, because unlike for blue and red, for which the pattern of chromas and values is quite similar, for yellow it differs radically.



Figure 3-18:value and chroma ranges for yellow and purple-blue demonstrate that
the natural colour space is highly irregular.Source:Handprint Media, 2001

(http://www.handprint.com/HP/WCL/color6.html)

According to the Munsell System (Munsell, 1966), any specific colour can be identified accurately by quoting the levels of hue, chroma and value. The Munsell Book of Colours (1929) displays a collection of coloured chips arranged according to these scales (as illustrated in figure 3-19). Each chip is identified numerically by its hue, value and chroma, and can be represented by means of the symbols H V/C (Hue, Value, Chroma), which is referred to as the Munsell notation (Munsell.com, 2002). The notation for a neutral colour is N V/. (with chroma being zero and therefore omitted).



Figure 3-19:The Munsell System of Colour NotationSource:http://www.colorsystem.com, 2002



Figure 3-20: Schematic diagram of the three axes of the Munsell colour system Source: Hall, 1979 (http://kiptron.psyc.virginia.edu/steve_boker/ColorVision2/node16.html)

Munsell hue, value and chroma can be varied independently, so all colours can be arranged according to these three attributes in a three-dimensional space (as illustrated in figure 3-20). However, because the variation in brightness among pure chromatic colours is too large for them to be arranged in sequence along the equator, a geometrically solid is unable to portray the opposing relationships between the colours, as we perceive them (Colorsystem.com, 2002).

Therefore the three-dimensional colour space can best be portrayed as a 'colour tree'. The neutral colours are placed along a vertical line, called the neutral axis, with black at the bottom, white at the top and greytones in between. The different hues are displayed at various angles around the neutral axis. The chroma scale is perpendicular to the vertical axis, extending outward from the centre. This three dimensional arrangement of colours is called the Munsell colour space. The relationship of the three scales in three-dimensional space is illustrated in figure 3-21 (Munsell.com, 2002).



As the highest chroma yellow colours have rather high values, while the highest chroma blue colours have lower values, the Munsell colour solid has an irregular shape, which is illustrated clearly in an early sketch of the colour tree represented in figure 3-22.

(http://munsell.com/munsell1.htm)



Source: Volume I of typed diary, March 20, p97, 1902 (in Kuehni, 2002, p25)

Closely resembling the Munsell Colour System is the Swedish Natural Colour System – NCS – (Hard and Sivik, 1979; Hard, 1975) which has also been used as a descriptive colour model (Sivik 1974a, 1975). The NCS is also based on an individual's perception of colour. According to this colour system, there exist six pure colour perceptions – Yellow, Red, Green, Blue, Black and White. The perception of any colour is based on the degree of its perceptual similarity to these six elementary reference colours, which are assumed to be a component of one's own visual system. Three parameters define a given hue – degree of whiteness, blackness and chromaticness (purity). Schematically, it is almost identical to the Munsell Colour System. It also uses brightness as a central Y-axis and colour purity or saturation as an X-axis. In fact there is evidence to suggest that a single relationship exists between the NCS and Munsell models, which appear both to be describing the same colour space (Judd and Nickerson, 1975).

3.3.5. CONCLUSION

From what was treated in this chapter, it should be clear that three colour-making attributes completely define any colour sensation: lightness or value identifies the proportion of the total illumination that is reflected by a surface; saturation or chroma indicates what proportion of the reflected wavelengths comes from a narrow span of the spectrum and hue refers to the location of these wavelengths within the total span of the colour circle.

The next chapter will reveal that in many colour studies, researchers have simply reported vague verbal descriptions of the colour samples used, hereby failing to provide adequate colour specifications and confusing the effects of specific colour attributes. This is a common, but major methodological caveat in studies on colour effects that we aim to address in the empirical study at hand.

We have colour categories stored in prior knowledge (Hoyer and MacInnis, 2001, p124). The category of things that are green, for example, includes grass, trees, leaves, and mint as members. These members have associations like refreshing, new, organic, peaceful, and springlike. Because of this category-based knowledge, consumers can make inferences about a stimulus based on its colour (i.e. category-based inferences). They may infer that a brand of green toothpaste that comes in a green package is refreshing, minty, and healthy (example adopted from Hoyer and MacInnis, 2001, p124). It is for this reason that Bio yoghurt by Danone, for example, is offered in green packaging and advertised with a predominantly green background. The fact that a bag of "M&M's" contains a disproportionate number of brown "M&M's" compared with the other colours, may also be due to marketers' finding that the brown colour makes consumers think that "M&M's" taste more "chocolate-like" (example adopted from Hoyer and MacInnis, 2001, p124). In a study concerning tomatoketchup, indeed, the darker-red the ketchup, the more spicy it was thought to be, and in contrast, the more orange, the milder the associated taste (Percy, Ketchum, MacLeod and Grove, 1974). Likewise, the colour of bread (white, light or dark) has been found relatively more important in determining consumer perceptions of its nutritional value than were price and nutritional information combined (Peterson, 1977) (for this reason some breads contain colouring, to make them appear darker, and hence more nutritious than they actually are (Consumer Reports, 1976)).

Because consumers can make inferences about an offer based on its colour or the colour of its package (Hoyer and MacInnis, 2001, p124), colour can be used as a marketing tool, in order to create, maintain and modify brand images in customers' minds (see Schmitt and Simonson, 1997). For this reason, the meanings associated with different colours are extremely important to marketers (McCracken, 1988; Schmitt and Simonson, 1997; Madden et al., 2000).

The existence of the Lanhan Act in the United States, which protects product colours as trademarks, demonstrates the significance of colour to convey meaning. Several companies have already taken legal steps to protect the colour of their brand (Lans, 1995). Think for instance of 'Belgacom' blue or 'Kodak' yellow.

Because category content varies with culture, the meaning associated with colours and hence inferences based on colour will also vary (Crowley, 1993). For example, white is usually associated with purity and cleanliness in Western Societies, but in Asian countries it signifies death. Green is a popular colour in Muslim countries, but is negatively perceived in Southwest Asia. Black has negative overtones in Japan, India and Europe, but is perceived positively in the Middle East.... Madden, Hewett and Roth (2000) explored the meanings associated with colours and colour combinations in eight countries and cross-cultural patterns in both similarity and dissimilarity in colour meaning associations.

The cognitive schemas and symbolic meanings, associated with specific colours, will not be discussed here in further detail, because they are not the main focus of the present empirical study, in which a 'classic' (i.e. affective) environmental psychology approach is pursued. Instead, we shall focus more specifically on the emotions elicited by colours, which will be elaborated in more detail in the next chapter.

Still, for more information on symbolic colour meaning associations, the interested reader is referred to Dériberé (1969); Sharpe (1974); Birren (1983), Madden, Hewett and Roth (2000), Funderburk (2001), ...

"Colour does more

than just give us objective information about our world:

It affects how we feel."

HERMAN MILLER

2001

3.5.1. INTRODUCTION

Our language, whether it be English, Dutch or French, is abound with expressions pointing to connections between colours and emotions (Terwogt and Hoeksema, 1995): in English, it is for instance possible to be purple with rage, or green with envy and when we are down, we are feeling blue... In het Nederlands kunnen we groen lachen,...mais en Français on rit jaune,....

Not only language, but also a large number of studies point to the associations between colours and emotions (e.g. Wexner, 1954; Schaie, 1961, 1962; Kreitler & Kreitler, 1972; Adams and Osgood, 1973; Byrnes, 1983; Valdez and Mehrabian, 1994; Terwogt and Hoeksma, 1995; Hemphill, 1996).

Evidence that relates colour to psychological affective states is also revealed by the studies on colour preference and mood tones. Undoubtedly one of the most intensive areas of colour research is that of colour preference. This field is characterised by more than a hundred years of investigations – the first empirical study having been conducted by Cohn in 1894 (Eysenck, 1941). Since that initial effort, numerous attempts have been made to determine the relationship between feelings/emotions and colours. Sharpe (1981), Ball (1965); Norman and Scott (1952) and Beach et al. (1988) provide excellent overviews of this research. Nevertheless, much of the research on colour and affect is weak due to a number of methodological shortcomings, which will first be examined in more detail.

3.5.2. METHODOLOGICAL CAVEATS

Introducing the subject, incorrectly, Hemphill (1996) asserts that: "few studies have focused on colour-emotion associations". Clearly, this is not exactly true. Valdez and Mehrabian (1994, p394) assess the situation more accurately by pointing out that: "despite the substantial body of experimental work in this area, results have failed to provide a thorough and general characterisation of relationships between colour and affect".

Indeed, despite considerable interest and work in the field, several authors note that much of the research on colour and affect is weak due to several methodological caveats (Gelineau, 1981; Beach et al., 1988; Valdez, 1993).

According to Valdez (1993), the methodological shortcomings in this regard can be grouped into two major categories. A first major area of caveats concerns the failure of many colourstudies to provide adequate "specifications or controls of colour stimuli" (e.g., absence of controls for saturation and brightness, while investigating effects of hue). A second area of concern involves the failure to use sufficiently reliable, valid or comprehensive "measures of emotional responses" to colour stimuli. In the following paragraphs these two methodological shortcomings will be discussed more deeply.

3.5.2.1. Colour Stimuli Specifications

With respect to the first area of methodological problems related to colour stimuli, it can be noted that many studies have simply reported vague verbal descriptions of the colour samples used. A common criticism of much work on colour-responses lies in not providing adequate operational definitions for stimulus conditions and being vague about measures taken to equate brightness and saturation when investigating hue effects (Norman and Scott, 1952; Beach et al., 1988; Valdez and Mehrabian, 1994). Early studies with regard to colour-emotion associations focused almost solely on the dimension of hue. Many researchers, selected colour stimuli that they felt best represented particular hues, such as red, green, yellow or blue.

A number of studies did not even use actual colour stimuli, but instead elicited subjects' responses to verbal labels of colours (words as "red" or "black") (e.g. Aaronson, 1970; Adams and Osgood, 1973).

Also many studies have failed to relate the colour samples used, to a standardized system of colour notation, such as the Munsell Colour System or the Swedish Natural Colour System – NCS (Hard and Sivik, 1979; Hard, 1975), which have been discussed earlier (in paragraph 3.3.4.). Some researchers specified one aspect (usually hue), but failed to specify the two additional characteristics of colour - saturation and brightness - necessary for a complete description of the colour samples used (see section 3.3.). This way, researchers have been testing differences in reactions to colour samples, confounding hue, saturation and brightness effects (Beach et al., 1988; Valdez, 1993; Valdez and Mehrabian, 1994). In studies where Munsell notations were provided for standardization and comparison purposes, the actual colours chosen were often highly saturated (Nakshain, 1964; Goodfellow and Smith, 1973; Jacobs and Hustmyer, 1974).

The illumination level has also often not been adequately controlled for (Nakashima, 1909; Washburn and Grose, 1921).

When coloured lights were used as stimuli, there appears to be an inherent problem arising from attempts to control for brightness (Beach et al., 1988; Valdez and Mehrabian, 1994). Equating red and blue light for luminance does not necessarily equate them for brightness (Neri et al., 1986). In the studies reviewed, various means were used to equate or control brightness –the method of limits (Hammes and Wiggins, 1962), the method of average error (Caldwell and Jones, 1985), simply asking subjects which of two colours was the lighter one (Wilson, 1966). Some researchers only mentioned that colours were matched for brightness without indicating which method was used (Stark et al., 1982; Nourse and Welch, 1971). Other authors made no mention of brightness at all (Erwin et al., 1961; Goldstein, 1942). Gerard (1958) and Neri et al. (1986) equated luminance, which does not necessarily equate for brightness. Also few studies gave the spectral distribution (saturation) of the coloured lights used as stimuli (Beach et al., 1988). Different means by which to control brightness and saturation or the failure to control for them at all, could certainly be a factor in the inconsistent findings noted among various studies (Beach et al., 1988).
A final methodological problem relating to the colour stimuli used, concerns the presentation order of colour samples, which has often not been counterbalanced (Nakashima, 1909; Washburn, McLean and Dodge, 1934). Yet, presentation and/or order effects were noted in a number of studies (Wilson, 1966; Smets, 1969; Nourse and Welch, 1971; Caldwell and Jones, 1985) (see Beach et al., 1988, p24, for a discussion).

3.5.2.2. Colour Responses and their Assessment

A second group of methodological problems relates to responses to colour (i.e. the dependent measure). As noted by Beach et al. (1988) and Valdez and Mehrabian (1994) it is difficult to abstract general patterns of findings from the colour-emotion literature, because experimenters have used highly limited assessments of reactions to colour. Response measures were often couched in introspective reports (Bullough, 1908; Nakashima, 1909) and almost uniformly, affective responses to colour have been conceptualised as a single effect rather than a multi-dimensional response (Crowley, 1993).

Some studies have used extremely rudimentary measurement techniques, such as simply requesting that subjects rank the colour samples in terms of a single evaluative dimension "perceived pleasantness" (e.g. Eysenck, 1941; Granger, 1955; Silver & McCulley, 1988; Madden et al., 2000), which is apt to elicit a different set of associations in different subjects. Indeed, some people may associate pleasantness with comfort and relaxation (i.e. pleasure plus low arousal), whereas others may associate pleasantness with excitement and elation (i.e. pleasure plus high arousal). The use of single terms to assess emotional reactions is therefore likely to have doubtful reliability and validity (Beach et al., 1988; Valdez and Mehrabian, 1994).

Other studies involved having subjects match verbal emotion labels to different colour samples (e.g. Wexner, 1954; Hemphill, 1996).

Still other studies have used adjective checklists with dubious reliability and validity to assess emotional reactions to colour. With this respect Valdez and Mehrabian (1994) caution again that single-emotion terms that refer vaguely to discrete emotional states (e.g. "exciting" or "comfortable") have doubtful reliability for assessing emotional reactions. Furthermore, they argue that, in the absence of a theoretical system that interrelates these discrete emotional states, it is hard to compare the emotional reactions to various colours in terms of these different single emotion terms. The instrument most often used to "tap" into connotative associations of objects/events is the semantic differential, introduced in 1957 by Osgood, Suci and Tannenbaum. Instead of looking for a single relationship between colour and pleasantness, with the advent of the semantic differential, colours were rated on a series of bipolar adjective scales (semantic differentials). These scales are usually divided into seven equal points, where the middle point is seen as neutral (0) while the end points represent opposite extremes (-3, +3). By means of a series of such bipolar scales, subjects had to judge/rate a number of colours, which provided another way to look at colours in terms of affective connotative qualities. The data gathered this way are usually summarized across subjects, yielding a matrix of scale x colour averages and computing correlations between colours and scales. With a large number of scales being employed, it is often necessary to subject the data to factor analysis in order to reduce the dimensionality and seek "common factors". These factors, in turn, represent a communality among the various scaled connotations of the colours. However, the particular meanings assigned to colours through the use of semantic differentials, rely on the specific descriptor scales an author chooses in any particular study. Although factor analysis is often used to identify the underlying dimensions of colour connotations, considerable inconsistencies have been found between researchers with regard to the number and names of identified factors (Beach et al., 1988). There is also considerable evidence to suggest that the stability of colour-meaning associations is dependent on the scale's relevance to a given colour (Sivik, 1974d). For example, one would predict much more consistent ratings across subjects to colours on the adjective pair "hot ----cold", than on the adjective pair "smooth ----- adequate" (a pair reported in one of the studies). Another concern of the semantic differential is the assumed bipolar nature of the adjectives. The works of Sivik (1974c) and Green and Goldfried (1965) show that the degree of perceived opposition between the two adjectives may vary from subject to subject. Thus, wherever researchers did measure multi-dimensional emotional reactions to colour, they have used a variety of affect inventories and semantic measures, with different and nonoverlapping dimensions, which makes it hard to compare them (Beach et al., 1988).

One prevalent dimension revealed systematically in semantic differential scales is the 'evaluative' dimension (good versus bad), which is closely related to preference and pleasure ratings (see Osgood, Suci and Tannenbaum, 1957; Mehrabian & Russell, 1974).

Another frequently discovered dimension is the 'activation' dimension. From the results of early studies, various authors cite support for the premise that certain colours have the capacity to excite and arouse an individual (stimulating brainwave activity, skin conductance, etc), while others have a calming and relaxing effect. Crowley (1993) points out that colours that enhance consumers' liking of a stimulus are not necessarily those colours that consumers find activating (or deactivating). These two dimensions of consumer response are shown to have different patterns across the visible spectrum (Crowley, 1993). This conceptualisation is consistent with Mehrabian and Russell's (1974; see also Russell and Pratt, 1980) contention that dimensions of "pleasure" and "arousal" are useful in describing emotional reactions to physical environments. Crowley (1993) found support for the existence of these two distinct dimensions in the impact of colour on shopping. In a similar vain, although expressed differently, Walters et al. (1982) and Gorn et al. (1997) use a two-dimensional view of arousal (Smith and Apter, 1975; Apter, 1976, 1981, 1982), to express the fact that arousal does not always need to be pleasurable or unpleasurable. According to Gorn et al. (1997) both feelings of excitement and relaxation can be pleasant and are likely to lead to favourable attitudes. On the other extreme, feelings of boredom and tension are unpleasant and likely to generate unfavourable attitudes. In the effects obtained for value and chroma on relaxation and excitement respectively, Gorn et al. (1997) found support for the existence of these two distinct dimensions in the effects of colour as an executional cue in advertising. In an attempt to better understand the effect of colour on human behaviour, prior research has examined both arousal perception (i.e., how active a stimulus is perceived to be), as well as the actual physiological arousal engendered by exposure to various colours. Investigators have often focused their attention on the latter, i.e. the physiological responses to colour (Kaiser, 1984; Beach et al., 1988). Two common physiological measures of arousal are changes in electrical activity in the brain (EEG) and changes in skin conductance or resistance (GSR). There are numerous ways to report EEG responses: e.g. amplitude of alpha waves, duration of alpha wave suppression, amount of EEG, etc... (Kaiser, 1984, presents an excellent discussion on this issue). The same can be said about GSR data. Different authors operationally define change scores in terms of responses during different time intervals after stimulus onset. Therefore much variability can exist in what is often thought of as a rigorous response measure. When one tries to compare findings from one study to another, different types of responses might be reported under the same general response name, making such comparisons virtually impossible (Kaiser, 1984; Beach et al., 1988).

Examining emotional reactions as a function of the dimensions of colour hue, saturation and brightness, Valdez and Mehrabian (1994) found support for the Pleasure-Arousal-Dominance model (Russell and Mehrabian, 1977), which argues that these three dimensions provide a general description of emotions and represent lowest common denominators of cognition. Although dominance is a dimension of emotional response that has almost completely been ignored by colour researchers, some of them previously found colour to elicit emotional responses that cannot be captured by pleasure or arousal dimensions, but seem to infer a dimension of dominance or potency (e.g. Osgood et al., 1957; Adams and Osgood, 1973; Sivik, 1974a).

Russell and Mehrabian (1977) provide evidence that the three independent bipolar PAD dimensions (Pleasure, Arousal and Dominance) are both necessary and sufficient to adequately define emotional states (see also Mehrabian, 1996 and Mehrabian, 1998). According to Valdez and Mehrabian (1994), this PAD Emotion Model can therefore be used to compare and contrast the findings from studies that have used non-overlapping dependent measures.

In an extensive study of colour meanings of isolated colours, Wright and Rainwater (1962) had subjects rate colours on 24 semantic differential scales. Originally six factors were extracted, but intercorrelations among them led the authors to propose a final four-dimensional connotative framework of happiness, forcefulness, warmth and elegance. Note that forcefulness also seems to relate to Russell and Mehrabian's dominance dimension.

Sivik's (1974a, 1975) work provides some of the most comprehensive insights into the field of colour meaning (Beach et al., 1988). Sivik uses the Swedish Natural Colour System – NCS – (Hard and Sivik, 1979; Hard, 1975) as his descriptive colour model, instead of the Munsell colour solid, but schematically, the two are almost identical. Sivik's (1974a, 1975) initial studies investigated colours in isolation: seventy-one colours (chosen to represent the NCS colour space) were judged on 26 semantic differential scales by a stratified sample of Swedish adults. Factor analysis yielded four factors: excitement, evaluation, potency and temperature. Emerging from Sivik's (1974a, 1975) initial studies is the complexity of the interaction process between stimulus colour properties and connotative associations. The use of isosemantic mapping graphically represents that interactive process.

Instead of summarizing his data in terms of scale-colour correlations, Sivik chose to present results in the form of isosemantic maps⁷. An isosemantic map is drawn on a colour triangle, which is a full section of the NCS solid axis for a given hue value. Isosemantic maps are constructed on each colour triangle for a given connotative factor. The isosemantic lines on a colour triangle are constructed by connecting the points of equal value.

From this review of the measures used to capture emotional responses to colour stimuli, it becomes obvious that in the absence of a theoretical system, it is hard to compare the emotional reactions to various colours. The next section on colour-emotion associations includes and discusses many empirical studies in detail for the purpose of placing the accomplishments and questions of this research into proper perspective. In order to be able to compare and contrast findings from these previous studies, a comprehensive system for the description of emotions is needed. Therefore, in accordance with Valdez and Mehrabian (1994), we will use the PAD (Pleasure-Arousal-Dominance) Emotion Model as a framework to describe and interrelate the findings from studies that have used a variety of verbal-report, physiological and behavioural measures bearing on emotions.

⁷ Guilford and Smith (1959) used a similar method which they termed isohedonic charts, where colour parameters were mapped on only one scale, pleasantness. Sivik has taken this concept and expanded it to incorporate factor scores abstracted from a number of scales representing a number of different connotative dimensions.

3.5.3. COLOUR-EVOKED PLEASURE

One colour effect deals with the evaluative or affective ("liking") response to colour. Several studies have examined affective or evaluative reactions to various colours, whether it be as a single response, a uni-dimensional pleasantness construct or a pleasantness dimension in a multidimensional emotion scale. Nevertheless, much of the earlier and even of the more recent work, dealing with colour preferences, has failed to control for the three dimensions of colour – hue, saturation and brightness – and is thus methodologically flawed (e.g. Dashiell, 1917; Eysenck, 1941; Birren, 1952; Wexner, 1954; Osgood et al., 1957; Jacobs and Suess, 1975; Walters et al., 1982; Boyatzis and Varghese, 1994; Hemphill, 1996; Etnier and Hardy, 1997; Madden et al., 2000). In the next paragraphs, these studies will be more carefully reviewed. More specifically, we will first present the effects of colour hue on pleasure responses, followed by the effects of colour saturation and brightness. Finally, we will summarise the conclusions of these previous studies in a concise manner.

3.5.3.1. Effects of Hue

Red, blue and green were usually reported as most preferred colours, while orange and yellow were judged to be less preferred (Guilford, 1934; Walton, Guilford and Guilford, 1933).

Attempting to study the contributions of hue, tint and saturation on affective judgements, Guilford (1934) and Guilford and Smith (1959), conducted some of the most systematic work in this area. Using Milton Bradley coloured papers, Guilford (1934) selected 18 hues being equal in tint (value or brightness) and chroma (saturation), 11 hues varying in value and low in saturation and 11 hues high in saturation. Each colour was judged on a nine-point pleasantness scale, from greatest possible unpleasantness (1) to greatest possible pleasantness (9). Thus, controlling for brightness and saturation, his findings revealed that the hues blue, green and red were more preferred, whereas yellow and orange appeared to be less preferred.

The work of Eysenck (1941) advocated a strong argument for a universal scale of preference (Beach et al., 1988). Eysenck calculated the average ranks for six saturated colours (not equated for brightness) from a number of studies encompassing the responses of 21,060

subjects and found the following colour ranking from most to least preferred: blue, red, green, violet, orange and yellow. With this regard he refers to a general factor of *"aesthetic appreciation"*. This empirical study has been cited extensively as it provides evidence for a universal and seemingly invariant preference ordering of colour. However, Beach et al. (1988) note that Eysenck's comparisons were made only for saturated colours and that there is no evidence that experimental conditions in the various studies sampled were comparable.

Wexner (1954) conducted a study dealing with associations between colour samples and words that describe feelings. She constructed a list of adjectives judged to be best examples of mood tones, displayed eight coloured papers (8,5" x 11") and asked subjects to select the one colour that best described a given emotional state. From her study she concluded that indeed definite colour-emotion associations exist. She found the colour red to be associated with "exciting" and "stimulating", both of which imply pleasure and high arousal (Valdez and Mehrabian, 1994). Blue was associated with "secure/comfortable" and "tender/soothing", which imply pleasure and low arousal (Valdez and Mehrabian, 1994). Orange was associated with "disturbing/distressed/upset", implying displeasure and high arousal. Although Wexner neither used standard specifications of her colour samples, nor controlled for brightness and saturation, her findings are generally in accord with other empirical findings.

Granger (1955) sought a more rigorous test of Eysenck's (1941) findings. He constructed sets of stimuli from Munsell colours, where one dimension would vary, while the other two were held constant for each of the attributes of hue, value and saturation. He found wavelength to be the major factor in determining preference (with 5B blue being most preferred and 5Y yellow being least preferred) and found no interaction effect among the different colour attributes.

The findings of two studies by Osgood et al. (1957) suggest that an acceptable colour is defined by the object with which it is associated. Conducting one of the initial studies on the meaning of colours, Osgood et al. (1957) studied selected colours in the context of objects. In a first study one of six colours or an achromatic condition (black-white) was applied to either one of five objects (shirt, ice-cream, rug, car, cake mix) or as background. The object was then rated on 20 semantic differential scales, representing evaluative, activity and potency factors. Although significant differences were found on the non-evaluative scales, no consistent effects were found on the evaluative ones. A significant colour x object interaction

was found on this common "evaluative" dimension. In terms of evaluation, yellow was rated as the "most favourable" colour. This is in contrast with previous studies using colour patches, which have found yellow to be generally disliked. The authors suggested that their unexpected finding may have been an artifact of the objects used in the study. For the other colour conditions no differences were observed. In a second study, colour was studied in relation to abstract sculptures. Again no significant difference was observed for evaluative scales, while for certain non-evaluative scales significant colour differences could be discerned. As in the first study the colour x object interaction was significant. For abstract sculptures, blue was the most preferred hue.

The work of Guilford and Smith (1959) provides a rigorous examination of the roles that hue, value and saturation played in pleasantness. A ten-point pleasantness scale was used to rate 316 stimulus papers; of these 295 were coloured stimulus papers matched on the three Munsell dimensions. Across all hues, the blue-green regions had the highest affective value (pleasantness) ratings, while the lowest were observed in the yellow, yellow-green regions. This pattern existed across all levels of brightness. Their studies yielded the following rank-ordering of hues, from most to least preferred: blue, green, purple, violet, red, orange, yellow.

Lawler and Lawler (1965) found that children "coloured with a yellow crayon after hearing a happy story and with a brown crayon after hearing a sad story", revealing a close link between colours and emotions.

Hogg (1969) had subjects rate single colours and colour pairs (manipulating degree of colour contrast) on twelve 7-point semantic differential scales. For single colours, the blue-purple region was generally preferred while the yellow-green region showed low preference ratings on the evaluative dimension (pleasant – unpleasant). With regard to colour pairs, the evaluation factor accounted for 21% of the variance. The colour pair is judged as more pleasant when both colours were blue, or when blue was present than if both colours were green or when green was present.

Helson and Lansford (1970) conducted a well-controlled experiment with regard to background colour effects on perceived colour pleasantness. 125 colour chips were presented

against 25 coloured backgrounds, under five different illumination sources, to ten subjects⁸, for a total of 156,250 ratings. Each subject was shown twelve colour chips on a background and made absolute judgements on a nine-point scale. Colour chips represented main regions from each Munsell hue dimension, with low, medium and high value and chroma. Colour hues, which were judged most pleasant (across all backgrounds and light sources), were blue, purple-blue, green and blue-green. Yellow and purple were ranked low, in addition to colours with strong yellow components.

Hopson, Cogan and Batson (1971) used white, black and grey as background colours and also found blue to be preferred most and yellow-red least.

In their (1973) cross-cultural study in which subjects had to rate colour concepts (words) using the semantic differential factors (Osgood et al., 1957), Adams and Osgood found the words "blue" and "green" to be rated as good across 23 cultures, whereas "yellow" was rated as bad. Compared to achromatic colours (greytones), chromatic colours were in general rated as good.

Sivik (1974a, 1975) had a stratified sample of Swedish adults judge seventy-one colours, chosen to represent the Swedish Natural Colour System's three-dimensional colour-space. Factor analysis, performed on the 26 semantic differential scales, yielded four factors: excitement, evaluation, potency and temperature. Analysing the evaluative factor, Sivik found that across all hues, dislike varied with hue. However, there appeared to be a complex interaction between hue, the degree of whiteness, blackness and chromaticness (purity or saturation), Sivik (1974a) found that across all three parameters (blackness, whiteness, purity), there were more instances of blue being judged positive or pleasant than any other hue. This observation might account for blue often being cited as the most preferred colour.

Mehrabian and Russell (1974) found that maximal pleasure associated with colours was reported in the green to blue region of the visible spectrum.

Contrary to expectations Gotz and Gotz (1974, 1975) found some yellows to be rated highly pleasant, while pinks were seen as most unpleasant.

⁸ The small number of subjects used in this study has often been cited as a concern (Beach et al., 1988)

Jacobs and Suess (1975) investigated the effects of four primary colours (red, yellow, green and blue) on anxiety (involving displeasure and high arousal). The colour hues were projected onto a large screen, however brightness and saturation levels were not controlled for. Scores on Spielberger, Gorsuch and Lushene's (1970) State-Anxiety Inventory served as the dependent variable. From this study higher state-anxiety scores were found to be associated with red and yellow than with blue and green. This finding is consistent with studies of colour preference, demonstrating that red and yellow are less pleasant than blue and green.

Using a two-dimensional view of arousal (Smith and Apter, 1975; Apter, 1976, 1981, 1982), Walters et al. (1982) found a link between red and felt 'excitement', and blue and felt 'relaxation'. Consistent with the generally accepted view that red is an exciting colour, whereas blue is relaxing (e.g., Guilford and Smith, 1959; Tom et al., 1987). Both feelings of excitement and relaxation are pleasant and likely to lead to favourable attitudes (Gorn et al., 1997).

Similarly, Bellizzi et al. (1983) found that subjects rated cool colour store environments as more attractive and more pleasant than warm colour store environments.

Several researchers have also noted discomfort under red illumination (Howett, 1985; Luria and Kobus, 1985).

In a cross-cultural setting, Reddy and Bennett (1985) found Americans to dislike yellow and green.

In a study among 121 undergraduates, focusing on the effects of colour hue on emotions, Valdez and Mehrabian (1994, Study 2) examined the pleasure elicited by 10 hues, using 5 replication sets with equal brightness and saturation levels in order not to confuse the effects (this way a total of 50 colour samples were used). Pleasure was measured according to Mehrabian's (1978) PAD emotion scale. Valdez and Mehrabian (1994) hypothesized short-wavelength hues (e.g. blue, green) to be more pleasant than long wavelength hues (e.g. yellow, orange). Because of previous contradictory findings with regard to the pleasantness of red, they hypothesized red to score neutral on pleasantness. To explore the effects of hue (10 levels) on averaged pleasure reactions (each colour sample was rated by nearly 25 subjects), a univariate analysis of variance (ANOVA) was used. This revealed a significant .01-level main

effect: F (9,80) = 21.21. Figure 3-23 depicts a plot of the mean pleasure responses to each of the 10 hues. In this figure the two complementary non-spectral hues, purple and red-purple, are inserted separately at the right-hand side of the graph. From this plot, extreme short wavelength colours seem to elicit more pleasure than extreme long wavelength colours, whereas middle wavelength hues appear to elicit less pleasure than either the extreme short or long wavelength hues.



Figure 3-23:Mean pleasure level as a function of colour wavelengthSource:Valdez and Mehrabian, 1994, Study 2, p402

According to Tukey's Multiple Comparison Procedure, pleasure levels for blue, blue-green, green, red-purple and purple were significantly greater than those for green-yellow, yellow

and yellow-red. Furthermore, pleasure levels of purple-blue and red were significantly greater than those for green-yellow and yellow. And finally, the pleasure level of yellow-red was significantly greater than that for yellow.

A regression analysis was used to test for a possible significant curviliniar relationship of pleasure (the dependent variable) to wavelength (the independent variable) for the eight spectral colours. (The two complementary colours purple and red-purple could not be included in the regression analysis, because they are not scaled alongside noncomplementary colours with respect to wavelength, as they do not exist within the visible spectrum of light). The regression analysis tested more specifically for effects of wavelength and (wavelength)² on pleasure ratings. The results of this regression analysis can be summarized in the following equation for raw pleasure scores: multiple regression coefficient = .68, with significance of effects assessed at the .05 level.

$$Pleasure = 1561 - 5.48 (Wavelength) + .0048 (Wavelength)^{2}$$

Figure 3-23 also shows predicted pleasure scores, computed from this equation. The plot of predicted pleasure values demonstrates that the preceding equation only provides a rough approximation of the obtained means: Pleasure-displeasure reactions to spectral colours were approximately a U-shaped function of wavelength, with yellows (green-yellow, yellow and red-yellow) at the bottom portion of the U. These findings were generally consistent with the hypotheses derived from a review of the literature. Nevertheless, the findings of Valdez and Mehrabian (1994, Study 2) provided a more concise way of describing the relation between hue and pleasure: Short-wavelength hues were rated as being the most pleasant, whereas intermediate wavelength hues were rated as the least pleasant. For the long wavelength hues yellow-red and red, this trend is reversed as they show increased pleasure ratings. The non-spectral colours red purple and purple elicited high pleasure ratings, comparable to those of the short-wavelength hues.

Testing nine different colours (pink, red, yellow, black, grey, green, blue, purple and brown), presented on 8.5"x11" papers, Boyatzis and Varghese (1994) concluded that children (4-6 years old) tended to associate positive emotions with bright colours. For the analyses, colours were in fact coded as dark or bright, with bright colours being pink, red, yellow, green, purple and blue and dark colours being black, brown and grey. The colour stimuli used, were

however not exactly specified and apparently effects of hue, saturation and brightness were confounded.

Analysing the colour-emotion associations among 40 undergraduate students, Hemphill (1996) found that bright colours (including white, pink, red, yellow, blue, purple and green) elicited mainly positive emotional associations and dark colours (including brown, grey and black) elicited mainly negative emotional associations. Here again the effects of the different colour attributes appear to be confused and colour stimuli were not clearly specified.

Results of a study by Etnier and Hardy (1997) on the influence of environmental colour on task performance found no main effect of colour on affect (as measured by the Zuckerman Inventory of Personal Reactions, Zuckerman, 1977), but indicated that colour and time⁹ did interact to impact positive affect, F (8, 232) = 10.90, p<.0001 and desire to cope, F (2,58) = 15.07, p<.0001. The findings lend some support to the idea that colour influences the organism depending on the task type – cognitive versus physical (Jokl, 1982). Positive affect decreased in the warm-coloured room more so than it did in the cool- and neutral-coloured rooms prior to performance of a cognitively demanding task. Thus subjects showed a less positive emotional response towards performing a cognitive task in the warm-coloured room. With regard to the physically demanding task, positive affect decreased less in the warm room prior to task performance than it did in the cool- and neutral-coloured rooms. Thus subjects showed a more positive emotional response towards performing the physical task in the warm-coloured room.

Madden et al. (2000) conducted a study in which respondents from eight countries evaluated ten colour-swatches, presented together with the colour-name. In this study the colour stimuli were not clearly specified according to a colour system, but were probably highly saturated. Brightness and saturation levels were not controlled for. Respondents were undergraduate students from Austria (n = 29), Brazil (n = 26), Canada (n = 29), Colombia (n = 48), Hong-Kong (n = 19), PRC (n = 31), Taiwan (n = 22) and the United States (n = 49). Colours were evaluated with regard to preference and with regard to colour meaning, using 20 item semantic differential scales. The mean liking ratings for the colours in the different countries are arrayed from most to least liked for each country and presented in table 3-1.

⁹ Affect was measured at several instances: when entering the main laboratory (1), before (2) and after (3) performing a cognitive task and before (4) and after (5) performing a motor task.

Aus	stria	Bra	Brazil		ada	Colo	mbia
Colour	Mean Liking	Colour	Mean Liking	Colour	Mean Liking	Colour	Mean Liking
Blue	6.59	White	6.34	Black	5.52	Blue	6.23
Green	5.86	Blue	6.12	Blue	5.48	White	5.54
White	5.52	Green	5.35	White	5.41	Green	5.38
Black	5.21	Black	5.23	Red	5.24	Red	5.10
Red	4.34	Red	5.00	Green	5.21	Black	5.02
Orange	4.34	Brown	4.50	Purple	4.03	Purple	3.83
Yellow	3.59	Gold	4.31	Yellow	3.79	Yellow	3.63
Brown	3.97	Purple	4.27	Orange	3.55	Gold	3.52
Purple	3.24	Gold	3.77	Gold	3.55	Brown	3.21
Gold	3.17	Orange	3.58	Brown	3.03	Orange	2.90
Hong	Kong	PH	RC	Tai	wan	United	States
Hong Colour	Kong Mean Liking	PH Colour	RC Mean Liking	Tai ^s Colour	wan Mean Liking	United Colour	States Mean Liking
Hong Colour	Kong Mean Liking	Pł Colour	RC Mean Liking	Tai ⁻ Colour	wan Mean Liking	United Colour	States Mean Liking
Hong Colour White	Kong Mean Liking 5.84	PF Colour Blue	RC Mean Liking 5.81	Tai ⁻ Colour Blue	wan Mean Liking 6.27	United Colour Blue	States Mean Liking 5.90
Hong Colour White Blue	Kong Mean Liking 5.84 5.37	PF Colour Blue White	RC Mean Liking 5.81 5.65	Tai [•] Colour Blue White	wan Mean Liking 6.27 6.23	United Colour Blue Green	States Mean Liking 5.90 5.63
Hong Colour White Blue Black	Kong Mean Liking 5.84 5.37 5.21	PI Colour Blue White Black	RC Mean Liking 5.81 5.65 5.26	Tai Colour Blue White Purple	wan Mean Liking 6.27 6.23 5.18	United Colour Blue Green Black	States Mean Liking 5.90 5.63 5.31
Hong Colour White Blue Black Red	Kong Mean Liking 5.84 5.37 5.21 5.21	PF Colour Blue White Black Red	RC Mean Liking 5.81 5.65 5.26 5.16	Tai Colour Blue White Purple Black	wan Mean Liking 6.27 6.23 5.18 4.82	United Colour Blue Green Black White	States Mean Liking 5.90 5.63 5.31 5.29
Hong Colour White Blue Black Red Yellow	Kong Mean Liking 5.84 5.37 5.21 5.21 5.21 5.05	PF Colour Blue White Black Red Green	RC Mean Liking 5.81 5.65 5.26 5.16 5.13	Tai Colour Blue White Purple Black Green	wan Mean Liking 6.27 6.23 5.18 4.82 4.82 4.82	United Colour Blue Green Black White Red	States Mean Liking 5.90 5.63 5.31 5.29 5.14
Hong Colour White Blue Black Red Yellow Purple	Kong Mean Liking 5.84 5.37 5.21 5.21 5.21 5.05 5.00	PI Colour Blue White Black Red Green Gold	RC Mean Liking 5.81 5.65 5.26 5.16 5.13 4.55	Tai Colour Blue White Purple Black Green Yellow	wan Mean Liking 6.27 6.23 5.18 4.82 4.82 4.82 4.64	United Colour Blue Green Black White Red Gold	States Mean Liking 5.90 5.63 5.31 5.29 5.14 3.94
Hong Colour White Blue Black Red Yellow Purple Gold	Kong Mean Liking 5.84 5.37 5.21 5.21 5.21 5.05 5.00 4.84	PF Colour Blue White Black Red Green Gold Yellow	RC Mean Liking 5.81 5.65 5.26 5.16 5.13 4.55 4.16	Tai Colour Blue White Purple Black Green Yellow Brown	wan Mean Liking 6.27 6.23 5.18 4.82 4.82 4.82 4.64 4.59	United Colour Blue Green Black White Red Gold Purple	States Mean Liking 5.90 5.63 5.31 5.29 5.14 3.94 3.90
Hong Colour White Blue Black Red Yellow Purple Gold Green	Kong Mean Liking 5.84 5.37 5.21 5.21 5.05 5.00 4.84 4.58	PI Colour Blue White Black Red Green Gold Yellow Orange	Mean Liking 5.81 5.65 5.26 5.16 5.13 4.55 4.16 4.13	Tai Colour Blue White Purple Black Green Yellow Brown Orange	wan Mean Liking 6.27 6.23 5.18 4.82 4.82 4.82 4.64 4.59 4.36	United Colour Blue Green Black White Red Gold Purple Brown	States Mean Liking 5.90 5.63 5.31 5.29 5.14 3.94 3.90 3.80
Hong Colour White Blue Black Red Yellow Purple Gold Green Brown	Kong Mean Liking 5.84 5.37 5.21 5.21 5.05 5.00 4.84 4.58 4.11	PI Colour Blue White Black Red Green Gold Yellow Orange Purple	Mean Liking 5.81 5.65 5.26 5.16 5.13 4.55 4.16 4.13 3.87	Tai Colour Blue White Purple Black Green Yellow Brown Orange Red	Mean Liking 6.27 6.23 5.18 4.82 4.82 4.64 4.59 4.36 4.32	United Colour Blue Green Black White Red Gold Purple Brown Orange	States Mean Liking 5.90 5.63 5.31 5.29 5.14 3.94 3.90 3.80 3.27
Hong Colour White Blue Black Red Yellow Purple Gold Green Brown Orange	Kong Mean Liking 5.84 5.37 5.21 5.21 5.21 5.05 5.00 4.84 4.58 4.11 3.89	PI Colour Blue White Black Red Green Gold Yellow Orange Purple Brown	Mean Liking 5.81 5.65 5.26 5.16 5.13 4.55 4.16 4.13 3.87 3.48	Tai Colour Blue White Purple Black Green Yellow Brown Orange Red Gold	Mean Liking 6.27 6.23 5.18 4.82 4.82 4.64 4.59 4.36 4.32 4.31	United Colour Blue Green Black White Red Gold Purple Brown Orange Yellow	States Mean Liking 5.90 5.63 5.31 5.29 5.14 3.94 3.90 3.80 3.27 3.26



With respect to the liking of black, green, red and white, there was no statistically significant difference across countries. Preferences for blue, brown, gold, orange and purple did differ significantly across cultures. Kendall's coefficient of concordance was used to assess similarities in the rankings from most to least preferred colours. The coefficient of agreement ranged from .38 to .78. In three-fourths of the country pairs, there was significant agreement on colour preference rankings. Blue was, overall, the most liked colour with a mean of 6.00

(on a 7-point scale). It was rated as the most preferred colour in five of the eight countries and as the second most liked colour in the three remaining countries. With regard to the semantic differential ratings, colours and semantic items were plotted in two-dimensional perceptual maps for each country. From these plots it is also apparent that blue, green and white are well liked across countries. The results obtained by Madden et al. (2000) indicate that in many parts of the world, consumers exhibit similarities in colour liking and colour meaning associations.

Although it is obvious that a lot of the early studies on the effects of hue on pleasure are methodologically flawed -confounding brightness and saturation effects with the effects of colour hue- there appears to be some consistency in the findings. Studies of colour preference have sometimes dichotomised colours into a warm group (red, orange, yellow) and a cool group (green, blue, violet: see Sharpe 1974 for a review). Cool colours (especially blue) are consistently preferred over warm colours (cf. Silver and McCulley, 1988), although some cross cultural differences have been found (see Wiegersma and Van Der Elst, 1988). More specifically, extreme short wavelength colours (blue, green) seem to be generally preferred, followed by the extreme long wavelength colour red. Yellow-green, yellow and orange appear to be generally disliked (Walton, Guilford and Guilford, 1933; Guilford, 1934; Eysenck, 1941; Wexner, 1954; Granger, 1955; Guilford and Smith, 1959; Helson and Lansford, 1970; Hopson, Cogan and Batson, 1971; Adams and Osgood, 1973; Sivik, 1974a, 1975; Mehrabian and Russell, 1974; Jacobs and Suess, 1975). Stated more formally, pleasure-displeasure reactions to spectral colours appear to approximate a U-shaped function of wavelength, with yellows (green-yellow, yellow and red-yellow) at the bottom portion of the U. Indeed, the findings of Valdez and Mehrabian (1994, Study 2) confirm that short-wavelength hues are rated as being the most pleasant, whereas intermediate wavelength hues are rated as the least pleasant. For the long wavelength hues yellow-red and red, this trend is reversed as they show increased pleasure ratings. The works of Eysenck (1941) and Madden et al. (2000) advocate a strong argument for a universal scale of preference. Nevertheless, Osgood et al. (1957), suggest that an acceptable colour is defined by the object with which it is associated. Studying selected colours in the context of objects, they found yellow to be rated as the "most favourable" colour, which is in contrast with the general finding, using colour patches, that yellow is mostly disliked.

THE EFFECTS OF COLOUR ON EMOTIONS: Table 3-2: The Pleasure-eliciting Properties of Colours

EFFECTS OF HUI	EFFECTS OF HUE							
Citation	Sample	Design	Stimuli	Dependent variable	Findings			
Guilford (1934)		Study designed in order to be able to distinguish the	Milton Bradley coloured papers 18 hues,	Nine-point pleasantness scale	The hues Blue, Green and Red appeared to be more preferred, whereas Yellow and Orange appeared to be less preferred			
		contributions of hue, value and saturation on affective judgements	equated for brightness and saturation		<u>From most to least preferred:</u> Blue-Green-Purple-Violet-Red-Orange-Yellow			
Eysenck (1941)	Meta-analyses encompassing the responses of 21,060 subjects	For a number of studies average ranks of colour preference were calculated	Six saturated colours (not equated for brightness)	Average preference rankings	Universal preference ordering of colour, from most to least preferred: Blue-Red-Green-Violet-Orange-Yellow Eysenck refers to a general factor of " <i>aesthetic appreciation</i> ".			
Wexner (1954)		Study concerning subjects' associations between colour samples and words that describe feelings	Eight coloured papers (not specified according to standard specifications & not controlled for brightness or saturation)	Words that describe mood tones	Red was found to be associated with "exciting" and "stimulating" and Blue with "secure/comfortable" and "tender/soothing", although these emotions vary on the 'arousal-dimension', they all do describe pleasant feelings. In contrast, Orange was associated with "disturbing/distressed/upset", implying displeasure and high arousal.			

Granger (1955)			Munsell colours with varied dimensions of hue, saturation and brightness	Preference ranks	Preference varies with the wavelength of the colourWith Blue most preferred, Yellow least preferredWavelength was found to be the major factor determining preference
Osgood et al. (1957)	Studied selected colours in the context of objects	<u>Study 1</u> Colours were applied to either one of five objects (shirt, ice-cream, rug, car, cake mix) <u>Study 2</u> Colours were applied to abstract sculptures	Six colours + black and white	An evaluative (pleasant-unpleasant) dimension identified from a 20 item semantic differential scale	A significant colour x object interaction was found for the evaluative dimension. The authors therefore suggest that an acceptable colour is defined by the object with which it is associated. For the 5 studied objects, yellow was rated as the most favourable colour (which may be due to the selected objects) and no differences were observed for the other colour conditions. With regard to the abstract sculptures, no significant difference was observed for the evaluative scales. Again the colour x object interaction was significant. With abstract sculptures, blue was the most preferred hue.
Guilford and Smith (1959)		Rigorous examination of the roles of hue, value and saturation on pleasantness	316 stimulus papers, 295 of which were matched on the three Munsell dimensions	Ten-point pleasantness scale	Across all hue regions, the Blue-Green regions had the highest pleasantness ratings, while the lowest were observed in the Yellow, Yellow-Green regions. This pattern existed across all levels of brightness
Hogg (1969)			Single colours and colour pairs, manipulating degree of contrast	An evaluative (pleasant-unpleasant) dimension identified from a twelve item 7- point semantic differential scale	The Blue-Purple region was generally preferred, while the Yellow- Green region showed low preference ratings on the evaluative dimension. Colour pairs were judged more pleasant when both colours were Blue or when Blue was present than when both colours were Green, or when Green was present.

Helson and Lansford (1970)	10 subjects provided a total of 156,250 ratings	Studied background colour effects on perceived pleasantness	125 colour chips, representing the main Munsell hue dimension regions, with low medium and high value and chroma, presented against 25 coloured backgrounds	Nine-point scale absolute preference judgements for coloured chips against backgrounds	Coloured chips against various coloured backgrounds judged most pleasant were: Blue, Purple-Blue, Green and Blue-Green. On the other hand Purple, Yellow and colours with strong Yellow components were ranked low.
Hopson, Cogan and Batson (1971)			Colours presented against white, black and grey backgrounds		Against white, black and grey backgrounds, Blue was found to be preferred most and Yellow-Red least.
Adams and Osgood (1973)	Subjects across 23 cultures		Colour names (words)	An evaluative (pleasant-unpleasant) dimension identified from a semantic differential scale	The words "Blue" and "Green" were found to be rated as good across 23 cultures, whereas the word "Yellow" was rated as bad.
Sivik (1974a)	A stratified sample of Swedish adults	Studied colours in isolation	71 colours chosen to represent the NCS colour space (based on the SwedishNatural Colour System)	An evaluative dimension, identified from a 26 item semantic differential scale	Although there appeared to be a complex interaction between hue, the degree of whiteness, blackness and purity, across all parameters, dislike appeared to vary with hue: there were more instances of blue being judged positive or pleasant than for any other hue. This observation might account for blue often being cited as the most preferred colour.
Mehrabian and Russell (1974)					Maximal pleasure associated with colours was found in the Green to Blue region of the visible spectrum.

Gotz and Gotz (1974, 1975)					Contrary to expectations, some Yellows were rated highly pleasant, while Pinks were seen as most unpleasant
Jacobs and Suess (1975)			Red Yellow Green Blue not controlled for brightness or saturation (projected onto a large screen)	Anxiety (involving high arousal and displeasure) (Spielberger, Gorsuch and Lushene's 1970 State-Anxiety Inventory)	Higher "anxiety" scores (involving displeasure) were found to be associated with Red and Yellow than with Blue and Green.
Walters et al. (1982)			(chroma and saturation were not controlled for)		Red was related to "felt excitement" and Blue was related to "felt relaxation".
Bellizzi et al. (1983) Experiment 2	Convenience sample 125 females	<u>Colour as a</u> <u>periferal cue</u> Laboratory experiment Manipulated background colours of projected furniture store interior	Red Yellow Green Blue & White (full strength hues)	Evaluation dimension of 20 item semantic differential scores	Subjects rated cool colour store environments as more attractive and more pleasant than warm colour store environments.
Howett (1985) Luria and Kobus (1985)					Discomfort appears to be experienced under Red illumination
Reddy and Bennett (1985)					Americans dislike Yellow and Green
Silver and McCulley (1988)				Colour preferences	Cool colours (especially blue) appear to be consistently preferred over warm colours (red, orange, yellow).

Bellizzi and Hite (1992) Experiment 2	Convenience sample 107 graduate marketing students	Laboratory experiment Manipulated background colours of projected furniture store interior	Red Blue (brightness and saturation not controlled)	Pleasure dimension of 18 item PAD scale (Mehrabian and Russell, 1974 as adapted by Donovan and Rossiter, 1982)	The Blue store environment was found to be more pleasurable than the Red store environment.
Crowley (1993)	Convenience sample 100 females	Laboratory experiment Manipulated background colours of projected furniture store interior	Blue Green Yellow Red (fully saturated)	Evaluation dimension of 20 item semantic differential scores	Found support for the hypothesis that consumer evaluations exhibit an increasingly positive linear trend from longer (warmer) to shorter (cooler) wavelengths.
Valdez and Mehrabian (1994) – Study 2	121 undergraduate students	Laboratory experiment Colour patches	50 colour samples: 10 hues, using 5 replication sets with equal brightness and saturation levels.	Pleasure, measured according to Mehrabian's (1978) PAD emotion scale	Pleasure levels for Blue, Blue-Green, Green, Red-Purple and Purple were significantly greater than those for Green-Yellow, Yellow and Yellow-Red. The pleasure level of Yellow-Red was significantly greater than that for Yellow. In fact a curvilinear relation seems to exist between colour wavelength and elicited pleasure-level: pleasure-displeasure reactions to spectral colours were approximately a U-shaped function of wavelength, with Yellows (Green-Yellow, Yellow and Red-Yellow) at the bottom portion of the U. Extreme short wavelength colours seem to elicit more pleasure than extreme long wavelength colours, whereas middle wavelength hues appear to elicit less pleasure than either the extreme short or long wavelength hues.

Etnier and Hardy (1997)		<u>Colour as a</u> <u>peripheral cue</u> Investigated the effect of environmental colour on the performance of cognitive and motor-tasks	Green, white or orange coloured room (not specified)	Positive affect (as measured by the Zuckerman Inventory of Personal Reactions 1977)	No main effect of room-colour on affect could be discerned, however, colour and time did appear to interact to impact positive affect. Positive affect decreased in the warm-coloured room more so than it did in the cool and neutral coloured rooms prior to performance of a cognitively demanding task. Thus subjects showed a less positive emotional response towards performing a cognitive task in the warm room. Positive affect decreased less in the warm room prior to performance of a physically demanding task than it did in the cool and neutral coloured rooms. Thus subjects showed a more positive emotional response towards performing a physical task in the warm room.
Madden et al. (2000)	Sample of undergraduate students from eight countries: Austria ($n = 29$), Brazil ($n = 26$), Canada ($n = 29$), Colombia ($n = 48$), Hong-Kong ($n = 19$), PRC ($n = 31$), Taiwan ($n = 22$) and the United States ($n = 49$).	Colour patches with colour names were evaluated by respondents	Black, blue, brown, gold, green, orange, purple, red, white and yellow. (standard colour shading options in Lotus Almipro word processing software, not clearly specified according to a colour system, probably highly saturated with brightness and saturation levels not controlled)	Colour preference ratings (like-dislike on a 7-point scale) Colour meaning (20 item semantic differential scale)	 Blue was, overall, the most liked colour. It was rated as the most preferred colour in five of the eight countries and as the second most liked colour in the three remaining countries. With regard to the semantic differential ratings, colours and semantic items were plotted in two-dimensional perceptual maps for each country. From these plots it is also apparent that blue, green and white are well liked across countries. The obtained results indicate that in many parts of the world, consumers exhibit similarities in colour liking and colour meaning associations. In three-fourths of the country pairs, there was significant agreement on colour preference rankings. With respect to the liking of black, green, red and white, there was no statistically significant difference across countries. Preferences for blue, brown, gold, orange and purple did differ significantly across cultures.

3.5.3.2. Effects of Saturation

Although early colour preference studies focused almost solely on the dimension of hue, Washburn (1911) found that the affective judgements to saturated colours were more positive than those assigned to tints (lighter colours) or shades (darker colours). However, it was not until the work of Guilford (1934) that any systematic attempt was made to study the contributions of hue, tint and saturation on affective judgements.

Attempting to study the contributions of hue, tint and saturation on affective judgements, Guilford (1934) and Guilford and Smith (1959), conducted some of the most systematic work in this area. Using Milton Bradley coloured papers, Guilford (1934) selected 18 hues being equal in tint (value or brightness) and chroma (saturation), 11 hues varying in value and low in saturation and 11 hues high in saturation. Each colour was judged on a nine-point pleasantness scale, from greatest possible unpleasantness (1) to greatest possible pleasantness (9). Saturated colours were found to be preferred over unsaturated ones.

Granger (1955) constructed sets of stimuli from Munsell colours, where one dimension would vary, while the other two were held constant for each of the attributes of hue, value and saturation. He found wavelength to be the major factor in determining preference (with 5B blue being most preferred and 5Y yellow being least preferred) but also noted that as saturation increased, so did preference, up to a point. Colours seen as too vivid were less preferred. Granger did not find an interaction effect among the different colour attributes.

The work of Guilford and Smith (1959) also provides a rigorous examination of the roles that hue, value and saturation play in pleasantness. A ten-point pleasantness scale was used to rate 316 stimulus papers; of these 295 were coloured stimulus papers matched on the three Munsell dimensions. Guilford and Smith found that the greater the saturation within a given hue, the greater its perceived pleasantness appeared to be, with the relationships tending to be curvilinear. Moreover, Guilford and Smith observed that colours at the levels of brightness which can be the most saturated, were most liked. Their data were presented in terms of isohedonic charts¹⁰. Isohedons, or lines of uniform affective value, for a given hue were plotted across varying values of brightness (y-axis) and saturation (x-axis). Representing the

¹⁰ A variation on this technique is isosemantic mapping (Sivik, 1975, 1974a), where colour is plotted in a similar manner in relation to semantic differential values.

data in this manner allowed predicting a likely pleasantness rating for colours defined by Munsell coordinates. With their laboratory experiments, Guilford and Smith (1959) demonstrated the complexity of the relationship between the three dimensions of colour as they relate to pleasantness.

Sampling 24 colours, Smets (1982) examined the contributions of brightness and saturation to pleasantness ratings in greater detail. She found all three attributes contributed to ratings along this dimension, with saturation accounting for 88% of the variance, whereas brightness accounted for only 12% of the variance.

In a comprehensive study of emotional reactions to colour among 250 undergraduates, Valdez and Mehrabian (1994) also found support for the hypothesis that pleasure is a positive correlate of saturation. Stepwise multiple regression analysis was used to explore the contributions of brightness and saturation to pleasure. The pleasure elicited by 76 colour stimuli was measured by means of Mehrabian's (1978) verbal-report PAD (Pleasure, Arousal, Dominance) scales. Averaged pleasure responses (each colour sample was rated by approximately 25 subjects) constituted the dependent variable, and brightness and saturation the independent variables in the regression analysis, yielding the following equation:

Pleasure = .69 Brightness + .22 Saturation

From this study, a colour's brightness and saturation levels appear to explain a substantial portion of the variance in the feelings of pleasure it induces (multiple regression coefficient = .69). As expected more saturated (more vivid and purer) colours induced greater feelings of pleasure in viewers. However, the regression analysis indicates the effect of brightness to be considerably stronger than the effect of saturation in determining pleasure responses to colour. The differential magnitudes of these two effects had not been anticipated based on previous research. Valdez and Mehrabian (1994) found no second-order curvilinear relationship between colour saturation and the elicited pleasure.

In their study involving 156 university undergraduates, Gorn et al. (1997) manipulated the colour of a big "swoosh" in the centre of an ad for a fictitious paint company, "Rainbow Paints", according to a 2 by 2 by 2 factorial design (hue x chroma x value). Using a two-dimensional view on arousal (Smith and Apter, 1975; Apter, 1976, 1981, 1982; Walters et al.,

1982), Gorn et al. (1997) found that respondents exposed to ads containing higher chroma colours reported a greater liking for the ad, and that this effect was mediated by greater feelings of excitement (involving high pleasure and high arousal) experienced. The greater feelings of excitement experienced involve a strong pleasure component, suggesting that increased saturation of colours may increase feelings of pleasure.

THE EFFECTS OF COLOUR ON EMOTIONS: Table 3-3: The Pleasure-eliciting Properties of Colours

EFFECTS OF SAT	URATION				
Citation	Sample	Design	Stimuli	Dependent variable	Findings
Washburn (1911)					Found that the affective judgements to saturated colours were more positive, than those assigned to tints or shades (which are less saturated).
Guilford (1934)		Study designed in order to be able to distinguish the contributions of hue, value and saturation on affective judgements	Milton Bradley coloured papers 11 hues varying in value and low in saturation and 11 hues high in saturation	Nine-point pleasantness scale	Found saturated colours to be preferred over unsaturated colours
Granger (1955)			Munsell colours with varied dimensions of hue, saturation and brightness	Preference ranks	Although wavelength was found to be the major factor determining preference, Granger also noted that, as saturation increased, so did preference, up to a point. Colours seen as too vivid were less preferred. Granger did not find an interaction effect among the different colour attributes.
Guilford and Smith (1959)		Rigorous examination of the roles of hue, value and saturation on pleasantness	316 stimulus papers, 295 of which were matched on the three Munsell dimensions	Ten-point pleasantness scale	They demonstrated the complexity of the relationship between the three dimensions of colour as they relate to pleasantness. Nevertheless, they found that the greater the saturation within a given hue, the greater its perceived pleasantness appeared to be, with the relationship tending to be curvilinear. They also observed that colours at the levels of brightness which can be the most saturated, were most liked.

Smets (1982)			24 Munsell colours, varying in saturation and brightness		Found saturation to account for 88% of the variance in pleasantness ratings.
Valdez and Mehrabian (1994) – Study 1	250 undergraduate students	Laboratory experiment with colour patches	76 colour samples : 10 hues, using 5 replication sets with equal brightness and saturation levels.	Pleasure, measured according to Mehrabian's (1978) PAD emotion scale	Found support for the hypothesis that pleasure is a positive correlate of saturation. As expected more saturated (more vivid and purer) colours induced greater feelings of pleasure. A colour's saturation level appears to explain a substantial portion of the variance in the feelings of pleasure it induces, however a regression analysis indicates the effect of brightness to be considerably stronger. No support was found for a second-order curvilinear relationship between saturation and the elicited pleasure.
Gorn et al. (1997)	156 university undergraduates	Manipulated the colour of a big "swoosh" in an advertisement	4 blue and 4 red colour samples, chosen according to a 2x2x2 factorial design (hue x chroma x value)	Feelings of excitement (involving high pleasure and high arousal)	Found that respondents exposed to ads containing higher chroma colours reported a greater liking for the ad and that this effect was mediated by greater feelings of excitement (involving high pleasure and high arousal) experienced.

3.5.3.3. Effects of Value

The case of achromatic colours:

With regard to achromatic colours such as white, grey and black (representing brightness variations only), Adams and Osgood (1973) found in their cross-cultural study, in which subjects had to rate colour concepts (words) using the semantic differential factors (Osgood et al., 1957), that the words grey and black were rated as bad, whereas the concept white was rated as good.

Birren (1978) and Sharpe (1974) found the colour black to be associated with anxiety, which implies displeasure and high arousal.

Apparently not controlling for value or chroma, Boyatzis and Varghese (1994) found that children tended to associate negative emotions with dark colours, such as black and grey.

Valdez and Mehrabian (1994, Study 3) used five achromatic colour samples (ranging from white to black) to explore the effect of brightness in achromatic colours on pleasure responses. A regression analyses with average pleasure (25 subjects rated each sample) as the dependent variable and brightness as the independent variable revealed the following equation, expressing the relationship of pleasure to brightness of achromatic colours (standardized variables, multiple regression coefficient = .71):

As hypothesized a positive relation could be revealed between brightness of achromatic colours and pleasure. Pleasure reactions increased as colour samples ranged from black, over greys of increasing brightness on to white. As expected black was rated as least pleasant, greys were assigned intermediate levels of pleasantness and white was perceived to be the most pleasant. The relationship between brightness and pleasantness was very strong (as can be determined by the beta weight of .71 in the preceding equation) and highly significant.

The case of chromatic colours

Although early colour preference studies focused almost solely on the dimension of hue, Washburn (1911) found that the affective value of tints (lighter colours) was preferred to shades (darker ones). However, it was not until the work of Guilford (1934) that any systematic attempt was made to study the contributions of hue, tint and saturation on affective judgements.

Eysenck (1941) calculated the average ranks for six saturated colours (not equated for brightness) from a number of studies encompassing the responses of 21,060 subjects. He noted that there appeared to be a direct relationship between liking of a given colour and its differentiation from white. In contrast to Washburn's (1911) findings, Eysenck's tentative conclusion was that preference of any colour varies inversely with the luminosity factor of that colour. Note however that his study involved only saturated colours.

Attempting to study the contributions of hue, tint and saturation on affective judgements, Guilford (1934) and Guilford and Smith (1959), conducted some of the most systematic work in this area. Using Milton Bradley coloured papers, Guilford (1934) selected 18 hues being equal in tint (value or brightness) and chroma (saturation), 11 hues varying in value and low in saturation and 11 hues high in saturation. Each colour was judged on a nine-point pleasantness scale, from greatest possible unpleasantness (1) to greatest possible pleasantness (9). From this study, Guilford (1934) found lighter colours to be preferred over darker ones.

The work of Guilford and Smith (1959) also provides a rigorous examination of the roles that hue, value and saturation play in pleasantness. A ten-point pleasantness scale was used to rate 316 stimulus papers; of these 295 were coloured stimulus papers matched on the three Munsell dimensions. Guilford and Smith found that as brightness increased, perceived pleasantness also increased, with the relationships tending to be curvilinear. However, a preferred hue pattern appeared to exist across all levels of brightness. Guilford and Smith also observed that colours at the levels of brightness which can be the most saturated, were most liked. Their data was presented in terms of isohedonic charts. With their laboratory experiments, Guilford and Smith (1959) demonstrated the complexity of the relationship between the three dimensions of colour as they relate to pleasantness.

In their (1973) cross-cultural study in which subjects had to rate colour concepts (words) using the semantic differential factors (Osgood et al., 1957), Adams and Osgood found that evaluation correlated in general strongly and positively with brightness.

In Sivik's (1974a, 1975) analysis of the evaluative factor, there appeared to be a complex interaction between hue, the degree of whiteness, blackness and chromaticness (purity or saturation). Generally, as the degree of blackness increased, colours were judged to be of a lower pleasantness or less preferred. Although this is a consistent finding, across all hues, dislike varied with hue. Sivik (1974a) found that across all three parameters (blackness, whiteness, purity), there were more instances of blue being judged positive or pleasant than any other hue. For example, to obtain the same degree of judged dislike between a dark blue and a dark yellow (brown), the blue must be much darker (have more blackness) than the yellow. Emerging from Sivik's (1974a, 1975) initial studies is the complexity of the interaction process between stimulus colour properties and connotative associations. Through the use of isosemantic mapping he graphically represents that interactive process.

Sampling 24 colours, Smets (1982) examined the contributions of brightness and saturation to pleasantness ratings in greater detail. She found all three attributes contributed to ratings along this dimension, with brightness accounting for 12% of the variance, whereas saturation accounted for 88% of the variance.

In a comprehensive study of emotional reactions to colour among 250 undergraduates, Valdez and Mehrabian (1994) also found support for the hypothesis that pleasure is a positive correlate of brightness. Stepwise multiple regression analysis was used to explore the contributions of brightness and saturation to pleasure. The pleasure elicited by 76 colour stimuli was measured by means of Mehrabian's (1978) verbal-report PAD (Pleasure, Arousal, Dominance) scales. Averaged pleasure responses (each colour sample was rated by approximately 25 subjects) constituted the dependent variable, and brightness and saturation the independent variables in the regression analysis, yielding the following equation:

From this study, a colour's brightness and saturation levels appear to explain a substantial portion of the variance in the feelings of pleasure it induces (multiple regression coefficient =

.69). As expected brighter (less dark) colours induced greater feelings of pleasure in viewers. Moreover, in contrast to Smet's (1982) findings, the regression analysis indicates the effect of brightness to be considerably stronger than the effect of saturation in determining pleasure responses to colour. The differential magnitudes of these two effects had not been anticipated based on previous research. Valdez and Mehrabian (1994) found no second-order curvilinear relationship between colour brightness and the elicited pleasure.

In their study involving 156 university undergraduates, Gorn et al. (1997) manipulated the colour of a big "swoosh" in the centre of an ad for a fictitious paint company "Rainbow Paints" according to a 2 by 2 by 2 factorial design (hue x chroma x value) but found (for red and blue hues) no support for the hypothesis that lower value, darker colours generate more unpleasant feelings (as measured by the items "unhappy", "irritated", "annoyed", loadings=0.74, 0.92, 0.90; Cronbach α =0.82). However, using a two-dimensional view on arousal (Smith and Apter, 1975; Apter, 1976, 1981, 1982; Walters et al., 1982), they did find that those exposed to ads containing higher value colours reported a greater liking for the ad and through this effect also a greater liking of the brand. The former effect appeared to be mediated by greater experienced feelings of relaxation (involving high pleasure and low arousal). The greater feelings of relaxation experienced involve a strong pleasure component, suggesting that increased lightness of colours may nevertheless increase feelings of pleasure.

THE EFFECTS OF COLOUR ON EMOTIONS: Table 3-4: The Pleasure-eliciting Properties of Colours

EFFECTS OF BRIGHTNESS Citation Sample Stimuli **Dependent** variable Findings Design The case of achromatic colours The words "Grey" and "Black" were found to be rated as bad across 23 Adams and Subjects Colour names An evaluative cultures, whereas the concept "White" was rated as good **Osgood** (1973) across 23 (words) (pleasant-unpleasant) dimension identified cultures from a semantic differential scale Birren (1978) Found the colour black to be associated with anxiety, implying and Sharpe displeasure and high arousal. (1974) **Boyatzis and** Not controlling Children tend to associate negative emotions with dark colours, such as Varghese (1994) for value or black and grey. chroma As hypothesized, a strong positive relation could be revealed between Valdez 125 Laboratory 5 achromatic Pleasure, measured &Mehrabian undergraduate experiment colour samples according to brightness of achromatic colours and pleasure. Pleasure reactions Colour patches (ranging from Mehrabian's (1978) increased as colour samples ranged from black, over greys of (1994) - Study 3students increasing brightness on to white, with black receiving the lowest and white to black) PAD emotion scale white receiving the highest pleasantness scores.

The case of chromatic colours

Washburn (1911)					Found that the affective value of tints (lighter colours) was preferred to shades (darker colours).
Guilford (1934)		Study designed in order to be able to distinguish the contributions of hue, value and saturation on affective judgements	Milton Bradley coloured papers 11 hues varying in value and low in saturation and 11 hues high in saturation	Nine-point pleasantness scale	Found lighter colours to be preferred over darker ones.
Eysenck (1941)	Meta-analyses encompassing the responses of 21,060 subjects	For a number of studies average ranks of colour preference were calculated	Six saturated colours (not equated for brightness)	Average preference rankings	Notes that there appears to be a direct relationship between liking of a given colour and its differentiation from white. His tentative conclusion was that preference of any colour varies inversely with the luminosity factor of that colour.
Guilford and Smith (1959)		Rigorous examination of the roles of hue, value and saturation on pleasantness	316 stimulus papers, 295 of which were matched on the three Munsell dimensions	Ten-point pleasantness scale	They demonstrated the complexity of the relationship between the three dimensions of colour as they relate to pleasantness. Nevertheless, they found that as brightness increased, perceived pleasantness also increased, with the relationship tending to be curvilinear
Lawler and Lawler (1965)			Happy versus sad story	Crayons chosen by children for colouring	Found that children coloured with a yellow crayon after hearing a happy story and with a brown crayon after hearing a sad story.
Adams and Osgood (1973)	Subjects across 23 cultures		Colour names (words)	An evaluative (pleasant-unpleasant) dimension identified from a semantic differential scale	Found that evaluation correlated in general strongly and positively with brightness.

Sivik (1974a)	A stratified sample of Swedish adults	Studied colours in isolation	71 colours chosen to represent the NCS colour space (based on the Swedish Natural Colour System)	An evaluative dimension, identified from a 26 item semantic differential scale	In Sivik's analysis of the evaluative factor, there appeared to be a complex interaction between hue, the degree of whiteness, blackness and purity. Generally as the degree of blackness increased, colours were judged to be of a lower pleasantness or less preferred. Although this is a consistent finding across all hues, dislike varied with hue. For example: to obtain the same degree of judged dislike between a dark blue and a dark yellow (brown), the blue must be much darker than the yellow as blue is generally more preferred.
Smets (1982)			24 Munsell colours, varying in saturation and brightness		Found brightness to account for 12% of the variance in pleasantness ratings.
Boyatzis and Varghese (1994)			Not controlling for value or chroma		Children tend to associate positive emotions with light colours, such as blue and yellow.
Valdez and Mehrabian (1994) – Study 1	250 undergraduate students	Laboratory experiment with colour patches	76 colour samples : 10 hues, using 5 replication sets with equal brightness and saturation levels.	Pleasure, measured according to Mehrabian's (1978) PAD emotion scale	Found support for the hypothesis that pleasure is a positive correlate of brightness. As expected brighter (less dark) colours induced greater feelings of pleasure. A colour's brightness level appears to explain a substantial portion of the variance in the feelings of pleasure it induces. A regression analysis indicates the effect of brightness to be considerably stronger than that of saturation No support was found for a second-order curvilinear relationship between brightness and the elicited pleasure.
Hemphill (1996)	40 undergraduate students		Not controlling for value or chroma	Colour-emotion association	Among adults (undergraduate students) bright colours (including white, pink, red, yellow, blue, purple and green) elicited mainly positive emotional associations and dark colours (including brown, grey and black) elicited mainly negative emotional associations.

Gorn et al. (1997)	156 university undergraduates	manipulated the colour of a big "swoosh" in an advertisement	4 blue and 4 red colour samples, chosen according to a 2x2x2 factorial design (hue x chroma x value)	Unpleasant feelings measured by the items: unhappy, irritated and annoyed (α =.82) Feelings of relaxation (involving high pleasure and low arousal)	Found no support for the hypothesis that lower value, darker colours generate more unpleasant feelings. However, they did find that those exposed to ads containing higher value colours reported a greater liking of the ad and through this effect also a greater liking of the brand. This effect appeared to be mediated by greater feelings of relaxation (involving high pleasure and low arousal) experienced.

3.5.3.4. Overview

Although it is obvious that a lot of the early studies on the effects of hue on pleasure are methodologically flawed - confounding brightness and saturation effects with the effects of colour hue - there appears to be some consistency in the findings. Extreme short wavelength colours (blue, green) seem to be generally preferred, followed by the extreme long wavelength colour red. Yellow-green, yellow and orange appear to be generally disliked (Walton, Guilford and Guilford, 1933; Guilford, 1934; Eysenck, 1941; Wexner, 1954; Granger, 1955; Guilford and Smith, 1959; Helson and Lansford, 1970; Hopson, Cogan and Batson, 1971; Adams and Osgood, 1973; Sivik, 1974a, 1975; Mehrabian and Russell, 1974; Jacobs and Suess, 1975). Stated more formally, pleasure-displeasure reactions to spectral colours appear to approximate a U-shaped function of wavelength, with yellows (greenyellow, yellow and red-yellow) at the bottom portion of the U. Indeed, the findings of Valdez and Mehrabian (1994, Study 2) confirm that short-wavelength hues are rated as being the most pleasant, whereas intermediate wavelength hues are rated as the least pleasant. For the long wavelength hues yellow-red and red, this trend is reversed as they show increased pleasure ratings. The works of Eysenck (1941) and Madden et al. (2000) advocate a strong argument for a universal scale of preference. Nevertheless, Osgood et al. (1957), suggest that an acceptable colour is defined by the object with which it is associated. Studying selected colours in the context of objects, they found yellow to be rated as the "most favourable" colour, which is in contrast with the general finding, using colour patches, that yellow is mostly disliked.

Although Washburn (1911) already suggested affective judgements to more saturated colours to be more positive, it was not until the work of Guilford (1934) that any systematic attempt was made to study the contribution of colour saturation to evoked pleasure. In this and subsequent studies, saturated colours were systematically found to be preferred over unsaturated ones. Granger (1955) noted that as saturation increased, so did preference, but only up to a certain point. He found colours seen as too vivid to be less preferred. Guilford and Smith (1959) also found that the greater the saturation within a given hue, the greater its perceived pleasantness appeared to be, with the relationships tending to be curvilinear. Smets (1982) found saturation to account for 88% of the variance in pleasure induced, whereas brightness accounted for only 12% of the variance. Although Valdez and Mehrabian (1994) also found support for the hypothesis that pleasure is a positive correlate of saturation,

contrary to Smets' (1982) findings, they found the effect of brightness to be considerably stronger than the effect of saturation. Moreover, contradicting the findings by Granger (1955) and Guilford and Smith (1959), Valdez and Mehrabian (1994) did not find a second-order curvilinear relationship between colour saturation and the elicited pleasure. Gorn et al. (1997) found that respondents exposed to ads containing higher chroma colours reported a greater liking for the ad, and that this effect was mediated by the experience of greater feelings of excitement, which involve pleasant feelings of high arousal.

With regard to achromatic colours such as white, grey and black (representing brightness variations only), empirical findings also reveal consistent results. Consistent with previous findings (Adams and Osgood, 1973; Sharpe, 1974; Birren, 1978; Boyatzis and Varghese, 1994), Valdez and Mehrabian (1994, Study 3) could confirm the hypothesised positive relation between brightness of achromatic colours and pleasure. Pleasure reactions increased as colour samples ranged from black, over greys of increasing brightness on to white. Black was rated as least pleasant, greys were assigned intermediate levels of pleasantness and white was the most pleasant. The relationship between brightness and pleasantness was very strong and highly significant.

Consistent with the effect of brightness for achromatic colours, lighter chromatic colours also seem to be preferred over darker ones. Although for saturated colours, Eysenck (1941) came to the tentative conclusion that preference of any colour varies inversely with the luminosity factor of that colour, more systematic studies investigating the contribution of brightness found exactly the opposite to be true: that indeed lighter colours are preferred to darker ones (Washburn, 1911; Guilford, 1934; Guilford and Smith, 1959, Adams and Osgood, 1973). Guilford and Smith (1959) found that, as brightness increased, perceived pleasantness also increased, with the relationships tending to be curvilinear. They also demonstrated the complexity of the relationship between the three dimensions of colour as they relate to pleasantness. In Sivik's (1974a, 1975) analysis of the evaluative factor, there also appeared to be a complex interaction between hue, the degree of whiteness, blackness and chromaticness (purity or saturation). However, generally, as the degree of blackness increased, colours were judged to be of a lower pleasantness or less preferred. Although this is a consistent finding, across all hues, dislike varied with hue (e.g. to obtain the same degree of judged dislike between a dark blue and a dark yellow (brown), the blue must be much darker (have more blackness) than the yellow). Examining the contribution of colour brightness to pleasantness
in greater detail, Smets (1982) found brightness to account for only 12% of induced pleasure variance, whereas saturation accounted for 88% of the variance. As expected, Valdez and Mehrabian (1994) also found that brighter (less dark) colours induced greater feelings of However, in contrast with Smets' (1982) findings, Valdez and pleasure in viewers. Mehrabian (1994) found the effect of brightness to be considerably stronger than the effect of saturation in determining pleasure responses to colour. Unlike Guilford and Smith (1959), Valdez and Mehrabian (1994) found no second-order curvilinear relationship between colour brightness and the elicited pleasure. Manipulating the colour of a big "swoosh" in the centre of an ad, Gorn et al. (1997) did not find support for the hypothesis that lower value or darker colours generate more unpleasant feelings (for red and blue hues). However, they did find that those exposed to ads containing higher value colours reported a greater liking of the ad and through this effect also a greater liking of the brand. The former effect appeared to be mediated by greater feelings of relaxation, which besides low arousal, also involve pleasure, thus suggesting that increased lightness of colours may nevertheless increase feelings of pleasure.

3.5.4. COLOUR-EVOKED AROUSAL

It has been widely accepted that people's feelings are affected by colour. In particular, colour is believed to affect the degree of felt arousal (e.g., Walters et al., 1982; Mikellides, 1990). From the results of early studies, various authors cite support for the premise that certain colours have the capacity to excite and arouse an individual, while others have a calming and relaxing effect.

In this regard Crowley (1993), points out that colours that enhance consumers' liking of a stimulus are not necessarily those colours that consumers find activating (or deactivating). These two dimensions of consumer response are shown to have different patterns across the visible spectrum. Referring to the theory of psychological reversals (Smith & Apter, 1975; Apter, 1976, 1981, 1982; Walters et al., 1982), Gorn (1997) also argues that arousal can be experienced as pleasant (i.e. excitement) or unpleasant (i.e. tension) and that a lack of arousal can also be experienced as pleasant (i.e. relaxing) or unpleasant (i.e. boring). In the consumer behaviour literature, colour arousal effects have only recently been addressed (Crowley, 1993; Valdez and Mehrabian, 1994; Gorn, 1997).

In an attempt to better understand the effect of colour on human behaviour, prior research has examined both arousal perception (i.e., how active a stimulus is perceived to be), as well as the actual physiological arousal engendered by exposure to various colours. Investigators have often focused their attention on the latter, i.e. the physiological responses to colour (Kaiser, 1984; Beach et al., 1988).

Two common physiological measures of arousal are changes in electrical activity in the brain (Electro Encephalogram - EEG) and changes in skin conductance or resistance (Galvanic Skin Response - GSR). With increased arousal, skin conductance increases while skin resistance decreases. Alpha waves are patterns of neural activity found during periods of relaxed wakefulness that decrease and become desynchronised during periods of arousal. This process is known as the alpha attenuation response, AAR (Ali, 1972). Thus arousal is often measured by changes in alpha wave frequency and/or amplitude (Kaiser, 1984; Wilson and Wilson, 1959). There are numerous ways to report EEG responses: e.g. amplitude of alpha waves, duration of alpha wave suppression, amount of EEG, etc... (Kaiser, 1984, presents an excellent discussion on this issue). The same can be said about GSR data. Different authors

operationally define change scores in terms of responses during different time intervals after stimulus onset. Therefore much variability can exist in what is often thought of as a rigorous response measure. When one tries to compare findings from one study to another, different types of responses might be reported under the same general response name, making such comparisons virtually impossible (Kaiser, 1984; Beach et al., 1988). Moreover, some of the original laboratory studies have been criticized because of their small and non-representative sample sizes, the failure to operationally define stimulus variables, confounding of stimulus and experimental design variables and extrapolating beyond the evidence provided by the data (Beach et al., 1988; Kaiser, 1984; Sharpe, 1981; Sucov, 1973; Norman and Scott, 1952).

Nevertheless, it has been generally acknowledged that some colours are more arousing and activating (stimulating brainwave activity, skin conductance, etc) while other colours tend to be more calming and deactivating.

These studies regarding the effects of colour on arousal will be reviewed in the next paragraphs. More specifically, we will first present the effects of colour hue on arousal responses, followed by the effects of colour saturation and brightness. Finally, the conclusions of these previous studies will be concisely summarized.

3.5.4.1. Effects of Hue

Physiological responses

Studies involving physiological reactions to colour have been motivated largely by the hypothesis that long-wavelength colours (e.g., red and yellow) are more arousing to short wavelength colours (e.g., blue and green).

According to Beach et al. (1988), Goldsstein (1942) is the most cited work in terms of how colour affects behaviour. Based on his observations while conducting experiments on a small number of patients (3-5) suffering from organic diseases of the central nervous system, Dr. Goldstein postulated an affective theory of colour, which he believed applied to all individuals. From his work with brain-damaged patients, he observed that in the presence of green, abnormal behaviours became less deviant, while in the presence of red, these

behaviours became exaggerated. The colour stimuli he used in these experiments were pieces of coloured paper, coloured rooms, coloured lights and coloured clothing. Based on his observations, Dr. Goldstein viewed red as having an "expansive" effect on the senses and of being capable of inducing a state of excitation in both emotional and motor behaviour. Green, on the other hand, had in his opinion a contractive nature and promoted tranquillity (Nakshian, 1964). Based on these premises, Dr. Goldstein (1942) felt that one's performance on certain motor tasks and judgements could be interfered with or disturbed by the colour red and facilitated by the colour green. However, neither numerical results, nor statistical analyses of his observations were ever presented (Beach et al., 1988; Nakshian, 1964; Norman and Scott, 1952; Goldstein, 1942).

According to Beach et al. (1988), one of the most comprehensive and stringent studies of the effects of coloured illumination on both subjective and physiological measures, was according to Beach et al. (1988) provided by Gerard (1958, 1959). In his often-cited empirical study, 24 male college students were exposed to three different lighting conditions (red, blue and white light, each projected on a screen during ten minutes in counterbalanced order). The hues blue and red were chosen as they represent extreme colours of the visible spectrum. Luminance of the translucent screen in the three light conditions was equated by the method of limits. Statistically significant differences could be revealed between the red and blue conditions for all physiological measures recorded during exposure, except for heartbeat. The blue illumination condition yielded lower blood pressure and palmar conductance, a lower respiration rate, increased alpha wave frequency (EEG) and a reduced eyeblink frequency as compared to the red illumination condition. Moreover, Gerard noted that subjects reported feeling more alert under the red illumination condition and more relaxed in the blue one, lending additional support to the sedative effect of blue. Responses to the white light varied, but were most often similar to those of the red light condition.

Erwin, Lerner, Wilson and Wilson (1961) exposed subjects to four different coloured lights (red, blue, green and yellow), for five minutes each, and found the duration of alpha wave onset to be shortest under the green condition. However, the authors could not reveal any significant difference in arousal (as measured by suppression of alpha waves) between the red, yellow and blue conditions. According to Beach et al. (1988), the authors did not discuss how (if at all) brightness was equated between the four colour stimuli, nor did they give much details regarding the colour stimuli used.

Among those investigators who have examined colour effects on arousal, Wilson (1966) was the first to suggest a possible relationship between arousal reactions and a colour's wavelength. As he found red to be significantly more arousing than green, based both on absolute skin conductance measures and conductance change (galvanic skin response) data, Wilson (1966) posited a U-shaped relationship between colour wavelength and arousal effects. According to Wilson (1966, p949) this relationship may help to explain why the psychological distance between red and violet is less than between either colour and green. This U-shaped hypothesis predicts that the colours with more extreme wavelengths are the most activating. Behind this hypothesis is a Darwinian logic (Wilson, 1966, p949) that more extreme wavelengths are arousing because they are near potentially harmful wavelengths, i.e. infrared (heat) and ultraviolet (sunburn). Wilson's (1966) subjects were exposed to highly saturated red and green slides. Five slides of each colour were shown in alternating order for one minute each. Using mean conductance level and change in conductance level as measures for arousal, he found support for the hypothesis that red is more arousing than green, with the effect being particularly apparent in the GSR data. His results showed both conductance scores and change scores to be higher in the red condition, demonstrating that red is more stimulating and more exciting. This study also revealed some confounding order effects (Beach et al., 1988). It should also be noted that neither colour brightness, nor saturation were controlled for in the study (Valdez and Mehrabian, 1994). When subjects were asked, as a check on brightness after the experiment, which of the two colours (red or green) was lighter, 18 out of the 20 responded green.

Contending that hues at the end of the visible spectrum (blue, purple) are more stimulating than those in the middle (green), as suggested by Wilson (1966), Nourse and Welch (1971) exposed fourteen subjects to green and violet light in alternating order (six one-minute exposures). Arousal was measured by the difference in skin conductance level at the initial exposure of a colour and the highest level reached in the first 12 seconds. A significant difference in GSR amplitude between the two colour conditions could be revealed, with GSR measures being higher in the violet light condition than in the green one. However, another factor may have been confounded with hue effects: instead of using filters, the lights were painted and appeared to differ not only in hue but to have two very different spectral compositions as well. Whereas the green light was a relatively pure spectral composition (high in saturation), the violet light had a spectral composition which covered the entire visible spectrum and was therefore low in saturation. Also the lights were presented in two

different configurations, with the green lights forming a '+'-shaped cross and the violet lights forming an 'x'. There was also a significant colour-order interaction: the response to violet was greatest for those subjects who had been shown green first and responses were only significantly different for the first presentation of a colour, and not for the remaining two, which may be explained by the subjects adaptation (habituation) on subsequent presentations. Nevertheless, this study is often cited as support for differential arousal properties of colour (Beach et al., 1988).

Ali (1972) investigated alpha wave recovery in male subjects being exposed to blue and red lights (equated for brightness) for either 5 or 10 minutes. Greater alpha recovery over time occurred under blue than under red light, which over ten minutes showed relatively little recovery. These findings led Ali to conclude that a higher level of arousal (cortical activity) takes place in red light, which then requires a longer time period before resting alpha activity can resume.

Jacobs and Hustmyer (1974) found GSR to be significantly affected by colour, while heartbeat and respiration measures yielded no difference. Red, yellow, green and blue coloured slides were shown to subjects for one minute each in counterbalanced order, with white slides projected between the different colours for one minute. Saturation was high for each colour, but brightness varied, with yellow being the most bright (8/) and the red being the least bright (3.9/). Arousal, measured in terms of GSR as the greatest change in skin conductance occuring within the first 15 seconds after stimulus onset, appeared to be highest in the red condition, followed by the green, yellow and blue conditions. Jacobs and Hustmyer (1974) found no significant difference between the red versus green condition and the blue versus yellow condition. They note that their findings agree with those reported by Gerard (1958) with regard to GSR and heartbeat, but are contrary to these of Gerard (1958) with regard to GSR their findings disagree with those of Wilson (1966). The authors claim discrepancies between this study and other studies to be due to the colour stimulus difference (lights versus painted surfaces).

Caldwell and Jones (1985) found no systematic effects of red, blue or white coloured lights on subjects' eyeblink frequency, skin conductance, finger pulse volume, heartbeat or EEG measures. However, as the stimulus was only presented for a maximum of 45 seconds, the length of exposure to the coloured light conditions may have been too short to detect subtle

changes. Colour effects appeared to be limited to the first presentation of a given colour. Nevertheless, no difference in percent of alpha activity or peak EEG frequency could be detected under red or blue light conditions.

Kuller (1985) found no evidence to support the notion that red interiors are more stressful than blue interiors. He observed no difference in EEG measures when individuals sat in a red or blue-walled room.

Based on previous research (Gerard, 1958; Lean, 1984), Etnier and Hardy (1997) hypothesized that there would be a statistically significant effect of environmental colour on physiological arousal, with the warm colour increasing arousal in comparison to the cool and neutral colours. In their own study on the influence of environmental colour on task performance Etnier and Hardy (1997) did not find a significant main effect of environmental colour (green, orange or white coloured room) on physiological arousal (heart rate, blood pressure). However the colour x time¹¹ interaction was significant for systolic blood pressure F(4,116)=275.03, p<.0001, diastolic blood pressure F(4,116)=18.46, p<.0001, and heart rate F(4,116)=1170.93, p<.0001.

It is evident that a great deal of inconsistency exists in the literature, arising from incomplete operational definitions of the stimulus conditions and the multifaceted nature of response measures defined as arousal (Beach et al., 1988). Whereas experimental studies involving physiological reactions to arousal (e.g., GSR, EEG) have been motivated largely by the hypothesis that long-wavelength colours are more arousing than short wavelength colours, numerous studies in this field could not confirm this contention (Erwin, Lerner, Wilson and Wilson, 1961; Jacobs and Hustmyer, 1974; Caldwell and Jones, 1985; Kuller, 1985; Etnier and Hardy, 1997). Nevertheless, this hypothesized positive relationship between colour wavelength and arousal is still generally proposed to be true, based on some supporting evidence (Goldstein, 1942; Gerard, 1958, 1959; Ali, 1972). However, researchers have often confined their studies to only two hues, often not even controlling for brightness or saturation, and just extrapolated the results without any further ado... Following a Darwinian logic, on the other hand, Wilson (1966) posited a U-shaped relationship between colour wavelength and arousal effects, predicting that the colours with more extreme wavelengths are the most

¹¹ Measures were taken at several instances: when entering the main laboratory (1), before (2) and after (3) performing a cognitive task and before (4) and after (5) performing a motor task.

activating. This contention has also received some support (Erwin, Lerner, Wilson and Wilson, 1961; Wilson, 1966; Nourse and Welch, 1971).

Arousal perception (i.e. semantic differential ratings)

Wexner (1954) conducted a study dealing with associations between colour samples and words that describe feelings. She constructed a list of adjectives judged to be best examples of mood tones and displayed eight coloured papers (8,5" x 11"). Subjects were asked to select the one colour that best described a given emotional state. From her study she concluded that indeed there exist definite colour-emotion associations. Wexner (1954) found the colour red to be associated with "exciting" and "stimulating", both of which imply pleasure and high arousal (Valdez and Mehrabian, 1994). Blue was associated with "secure/comfortable" and "tender/soothing", which imply pleasure and low arousal (Valdez and Mehrabian, 1994). Orange was associated with "disturbing/distressed/upset", implying displeasure and high arousal. However, Wexner neither used standard specifications of her colour samples, nor controlled for brightness or saturation. Nevertheless, her findings are generally in accord with other research.

Conducting one of the initial studies on the meaning of colours, Osgood et al. (1957) studied selected colours in the context of objects. In a first study one of six colours or an achromatic condition (black-white) was applied to either one of five objects (shirt, ice-cream, rug, car, cake mix) or as background. The object was then rated on 20 semantic differential scales, representing evaluative, activity and potency factors. Activity seemed to be related to the hue dimension – red and yellow were seen as active, while green, violet and blue were passive.

Wright and Rainwater (1962) undertook an extensive study of colour meanings of isolated colours. They had over 1200 subjects rate 50 3" mat surface colours on 24 semantic differential scales. Each subject rated only one colour on all 24 scales. Hue, value and saturation were varied. They found that redness corresponds with greater perceived warmth and less tranquillity and that excitation may be a linear function of hue.

Investigating the emotional stereotypes of colours, Aaronson (1970) had subjects rate colour names to 12 adjectives from the Emotion Profile Index (forced choice adjective checklist

presented as paired comparisons). He interprets his data as supportive of reliable colouremotion stereotypes; however, he never presents the actual colour-emotion stereotypes other than for the single descriptor "activation". He cites red as being high in activation and blue being low.

In their cross-cultural study in which subjects had to rate colour concepts (words) using the semantic differential factors (Osgood et al., 1957), Adams and Osgood (1973) found red to be rated as active across 23 cultures.

In a very comprehensive study, Sivik (1974a, 1975) examined seventy-one colours selected to represent the colour space of the Swedish Natural Colour System - NCS (Hard and Sivik, 1979; Hard, 1975). Judgements on 26 semantic differential scales by a stratified sample of Swedish adults revealed four underlying factors: excitement, evaluation, potency and temperature. From his findings, Sivik concluded that, with respect to the excitement factor, no difference was exhibited between hues with equal chromaticness. This finding contradicts the long-held stereotype that warm colours, such as red and yellow, are exciting, while cool colours, such as blue and green, are calming. Perceived colour-temperature ratings did show a strong hue component. Across all three parameters, yellow and yellow-red regions are judged warmest while blues and blue-greens are judged as coldest. However, the degree of perceived warmth and coldness also changed with varying amounts of blackness, whiteness and chromaticness. Sivik hypothesizes that the reason why red and yellow have been labelled as exciting, might be attributed to the fact that pure, saturated reds and yellows are more prevalent than pure and highly saturated greens and blues. This was also noted by Beach et al. (1988). Sivik notes that one can have a dull red, or an exciting green, based on the purity of the colour. Emerging from Sivik's (1974a, 1975) initial studies is the complexity of the interaction process between stimulus colour properties and connotative associations. Instead of summarizing his data in terms of scale-colour correlations, Sivik graphically represents that interactive process through the use of isosemantic mapping

Measuring arousal by means of a paper and pencil anxiety test (anxiety being an emotional state involving displeasure and high arousal), Jacobs and Suess (1975) obtained support for the hypothesis that red is more arousing than blue. They assigned subjects to one of four colour conditions. Similar to the study reported by Jacobs and Hustmyer (1974), red, yellow, green and blue coloured slides were projected on a large screen. Scores on Spielberger,

Gorsuch and Lushene's (1970) State-Anxiety Inventory served as the dependent variable. Subjects were administered the paper and pencil anxiety test three times. Between each administration, a colour slide was projected during five minutes. Significantly higher anxiety scores were obtained in the red and yellow colour conditions than in the green and blue ones. This finding is consistent with studies of colour arousal, demonstrating that red and yellow are more arousing than blue and green. Still, brightness and saturation levels were not controlled for.

Citing anecdotal stories regarding colour decisions based on assumed behavioural responses, Porter and Mikellides (1976a) describe green as being calm and restful. Birren (1978) and Sharpe (1974) found the colour red to be associated with excitation.

Consistent with the generally accepted view that red is an exciting colour, whereas blue is relaxing (e.g., Guilford and Smith, 1959), Walters et al. (1982), using a two-dimensional view of arousal (Smith and Apter, 1975; Apter, 1976, 1981, 1982), found a link between red and felt 'excitement', and blue and felt 'relaxation'. Chroma and saturation were, nevertheless, not controlled for in this study either.

Profusek and Rainey (1987) investigated the effects of rooms painted in red versus Baker-Miller pink on state anxiety (which implies feelings of low pleasure and high arousal). As hypothesized, pink elicited less anxiety than red. Colour brightness and saturation level were however not controlled for. Actually Baker-Miller pink was much lighter than red, thus hue and value effects were confounded.

Weller and Livingstone (1988) investigated the effects of the colour of paper (blue, pink, white), on which a text about rape and murder incidents was presented, on emotional reactions to these events. Subjects reported being less upset when these incidents were presented on pink paper than when they were described on blue or white paper. Again brightness and saturation were not controlled in this study.

In a study among 121 undergraduates, focussing on the effects of colour hue on emotions, Valdez and Mehrabian (1994, Study 2) examined the arousal elicited by 10 hues, using 5 replication sets with equal brightness and saturation levels in order not to confuse the effects (this way a total of 50 colour samples were used). Arousal was measured according to Mehrabian's (1978) PAD emotion scale. Long-wavelength hues were hypothesized to be more arousing than short-wavelength hues. A univariate analysis of variance (ANOVA) was used to explore the effects of hue (10 levels) on averaged arousal reactions (each colour sample was rated by nearly 25 subjects), revealing a significant .01-level main effect: F (9,80) = 3.80. Figure 3-24 demonstrates a plot of the mean arousal responses to each of the 10 hues. The two non-spectral hues purple and red-purple, are depicted separately at the right of the graph.



Figure 3-24:Mean arousal level as a function of colour wavelengthSource:Valdez and Mehrabian, 1994, Study 2, p403

No support was found for the hypothesis that long-wavelength hues are more arousing than short wavelength hues. Tukey's Multiple Comparison Procedure could only reveal that the mean arousal level for green-yellow was significantly higher than the mean arousal levels for purple-blue, yellow-red and red-purple and that the mean arousal level for blue-green was significantly higher than the mean arousal level for purple-blue.

Furthermore, a regression analysis was used to test for a possible significant parabolic relationship of arousal (the dependent variable) to wavelength (the independent variable). However, also here no significance was obtained. The obtained results relating hue and arousal were generally weak and non-significant. The only noteworthy generalization is that the green hues (green-yellow, blue-green, and green) seem to elicit the highest arousal reactions from subjects.

Gorn et al. (1997) manipulated the colour of a big "swoosh" in the centre of an ad for a fictitious paint company "Rainbow Paints" according to a 2 by 2 by 2 factorial design (hue x chroma x value). Using a two-dimensional view on arousal (Smith and Apter, 1975; Apter, 1976, 1981, 1982; Walters et al., 1982), they found partial support for the hypothesis that respondents exposed to the colour red are more excited than those exposed to the colour blue and that those exposed to the colour blue are more relaxed. Results from a study among 156 university undergraduates revealed that those exposed to the red hue reported stronger feelings of excitement. However only the results for the rating scale measure were statistically significant, not those of the verbal protocol measure¹² (rating scale: F(1,138)=4.98, p<0.05, omega square=0.026, means 1.97 versus 2.53; verbal protocol: F(1,137)=3.10, p>0.05). The results also show that those exposed to the blue hue reported experiencing greater feelings of relaxation (rating scale means = 4.31 vs. 4.04), however the effect of hue on feelings of relaxation fails to reach statistical significance on both the rating scale and the verbal protocol measures. Because the effects revealed for hue were small as compared to those for chroma and value, and much of the prior research has reported significant hue effects, Gorn et al. (1997) conducted a follow up study, comparing red and blue at very high saturation (5 Blue at value level 4 and chroma level 10 and 5 Red at value level 4 and chroma level 14). Because by doing so the stimuli varied not only in hue, but also in saturation, the effects of saturation and hue are confound. Submitting respectively 18 and 19 undergraduate business students to each condition, the results revealed that under these new conditions hue did not significantly

¹² The verbal protocol measure consisted of having respondents list all the feelings they had while viewing the ad. Subsequently these feeling responses were coded by two judges into three separate categories: 'excitement', 'relaxation' and 'unpleasantness', in parallel to the three feeling dimensions measured by the rating scales. All responses were classified, scored (+ or -) and summed for each category. Responses obtained from this verbal protocol-measure were significantly correlated with the scores on the rating scale measures (p < .05).

affect feelings of relaxation or excitement, nor ad or brand attitudes. They conclude that taken together, the results of both studies suggest that, when used as an executional cue in an ad, the hue (red vs. blue) of a colour at best has modest effects on feelings.

Madden et al. (2000) conducted a study in which respondents from eight countries evaluated ten colour-swatches, presented together with the colour-name. In this study the colour stimuli were not clearly specified according to a colour system. Brightness and saturation levels were not controlled for. The colour patches were probably high in saturation. Respondents were undergraduate students from Austria (n = 29), Brazil (n = 26), Canada (n = 29), Colombia (n = 48), Hong-Kong (n = 19), PRC (n = 31), Taiwan (n = 22) and the United States (n = 49). Colour meanings were captured using 20 item semantic differential scales. Colours and semantic items were plotted in two-dimensional perceptual maps for each country. From these plots it is apparent that "a spectrum of colour meaning" can be identified along an arousal or activation dimension: with red on the one hand, ranging over a gold-orange-yellow cluster, over a black-brown-purple cluster to a blue-green-white cluster on the other hand. The meaning associations along this spectrum run from "active", "hot" and "vibrant" (associated with red) to "calming", "gentle" and "peaceful" (associated with a blue-green-white cluster). The results obtained by Madden et al. (2000) indicate that in many parts of the world, consumers exhibit similarities in colour meaning associations.

Thus, studies using semantic differential scales to capture "perceptions of arousal" have more systematically found excitement and activity to be related to the hue dimension, with red and yellow generally seen as active, stimulating and exciting, while green, violet and blue are perceived to be passive, relaxing, soothing and calm (Wexner, 1954; Osgood et al., 1957) Wright and Rainwater, 1962; Aaronson, 1970; Adams and Osgood, 1973; Sharpe, 1974; Jacobs and Suess, 1975; Porter and Mikellides, 1976a; Birren, 1978; Walters et al., 1982; Madden et al., 2000). However, contradicting this long-hold stereotype that warm colours are exciting, while cool colours are calming, Sivik (1974a, 1975) concluded on the basis of his comprehensive study, that no difference is exhibited between hues with equal chroma on the excitement factor, thus revealing the complexity of the interaction process between stimulus colour properties and connotative associations. Likewise, Valdez and Mehrabian (1994, Study 2) could not support the hypothesis that long-wavelength hues are more arousing than short wavelength hues either. They found the obtained results, relating hue and arousal, to be generally weak and non-significant. Unexpectedly, they found the green hues (green-yellow,

blue-green, and green) to elicit the highest arousal reactions from subjects. Gorn et al. (1997) conclude from their study with regard to the effects of colour in advertising, that when used as an executional cue in an ad, the hue (red vs. blue) of a colour at best has modest effects on feelings of arousal.

Studies using both approaches (physiological responses and perceptions of arousal) have found that certain colours, are indeed more activating than other colours (e.g. Nakshian, 1964; Clynes and Kohn, 1968; Sallis and Buckalew, 1984; Jacobs and Nordan, 1979, see Crowley, 1993). However there is a lot of inconsistency in the findings as to what colours are more and less arousing and whether there is indeed a relationship between colour wavelength and its arousal eliciting quality.

THE EFFECTS OF COLOUR ON EMOTIONS: Table 3-5: The Arousal-eliciting Properties of Colours

EFFECTS OF HUE

Citation	Sample	Design	Stimuli	Dependent variable	Findings
Goldstein (1942)	Experiments conducted on a small number of patients (3-5) with brain-damage	Subjects exposed to pieces of coloured paper, coloured rooms, coloured lights and coloured clothing	Red and Green (not equated for brightness or saturation)	Observations of deviant behaviour	In the presence of green, abnormal behaviours became less deviant, while in the presence of red, these behaviours became exaggerated.
Wexner (1954)		Study concerning subjects' associations between colour samples and words that describe feelings	Eight coloured papers (not specified according to standard specifications & not controlled for brightness or saturation)	Words that describe mood tones	Orange was found to be associated with "disturbing/distressed/upset", implying high arousal (and displeasure). Red was found to be associated with "exciting" and "stimulating", (implying high arousal and pleasure) whereas Blue was associated with "secure/comfortable" and "tender/soothing" (implying low arousal and pleasure)
Osgood et al. (1957)	Studied selected colours in the context of objects	Study 1 Colours were applied to either one of five objects (shirt, ice-cream, rug, car, cake mix) <u>Study 2</u> Colours were applied to abstract sculptures	Six colours + black and white	An activation dimension identified from a 20 item semantic differential scale	Activity seemed to be related to the hue dimension: red and yellow were seen as active, while green, violet and blue were passive.

Gerard (1958, 1959)	24 male college students	Subjects exposed to three different coloured lighting conditions, each projected on a screen for 10 minutes in counterbalanced order	Red, Blue and White light (with luminance equated by the method of limits)	Physiological measures: - blood pressure - palmar conductance - respiration rate - alpha wave frequency (EEG) - eyeblink frequency - heart beat	Statistically significant differences could be revealed between the red and blue conditions for all physiological measures, except for heart beat. The blue illumination condition yielded lower blood pressure and palmar conductance, a lower respiration rate, increased alpha wave frequency (EEG) and a reduced eyeblink frequency as compared to the red illumination condition. Moreover, Gerard noted that subjects reported feeling more alert under the red illumination condition and more relaxed in the blue one. Responses to the white light varied, but were most often similar to those in the red light condition.
Erwin, Lerner, Wilson & Wilson (1961)		Subjects exposed to four different coloured lights for 5 minutes	Red, Blue, Green and Yellow light (not specified nor equated for luminance)	<u>Physiological</u> <u>measure:</u> - suppression of alpha waves	Found the duration of alpha wave onset to be shortest under the green condition. No significant difference was found between the other conditions.
Wright and Rainwater (1962)	over 1200 subjects	An extensive study of colour meanings of isolated colours	50 3" mat surface colours (with hue, value and saturation varied)	24 semantic differential scales (showiness, calmness, tranquility,)	Found that redness corresponds with less tranquillity and that excitation may be a linear function of hue.
Wilson (1966)		Subjects were exposed to highly saturated red and green slides. (five slides of each colour were shown in alternating order for one minute each)	highly saturated red and green slides (neither colour brightness, nor saturation were controlled for)	Physiological measures: GSR measure - mean conductance level - change in conductance level	 Wilson posited a U-shaped relationship between colour wavelength and arousal, predicting that colours with more extreme wavelengths are the most activating. He found support for the hypothesis that red is more arousing than green, with the effect being particularly apparent in the GSR data. His results showed both conductance scores and change scores to be higher in the red condition, demonstrating that red is more stimulating and more exciting. This study also revealed some confounding order effects.

Aaronson (1970)		Investigated the emotional stereotypes of colours	colour names	12 adjectives from the Emotion Profile Index (forced choice adjective checklist presented as paired comparisons)	He cites red as being high in activation and blue as being low in activation.
Nourse and Welch (1971)	Sample of 14 subjects	Exposed subjects to green and violet light in alternating order (six one- minute exposures).	Green and Violet light (not only differing in hue, but also in spectral composition, i.e. purity)	Physiological measure: (GSR) Arousal was measured by the difference in skin conductance level at the initial exposure of a colour and the highest level reached in the first 12 seconds.	A significant difference in GSR amplitude between the two colour conditions could be revealed, with GSR measures being higher in the violet light condition than in the green one, supposedly offering support for the contention that hues at the end of the visible spectrum (blue, purple) are more stimulating than those in the middle (green), Also a significant colour-order interaction was noted, which may be explained by the subjects adaptation (habituation) on subsequent presentations.
Ali (1972)	Male subjects	Exposed subjects to blue and red lights for either 5 or 10 minutes	Red and Blue lights, equated for brightness	<u>Physiological</u> <u>measure:</u> alpha wave recovery	Greater alpha wave recovery over time occurred under blue than under red light. Thus, a higher level of arousal (cortical activity) appears to take place in the red light
Adams and Osgood (1973)	Subjects across 23 cultures		Colour names (words)	An activation dimension identified from a semantic differential scale	The word "Red" was found to be rated as active across 23 cultures.

Jacobs and Hustmyer (1974)		Subjects were shown different coloured slides for one minute each in counterbalanced order, with white slides projected between the different colours for one minute	Red, yellow, green and blue coloured slides, projected on a large screen (saturation was high for each colour, but brightness varied, with yellow being the most bright (8/) and red being the least bright (3.9/))	Physiological measures: - GSR (the greatest change in skin conductance occuring within the first 15 seconds after stimulus onset) - heart beat - respiration rate	Found GSR to be significantly affected by colour, while heart rate and respiration measures yielded no difference. Arousal, measured in terms of GSR, appeared to be highest in the red condition, followed by the green, yellow and blue conditions. They found no significant difference between the red versus green condition nor between the blue versus yellow condition.
Sivik (1974a)	A stratified sample of Swedish adults	Studied colours in isolation	71 colours chosen to represent the NCS colour space (based on the Swedish Natural Colour System)	An excitement dimension, identified from a 26 item semantic differential scale	From his findings, Sivik concludes that no difference was exhibited between hues with equal chromaticness on the excitement factor.This finding contradicts the long-held stereotype that warm colours such as red and yellow are exciting, while cool colours, such as blue and green are calming.Sivik hypothesizes that the reason why red and yellow have been labelled as exciting might be attributed to the fact that pure, saturated reds and yellows are more prevalent than pure and highly saturated greens and blues.

Jacobs and Suess (1975)	Subjects were assigned to one of four colour conditions : different coloured slides were projected on a large screen for five minutes	Red, yellow, green and blue coloured slides were projected on a large screen (brightness and saturation levels were not controlled for)	State-Anxiety Inventory (Spielberger, Gorsuch and Lushene, 1970) (anxiety is an emotional state involving displeasure and high arousal)	Obtained support for the hypothesis that red is more arousing than blue. Significant higher anxiety scores were obtained in the red and yellow colour conditions than in the green and blue ones. This finding is consistent with studies of colour arousal, demonstrating that red and yellow are more arousing than blue and green.
Walters et al. (1982)		(chroma and saturation were not controlled for)		Red was related to "felt excitement" and Blue was related to "felt relaxation".
Caldwell and Jones (1985)	Subjects exposed to three different coloured lighting conditions, each for 45 seconds	Red, Blue and White light	<u>Physiological</u> <u>measures:</u> - finger pulse volume - skin conductance - alpha wave frequency (EEG) - eye-blink frequency - heart beat	Found no systematic effect of red, blue or white coloured lights on subjects' eye-blink frequency, skin conductance, finger pulse volume, heart rate or EEG measures. However, as the stimulus was only presented for a maximum of 45 seconds, the length of exposure to the coloured light conditions may have been too short to detect subtle changes. Colour effects appeared to be limited to the first presentation of a given colour. Nevertheless, no difference in percent of alpha activity or peak EEG frequency could be detected under red or blue light conditions.
Kuller (1985)	Subjects were assigned to a red or blue walled room	Red or blue- walled room	Physiological measures: EEG measures	Found no evidence to support the notion that red interiors are more stressful than blue interiors. He observed no difference in EEG measures when individuals sat in a red or blue-walled room.
Profusek and Rainey (1987)	Subjects were exposed to rooms painted in red versus Baker- Miller pink	Red or Baker- Miller pink walled room	State anxiety	As hypothesized, pink elicited less anxiety than red.

Weller and Livingstone (1988)		<u>Colour as a</u> <u>peripheral cue</u> Investigated the effects of the colour of paper on which text about rape and murder incidents was presented	Blue, Pink & White coloured paper (brightness and saturation were not controlled)	Emotional reactions to rape and murder incidents.	Subjects reported being less upset when rape and murder incidents were presented on pink paper than when they were described on blue or white paper.
Valdez and Mehrabian (1994) – Study 2	121 undergraduate students	Laboratory experiment Colour patches	50 colour samples: 10 hues, using 5 replication sets with equal brightness and saturation levels.	Arousal, measured according to Mehrabian's (1978) PAD emotion scale	A significant main effect of hue on arousal could be revealed. However no support was found for the hypothesis that long-wavelength hues are more arousing than short wavelength hues. The mean arousal level for green-yellow was significantly greater than the mean arousal levels for purple-blue, yellow-red and red-purple and the mean arousal level for blue-green was significantly greater than the mean arousal level for purple-blue. The obtained results, relating hue and arousal were generally weak and non-significant. The only noteworthy generalization is that the green hues (green-yellow, blue-green and green) seem to elicit the highest arousal reactions from subjects.
Etnier and Hardy (1997)		<u>Colour as a</u> <u>periferal cue</u> Investigated the effect of environmental colour on the performance of cognitive and motor-tasks	Blue-Green, white or orange coloured room	 <u>Physiological</u> <u>arousal:</u> systolic blood pressure diastolic blood pressure heart beat. 	Did not find a significant main effect of environmental colour on physiological arousal. However the colour x time interaction was significant for systolic blood pressure, diastolic blood pressure and heart beat.

Gorn et al. (1997)	<u>Study 1</u> 156 university undergraduates <u>Study 2</u> 18 and 19 undergraduate students	manipulated the colour of a big "swoosh" in an advertisement	Study 1 4 blue and 4 red colour samples, chosen according to a 2x2x2 factorial design (hue x chroma x value) Study 2 Higly saturated blue 5B 4/10 and red 5R 4/14	Using a two- dimensional view on arousal (Smith and Apter, 1975 ; Apter, 1976, 1981, 1982; Walters et al., 1982) (with excitement versus boredom and tension versus relaxation) Both a rating scale measure as a verbal protocol measure were used	Found partial support for the hypothesis that respondents exposed to the colour red are more excited than those exposed to the colour blue and that those exposed to the colour blue are more relaxed. Those exposed to the red hue reported stronger feelings of excitement, however only the results for the rating scale measure were statistically significant, not those of the verbal protocol measure. The results also show that those exposed to the blue hue reported experiencing greater feelings of relaxation, however the effect of hue on feelings of relaxation fails to reach statistical significance on both the rating scale and the verbal protocol measures. Because in study 2, the stimuli varied not only in hue, but also in saturation, the effects of saturation and hue are confound. The results revealed that under these new conditions hue did not significantly affect feelings of relaxation or excitement, nor add or brand attitudes. They conclude that taken together, the results of both studies suggest that, when used as an executional cue in an ad, the hue (red vs. blue) of a colour at best has modest effects on feelings.
Madden et al. (2000)	Sample of undergraduate students from eight countries: Austria ($n = 29$), Brazil ($n = 26$), Canada ($n = 29$), Colombia ($n = 48$), Hong-Kong ($n = 19$), PRC ($n = 31$), Taiwan ($n = 22$) and the United States ($n = 49$).	Colour patches with colour names were evaluated by respondents	Black, blue, brown, gold, green, orange, purple, red, white and yellow. (standard colour shading options in Lotus Almipro word processing software, not clearly specified according to a colour system, probably highly saturated with brightness and saturation levels not controlled)	Colour meaning (20 item semantic differential scale)	Colours and semantic items were plotted in two-dimensional perceptual maps for each country. From these plots "a spectrum of colour meaning" can be identified along an arousal or activation dimension: with red on the one hand, ranging over a gold-orange- yellow cluster, over a black-brown-purple cluster to a blue-green-white cluster on the other hand. The meaning associations along this spectrum run from "active", "hot" and "vibrant" (associated with red) to "calming", "gentle" and "peaceful" (associated with a blue-green- white cluster). The results obtained by Madden et al. (2000) indicate that in many parts of the world, consumers exhibit similarities in colour meaning associations.

3.5.4.2. Effects of Saturation

Remarkably, none of the studies dealing with physiological reactions to colour have investigated these reactions in relation to colour brightness and saturation levels.

Nevertheless, using semantic differential ratings, Wright and Rainwater (1962) did undertake an extensive study of colour meanings, incorporating the effects of brightness and saturation. They had over 1200 subjects rate 50 3" mat surface colours on 24 semantic differential scales. Each subject rated only one colour on all 24 scales. Hue, value and saturation were varied. Originally six factors were extracted, but intercorrelations among them led the authors to propose a final four-dimensional connotative framework of happiness, forcefulness, warmth and elegance. Across all scales, saturation was found to exert the strongest influence in terms of colour meanings. The relation of activity to colour appeared to be a direct function of saturation. "Showiness", according to Valdez and Mehrabian (1994) assumed to be indicative of the arousing quality of a colour, correlated positively with saturation, suggesting that arousal is a positive correlate of colour saturation.

Hogg (1969) had subjects rate single colours and colour pairs (manipulating the degree of colour contrast) on twelve 7-point semantic differential scales. For single colours, results showed 70.5% of the variance to be attributable to a factor labelled colour obtrusiveness, which was highly correlated with saturation. For colour pairs, sixty-one percent of the variance was accounted for by an activity-potency factor. Saturation was again the most important component of this factor: the more saturated a colour, the more potent/active it was perceived to be.

In their cross-cultural study in which subjects had to rate colour concepts (words) using the semantic differential factors (Osgood et al., 1957), Adams and Osgood (1973) found activity to be associated strongly with colour (versus no colour), which may imply that more saturated colours are perceived as more active than less saturated colours, which seem to contain less colour pigment.

Providing some of the most comprehensive insights into the field of colour meaning (according to Beach et al., 1988), Sivik (1974a, 1975) uses the Swedish Natural Colour System – NCS – (Hard and Sivik, 1979; Hard, 1975) as his descriptive colour model. His

initial studies investigated colours in isolation. Seventy-one colours (chosen to represent the NCS colour space) were judged on 26 semantic differential scales by a stratified sample of Swedish adults. Factor analysis yielded four factors: excitement, evaluation, potency and temperature. Emerging from Sivik's (1974a, 1975) initial studies is the complexity of the interaction process between stimulus colour properties and connotative associations. Sivik hypothesizes that the reason why red and yellow have been labelled as exciting, might be attributed to the fact that pure, saturated reds and yellows are more prevalent than pure and highly saturated greens and blues. Thus he implies that colour saturation is positively correlated to excitement.

In a comprehensive study of emotional reactions to colour among 250 undergraduates, Valdez and Mehrabian (1994, Study 1) also found support for the hypothesis that arousal is a positive correlate of saturation. Stepwise multiple regression analysis was used to explore the contributions of brightness and saturation to arousal. The arousal elicited by 76 colour stimuli was measured by means of Mehrabian's (1978) verbal-report PAD (Pleasure, Arousal, Dominance) scales. Averaged arousal responses (each colour sample was rated by approximately 25 subjects) constituted the dependent variable, and brightness and saturation the independent variables in the regression analysis, yielding the following equation:

Arousal = -.31 Brightness + .60 Saturation

From this study, a colour's brightness and saturation levels appear to explain a substantial portion of the variance in the feelings of arousal it induces (multiple regression coefficient = .73). More specifically, the equation presented above indicates that more saturated colours induce greater feelings of arousal in viewers. Furthermore, the regression analysis indicates the effect of saturation to be considerably stronger than the effect of brightness in determining arousal responses to colour. Indeed, the contribution of saturation to arousal is almost twice the magnitude of the effect of brightness on arousal. Thus the hypothesized positive relationship between saturation and arousal could be confirmed: more saturated (more vivid and purer) colours appear to elicit greater feelings of arousal.

Gorn et al. (1997) manipulated the colour of a big "swoosh" in the centre of an ad for a fictitious paint company "Rainbow Paints" according to a 2 by 2 by 2 factorial design (hue x chroma x value). Using a two-dimensional view on arousal (Smith and Apter, 1975; Apter,

1976, 1981, 1982; Walters et al., 1982), Gorn et al. (1997) found support for the hypothesis that ads with higher chroma colours induce greater feelings of excitement, without feelings of relaxation being negatively affected. Their results reveal that when the chromatic strength of the coloured "swoosh" in the ad was high as opposed to low, the message recipient felt more excited (rating scale: F(1.138)=4.31, p<0.05, omega square=0.021, means 2.47 versus 2.00; verbal protocol: F(1,137)=5.96, p<0.05, omega square = 0.033, means =0.22 versus 0.03).

THE EFFECTS OF COLOUR ON EMOTIONS: Table 3-6: The Arousal-eliciting Properties of Colours

EFFECTS OF SATURATION

Citation	Sample	Design	Stimuli	Dependent variable	Findings
Wright and Rainwater (1962)	Over 1200 subjects	An extensive study of colour meanings of isolated colours	50 3" mat surface colours (with hue, value and saturation varied)	24 semantic differential scales (showiness, calmness, tranquillity,)	Across all scales, saturation was found to exert the strongest influence in terms of colour meanings. The relation of activity to colour appears to be a direct function of saturation. "showiness", assumed to be indicative of the arousing quality of a colour, correlated positively with saturation.
Hogg (1969)			Single colours and colour pairs (manipulating degree of colour contrast)	Twelve 7-point semantic differential scales.	For single colours, results showed 70.5% of the variance to be attributable to a factor labelled colour obtrusiveness, which was highly correlated with saturation. For colour pairs, sixty-one percent of the variance was accounted for by an activity-potency factor. Saturation was again the most important component of this factor – the more saturated a colour, the more potent/active it was perceived to be.
Adams and Osgood (1973)	Subjects across 23 cultures		Colour names (words)	An activation dimension identified from a semantic differential scale	Found activity to be associated strongly with colour (versus no colour).
Sivik (1974a)	A stratified sample of Swedish adults	Studied colours in isolation	71 colours chosen to represent the NCS colour space (based on the Swedish Natural Colour System)	An excitement dimension, identified from a 26 item semantic differential scale	Sivik hypothesizes that the reason why red and yellow have been labelled as exciting might be attributed to the fact that pure, saturated reds and yellows are more prevalent than pure and highly saturated greens and blues.

Valdez and Mehrabian (1994) – Study 1	250 undergraduate students	Laboratory experiment with colour patches	76 colour samples : 10 hues, using 5 replication sets with equal brightness and saturation levels.	Arousal, measured according to Mehrabian's (1978) PAD emotion scale	Found support for the hypothesis that arousal is a positive correlate of saturation. Stepwise multiple regression analysis was used to explore the contribution of brightness and saturation to arousal, yielding the following equation: Arousal =31 Brightness + .60 Saturation A colour's brightness and saturation levels appear to explain a substantial portion of the variance in the feelings of arousal induced (multiple regression coefficient = .73). More specifically, the equation presented above indicates that more saturated (more vivid and purer) colours induce greater feelings of arousal. Furthermore, the regression analysis indicates the effect of saturation to be considerably stronger than the effect of brightness in determining arousal responses to colour (the contribution of saturation to arousal is almost twice the magnitude of the effect of brightness on arousal).
Gorn et al. (1997)	<u>Study 1</u> 156 university undergraduates	manipulated the colour of a big "swoosh" in an advertisement	Study 1 4 blue and 4 red colour samples, chosen according to a 2x2x2 factorial design (hue x chroma x value)	Using a two- dimensional view on arousal (Smith and Apter, 1975 ; Apter, 1976, 1981, 1982; Walters et al., 1982) (with excitement versus boredom and tension versus relaxation) Both a rating scale measure and a verbal protocol measure were used	Found support for the hypothesis that ads with higher chroma colours induce greater feelings of excitement, but not relaxation. Their results reveal that when the chromatic strength of the coloured "swoosh" in the ad was high, as opposed to low, the message recipient felt more excited, but not more relaxed.

3.5.4.3. Effects of Value

The case of achromatic colours:

The effect of brightness in achromatic colours such as white, grey and black (representing brightness variations only) on arousal responses was originally unclear.

Adams and Osgood (1973) found in their cross-cultural study, in which subjects had to rate colour concepts (words) using the semantic differential factors (Osgood et al., 1957), that the words grey and black were rated as inactive, implying low arousal.

On the contrary, Birren (1978) and Sharpe (1974) found the colour black to be associated with anxiety, implying displeasure and high arousal.

Psychophysiological research suggests that white has a calming effect in that white light seems to produce the least amount of tension in the form of hand tremor (James and Domingos, 1953). More recently, and consistent with the work of James and Domingos (1953), Ott (1976) reports that students in classrooms lit with full-spectrum white light were calmer and less fidgety than those in other lighting conditions.

In a more systematic study, Valdez and Mehrabian (1994, Study 3) used five achromatic colour samples (ranging from white to black) to explore the effect of brightness in achromatic colours on arousal responses. Arousal was in general hypothesized to be a positive correlate of brightness, however, with regard to achromatic colours no explicit hypothesis was given. A nonlinear regression analyses with average arousal (25 subjects rated each sample) as the dependent variable and brightness as the independent variable revealed the following equation, expressing the relationship of arousal (raw values) to brightness of achromatic colours (multiple regression coefficient = .47):

Arousal =
$$8 - .6915$$
 (Brightness) + .0073 (Brightness)²

A plot of the actual and predicted mean arousal responses as a function of brightness is given in figure 3-25.



Source: Valdez and Mehrabian, 1994, Study 3, p405

Figure 3-25 demonstrates that arousal reactions to achromatic colours tend to form a U-shaped function of brightness: black elicited the greatest arousal response, which diminished steadily for the three successive greys of increasing brightness, but increased again to an intermediate value for white. Figure 3-25 also shows that the obtained arousal means for all five levels of brightness were predicted extremely well by the proposed equation.

The case of chromatic colours:

As was the case for saturation, none of the studies dealing with physiological reactions to colour have investigated these reactions in relation to colour brightness levels.

Nevertheless, using semantic differential ratings, Wright and Rainwater (1962) did undertake an extensive study of colour meanings, incorporating the effects of brightness and saturation. They had over 1200 subjects rate 50 3" mat surface colours (with varying hue, value and saturation) on 24 semantic differential scales. Each subject rated only one colour on all 24 scales. Originally six factors were extracted, but intercorrelations among them led the authors to propose a final four-dimensional connotative framework of happiness, forcefulness, warmth and elegance. "Showiness", according to Valdez and Mehrabian (1994) assumed to be indicative of the arousing quality of a colour, correlated positively with brightness, whereas "calmness", indicative of the non-arousing quality of a colour, was found to correlate negatively with brightness, suggesting that arousal is a positive correlate of colour brightness.

Profusek and Rainey (1987) found that rooms painted in Baker-Miller pink, a high value colour, namely a "whitish" hue of the red family of colours, had a more relaxing and calming effect than red-painted rooms (see also Schauss, 1985; Bennett et al., 1991).

Contrasting these previous findings, in a comprehensive study of emotional reactions to colour among 250 undergraduates, Valdez and Mehrabian (1994, Study 1) did not find support for the hypothesis that arousal is a positive correlate of brightness. On the contrary, they revealed a negative relationship between colour brightness and induced arousal. In order to explore the contributions of brightness and saturation to arousal, stepwise multiple regression analysis was used. The arousal elicited by 76 colour stimuli was measured by means of Mehrabian's (1978) verbal-report PAD (Pleasure, Arousal, Dominance) scales. Averaged arousal responses (each colour sample was rated by approximately 25 subjects) constituted the dependent variable, and brightness and saturation the independent variables in the regression analysis, yielding the following equation:

Arousal = -.31 Brightness + .60 Saturation

From this study, a colour's brightness and saturation levels appear to explain a substantial portion of the variance in the feelings of arousal it induces (multiple regression coefficient = .73). However, contradicting the hypothesized positive relationship, the equation indicates that less bright (i.e. darker) colours induce greater feelings of arousal in viewers. Nevertheless, the effect of brightness appears to be considerably weaker than the effect of saturation in determining arousal responses to colour. One reason for this contradicting

finding may have been that the latter hypothesis between brightness and arousal was inferred by Valdez and Mehrabian (1994) from reports that used experimental methods that confounded brightness and saturation levels while testing for the effect of brightness. Previous studies tended to select highly saturated bright colours, when sampling for bright colours, assuming that bright colours are also more saturated. This way confounding colour brightness and saturation, earlier researchers may have incorrectly attributed the greater arousal response to highly saturated bright colour samples to brightness, rather than to saturation. This error may have been possible, because the contribution of saturation to arousal is almost twice the magnitude of the effect of brightness). From the results of the study by Valdez and Mehrabian (1994, Study 1), thus, unexpectedly, darker (less bright) colours seem to induce more arousal.

Furthermore, Valdez and Mehrabian (1994) found a second-order curvilinear relationship between colour brightness and the elicited arousal.

$$Arousal = 8.724 - .62 (Brightness) + .007173 (Brightness)^{2}$$

The multiple regression coefficient for this equation is .52. Actual mean arousal values and those predicted from the equation above were plotted against brightness (for each of the six brightness values sampled) and showed extremely close agreement (see figure 3-26).



Both the actual and predicted plots demonstrate that arousal declines steeply and monotonically with increasing brightness up to a brightness value of 43, beyond which arousal reversed and increased slightly for the highest brightness value. In other words, arousal appears to decrease as colours range from dark to light, but this trend is slightly reversed for the lightest colours.

Manipulating the colour of a big "swoosh" in the centre of an ad for a fictitious paint company "Rainbow Paints" according to a 2 by 2 by 2 factorial design (hue x chroma x value), Gorn et al. (1997) used a two-dimensional view on arousal (Smith and Apter, 1975; Apter, 1976, 1981, 1982; Walters et al., 1982). In agreement with the results obtained by Valdez and Mehrabian (1994), they found support for the hypothesis that lighter, higher value colours are more relaxing than lower value, darker colours, without feelings of excitement being affected. In their study involving 156 university undergraduates, Gorn et al. (1997) found that those exposed to ads containing higher value colours (for red and blue hues) reported experiencing greater feelings of relaxation (rating scale: F(1.138)=9.22, p<0.01, omega square=0.054, means 3.70 versus 4.67; verbal protocol: F(1,137)=5.62, p<0.05, omega square = 0.04, means =0.08 versus 0.85), but no difference in the level of felt excitement (rating scale: F(1.138)=2.95, p>0.05; verbal protocol: F>1).

THE EFFECTS OF COLOUR ON EMOTIONS: Table 3-7: The Arousal-eliciting Properties of Colours

EFFECTS OF BRIGHTNESS Citation Stimuli Dependent variable Findings Sample Design The case of achromatic colours The words "Grey" and "Black" were rated as inactive across 23 Subjects Adams and Colour names An activation **Osgood** (1973) across 23 (words) dimension identified cultures. cultures from a semantic differential scale **Birren (1978)** Found the colour black to be associated with anxiety, implying displeasure and high arousal. and Sharpe (1974) Laboratory A curvi-linear relationship could be revealed between the brightness of Valdez and 125 5 achromatic Arousal. measured undergraduate experiment colour samples according to achromatic colours and the arousal evoked. Arousal reactions to Mehrabian Mehrabian's (1978) achromatic colours tend to form a U-shaped function of brightness: (1994) -Study 3 students Colour patches (ranging from black elicited the greatest arousal response, gradually diminishing for white to black) PAD emotion scale lighter greys, but increasing again for white to an intermediate level of arousal. The case of chromatic colours Tension in the form White seems to have a calming effect in that it appears to produce the James and Domingos (1953) of hand tremor least amount of tension in the form of hand-tremor Wright and 50 3" mat 24 semantic Found that the relation of activity to colour appears to be a direct over 1200 An extensive function of brightness: "showiness", indicative of the arousing quality Rainwater (1962) subjects study of colour surface colours differential scales of a colour, correlated positively with brightness. meanings of (with hue, value (showiness, Furthermore, "calmness", indicative of the non-arousing quality of a isolated colours and saturation calmness. colour, correlated negatively with brightness. varied) tranquility,...)

Ott (1976) Students	Different coloured light lit classrooms			Students in classrooms lit with full-spectrum white light were calmer and less fidgety than those in other lighting conditions.
Profusek and Rainey (1987)	Subjects were exposed to rooms painted in red versus Baker- Miller pink	Red or Baker- Miller pink walled room	State anxiety	As hypothesized, Baker-Miller pink (a 'whitish' hue of the red family of colours) had a more relaxing and calming effect than red-painted rooms.
Valdez and 250 Mehrabian undergradua (1994) – Study 1 students	Laboratory e experiment with colour patches	76 colour samples : 10 hues, using 5 replication sets with equal brightness and saturation levels.	Arousal, measured according to Mehrabian's (1978) PAD emotion scale	Contrary to expectations, arousal appeared to be a negative correlate of brightness. Stepwise multiple regression analysis was used to explore the contribution of brightness and saturation to arousal, yielding the following equation: Arousal =31 Brightness + .60 Saturation A colour's brightness and saturation levels appear to explain a substantial portion of the variance in the feelings of arousal induced (multiple regression coefficient = .73). More specifically, the equation presented above indicates that less bright (i.e. darker) colours induce greater feelings of arousal. However, the regression analysis indicates the effect of saturation to be considerably stronger than the effect of brightness to arousal is about half the magnitude of the effect of brightness on arousal). Furthermore, a second-order curvilinear relationship was found between colour brightness and the elicited arousal: Arousal = 8.724 – .62 (Brightness) + .007173 (Brightness) ² (multiple regression coefficient of .52) Thus, arousal appears to decline steeply and monotonically with increasing brightness up to a brightness value of 43, beyond which arousal reversed and increased slightly for the highest brightness value. In other words, arousal appears to decrease as colours range from dark to light but this trend is slightly reversed for the lightest colours

Gorn et al. (1997)Study 1 156 university undergraduatesmanipulated the colour of a big "swoosh" in an advertisementStudy 1 4 blue and 4 red chosen according to a 2x2x2 factorial design (hue x chroma x value)Using a two- dimensional view on arousal (Smith and Apter, 1975; Apter, 1976, 1981, 1982; Walters et al., 1982)Found support for the hypothesis that lighter, higher value colours are more relaxing than lower value, darker colours, without feelings of excitement being affected.Gorn et al. (1997)Study 1 undergraduatesStudy 1 4 blue and 4 red chosen according to a 2x2x2 factorial design (hue x chroma x value)Using a two- dimensional view on arousal (Smith and Apter, 1975; Apter, 1976, 1981, 1982; Walters et al., 1982)Found support for the hypothesis that lighter, higher value colours are more relaxing than lower value, darker colours, without feelings of excitement being affected.Both a rating scale measure and a verbal protocol measure were usedStudy 1 4 blue and 4 red colour samples, colour samples, colours and text text text text text text text text						
	Gorn et al. (1997)	<u>Study 1</u> 156 university undergraduates	manipulated the colour of a big "swoosh" in an advertisement	Study 1 4 blue and 4 red colour samples, chosen according to a 2x2x2 factorial design (hue x chroma x value)	Using a two- dimensional view on arousal (Smith and Apter, 1975; Apter, 1976, 1981, 1982; Walters et al., 1982) (with excitement versus boredom and tension versus relaxation) Both a rating scale measure and a verbal protocol measure were used	Found support for the hypothesis that lighter, higher value colours are more relaxing than lower value, darker colours, without feelings of excitement being affected. They actually found that those exposed to ads containing higher value colours (for red and blue hues) reported experiencing greater feelings of relaxation, but no difference in the level of felt excitement.

3.5.4.4. Overview

With regard to the effect of colour hue on arousal, a great deal of inconsistency exists in the literature, arising from incomplete operational definitions of the stimulus conditions and the multifaceted nature of response measures defined as arousal (Beach et al., 1988; Valdez and Mehrabian, 1994). Studies involving physiological reactions to arousal have been motivated largely by the hypothesis that long-wavelength colours (e.g. red and yellow) are more arousing than short wavelength colours (e.g. blue and green). Despite the numerous studies in this field that could not confirm this contention (Erwin, Lerner, Wilson and Wilson, 1961; Jacobs and Hustmyer, 1974; Caldwell and Jones, 1985; Kuller, 1985; Etnier and Hardy, 1997), it still is generally accepted, based on some supporting evidence (Goldstein, 1942; Gerard, 1958, 1959; Ali, 1972). However, researchers often have confined their studies to only two hues, often not even controlling for brightness or saturation and just extrapolated the results without any further ado... Following a Darwinian logic, on the other hand, Wilson (1966) posited a U-shaped relationship between colour wavelength and arousal effects, predicting that the colours with more extreme wavelengths are the most activating. This contention has also received some support (Erwin, Lerner, Wilson and Wilson, 1961; Wilson, 1966; Nourse and Welch, 1971). Studies using semantic differential scales to capture "perceptions of arousal", have more systematically found excitement and activity to be related to the hue dimension, with red and yellow generally seen as active, stimulating and exciting, while green, violet and blue are perceived to be passive, relaxing, soothing and calm (Wexner, 1954; Osgood et al., 1957 Wright and Rainwater, 1962; Aaronson, 1970; Adams and Osgood, 1973; Sharpe, 1974; Jacobs and Suess, 1975; Porter and Mikellides, 1976a; Birren, 1978; Walters et al., 1982; Madden et al., 2000). However, contradicting this long-held stereotype that warm colours are exciting, while cool colours are calming, Sivik (1974a, 1975) concluded on the basis of his comprehensive study, that no difference is exhibited between hues with equal chromaticness on the excitement factor. Sivik hypothesizes that the reason why red and yellow have generally been labelled as exciting, might be attributed to the fact that pure, saturated reds and yellows are more prevalent than pure and highly saturated greens and blues. He notes that one can have a dull red, or an exciting green, based on the purity of the colour. Emerging from Sivik's (1974a, 1975) initial studies is the complexity of the interaction process between stimulus colour properties and connotative associations. Likewise, Valdez and Mehrabian (1994, Study 2) could not support the hypothesis that longwavelength hues are more arousing than short wavelength hues either. Conducting a very

systematic and comprehensive study, they found the mean arousal level for green-yellow to be significantly greater than the mean arousal levels for purple-blue, yellow-red and redpurple and the mean arousal level for blue-green to be significantly greater than the mean arousal level for purple-blue. A regression analysis to test for a possible significant parabolic inverted-U shaped relationship of arousal to wavelength (note that this is the opposite of Wilson's 1966 contention), revealed no significant effect either. The obtained results, relating hue and arousal, were generally weak and non-significant. The only noteworthy generalization is that, unexpectedly, the green hues (green-yellow, blue-green, and green) seem to elicit the highest arousal reactions from subjects. Gorn et al. (1997) conclude from their study with regard to the effects of colour in advertising, that when used as an executional cue in an ad, the hue (red vs. blue) of a colour at best has modest effects on feelings of arousal. Thus, studies using both approaches (physiological responses and perceptions of arousal) have found that certain colours, are indeed more activating than other colours (e.g. Nakshian, 1964; Clynes and Kohn, 1968; Sallis and Buckalew, 1984; Jacobs and Nordan, 1979, see Crowley, 1993). However there is a lot of inconsistency in the findings as to what colours are more and less arousing and whether there is indeed a relationship between colour wavelength and its arousal eliciting quality.

Remarkably, none of the studies dealing with physiological reactions to colour have investigated these reactions in relation to colour saturation levels. Nevertheless, studies based on semantic differential ratings generally suggest that arousal is a positive correlate of colour saturation (Wright and Rainwater, 1962; Hogg, 1969; Adams and Osgood, 1973; Sivik,1974a, 1975). Wright and Rainwater (1962) even found colour saturation to exert the strongest influence in terms of colour meanings. Valdez and Mehrabian (1994, Study 1) also found a colour's saturation level to explain a substantial portion of the variance in the feelings of arousal it induces, with more saturated (more vivid and purer) colours evoking greater feelings of arousal in viewers. Moreover, conform to the findings by Wright and Rainwater (1962), Valdez and Mehrabian (1994, Study 1) noticed that the effect of saturation is considerably stronger than the effect of brightness in determining arousal responses to colour (almost twice the magnitude). Consistent with these findings, Gorn et al. (1997) also found support for the hypothesis that ads with higher chroma colours induce greater feelings of excitement.
The effect of brightness in achromatic colours on arousal responses was originally unclear. Contrary to Adams and Osgood's (1973) finding that the words grey and black were rated as inactive, implying low arousal, Birren (1978) and Sharpe (1974) found the colour black to be associated with anxiety, implying displeasure and high arousal. Psychophysiological research suggests that white has a calming effect in that white light seems to produce the least amount of tension in the form of hand tremor (James and Domingos, 1953). Consistent with the work of James and Domingos (1953), Ott (1976) reports that students in classrooms lit with full-spectrum white light were calmer and less fidgety than those in other lighting conditions. In a more systematic study, Valdez and Mehrabian (1994, Study 3) found that arousal reactions to achromatic colours tend to form a U-shaped function of brightness, with black eliciting the greatest arousal response, which diminishes steadily for the three successive greys of increasing brightness, but increases again to an intermediate value for white.

Also with regard to chromatic colours, the effect of brightness was originally unclear. None of the studies dealing with physiological reactions to colour have investigated these reactions in relation to colour brightness levels. Earlier studies, based on semantic differential ratings, seemed to indicate that arousal is a positive correlate of colour brightness (Wright and Rainwater, 1962; Profusek and Rainey, 1987). More recent studies, however, found exactly the opposite to be true (Valdez and Mehrabian, 1994 - Study 1; Gorn et al., 1997). Indeed, conducting a very comprehensive and systematic study, Valdez and Mehrabian (1994) revealed a negative relationship between colour brightness and induced arousal, indicating that less bright (i.e. darker) colours induce greater feelings of arousal in viewers. From this study the effect of brightness also appears to be considerably weaker than the effect of saturation in determining arousal responses to colour. Furthermore, Valdez and Mehrabian (1994) also found a second-order curvilinear relationship between colour brightness and the elicited arousal. Arousal declined steeply and monotonically with increasing brightness up to a brightness value of 43, beyond which arousal reversed and increased slightly for the highest brightness value. In other words, arousal appears to decrease as colours range from dark to light, but this trend is slightly reversed for the lightest colours. In agreement with the results obtained by Valdez and Mehrabian (1994), Gorn et al. (1997) also found support for the hypothesis that lighter, higher value colours are more relaxing than lower value, darker colours, without feelings of excitement being affected. Using a two-dimensional view on arousal, they found that those exposed to ads containing higher value colours (for red and blue

hues) reported experiencing greater feelings of relaxation, but no difference in the level of felt excitement.

3.5.5. COLOUR-EVOKED DOMINANCE

Dominance is a dimension of emotional response that has almost completely been ignored by colour researchers. However, examining emotional reactions to colours, Valdez and Mehrabian (1994) found support for the existence of a dominance dimension, in correspondence with the Pleasure-Arousal-Dominance Emotion Model introduced by Mehrabian and Russell (1974), (see also Russell and Mehrabian, 1977). Also earlier, researchers already found colour to elicit emotional responses that cannot be captured by pleasure or arousal dimensions, but seem to infer a dimension of dominance or potency (e.g. Osgood et al., 1957; Adams and Osgood, 1973; Sivik, 1974a). In the next paragraphs, these studies regarding the effects of colour on dominance will be reviewed. More specifically, we will first present the effects of colour hue on dominance responses, followed by the effects of colour saturation and brightness. Finally, the conclusions of these previous studies will be briefly summarized.

3.5.5.1. Effects of Hue

In a very comprehensive cross-cultural study in which subjects from 23 different cultures rated colour concepts (e.g. the words "blue", "green", "yellow") using the semantic differential factors (Osgood et al., 1957), Adams and Osgood (1973) found the following chromatic hue effects on perceived 'potency': across the 23 samples yellow was rated as weak, whereas red was rated as strong.

Birren (1978) and Sharpe (1974) found the colour red to be associated with aggression and excitation, whereas green was associated with withdrawal.

In a more recent study among 121 undergraduates, focussing on the effects of colour hue on emotions, Valdez and Mehrabian (1994, Study 2) examined the dominance elicited by 10

hues, using 5 replication sets with equal brightness and saturation levels in order not to confuse the effects (this way a total of 50 colour samples were used). A univariate analysis of variance (ANOVA) was used to explore the effects of hue (10 levels) on averaged dominance reactions (each colour sample was rated by nearly 25 subjects), revealing a significant .01-level main effect: F (9,80) = 3.06. Figure 3-27 depicts a plot of the mean dominance responses to each of the 10 hues. From this plot, middle wavelength hues appear to elicit more dominance than either the extreme short or long wavelength hues.



Yet, the obtained results were generally weak and non-significant. According to Tukey's Multiple Comparison Procedure the only significant differences in mean dominance ratings

that could be revealed were that green-yellow and yellow were rated as significantly more dominant than red-purple.

The studies above demonstrate that empirical findings are inconsistent with regard to hue effects of colour on the dominance induced in viewers. The results obtained by Valdez and Mehrabian (1994) suggest the exact opposite of the earlier findings by Adams and Osgood (1973) who did not use actual colour stimuli but colour names, and by Birren (1978) and Sharpe (1974). A plausible explanation may be the failure of the earlier colour-studies to provide adequate specifications or controls of colour stimuli (i.e. absence of controls for saturation and brightness, while investigating effects of hue). Another possible explanation for these different findings could also be that colours that are rated as weak elicit feelings of dominance in viewers and vice versa (colours that are rated as strong might elicit feelings of submissiveness). However, further research is needed to find out if this is actually the case.

THE EFFECTS OF COLOUR ON EMOTIONS: Table 3-8: The Dominance-eliciting Properties of Colours

EFFECTS OF HUE

Citation	Sample	Design	Stimuli	Dependent variable	Findings
Adams and Osgood (1973)	Subjects across 23 cultures		Colour names (words)	A potency dimension identified from a semantic differential scale	Across the 23 samples, "yellow" was rated as weak, whereas "red" was rated as strong.
Birren (1978) and Sharpe (1974)					Found the colour "red" to be associated with aggression and excitation, whereas "green" was associated with withdrawal.
Valdez and Mehrabian (1994) – Study 2	121 undergraduate students	Laboratory experiment Colour patches	50 colour samples: 10 hues, using 5 replication sets with equal brightness and saturation levels.	Dominance, measured according to Mehrabian's (1978) PAD emotion scale	A univariate analysis of variance (ANOVA) was used to explore the effects of hue (10 levels) on averaged dominance reactions (each colour sample was rated by nearly 25 subjects), revealing a significant .01-level main effect : middle wavelength hues appear to elicit more dominance than either the extreme short or long wavelength hues. However, the obtained results were generally weak and non-significant. The only significant differences in mean dominance ratings that could be revealed were that green-yellow and yellow were rated as significantly more dominant than red-purple.

3.5.5.2. Effects of Saturation

Conducting one of the initial studies on the meaning of colours, Osgood et al. (1957) studied selected colours in the context of objects. In a first study one of six colours or an achromatic condition (black-white) was applied to either one of five objects (shirt, ice-cream, rug, car, cake mix) or as background. The object was then rated on 20 semantic differential scales, representing evaluative, activity and potency factors. Potency was found to be directly dependent upon a colour's saturation.

Hogg (1969) had subjects rate single colours and colour pairs (manipulating degree of colour contrast) on twelve 7-point semantic differential scales. For single colours, results showed 70.5% of the variance to be attributable to a factor labelled colour obtrusiveness, which was highly correlated with saturation. For colour pairs, sixty-one percent of the variance was accounted for by an activity-potency factor. Saturation was again the most important component of this factor – the more saturated a colour, the more potent/active it was perceived to be.

Using the Swedish Natural Colour System – NCS (Hard and Sivik, 1979; Hard, 1975) as his descriptive colour model, Sivik (1974a, 1975) initially investigated colours in isolation. Seventy-one colours (chosen to represent the NCS colour space) were judged on 26 semantic differential scales by a stratified sample of Swedish adults. Factor analysis yielded four factors: excitement, evaluation, potency and temperature. Sivik found that as chromaticness increased, so did perceived potency for all hues except for yellow. Also for the potency dimension a complex interaction appeared to exist between the three NCS parameters and a given hue.

In a comprehensive study of emotional reactions to colour among 250 undergraduates, Valdez and Mehrabian (1994) also found support for the hypothesis that dominance is a positive correlate of saturation. Stepwise multiple regression analysis was used to explore the contributions of brightness and saturation to dominance. The dominance elicited by 76 colour stimuli was measured by means of Mehrabian's (1978) verbal-report PAD (Pleasure, Arousal, Dominance) scales. Averaged dominance-submissiveness responses (each colour sample was rated by approximately 25 subjects) constituted the dependent variable, and brightness and

saturation the independent variables in the regression analysis, yielding the following equation:

Dominance = -.76 Brightness + .32 Saturation

From this study, a colour's brightness and saturation levels appear to explain a substantial portion of the variance in the feelings of dominance it induces (multiple regression coefficient = .87). More specifically, the equation above indicates that more saturated colours induce greater feelings of dominance in viewers. Thus, more saturated (more vivid and purer) colours appear to elicit feelings of dominance, strength or boldness. However, the regression analyses indicates the effect of brightness to be considerably stronger than the effect of saturation in determining dominance responses to colour.

THE EFFECTS OF COLOUR ON EMOTIONS: Table 3-9: The Dominance-eliciting Properties of Colours

EFFECTS OF SATURATION

Citation	Sample	Design	Stimuli	Dependent variable	Findings
Osgood et al. (1957)	Studied selected colours in the context of objects	Study 1 Colours were applied to either one of five objects (shirt, ice-cream, rug, car, cake mix) Study 2 Colours were applied to abstract sculptures	Six colours + black and white	A potency dimension identified from a 20 item semantic differential scale	Potency was found to be directly dependent upon a colour's saturation
Hogg (1969)			Single colours and colour pairs (manipulating degree of colour contrast)	Twelve 7-point semantic differential scales	For single colours, results showed 70.5% of the variance to be attributable to a factor labelled colour obtrusiveness, which was highly correlated with saturation. For colour pairs, sixty-one percent of the variance was accounted for by an activity-potency factor. Saturation was again the most important component of this factor – the more saturated a colour, the more potent/active it was perceived to be. Also the greater the contrast in perceived value, the more potent and active a pair was judged to be.
Sivik (1974a)	A stratified sample of Swedish adults	Studied colours in isolation	71 colours chosen to represent the NCS colour space (based on the Swedish Natural Colour System)	A potency dimension, identified from a 26 item semantic differential scale	A complex interaction exists between hue, value and saturation. However, as chromaticness increased, so did perceived potency for all hues except for yellow.

Valdez and Mehrabian (1994) – Study 1	250 undergraduate students	Laboratory experiment with colour patches	76 colour samples : 10 hues, using 5 replication sets	Dominance- submissiveness, measured according to Mehrabian's	Stepwise multiple regression analysis was used to explore the contributions of brightness and saturation to dominance, yielding the following equation: Dominance = 76 Brightness + $.32$ Saturation.
			with equal brightness and saturation levels.	(1978) PAD emotion scale	From this study, a colour's brightness and saturation levels appear to explain a substantial portion of the variance in the feelings of dominance it induces (multiple regression coefficient = $.87$). More specifically, the equation above indicates that more saturated (more vivid and purer) colours induce greater feelings of dominance, strength or boldness in viewers.
					Yet, the regression analysis indicates the effect of brightness to be considerably stronger than the effect of saturation in determining dominance responses to colour.

3.5.5.3. Effects of Value

The case of achromatic colours:

With regard to achromatic colours such as white, grey and black (representing brightness variations only) empirical findings reveal much more consistent results.

Conducting a study dealing with associations between colour samples and words that describe feelings, Wexner (1954) found black to be associated with "powerful/strong/masterful", implying high dominance.

Adams and Osgood (1973) also found black to be rated as strong, whereas white and grey were generally rated as weak.

Frank and Gilovich (1988) investigated the effects of black versus non-black uniforms of professional football and hockey teams on aggressive behaviour. They found that black uniforms, compared with non-black uniforms, not only were associated with greater degrees of perceived aggression, but also led to higher levels of player aggressiveness.

Valdez and Mehrabian (1994, Study 3) used five achromatic colour samples (ranging from white to black) to explore the effect of brightness in achromatic colours on dominance responses. A non-linear regression analyses with average dominance (25 subjects rated each sample) as the dependent variable and brightness as the independent variable revealed the following equation, expressing the relationship of dominance to brightness of achromatic colours (multiple regression coefficient = .65):

Dominance = 25 - 1.2675 (Brightness) + .0088 (Brightness)²

A plot of the actual and predicted mean dominance responses as a function of brightness is given in figure 3-28.



Source:

Valdez and Mehrabian, 1994, Study 3, p405

Consistent with previous findings, black elicited the highest level of dominance, greys induced intermediate levels of dominance and white elicited the lowest level of dominance. The plotted results support the hypothesized negative relationship, but reveal a parabolic relation. Although dominance decreased monotonically with increasing brightness, the slope became less steep for brighter colours. Figure 3-28 also illustrates that the obtained dominance means for all five levels of brightness were predicted extremely well by the equation above.

The case of chromatic colours:

In their (1973) cross-cultural study in which subjects had to rate colour concepts (words) using the semantic differential factors (Osgood et al., 1957), Adams and Osgood found 'potency' to be correlated positively with darkness.

Sivik (1974a, 1975) used the Swedish Natural Colour System – NCS – (Hard and Sivik, 1979; Hard, 1975) as his descriptive colour model. Seventy-one colours (chosen to represent the NCS colour space) were judged on 26 semantic differential scales by a stratified sample of Swedish adults. Factor analysis yielded four factors: excitement, evaluation, potency and temperature. Blackness seemed to be the determinant of potency across all hues, but the extent of its effect varied among them. Also for the potency dimension a complex interaction appears to exist between the three NCS parameters and a given hue.

Damhorst and Read (1986), who investigated the effects of female job applicants' dark versus light clothing on male raters, found that men rated models who wore dark jackets as more powerful and competent than models who wore light jackets. Moreover, brightness of clothing was found to be even more important than facial expressions in determining judgements of 'potency'.

In a comprehensive study of emotional reactions to colour among 250 undergraduates, Valdez and Mehrabian (1994) also found support for the hypothesis that dominance is a negative correlate of brightness. Stepwise multiple regression analysis was used to explore the contributions of brightness and saturation to dominance. The dominance elicited by 76 colour stimuli was measured by means of Mehrabian's (1978) verbal-report PAD (Pleasure, Arousal, Dominance) scales. Averaged dominance-submissiveness responses (each colour sample was rated by approximately 25 subjects) constituted the dependent variable, and brightness and saturation the independent variables in the regression analysis, yielding the following equation:

Dominance = -.76 Brightness + .32 Saturation

From this study, a colour's brightness and saturation levels appear to explain a substantial portion of the variance in the feelings of dominance it induces (multiple regression coefficient

= .87). More specifically, the equation above indicates that less bright (i.e. darker) colours induce greater feelings of dominance in viewers. Thus, darker (i.e. less bright) colours appear to elicit feelings of dominance, strength or boldness. Moreover, the regression analyses indicates the effect of brightness to be considerably stronger than the effect of saturation in determining dominance responses to colour.

Furthermore, Valdez and Mehrabian (1994) found a second-order curvilinear relationship between colour brightness and the elicited dominance.

Dominance = 28.156 - 1.66 (Brightness) + .016 (Brightness)²

The multiple regression coefficient for this equation is .88. Actual mean values of dominance and those predicted from the equation above were plotted against brightness (for each of the six brightness values sampled) and showed extremely close agreement (see figure 3-29).



Figure 3-29: Actual and predicted average dominance levels as functions of colour brightness

Source: Valdez and Mehrabian, 1994, study 1.

Both the actual and predicted plots demonstrate that dominance declines steeply and monotonically with increasing brightness up to a brightness value of 43, beyond which dominance levels off. In other words, dominance appears to decrease as colours range from dark to light, but levels off for the lightest colours.

THE EFFECTS OF COLOUR ON EMOTIONS: Table 3-10: The Dominance-eliciting Properties of Colours

EFFECTS OF BRIGHTNESS Citation Sample Design Stimuli **Dependent** variable Findings The case of achromatic colours Wexner (1954) Study concerning Eight coloured Words that describe Found black to be associated with "powerfull/strong/masterful", mood tones implying high dominance. subjects' papers associations (not specified between colour according to samples and standard words that specifications) describe feelings The words "White" and "Grey" were found to be rated as weak across Subjects Colour names A potency dimension Adams and **Osgood** (1973) identified from a 23 cultures, whereas "Black" was rated as strong. across 23 (words) cultures semantic differential scale Black versus Observation of They found that black uniforms, compared with non-black uniforms, Frank and Investigated the Gilovich (1988) not only were associated with greater degrees of perceived aggression, effects of black non-black aggressive behaviour but also led to higher levels of player aggressiveness. uniforms of versus non-black professional uniforms of professional football and football and hockey teams hockey teams on aggressive behaviour

Valdez	125	Laboratory	5 achromatic	Dominance,	A nonlinear regression analysis revealed the following equation,
&Mehrabian	undergraduate	experiment	colour samples	measured according	expressing the relationship of dominance to brightness of achromatic
(1994) – Study 3	students	Colour patches	(ranging from	to Mehrabian's	colours (multiple regression coefficient = .65):
			white to black)	(1978) PAD emotion	Dominance = $25 - 1.2675$ (Brightness) + .0088 (Brightness) ²
				scale	
					Consistent with previous findings, black elicited the highest level of
					dominance, greys induce intermediate levels of dominance and white
					elicited the lowest level of dominance. Although dominance decreased
					monotonically with increasing brightness, the slope became less steep
					for brighter colours.

The case of chromatic colours							
Adams and Osgood (1973)	Subjects across 23 cultures		Colour names (words)	A potency dimension identified from a semantic differential scale	Found 'potency' to be correlated positively with darkness		
Sivik (1974a)	A stratified sample of Swedish adults	Studied colours in isolation	71 colours chosen to represent the NCS colour space (based on the Swedish Natural Colour System)	A potency dimension, identified from a 26 item semantic differential scale	Blackness seemed to be the determinant of potency across all hues, but the extent of its effect varied among them.		
Damhorst & Read (1986)		Investigated the effects of female job applicants' dark versus light clothing on male raters	dark versus light clothing	female job applicants' ratings by male raters	Found that men rated models who wore dark jackets as more powerful and competent than models who wore light jackets. Moreover, brightness of clothing was found to be even more important than facial expressions in determining judgements of 'potency'.		

Valdez and Mehrabian (1994) – Study 1	250 undergraduate students	Laboratory experiment with colour patches	76 colour samples : 10 hues, using 5 replication sets with equal	Dominance, measured according to Mehrabian's (1978) PAD emotion scale	Stepwise multiple regression analysis was used to explore the contributions of brightness and saturation to dominance, yielding the following equation: Dominance =76 Brightness + .32 Saturation.
			brightness and saturation levels.		explain a substantial portion of the variance in the feelings of dominance it induces (multiple regression coefficient = $.87$). More specifically, the equation above indicates that less bright (i.e. darker) colours induce greater feelings of dominance, strength or boldness in viewers. Moreover, the regression analysis indicates the effect of brightness to be considerably stronger than the effect of saturation in determining dominance responses to colour.
					Furthermore, Valdez and Mehrabian (1994) found a second-order curvilinear relationship between colour brightness and the elicited dominance: (multiple regression coefficient of .88). Dominance = $28.156 - 1.66$ (Brightness) + .016 (Brightness) ² .
					Dominance appears to decline steeply and monotonically with increasing brightness, up to a brightness value of 43, beyond which dominance levels off. In other words, dominance seems to decrease as colours range from dark to light, but levels off for the lightest colours.

3.5.5.4. Overview

Previous studies with regard to the effect of colour-hue on the dominance induced in viewers appear to demonstrate that empirical findings are inconsistent. Not using actual colour stimuli but colour names, Adams and Osgood (1973) found yellow to be rated as weak. Birren (1978) and Sharpe (1974) found green to be associated with withdrawal. In contrast they both found red to be rated as strong and associated with aggression (Adams and Osgood, 1973; Birren, 1978 and Sharpe, 1974). Although the results concerning the impact of colour-hue on dominance, obtained by Valdez and Mehrabian (1994), were generally weak and non-significant, they suggest the exact opposite. According to their findings, middle-wavelength hues appear to elicit more feelings of dominance in viewers than either the extreme short or long wavelength hues. A plausible explanation may be the failure of the earlier colour-studies to provide adequate specifications or controls of colour stimuli (i.e. absence of controls for saturation and brightness, while investigating effects of hue). Another possible explanation for these different findings could be that colours that are rated as weak elicit feelings of submissiveness). However, further research is needed to find out if this actually is the case.

Concerning the effect of colour saturation on dominance, empirical findings reveal much more consistent results. Potency was found to be directly dependent upon a colour's saturation (Osgood et al., 1957). The more saturated a colour, the more potent it is perceived to be (Hogg, 1969). Sivik (1974a, 1975) also found that as chromaticness increased, so did perceived potency. However he found this not to be true for yellow, revealing a complex interaction between the colour attributes for the potency dimension as well. Valdez and Mehrabian (1994) also found support for the hypothesis that dominance is a positive correlate of saturation. Apparently, more saturated (more vivid and purer) colours appear to elicit greater feelings of dominance, strength or boldness in viewers. Although they found that a colour's brightness and saturation levels appear to explain a substantial portion of the variance in the feelings of dominance it induces, a regression analyses indicates the effect of brightness to be considerably stronger than the effect of saturation.

With regard to achromatic colours, such as white, grey and black (representing brightness variations only) empirical findings also reveal consistent results. Consistent with previous findings (Wexner, 1954; Adam and Osgood, 1973; Frank and Gilovich, 1988), Valdez and

Mehrabian (1994) found that black elicited the highest level of dominance, greys induced intermediate levels of dominance and white elicited the lowest level of dominance. They did not only find support for a hypothesized negative relationship, but they also revealed a parabolic relation. Although dominance decreased monotonically with increasing brightness, the slope became less steep for brighter colours.

There is also a general consensus on the effects of colour brightness on feelings of dominance, as far as chromatic colours are concerned. Adams and Osgood (1973) found 'potency' to be correlated positively with darkness. Also Sivik (1974a, 1975) found blackness to be a determinant of potency across all hues, but the extent of its effect seemed to vary among them, revealing a complex interaction among the colour attributes. Damhorst and Read (1986) found brightness of clothing to be even more important than facial expressions in determining judgements of 'potency'. Indeed, men rated models who wore dark jackets as more powerful and competent than models who wore light jackets. In a comprehensive study of emotional reactions to colour among 250 undergraduates, Valdez and Mehrabian (1994) also found support for the hypothesis that dominance is a negative correlate of brightness. Less bright (i.e. darker) colours were found to induce greater feelings of dominance and strength in viewers. Moreover, the effect of brightness was found to be considerably stronger than the effect of saturation in determining dominance responses to colour. Furthermore, Valdez and Mehrabian (1994) found a second-order curvilinear relationship between colour brightness and the elicited dominance. Dominance appeared to decline steeply and monotonically with increasing brightness up to a brightness value of 43, beyond which dominance levels off. In other words, dominance appears to decrease as colours range from dark to light, but levels off for the lightest colours.

"The key lies in not looking for the magic link between colour and emotions,

but in exploiting the ways that colour affects one's appreciation

of objects and people involved in the setting"

BEACH, WISE & WISE

1988, p51

Colour can affect behaviour in one of three ways. First of all, it is generally accepted that colour can be used to convey meaning. By using colour-associations, thoughts and beliefs can be influenced, which may in turn impact attitudes, intentions and behaviours. Secondly, as mood can bias attitudes in a mood-congruent direction and colour can have a major impact on mood, it is clear that marketers and interior designers can use colour to actually influence moods and attitudes, and thus intentions and consumer behaviour. Finally, colour-evoked mood can also affect behaviour directly. Moreover, it has been suggested by industrial psychologists that the appropriate use of colour can enhance the overall quality of the environment and, thus, influence performance (Mahnke, 1981; Venolia, 1988; Ward, 1995; Pressly and Heesacker, 2001).

3.6.1. THE APPLICATION OF "COLOUR THEORY" TO REAL WORLD SITUATIONS

As colours applied to designed settings have been assumed to possess the same affecteliciting qualities as those revealed by laboratory experiments, they have been judged capable of influencing the emotional state and behaviour or performance of an environment's occupant (Etnier and Hardy, 1997). In the Human Factors Design Handbook (Woodson, 1981), one and a half pages deal with "typical colours and their effects on humans". Many prisons, hospitals, companies and schools have adopted systematic colour schemes, which may have been designed to produce particular mood states or behaviours in their inhabitants. Although Judd (1971) has suggested that the results of experimental colour research could serve as a tool (although crude and limited) by which one might begin to make colour decisions in the built environment, a severe lack of knowledge about colour research has been noted among specialists, like architects, interior designers and colour consultants (Janssens and Mikellides, 1998). The literature with regard to the effects of colours in the environment, is replete with anecdotal evidence, rather than based on systematic investigation. While there is no shortage of suggestions, they often lack empirical verification (Gorn et al., 1997).

As colour psychologists have discovered that warm colours such as red, orange and yellow tend to be more stimulating and exciting, whereas cool colours such as blue, green and violet tend to be more soothing, divergent colour applications have often been based on this distinction.

Thus, high-energy, warm colours, which are considered to stimulate and encourage excitement, are deemed more appropriate in places such as health clubs and fast-food restaurants, where high levels of energy and activity are generally desirable (Lindsay and Norman, 1973). Bellizzi et al. (1983) found warm colours to be more likely to draw customers to an outlet, but caution that they can also create tension. When the goal is to stimulate quick purchases or activity, warm colours are suggested to be more appropriate. Discount stores such as Target and Kmart often use a red-based colour scheme. Kmart recently switched from light blue to red to remove a disadvantage in this area (Hoyer and MacInnis, 2001, p164-165).

A newspaper clipping (Argue, 1991) asserts that red is also supposed to be an effective colour to use in restaurants, "because it leads people to eat more". According to Van Bergen (1995) Mc Donald's initially used a high chroma red, but, because too much red led customers to complain of headaches, they switched it to a "more relaxing" higher value pastel-shade of red. Wienerschnitzel, a hot dog restaurant chain with 350 locations in the United States, changed, the colour of its buildings, as advised by Wagner, the creator of the Colour Research Institute. After adding a little orange to the colour, which had to convey the message that the chain sold inexpensive hot-dogs, Wienerschnitzel reported a 7% increase in sales (Lane, 1991).

Red supposedly also makes people lose track of time, making it a good colour for use in casinos (Argue, 1991).

According to Humphrey (1976) one's response to red is a reflexive one. He argues that it serves the purpose of preparing (arousing) one to take some form of action, which is defined by the context. He notes that red is the most common colour signal in nature, arising from the fact that it contrasts well with both green foliage and the blue sky. Problems arise from the ambiguity of its signal: it can signal either approach (sexual display, edible food) or avoidance (aggressive behaviour, poisonous substance) (Mikellides, 1979; Beach et al., 1988).

Cool colours, on the other hand, are considered as restful and calming, and thus are deemed to be more appropriate in places such as hotel rooms, resorts, spas or doctors' offices, where it is

desirable for consumers to feel calm and relaxed (Lindsay and Norman, 1973; Bellizzi et al., 1983; Hoyer and MacInnis, 2001, p164-165). Cool colours are also advised in places where consumers are supposed to spend some time deliberating, such as banks, or stores that sell expensive consumer durables. In contrast to what may be expected, the systematic use of green in hospitals is not initiated by its relaxing qualities. Instead, green has been used in hospitals as a means of minimizing the after images¹³ experienced by staff members in the operating rooms (Beach et al., 1988).

Nevertheless, based on the attributions to the colour green as being calm and restful, it has been selected for use in redecorating the main cell block and solitary confinement areas at Alcatraz (Porter and Mikellides, 1976a).

Miller-Baker Pink (a kind of bubble gum pink; R:255, G:145; B:175), also known as "drunk tank pink", is also used in jails to calm violent prisoners, as this particular (low saturated, bright) colour was also found to have a tranquilizing effect and to suppress aggressive behaviour (Profusek and Rainey, 1987; Schauss, 1979; 1985; Color Voodoo, 1999; Honolulu Star Bulletin, 1999). The colour is now being used in more than 1400 government hospitals and correctional facilities in the U.S. (Walker, 1991).

The systematic application of colour has also been evident in sports. George Lumkin, associate head coach of the University of Hawaii football team, observed in 1991 that visitor locker rooms in the Universities of Iowa and Colorado were painted pink in the belief that the colour made players passive (Honolulu Star Bulletin, 1999). In an attempt to influence his own team's emotions and ultimately to influence his team's performance, Alonzo Stagg, a former head football coach at the University of Chicago, purposefully used a blue dressing room for rest periods and a red dressing room for fight talks (as cited in Etnier and Hardy, 1997). The Western Athletic Conference (WAC) has now enacted the rule that a visiting team's locker room cannot be painted in a different colour than the one of the home team. Coaches try to use colour to their advantage not only by altering the colour of the locker rooms, but also by experimenting with the colour of the uniforms and helmets (Zurcher, 2001). Frank and Gilovich (1988) investigated the effects of black versus non-black uniforms

¹³ Opponent colour processing is responsible for the fact that we can experience afterimages. When looking at the colour red, for example, for a longer period of time, our eyes get adapted to seeing this red colour. If we subsequently turn to look at a white wall, we will experience seeing a green afterimage (the opponent of red) (we refer to Levine (2000), for more details on this visual phenomenon).

of professional football and hockey teams on aggressive behaviour. They found that black uniforms, compared with non-black uniforms, not only were associated with greater degrees of perceived aggression, but also led to higher levels of player aggressiveness.

Nevertheless, while the application of "colour theory" to real-world situations has been extensive, research examining the effect of environmental colour on actual behaviour and performance has been sporadic and inconclusive (Etnier and Hardy, 1997).

3.6.2. EMPIRICAL EVIDENCE OF COLOUR-BEHAVIOUR EFFECTS

Although some sporadic studies have been conducted to investigate the effects of colour with regard to behaviour, regretfully much of this research is weak due to one major methodological drawback: the failure to provide adequate specifications and controls of colour stimuli (i.e. absence of controls for saturation and brightness, while investigating hue effects). Moreover, there is a lack of systematic research on the topic.

In the next paragraphs an overview will be presented on some of the empirical findings with regard to the effects of colour on actual behaviour. First the impact of environmental colour on task performance will be examined. Subsequently, the effect of colour on time perceptions and actual time spending will be addressed. And finally we will review some of the findings concerning the effect of colour on actual behaviours such as voting behaviour, cooperative behaviour and gambling behaviour.

3.6.2.1. Task-performance

Researchers examining the influence of colour on task performance have typically hypothesized that the relationship between colour and performance is partially dependent upon the classification of the task as a gross motor skill, a fine motor skill or a cognitive skill (Jokl, 1982; Etnier and Hardy, 1997). Colours with long wavelengths (e.g., red, orange) have been hypothesized to facilitate the performance of gross motor tasks, whereas colours with short wavelengths (e.g., green, blue) have been hypothesized to facilitate the performance of fine motor tasks and cognitive tasks (Jokl, 1982; Etnier and Hardy, 1997).

Although research on the influence of environmental colour on cognitive task performance has been scarce (Pressey, 1921; Wohlfahr, 1986), results of these studies indicated that colour did not influence cognitive task performance.

Substantially more research has been dedicated to the assessment of the effect of colour on the performance of motor skills (Goodfellow and Smith, 1973; Green et al., 1982; James and Domingos, 1953; Nakshian, 1964).

The most cited work in terms of how colour affects behaviour, is provided by Goldstein (1942). From his work with brain-damaged individuals, he observed that in the presence of green abnormal behaviours (such as trembling or an unstable pace) became less deviant, while in the presence of red, these behaviours became exaggerated. Based on these observations Dr. Goldstein postulated an affective theory of colour, which he believed applied to all individuals. He viewed red as having an "expansive" effect on the senses and of being capable of inducing a state of excitation in both emotional and motor behaviour. Green, on the other hand, had in his opinion a contractive nature and promoted tranquillity (Nakshian, 1964). Based on these premises, Dr. Goldstein (1942) felt that one's performance on certain motor tasks and judgements could be interfered with or disturbed by the colour red and facilitated by the colour green. However, neither numerical results, nor statistical analyses of his observations were ever presented (Beach et al., 1988; Nakshian, 1964; Norman and Scott, 1952; Goldstein, 1942).

In a study among 48 subjects, who had to perform a series of nine perceptual and motor tasks¹⁴ under red, green and grey colour conditions, Nakshian (1964) found little evidence to support Goldstein's (1942) premise that a red surrounding impaired and a green one facilitated certain motor performances and judgement abilities. The environmental colour conditions he used in his experiment were coloured partitions on a tabletop apparatus. Although the colours were matched on lightness according to the Munsell colour System, the red was more saturated than the green (5R 4.5/12 vs. 7.5G 4.5/6). Nevertheless, significant

¹⁴ The perceptual tasks studied include judgements of length and time. The motor tasks examined were comprised of fine motor tasks (such as hand tremor and tweezer dexterity) and gross motor tasks (such as most comfortable arm position, outward arm movement and motor inhibition).

differences could be revealed for only two out of the nine tasks: hand tremor and motor inhibition¹⁵. Performance for both tasks was better under the green condition than under the red.

Hammes and Wiggins (1962) found that colour illumination had no effect on high- or lowanxious subjects in a perceptual motor steadiness task under three different lighting conditions (red, blue and white). Red and blue illumination were produced by Wratten filters and brightness among the three lighting conditions was equated by the method of limits. Hammes and Wiggins (1962) thus found no support for the hypothesis that subjects perform best under blue illumination, due to its calming nature.

In a study among 25 females who had to perform two psychomotor tasks, Goodfellow and Smith (1973) found no differences in performance across five colour conditions. The colour conditions used involved a tabletop booth, painted in either red, blue, green, yellow or gray, with medium brightness and high saturation, matched to Munsell notations.

Neri, Luria and Kobus (1986) had eight subjects engage in a CRT (Cathode Ray Tube) reaction task, under four different illumination conditions (red, blue, low level white and no light). Subjects had to select the appropriate button, representing the quadrant on a CRT screen in which a stimulus was initially presented, as quickly as possible. In addition to ambient lighting, both CRT background colours and target colours were manipulated. The manipulated ambient lights, falling on the CRT screen were equated for brightness, but had rather low illumination levels. No significant differences in mean reaction times could be detected under the four ambient lighting conditions.

Hamid and Newport (1989) studied the effect of colour on mood and physical strength in preschool children. Results indicated that children exhibited more positive moods and greater strength when in a pink room than when in a blue room.

Investigating the effect of colour (red, green and white) on mood and task productivity, Kwallek and Lewis (1990) found that workers in red offices made the fewest errors, whereas workers in white offices made the most errors. Moreover, workers in red offices reported less

¹⁵ In both tasks inhibitory control over motor performance was measured.

confusion than workers in green offices did. Nevertheless, employees in red offices reported the colour red to be more distracting than those in white offices.

According to Etnier and Hardy (1997), the results of the studies by James and Domingos (1953), Nakshian (1964), and Green et al. (1982) lend some support to Jokl's (1982) colour theory. However, since the tasks that were used in these studies were not clearly classifiable as fine or gross motor tasks, it remains unclear how colour influences performance on these tasks.

Etnier and Hardy, (1997) examined the effect of environmental colour on performance of mentally and physically demanding tasks. The tasks, a fine motor task with a large cognitive component (the McCloy Blocks Test¹⁶, McCloy, 1942) and a gross motor task (the Wingate Anaerobic Test¹⁷) had to be performed in a cool-coloured (dark saturated blue-green) room, a warm-coloured (saturated reddish-orange) room and a neutral coloured white room. Colours were chosen from Sherwin Williams Decorator Colours, with the intention to keep brightness and saturation equal, so that the rooms differed only in hue. 30 undergraduate volunteers (17 females and 13 males) participated in the experiment. In accordance with Jokl's (1982) hypothesis, it was hypothesized that the warm colour would improve the performance on the gross motor task, and would hinder performance on the fine motor task. On the other hand it was hypothesized that the cool colour would improve performance on the fine motor task and would hinder performance on the gross motor task. In contrast to results found by other researchers (Green et al., 1982; James and Domingos, 1953), Etnier and Hardy (1997) found that environmental colour did not directly affect performance of either a fine motor task or a gross motor task. However, colour and time¹⁸ did interact to impact positive affect, F (8, 232) = 10.90, p<.0001 and desire to cope, F (2,58) = 15.07, p<.0001 (as measured by the Zuckerman Inventory of Personal Reactions), suggesting the possibility that tasks of a longer

¹⁶ The McCloy Blocks Test (McCloy, 1942) consists of an arrangement of blocks with different colours on the top and bottom (red, blue, yellow or green), which are supposed to be picked up according to specific rules. In deciding which block to select next, a mental sequence of colours needs to be followed. The time needed to complete the task is recorded to measure one's performance on the test. The authors note that the colours on the blocks could potentially confound the effects of environmental room colour, but assume that this would not systematically confound the results of their study.

¹⁷ The Wingate Anaerobic Test measures one's maximal anaerobic power with regard to the large muscle groups of the legs. The test involves a 30 sec. sprint on a cycle ergometer (Vandewalle, Peres & Monod, 1987). The number of revolutions is recorded as a measure of performance.

¹⁸ Affect was measured at several instances: when entering the main laboratory (1), before (2) and after (3) performing a cognitive task and before (4) and after (5) performing a motor task.

duration may be indirectly influenced by environmental colour. Positive affect decreased in the warm-coloured room more so than it did in the cool- and neutral-coloured rooms prior to performance of the cognitively demanding task. Thus subjects showed a less positive emotional response towards performing the cognitive task in the warm-coloured room. On the other hand, positive affect decreased to a lesser extent in the warm-coloured room prior to performance of the physically demanding task than it did in the cool- and neutral-coloured rooms. Thus subjects showed a more positive emotional response towards performing the physical task in the warm-coloured room. While these effects on positive affect did not influence performance in this study, it is possible that a task of a longer duration could make positive affect necessary for successful performance and thus the influence on positive affect could impact subsequent task performance. Additionally, after performance of the physically demanding task, desire to cope was found to increase more in the warm-coloured room than in either the cool- or neutral-coloured rooms. Again, while this effect did not influence performance, it is possible that in a situation in which repeated performance of a physically demanding task is required, the influence on desire to cope could impact subsequent task performance. These findings lend some support to the idea that colour influences the organism in a manner, which is dependent on the task type (Jokl, 1982).

Wolson and Case (2000) found players of a computer game with a blue screen to improve gradually over the session of five consecutive games, whereas players with a red screen peaked in the middle of the session and then deteriorated. A similar pattern for heart rate could be discerned, suggesting a mediating role of arousal in the effect. Players in the blue condition were found to make more errors than those confronted with the red background. Although varying sound had little impact, the red/loud combination was associated with feelings of "excitement" and "playing well", although this was not reflected systematically in the players' performance.

From this review of empirical studies on the effects of colour on task performance it is obvious that findings are inconclusive and that more research on the topic is called for.

3.6.2.2. Time Perceptions and Time Spending

In addition to hypothesizing specific effects of colour on psychomotor performance, Goldstein (1942) also postulated differences in psychophysical judgments of time passage under the colours red and green. Time distortions have since been investigated under more rigorous experimental settings (Beach et al., 1988).

The passage of time has been observed to be overestimated in a room painted with warm colours and underestimated in a cool-coloured room (National Aeronautics and Space Administration, Johnson Spacecraft Center, 1976).

Asking respondents to estimate the amount of time spent under two different lighting conditions and whether time exposure to a second light was equal to that of a first, provided in both cases shorter time estimations for the red as compared to the blue light conditions (Smets, 1969). However, Smets noted strong order effects in her data: perceived time duration was shorter for the first exposed colour light, whether it was red or blue.

Humphrey (1971) and Humphrey and Keeble (1977) found that monkeys seemed to prefer to spend more time in a blue light condition than in a red light condition. They also found that monkey responses were much faster in the red than in the blue light condition. They account for these findings by proposing that time appears to pass more quickly in red light than in blue light, referring to a kind of "subjective clock".

Srivastava and Peel (1968) conducted an experiment in which they compared time spending and movement patterns in the attendance of art museum exhibits where the walls were painted light beige or dark brown. Visitors in the dark room as compared to those in the light room were found to spend less time in the room, but to cover twice as much area and to take more footsteps (as measured by a hodometer which records footfalls unobtrusively).

Equalizing brightness through the method of averaged error, Caldwell and Jones (1985) found no significant difference in counting rates under red, blue or white illumination, nor did they observe differences under the three lighting conditions when subjects were required to estimate 35 and 45 second time intervals. They did however also find significant presentation effects in both experiments. A post-experiment interview revealed that sixty percent of the subjects reported they thought to count faster in the red light condition than in the blue one.

Bellizzi, Crowley and Hasty (1983) suggested – based on their findings – that customers may shorten their shopping time when in a tense, unpleasant environment and discourage for this reason the use of red and other warm colours inside the retail store. In a first experiment, however, Bellizzi and Hite (1992) found display background colour not significantly related to actual shopping time. Nevertheless, in a second experiment, approach behaviour intentions -including two time-related items (i.e. the intension to browse around and spend time in the store)- were found to be more favourable in a blue as compared to a red store.

Beach et al. (1988) conclude in a review that if there is a distortion in subjective time estimation due to colour, it is not a very robust phenomenon.

3.6.2.3. Miscellaneous Behaviours

Very unsystematically, a number of studies have examined the effect of colour on a range of different behaviours, such as gambling behaviour, voting behaviour, food choice etc... Again, most of these studies are methodologically flawed and unsystematic with regard to the selection and specification of colour stimuli used. Nevertheless, they provide some evidence on actual colour-behaviour effects.

Stark, Saunders and Wookey (1982) investigated the influence of ambient lighting on gambling behaviour. Twenty-eight subjects played in groups of seven a variation on the game of three card brag under blue and red ambient light conditions (matched for brightness, but otherwise unspecified). The authors found that significantly more money was waged and more bets were placed in the red light condition. Yet, the difference in money waged between the groups amounted to less than one pence (Beach et al., 1988). An interaction was also observed: as time increased, subjects waged more money, increased the number of bets, and these behaviours were more prevalent in the red light condition. From these findings, Stark, Saunders and Wookey (1982) concluded that red had a more excitatory effect on gambling behaviour and that people's behaviour became riskier in red light. However, they did not provide information regarding the average sums of money won or lost in each light condition.

Beach et al. (1988) critically reviewed the study and cautioned casino's for major redecorating decisions based on these findings.

Garrett and Brooks (1987) found that ballot colour affected voting behaviour when the candidates' sex was not specified. In that case men showed greater preference for candidates whose positions were printed on green ballots, than for those whose positions were printed on pink ballots. Female subjects showed the reverse preferences.

Roll's (1985) findings suggest that colour is an important variable in food selection. Walsh and Toma (1990) found children to prefer and choose candies (fruit-flavoured as well as candy-coated chocolate candies) on the basis of their colour. A main effect found for colour indicated that childrens' candy preferences ranged from red over green to orange with yellow being the least preferred.

Read et al. (1999) found changes in wall colour (red vs. white) and ceiling height to impact preschool children's cooperative behaviour.

Research on the influence of colour in the store environment demonstrates that colour also appears to influence shopper's emotions, beliefs, attitudes and behaviour (Bellizzi, Crowley and Hasty, 1983; Middlestadt, 1990; Bellizzi and Hite, 1992; Crowley, 1993). Indeed, colour appears to have the ability to physically attract consumers toward a retail display (Bellizzi, Crowley and Hasty, 1983). Furthermore, the colour in the store environment also appears to affect consumers' attitude towards buying a product (Middlestadt, 1990); it can influence simulated purchases (Bellizzi and Hite, 1992) and purchasing rates (Bellizzi and Hite, 1992), time spent in the store (Bellizzi and Hite, 1992), pleasant feelings (Bellizzi and Hite, 1992; Crowley, 1993), arousal (Crowley, 1993), product beliefs (Middlestadt (1990) and store and merchandise image (Bellizzi, Crowley and Hasty, 1983; Crowley, 1993). These studies will be discussed in more detail in the next chapter.

Chapter 4

Effects of Colour in the Store Environment: Review of Empirical Results, Research Model and Hypotheses



CHAPTER STRUCTURE

Chapter 4: Effects of Colour in the Store Environment: Review of Empirical Results, Research Model and Hypotheses

- 4.1. Introduction
- 4.2. Research Objectives
- 4.3. The Need to Study Colour in Context
- 4.4. The Influence of Colour In The Store Environment: Empirical Evidence
 - 4.4.1. Empirical Evidence provided by Bellizzi, Crowley and Hasty (1983) : "The Effects of Colour in Store Design".
 - 4.4.2. Empirical Evidence provided by Middlestadt (1990) : "The Effect of Background and Ambient Colour on Product Attitudes and Beliefs".
 - 4.4.3. Empirical Evidence provided by Bellizzi and Hite (1992) : "Environmental Colour, Consumer Feelings and Purchase Likelihood".
 - 4.4.4. Empirical Evidence provided by Crowley (1993) : "The Two-Dimensional Impact of Colour on Shopping".
 - 4.4.5. Overview of Empirical Evidence
- 4.5. The Influence of Colour in the Store Environment: Research Model
- 4.6. The Influence of Colour in the Store Environment: Hypotheses
 - 4.6.1. Store Colour-evoked Pleasure
 - 4.6.1.1. Store Colour Hue
 - 4.6.1.2. Store Colour Brightness
 - 4.6.1.3. Store Colour Saturation
 - 4.6.2. Store Colour-evoked Arousal
 - 4.6.2.1. Store Colour Hue
 - 4.6.2.2. Store Colour Brightness
 - 4.6.2.3. Store Colour Saturation
 - 4.6.3. Store Colour-evoked Dominance
 - 4.6.3.1. Store Colour Hue 4.6.3.2. Store Colour Brightness 4.6.3.3. Store Colour Saturation
 - 4.6.4. The Impact of Store Colour-evoked Emotions on Approach-Avoidance Behaviours

- The Moderating Role of Demographic Characteristics 4.7.

 - 4.7.1. Age 4.7.2. Gender
4.1. INTRODUCTION

This chapter focuses on the effects of colour in the store environment. First, the research objectives of this study are outlined in section 4.2. In section 4.3. the need is established to study colour in context and specifically to study the effects of colours applied to environments. Section 4.4. presents the available empirical evidence regarding the influence of store-interior colour, provided respectively by Bellizzi, Crowley and Hasty (1983), Middlestadt (1990), Bellizzi and Hite (1992) and Crowley (1993). In section 4.5. the applied research model, which is based on environmental psychology, is proposed. Subsequently, in section 4.6. our hypotheses with regard to store-colour-evoked emotions and the impact of store-colour-evoked emotions on approach-avoidance behaviours are presented. Finally, in section 4.7. the possible moderating roles of the demographic characteristics, age and gender, are discussed.

4.2. RESEARCH OBJECTIVES

Research on store atmospherics is predominantly based on environmental psychology and specifically on the Mehrabian and Russell (1974) Stimulus-Organism-Response model, as introduced to the study of store atmophere by Donovan and Rossiter (1982). In this context it has been proposed that approach/avoidance behaviours of customers are largely determined by individual internal (cognitive, emotional and physiological) responses to the store environment (Bitner, 1992). Thus, atmospheric variables have been found to influence a wide variety of consumer evaluations and behaviours (see Turley and Milliman, 2000 for a review).

The study, presented here, aims to fill a gap in this extensive stream of research concerning the impact of store atmospherics on the consumer. In their suggestions for further research with regard to atmospheric effects on shopping behaviour, Turley and Milliman (2000) draw the attention to store interior colour, which has not received the attention it probably deserves. As research on the impact of colour in the store environment is scarce and methodologically flawed, it is our challenge to assess the impact of this store design element more thoroughly.

Although it has been demonstrated that environmental colour can actually influence the emotions, beliefs, attitudes and behaviours of retail shoppers (e.g. Bellizzi et al., 1983; Middlestadt, 1990; Bellizzi and Hite, 1992; Crowley, 1993), it is still unclear how exactly colour affects approach behaviour.

Moreover, studies on the subject are flawed because they failed to provide adequate specifications and controls of colour stimuli, a caveat noted to be common among colour research (Gelineau, 1981, Beach et al., 1988; Valdez and Mehrabian, 1994). Indeed, colour stimuli have not been specified according to a standard colour system, but only vague verbal descriptions have been reported. Thus, hue effects have been tested, without controlling for saturation or brightness, this way confounding the effects of the three colour attributes. Moreover, only effects of colour-hue have been investigated in a retailing context, whereas effects of saturation and brightness have been ignored completely. Yet, these neglected colour dimensions may very well be the major determining factors in colour effects on human feelings and behaviour (Valdez and Mehrabian, 1994; Gorn et al., 1997).

In fact, Bellizzi, Crowley and Hasty note in their 1983 paper that the fully saturated colours used in their study may be too strong for many design applications and that it still has to be determined whether less saturated versions of these colours would have similar effects. Up to now, however, this simple question still has not been investigated yet. Indeed the scarce research on the topic published thus far, focused on the effects of hue, without considering effects of brightness and saturation and worse, not even controlling for them. Also Bellizzi and Hite (1992) call for further research into the area in order to develop an understanding of how colour may affect consumers. They also suggest to study the effects of other colours and "other" reds and blues, referring to reds and blues with different value and saturation levels.

In the current research endeavour we will attempt to address these vital issues. Two basic objectives are contemplated. First and foremost, a more accurate methodology will be proposed for assessing the impact of colour in the store environment on shopper behaviour. Applying an environmental psychology approach, the proposed study aims to assess (1) the direct effect of colour in the store environment on the emotions experienced in the store (notably pleasure, arousal and dominance) and (2) the direct and indirect effect of colour on consumers' appoach/avoidance behaviour through the store colour-evoked emotions. In particular, the specific effects of colour hue, saturation and brightness in the store design will be investigated. Moreover, the moderating impact of individual differences with regard to atmospheric responsiveness will be assessed.

Although Bellizzi et al. (1983) hoped that their initial study would serve as a catalyst to encourage additional research interest in the role of colour in retail store management, they did not have many academic adherents. With this research project we hope not only to gain a deeper understanding of the effects of colour in the store environment, but also to generate some revived academic interest into the subject.

Still, the booming retailing industry already shows a stupendous interest in the research topic (Brengman, 2002; Retail-Update, 2002; POPAI-Benelux; Cahan, 2002). Indeed, the fact that almost two thirds of all purchase decisions are made in the store (POPAI, 1998) has raised the attention to the point-of-purchase and to store atmospherics. From a more practical point of view, this research is intended to assist retailers in selecting and avoiding certain colours in order to produce desired consumer moods and in-store behaviours.

"There is little systematic empirical research in marketing on the effects of colour"

> GORN et al. 1997, p1387

4.3. THE NEED TO STUDY COLOUR IN CONTEXT

Norman and Scott (1952) criticize that a "primary defect" of colour preference research is that the colour-stimuli used almost always involve colour patches or colour names in isolation, ignoring the fact that colour is almost always associated with some object or context. According to Norman and Scott (1952), judged preferences for colours presented as small chips reveal little about preferences for coloured objects or environments.

An early study by Washburn, McLean and Dodge (1934) already revealed that even the size of the colour stimulus could affect preference judgements. They found that preference ratings for some colours (yellow and orange) varied with stimulus size, the smaller size (5x5 cm) being preferred while the larger size (25x25 cm) was judged as unpleasant. This may be yet another reason for inconsistencies reported by various researchers on colour preferences (Beach et al., 1988).

The findings of two studies by Osgood et al. (1957) also suggest that colour acceptance is defined by the object with which it is associated. Conducting one of the initial studies on the meaning of colours, Osgood et al. (1957) studied selected colours in the context of objects. In a first study one of eight colours was applied to either one of five objects (shirt, ice-cream, rug, car, cake mix) or as background. With regard to the "evaluative" dimension, a significant colour x object interaction was identified. In terms of evaluation, yellow was rated as the "most favourable" colour, in contrast with previous studies using colour patches, which have found yellow to be generally disliked. The authors suggested that their unexpected finding may have been an artifact of the objects used in the study. In a second study, colour was studied in relation to abstract sculptures. As in the first study, also with regard to abstract sculptures, the colour x object interaction was found to be significant. In this case, blue was the most preferred hue.

In the opinion that more work needed to be done, such as testing colours applied to objects, Guilford and Smith (1959) caution their readers with regard to applying their findings to real world situations.

Taft (1997) compared evaluative semantic ratings (beautiful-ugly, elegant-vulgar, louddiscrete, masculine-feminine and warm-cold) of colour chips and five objects of the same colour for 13 colours. In contrast to the findings by Osgood et al. (1957), Taft's analyses indicate that generally few significant differences exist between chips and object ratings for the same colour. The results of this study have implications for the generalizability of the results of earlier colour meaning research and for the use of colour chips in colour planning.

Rather sceptical as to the effect of colour in the environment, Beach et al. (1988), nevertheless, point out, that studying colours in isolation (e.g. colour boards or colour samples) provides us with little information, other than how a subject at that moment rates a given colour on a given scale. They argue that colours need to be studied in context, especially in environmental contexts, and feel that even then caution must be taken with respect to the generalizability of the findings. Whereas Judd (1971) suggested that the results of experimental colour research could serve as a tool by which one might begin to make colour decisions in the 'built environment' (i.e. architectural design), Beach et al. (1988, p99) argue that *"it is fruitless to build up to a theory of colour application in settings from highly reductionist experiments alone"*. In an extensive literature review on the effects of colour, Beach et al. (1988) note that the evaluation of colours needs to be studied in the context of a given object or situation to yield most meaningful results.

However, as studies on colour preference and colour meaning still systematically rely on the use of colour chips, and the study of colours in environmental contexts is scarce, it remains a critical question whether preferences for and meanings of isolated colour-chips are generalizable to contextual colours, such as store colour.

Anyway, as findings from early laboratory experiments on the affective and connotative qualities of colours did become spread, these associations were believed to hold when applied to environmental settings as well. Thus colours were believed capable of influencing the emotional state of an environment's occupant (e.g. Birren, 1983). For this reason specific colours have been applied in sports locker rooms, hospitals, jail cells, fast-food restaurants and other places. With regard to the "effects" of colours in the environment, however, the literature seems to contain mostly anecdotal evidence, revealing a severe lack of systematic empirical research.

As it is the aim of this particular study to investigate the effects of colour in the store environment, we will first review the studies on colour effects that have been conducted in a retail setting more carefully. "Marketing and retail managers should be aware that colour can have negative or positive marketplace implications that may affect not only mood or feelings, but also in-store behaviour"

> BELLIZZI & HITE 1992, p361

4.4. THE INFLUENCE OF COLOUR IN THE STORE ENVIRONMENT: EMPIRICAL EVIDENCE

Two decades ago, Bellizzi, Crowley and Hasty (1983) observed that while many studies had investigated the effects of colour, little research had been addressed to the specific use of colour in retail applications. They suggested that retailers should be careful in attempting to create exciting and attracting store environments, because too much excitement and attraction may lead to shopper avoidance of uncomfortable environments. Up until this day only four laboratory experiments have been published on the influence of colour on retail shoppers. Although scarce (Turley and Milliman, 2000), research on the influence of colour in the store environment demonstrates that colour appears to influence shoppers' emotions, beliefs, attitudes and behaviour (Bellizzi, Crowley and Hasty, 1983; Middlestadt, 1990; Bellizzi and Hite, 1992; Crowley, 1993). Indeed, colour appears to have the ability to physically attract consumers towards a retail display (Bellizzi, Crowley and Hasty, 1983). Furthermore, the colour in the store environment also appears to affect consumers' attitude towards buying a product (Middlestadt, 1990); it can influence simulated purchases (Bellizzi and Hite, 1992) and purchasing rates (Bellizzi and Hite, 1992), time spent in the store (Bellizzi and Hite, 1992), pleasant feelings (Bellizzi and Hite, 1992; Crowley, 1993), arousal (Crowley, 1993), product beliefs (Middlestadt (1990) and store and merchandise image (Bellizzi, Crowley and Hasty, 1983; Crowley, 1993).

In spite of the fact that these studies have found that colour influences the behaviour of retail shoppers, it is still unclear how exactly colour affects shoppers' emotions and behaviour. As mentioned previously, these past studies are flawed, because they have a major methodological drawback in that they failed to provide adequate specifications or controls of colour stimuli (e.g., absence of controls for saturation and brightness while investigating effects of hue), a caveat noted to be common among colour research (Gelineau, 1981, Beach et al., 1988; Valdez and Mehrabian, 1994). In fact, colour stimuli can be characterized completely in terms of hue (i.e., wavelength), brightness or value (i.e., black-to-white quality) and saturation or chroma (i.e., purity or vividness, with lower saturation colours containing more grey) (Munsell, 1966; Valdez and Mehrabian, 1994). In the studies by Bellizzi, Crowley and Hasty (1983), Middlestadt (1990), Bellizzi and Hite (1992) and Crowley (1993) however, only vague verbal descriptions are reported with regard to the colour stimuli used. Moreover, only effects of hue have been investigated, but saturation and brightness have not been

controlled for. Effects of saturation or brightness have never been tested in a retailing context although these neglected colour dimensions may well be the major determining factors in colour effects on human feelings and behaviour (Valdez and Mehrabian, 1994; Gorn et al., 1997).

Nevertheless, a comprehensive review of the methodologies used in these previous studies and the findings obtained may be interesting to situate our own empirical study and will be presented next.

4.4.1. EMPIRICAL EVIDENCE PROVIDED BY BELLIZZI, CROWLEY AND HASTY (1983): "The effects of colour in store design".

The first and most cited empirical study with respect to the impact of colour on retail shoppers was conducted by Bellizzi, Crowley and Hasty (1983). Their results suggest that colours can physically attract shoppers toward a retail display and that they may affect consumer perceptions of the store environment and merchandize carried.

A convenience sample of 125 female subjects, aged between 18 and 64, was randomly assigned to 5 colour conditions in a laboratory setting, in order to investigate whether the colour of a retail store display area affects approach orientation, physical attraction and store and merchandize image.

To bring the participants in the right frame of mind during *the first phase of the experiment*, each subject, consecutively, was handed a lightweight chair upon entering the experimental room, together with a distracting questionnaire related to shopping habits. Subjects were observed and photographed through a one-way mirror while filling out this initial questionnaire. For each colour condition, fabric panels in five experimental colours alternately covered the entire far wall of the experimental area. The five experimental colours used, were full strength hues taken from the "Color Vu Corporation Standard Color Chart": Red (#1), Yellow (#57), Green (#161), Blue (#129) and White. The floor and other walls of the experiment room were covered in M/7 gray. Approach orientation was operationally defined as the direction the subjects faced in relation to the experimental far wall upon entering the experimental room. It was recorded where and how participants sat in relation to this coloured wall. This was recorded on a six-point scale with those directly facing the

coloured wall on the one extreme receiving a score of six and those facing the exact opposite direction on the other extreme receiving a score of zero. Subjects facing other directions were assigned scores from 1 to 5, depending on the angle with the experimental wall. In measuring approach orientation the authors were attempting to measure a construct linked to attention (i.e. the attention-grabbing quality of the colour). The physical attraction of the colour was defined as the linear distance that subjects sat from the experimental wall, measured in inches with a standard tape measure placed at a right angle of the experimental wall.

Although no relationship could be revealed between colour and approach orientation, colour was found to be associated with physical attraction (ANOVA, p=.043). Subjects participating in the experiment sat closest to the yellow wall and furthest from the green wall. Duncan's multiple-range test indicated that the distance that subjects sat from the yellow wall significantly differed from all the other treatments except for red. Warm (yellow, red) coloured walls were found to generate a stronger physical attraction on the subjects than cool (green, blue) coloured walls (t-test, p=.02). Colour preference and the interaction between colour treatment and colour preference were not significant. Subjects sat significantly closer to the warm walls regardless of colour preference.

Subsequently, in the *second phase of the experiment*, subjects were presented two slides projected on an 8-foot square screen on the near wall of the experimental room. The first slide depicted the front of a furniture store and the second one represented an architectural drawing of the interior of the store, with for each colour condition a different colour fabric placed underneath the slide, creating the effect of background colours. These background colours matched the colours of the far wall of the experimental room. A questionnaire, consisting of 20 7-point bipolar items, was administered to capture perceptions of the store environment and merchandize. The first thirteen of these items were taken from the "Judgements of Environmental Quality Scale" (Fisher, 1974), a scale designed to assess general perceptions of various aspects of the retail store and merchandise. Information on age, occupation and colour preference was also requested.

With regard to the influence of colour on perceptions of the retail environment and merchandise, seven of the twenty items were found to have a significant influence (at the .10 level). Duncan's multiple-range test indicated that the red store interior was seen as

significantly more colourful than the other store interiors. Yellow, blue, green and white interiors did not score significantly different on the drab/colourful scale, nor were warm colours considered more or less colourful than cool colours. With regard to the negative/positive scale, the red and the blue colour treatments significantly differed in their assigned scores: the red interior was seen as more negative, while the blue interior was judged to be more positive. More general, warm coloured interiors were judged more negative than cool coloured interiors. The red interior was also found to be more tense than the other treatments. The Duncan test also indicated that yellow was seen as more tense than blue, but not green, and that blue was found to be more relaxed than red and yellow. In general warm interiors were judged more tense than cool interiors. Blue was found just as lively as red, while white was definitely judged less lively... In general warm coloured interiors were thought to carry more up-to-date merchandize than the cool coloured stores. In fact merchandise in the red store environment was rated as most up-to-date.

To summarize, certain perceptive differences were noted: warm colours, red in particular, were viewed as bright and colourful yet also as negative and tense. On the contrary cool colours, blue in particular, were rated as positive and relaxing. In general, cool coloured interiors were viewed as more attractive than warm coloured interiors and respondents also associated the cool coloured stores with a more pleasant shopping environment. There was no perceived difference between the colour treatments among perceptions of merchandise prices, merchandise quality and friendliness of store personnel.

After studying the colour-connotations with regard to the individual items, the 20 items were also factor analysed using VARIMAX rotation, revealing three meaningful factors:

- a non-evaluative "activity" dimension (i.e. unlively/lively, unmotivating/motivating, boring/stimulating, confer Osgood, Suci and Tannenbaum's "activity aspect of meaning", 1957).
- (2) an "evaluation" dimension (i.e. negative/positive, unattractive/attractive, tense/ relaxed, uncomfortable/comfortable, bad/good and unpleasant/pleasant), confer Osgood, Suci and Tannenbaum's "evaluative aspect of meaning", 1957).
- (3) a price-quality dimension (i.e. low price/high price and low quality/high quality)

The red coloured store scored significantly higher on the activity dimension than the other treatments (Duncan's test). In general the warm coloured stores were perceived as significantly more active than the cool coloured stores. On the evaluative dimension, the blue coloured store received a significantly higher score than the stores in other colours (except for green and white). In general the cool coloured stores were rated significantly higher on the evaluation dimension than the warm coloured stores. No significant differences were found between the colour treatments on the price-quality dimension.

Thus warm colours, red in particular, appear to score high on an activity factor, but low on an evaluative factor. On the contrary cool colours, blue in particular, were rated as favourable on an evaluative factor and low on an activity factor. Colours used in the study did not affect price-quality perceptions.

According to the authors the most interesting finding of this study is that, regardless of colour preference, subjects appear to be physically drawn to warm colours; however they feel that warm-coloured environments are generally unpleasant. Subjects perceived red and warm coloured store environments as tense and negative, while cool coloured store environments were perceived as calm and positive. Therefore, it is suggested to use warm colours for display windows and a retail store's exterior design. Using red and other warm colours inside the retail store is discouraged, because customers may shorten their shopping time when in a tense, unpleasant environment and the shopping trip may even be reduced up to the point where purchase decisions are postponed. It is interesting to note that the colours used in the study did not affect perceptions of price nor quality of merchandize.

4.4.2. EMPIRICAL EVIDENCE PROVIDED BY MIDDLESTADT (1990): "The Effect of Background and Ambient Colour on Product Attitudes and Beliefs".

Following a more cognitive approach (Fishbein and Ajzen, 1975), arguing that attitudes, intentions and behaviours are determined by underlying beliefs and that change ultimately occurs by changing this cognitive structure, Middlestadt (1990) investigated the effects of blue and red background colours on product beliefs. In her study, Middlestadt (1990) does not focus on the effects of colour per se, but rather on the process by which colour produces

attitude change. She suggests that belief-based change processes may be responsible for some of the changes previously assumed to be affect-based.

84 female undergraduate students participated in this study. Subjects were, in goups of about 6 to 10, randomly assigned to one of two experimental conditions, red or blue. In each condition three products were subsequently presented: a bottle of perfume, a gold and silver pen and a bottle of mineral water, projected against either a red or blue background. Not only the background colour of the slides was manipulated, but also the surrounding (i.e. ambient) colour of the room. For this purpose additional lights in the slide-background colour (red or blue) were projected at the ceiling in the front of the room. No attempt was made to control the luminance of the projected light and it was not specified what exact hues were used, nor was there a control for saturation or brightness of the hues.

For each product, the attitude towards buying the products was assessed directly by means of six seven-point bipolar semantic differential items (e.g. buying the pen would be pleasant/unpleasant, enjoyable/unenjoyable, I like/I dislike, good/bad, wise/foolish, beneficial/harmful). The underlying cognitive structure was measured as recommended by Ajzen and Fishbein (1980). For each of eight salient consequences of performing the purchase behaviour, a belief item assessed the likelihood that this outcome would occur if the product was bought and an evaluation item assessed the evaluation of the outcome (i.e. each respondent indicated on a seven-point bipolar scale, how good or bad the outcome was).

No significant effects of the colour manipulations could be revealed on attitudes or beliefs with respect to two of the products, the perfume and the mineral water. In contrast, a significant effect was found of colour on the attitude towards buying the pen pictured in the slide. Respondents who were exposed to the pen with the blue background and ambient colour had a more positive attitude towards buying the product than those exposed to the pen with the red background and ambient colour. Over the six items, the multivariate F (Approx. F (6, 77)=3.21) was statistically significant at beyond the .01 level. For each of the individual items the relation was in the hypothesized direction, however, only for two of them (pleasant/unpleasant and enjoyable/unenjoyable) the mean difference was statistically significant at beyond the .01 level.

For the pen a difference in attitude towards buying the product was found between the two colour conditions. Middlestadt (1990) investigated whether this effect could be mediated by an effect on the underlying cognitive structure. For this reason the sum of the eight beliefs times evaluation was calculated, which provides an indirect assessment of attitude and represents the evaluative implications of the underlying cognitive structure. The correlations of this measure with the individual semantic differential items were all statistically significant ranging from .45 to .59. Also for this measure, those exposed to the pen in the blue condition were more positive towards buying the pen than those who were assigned the red condition (F (1,82)=63.57, significant at beyond the .01 level). This difference implies that the cognitive structure of those in the blue condition had significantly more positive evaluative implications than those in the red condition; i.e. they believed that buying the pen was likely to lead to positive outcomes and unlikely to lead to negative ones.

Concerning the effect of colour on beliefs and evaluations of individual outcomes, multivariate analyses of variance on the individual outcomes revealed a significant effect of colour over the eight beliefs (Approx. F (8, 75)=3.14) at beyond the .01 level, but no significant effect over the eight evaluations (Approx. F (8, 75)=1.32). The background and ambient colour appeared to have the strongest effect on two beliefs: the degree to which buying the pen was found to be something which is elegant and which is unique. Compared to those in the red condition, subjects in the blue condition believed it to be more likely that the pen was elegant as well as unique. With regard to the beliefs that buying the pen would mean buying something of good quality and something which has a pleasant colour, marginal, however not significant, the pen presented in the blue condition was perceived as slightly more likely to be more expensive than the pen presented in the red condition.

To summarize, in this study, it was shown that background and ambient colour had an effect on the attitude towards buying a product. Respondents presented a slide with a pen against a blue background (with blue ambient colour) exhibited a more positive attitude towards buying the product than those who were shown the same pen against a red background (with red ambient colour). It was also demonstrated that subjects in the blue condition also held different beliefs about buying the product. Thus although background and ambient colour would seem, on the surface, to carry no product information, it was found to affect the underlying belief structure as well as the attitude towards the product. Middlestadt (1990) attempted to determine how colour could produce this effect on product attitudes. Since the background colour does not explicitly convey any product information, it might be argued that the affect associated with the colour is directly transferred to the product and thus created a positive attitude. Middlestadt measured the underlying cognitive structure in detail and demonstrated that the colour affected the subject's beliefs about buying the product. Compared to subjects in the red condition, those who were exposed to the pen in the blue condition believed more strongly that buying the pen would lead to two positive outcomes: buying a pen which was elegant and which was unique. This difference in beliefs resulted in a cognitive structure that was more positive for those exposed to the pen against a blue background.

Thus it could be argued that the change in attitude toward buying the pen found in this study was caused by a belief-based change process. However, the data do not prove that a cognitive difference mediated the attitude and behaviour effects found. As argued by Gorn (1982), it is possible that the change in the beliefs occurred after and in justification of the change in attitude.

4.4.3. EMPIRICAL EVIDENCE PROVIDED BY BELLIZZI AND HITE (1992): "Environmental colour, consumer feelings and purchase likelihood".

Bellizzi and Hite (1992) tested the effects of red and blue in a shopping-related context. They selected red and blue because of their opposite colour properties. Two laboratory experiments were conducted to test hypotheses concerning the effects of colour to induce moods or feelings that subsequently may induce behaviour or behavioural intentions in a simulated purchase context. For both experiments retail environments were simulated using predominantly red or blue colours. In the first experiment a behavioural dependent measure was used. In the second experiment, in addition to the behavioural intention measure, several mood-related measures were included in order to identify the specific mood-states that may be induced by different colour environments.

Experiment 1:

During the course of the first experiment 70 female subjects were, one-by-one, exposed to one of two experimental conditions. Either a red or a blue retail environment was created, by

projecting five slides on 5-foot square screens, placed side by side along the wall of the experimental room. On the slides four different models of console televisions were depicted along with descriptions. The televisions were pictorially very similar, screen sizes and cabinets were the same for each set; however features and sizes were different. The fifth slide represented a no-purchase option. All the slides had either a red or a blue background, depending on the experimental condition. The colour stimuli used were Canson Paper – red #505 and Canson Paper – blue #595. The colours were clearly specified, however they were not controlled for saturation or brightness.

Subjects were told to assume that they had decided to buy a console (not big screen) model television and that they could start shopping in the experimental store. During this simulated shopping trip participants were asked to examine different televisions and to decide which one they would buy. To help them in their decision, a handout with descriptions of the T.V. features highlighted in the displays was given. Finally subjects had to indicate which television-set they would buy. In case they did not feel comfortable in selecting one of the television sets, assuming they had decided to buy one, subjects were also given a "no-buy" option, which would give them the possibility to shop around in another store before making a decision. The subjects were unobtrusively timed while shopping with an electronic stopwatch. After making their purchase selection, they were asked to complete a short questionnaire.

The effects of the red and blue displays on two dependent variables (purchase/non-purchase and shopping time) were examined. Based on the finding that blue is perceived as more positive, pleasant and calm (Bellizi et al., 1983), it was assumed that a more positive, pleasant and calm environment would make shoppers feel more comfortable and, therefore more likely to make a purchase. With their first experiment Bellizzi and Hite (1992) found support for the hypothesis that the blue display would produce higher purchase rates (lower postponement rates) than the red display. A total of 39% of the shoppers in the red store postponed selection, compared to only 18% of the shoppers in the blue store (with χ^2 significant at the .05 level). During the simulated shopping trip, shoppers in the red store spent an average of 314\$, which was significantly less than the average spending of 458\$ in the blue store. The findings also revealed that the shoppers in the red store. One half of the shoppers viewing the blue displays selected the most expensive set (\$599), compared to only 19% of the shoppers viewing the red displays (t-test, significant at the .01 level).

Because red was expected to be more distracting (Gerard, 1957) than blue and to increase task difficulty and anxiety, Bellizzi and Hite (1992) hypothesized that shopping time would be longer in the red display condition and shorter in the blue display condition (Jacobs & Suess, 1975; Goldstein, 1939; Smets, 1969). However, they found no support for this hypothesis. The colour treatment was not found to be significantly related to shopping time. Shoppers spent an average of 169 seconds in the red store as compared to 145 seconds in the blue store, which was in the hypothesized direction, however the difference was not significant. In retrospect, Bellizzi and Hite (1992) admitted that using time as an indicator of distraction may not have been appropriate. They suggest that an alternate hypothesis could be advanced which would link the comfort induced by the blue environment with increased shopping time. Indeed, Bellizzi, Crowley and Hasty (1983) suggested, based on their findings, that customers may shorten their shopping time when in a tense, unpleasant environment and discourage for this reason the use of red and other warm colours inside the retail store. Bellizzi and Hite (1992) suggest that although distraction may always be negative, arousal may be desired and only become negative at extreme levels. They suggest that future research should attempt to measure colour effects such as distraction and arousal more directly (rather than using shopping time as a proxy).

Experiment 2

107 graduate marketing students participated in the second experiment. Again two colour stimuli were utilized: subjects were randomly assigned to either a red or a blue colour condition. This time the stimulus involved the projection of an architectural drawing of the interior of a furniture store transferred on a transparency with coloured fabrics underneath, creating the effect of a background colour. The colours were full-strength hues taken from Color Vu Corporation Standard Colour Chart red (#1) and blue (#129). Note that they were not controlled for saturation nor brightness.

A questionnaire was administered containing three major sections. The first section assessed consumers' emotional state while in the store. Subjects were instructed to respond on how they would feel while they would be shopping in the store depicted on the slide. The measurement items included the semantic differential pleasure-arousal-dominance (PAD) measures of Mehrabian and Russell (1974) (as modified for retail environments by Donovan and Rossiter, 1982). The second section included the 8-item approach/avoidance (purchase) intentions used by Mehrabian and Russell (1974), as adapted by Donovan and Rossiter

(1982). The last section included a demographic item (subjects' gender), a shopping interest item and an item involving colour blindness.

Purchase intention, as measured by the approach/avoidance scale, was expected to be higher in the blue coloured store, than in the red coloured store, because both the arousal and negative evaluation properties of red, were expected to enhance anxiety and, therefore uneasiness with shopping in the red store environment. The blue environment was expected to produce more pleasurable as well as less arousing feelings. The dominance scale was administered in an exploratory fashion.

As expected, subjects in the blue environment expressed a greater intention to shop, browse and buy in the simulated store. Five out of the eight items in the approach-avoidance scale produced statistically significant results (at the .09 level or lower), favouring the blue environment. Moreover, all items generated results in the expected direction. An average score (α =.86) (as used by Donovan and Rossiter, 1982), produced a significant result at the .045 level, indicating greater intention to approach in the blue store and to avoid in the red store.

With regard to the PAD scores, factor analysis with varimax rotation identified 4 factors: pleasure (α =.89), arousal (α =.88), dominance (α =.79) and a freedom-of-movement factor (α not mentioned – explaining only 5.5% of the variance). Colour effects were observed on the pleasure dimension only and not on the arousal, dominance or freedom-of-movement dimensions. Respondents were found to report a more pleasant feeling in the blue store than in the red store (p<.01). Not only were more pleasurable scores reported for the blue environment, purchase intentions were also linked to more pleasurable feelings.

No gender difference with regard to approach-avoidance and the colour conditions could be revealed.

The results of experiments 1 and 2 clearly show more positive consumer reactions to blue. The use of blue displays, instead of red ones, appears to have a favourable effect on simulated purchases (experiment 1) and approach/avoidance behaviour (experiment 2). Arousal could not be identified as a possible causal link, since no effect of colour on arousal could be revealed for the two hues (blue and red) used in this study. Therefore, experiment 2 suggests

that colour-evoked pleasure is solely responsible for the outcome. The identification of an evaluative link (good/bad; negative/positive), was also reported by Bellizzi et al. (1983), who also concluded that consumers view red retail environments as more negative and unpleasant than blue ones. In contrast to the findings by Bellizzi and Hite (1992), Bellizzi et al. (1983), did find warm coloured store environments (red in particular) also to be more active or arousing than cool coloured store environments (such as blue).

The work of Bellizzi and Hite (1992) goes beyond that of Bellizzi et al. (1983) by demonstrating the impact on a behavioural consumer reaction, simulated purchase (experiment 1) and approach behaviour intentions (experiment 2). Bellizzi and Hite (1992, p361) notice that: "marketing and retail managers should be aware that colour can have negative or positive marketplace implications that may affect not only mood or feelings, but also in-store behaviour".

Bellizzi and Hite (1992) warn, nevertheless, that because of the artificial laboratory environments used in both experiments, practical retail recommendations can be advanced only with great caution. Although in experiment 2 a different product was selected, a different set of stimuli was used and a separate sample was obtained, the results of this second experiment are consistent with those of experiment 1 and yield support for the overall conclusion regarding the positive effects of blue and the negative effects of red. Moreover, they also suggest an explanation for the findings. Still, Bellizzi and Hite (1992) call for further research into the area in order to develop an understanding of how colour may affect consumers. They also suggest to study the effects of other colours and other reds and blues.

4.4.4. EMPIRICAL EVIDENCE PROVIDED BY CROWLEY (1993): "The two-dimensional impact of colour on shopping"

Crowley (1993) studied 100 females (age 18-64, convenience sample). For the testing conditions four fully saturated colours (i.e. pure colours, containing no black or white to dilute the colour) were used: blue, green, yellow and red. The colour of the stimulus was operationalized as the background colour of a furniture store (with the walls, floor and ceiling of the store depicted in the treatment colour). The furniture itself was neutral in colour. The questionnaire used to measure perceptions of the store environment and merchandise

consisted of twenty semantic differential items (cfr. Bellizzi et al., 1983). The first thirteen of these items were taken from the "Judgements of Environmental Quality Scale" (Fisher, 1974), a scale designed to assess general perceptions of environmental quality. The remaining questions related to the subject's perceptions of various aspects of the retail store and merchandise.

To examine the existence of distinct components within consumer responses to the stimuli, the twenty semantic differential scales were subjected to principle components analysis using VARIMAX rotation, revealing two key factors:

- (1) a non-evaluative "activation" dimension (i.e. motivating, modern, colourful, stimulating, cheerful, lively and bright, with coefficient alpha = .85), explaining 23.1% of the total variance.
- (2) an "evaluation" dimension (i.e. positive, attractive, relaxed, comfortable, good and pleasant, with coefficient alpha = .91), explaining 22.2% of the total variance (confer Osgood, Suci and Tannenbaum's "evaluative aspect of meaning" (bad/good, tense/relaxed, unpleasant/pleasant) (1957).

Thus in consumers' ratings of the store environment two distinct dimensions were found: an activation and an evaluation dimension. These two dimensions have also been identified by Russell and Pratt (1980).

The hypothesis that consumer evaluations exhibit an increasingly positive linear trend from longer (warmer) to shorter (cooler) wavelengths was supported. Evaluations became steadily more positive as wavelength moved from red to blue.

Support was also found for the hypothesis that the activation dimension within consumers' response to colour exhibits a U-shaped pattern across wavelengths. Retail store environments in the more extreme wavelength colours (red and blue) were perceived as more active environments.

The results of the study also indicate that colour can affect perceptions of the merchandise within the store environment. Perceptions of merchandise quality did not differ across the colour treatments, although these perceptions were significantly correlated with the composite ratings on the activation dimension (r = -.41, p<.01) and the evaluation dimension (r = .38,

p<.01). A linear trend in the treatment means was found for perceptions of merchandise style (p<.05) and merchandise price (p<.10), with merchandise in the red environment being perceived as most up-to-date and higher in price. Merchandise in the green environment was perceived as relatively outdated. Regarding merchandise selection, a quadratic trend in the means was found (p=.08), with the blue environment being the most effective for enhancing these perceptions. Thus, while perceptions of merchandise quality were not significantly impacted by store colour, other aspects of merchandise perceptions (such as style and price perceptions), appeared to be partially driven by the colour of the store environment.

A limitation of this study is that only 4 colours were tested from the visible spectrum, whereas the hypothesis implies a continuous function across the visible spectrum.

Crowley (1993) suggested that the activation dimension may be more relevant to impulse buying situations. He suggests that an activated consumer may be more likely to engage in impulse buying and that for this type of situation the more activating colours such as red and blue are most appropriate, while moderate wavelength colours such as green should be avoided. Crowley (1993) also suggested that for some consumer behaviour contexts, such as waiting in line, a lower level of activation may be more desirable. Crowley (1993) suggested that the evaluative dimension may be more relevant in other contexts. For example to obtain more positive responses to constructs such as attitude towards an advertisement, shorter wavelength colours, such as blue are likely to be most effective. Crowley (1993) suggested that in retail environments such as financial institutions, upscale jewelry stores and exclusive restaurants creating an evaluatively pleasing atmosphere is more important than activating the consumer.

Crowley (1993) described the two-dimensional impact of colour as a function of colour wavelength. The results of his study support that evaluative effects are most positive at the short wavelength (blue) end of the visible spectrum, while the activation response engendered by colour exhibits a U-shaped pattern across wavelengths.

Crowley (1993) encourages further research to develop our understanding of consumer response to colour in a wider variety of contexts.

4.4.5. OVERVIEW OF EMPIRICAL EVIDENCE

A comprehensive overview of the few studies on the impact of colour in a retail context is presented in the following tables.

Actually, store colour has not been found to affect price-quality perceptions (Bellizzi et al., 1983), however, warm-coloured interiors (especially red) were thought to carry more up-todate merchandize than cool coloured interiors. Middlestadt (1990) also found no significant effects of ambient and background colour on attitudes or beliefs for perfume or mineral water. However, the attitude towards buying a pen did significantly differ when presented against a blue or a red background. Middlestadt (1990) found respondents exposed to a pen with a blue background and ambient colour to have a more positive attitude towards buying the product than those exposed to the pen with a red background and ambient colour. As opposed to respondents in the red condition, those in the blue condition believed that buying the pen was more likely to lead to positive outcomes (especially elegance and uniqueness) and less likely to lead to negative ones. Crowley's (1993) findings also indicate that colour can affect perceptions of the merchandize within the store environment, however, for merchandize quality no direct relationship with store colour was found. On the other hand, merchandize style and price perceptions did show an increasingly positive linear trend with increasing wavelengths. With regard to merchandize selection, a quadratic trend was found, with the blue environment being the most effective for enhancing these perceptions. Although Crowley (1993) could not discern a direct relationship between store colour and perceptions of merchandize quality, these perceptions appeared to be significantly correlated with an evaluation and activation dimension of judgements of environmental quality.

With regard to colour effects on actual approach behaviour, Bellizzi et al. (1983) found subjects to be physically drawn to warm colours, regardless of colour preference (Bellizzi et al., 1983). In contrast to cool colours, warm colours (red in particular) appeared to score high on an activity factor (Bellizzi et al., 1983). Confining their study to red and blue hues, Bellizzi and Hite (1992) however, did not observe a colour effect on the arousal dimension. Crowley (1993) eventually, found support for the hypothesis that the activation dimension within consumer's response to colour exhibits a U-shaped pattern across wavelengths. It has to be noted though, that in all of these studies fully saturated hues were used as stimuli, which were not controlled for brightness or saturation.

With concern to an evaluative factor, warm colours (red in particular) appear to score low, in contrast to the less activating cool colours, which appear to be more preferred (Bellizzi et al., 1983). Bellizzi and Hite (1992) also found that a blue store environment was found more pleasant than a red store environment. The hypothesis that consumer evaluations exhibit an increasingly positive linear trend from longer (warmer) to shorter (cooler) wavelengths was supported by Crowley (1993) as well.

In a first experiment, Bellizzi and Hite (1992) also found blue displays to produce higher purchase rates (lower postponement rates) than red displays. Moreover, shoppers in the blue store were found to spend on average significantly more than those in the red store and they were found to select the most expensive TV-set significantly more often. However, contrary to expectations they did not find the colour treatment to be significantly related to shopping time. In a second experiment, Bellizzi and Hite (1992) also found, as expected, that subjects in a blue environment expressed a greater intention to shop, browse and buy in a simulated store (i.e. they exhibited more approach behaviour). Moreover, in this follow-up study, they found purchase intentions to be linked to more pleasurable feelings, revealing an indirect effect of store colour on approach behaviour through colour-evoked feelings as well. Bellizzi and Hite (1992) observed only colour effects on the pleasure dimension, and not on the arousal or dominance dimensions. This could, however, be due to the fact that only two fully saturated hues (red and blue) were investigated.

From these initial studies it is clear that there are indeed apparent effects of store-colour on attitudes and beliefs, in-store emotions and shopping behaviour. Nevertheless, more research is needed to grasp these effects more clearly. It is obvious that the effects of more colours need to be investigated, taking into account not only hue effects, but also brightness and saturation effects, which have been ignored up until now, but appear to have a significant impact on the emotions elicited by colours (e.g. Valdez and Mehrabian, 1994; Gorn et al., 1997).

				DEPENDENT VARIABLES			
Empirical Study	Research	Sample	Colour	Internal Responses		Approach	Findings
Empirical Study	Design	Sample	Stimuli	Emotions	Cognitions	Behaviour	T indings
Bellizzi, Crowley and Hasty (1983) "The effects of colour in store design" Journal of Retailing	Laboratory experiment: <u>Experiment 1:</u> Manipulated wall colours in experimental area <u>Experiment 2:</u> Manipulated background colours of projected furniture store interior	Convenience sample 125 females age 18-64	Four full strength hues and white: <i>"Color Vu Corporation Standard Color Chart":</i> - Red (#1) - Yellow (#57) - Green (#161) - Blue (#129) - White	Experiment 2: Semantic differen et al. 1957) 20 items (13 items from the 'Jue environmental quality- 1974) 3 dimensions - evaluation - activation - price/quality	tial (cf. Osgood dgements of scale' – Fisher	Experiment 1: - Approach orientation (i.e. attention) - Physical Attraction	 Experiment 1: No relation between colour and attention/approach orientation Subjects appear to be physically drawn to warm colours (regardless of colour preference) Experiment 2: Warm colours (red in particular) appeared to score high on an activity factor, but low on an evaluative factor (i.e. tense and unpleasant) Cool colours (blue in particular) appeared to score low on an activity factor, but high on an evaluative factor (i.e. relaxed and pleasant) The colours used in the study did not affect price-quality perceptions, however warm coloured interiors (especially red) were thought to carry more up-to-date merchandize than cool coloured interiors.

Table 4-1: Empirical evidence provided by Bellizzi, Crowley and Hasty (1983)

Table 4-2: Empirical evidence provided by Middlestadt (1990)

				DEPENDENT VARIABLES			
Empirical Study	Research	Sample	Colour	Internal R	esponses	Approach	Findings
Empirical Study	Design	Sampie	Stimuli	Emotions	Cognitions	Behaviour	1 munigs
Middlestadt (1990) "The effect of background and ambient colour on product attitudes and beliefs" <i>Advances in</i> <i>Consumer</i> <i>Research</i>	Laboratory experiment: Manipulated ambient and background colours of product displays (perfume, pen and mineral water)	Convenience sample 84 female undergraduate students	- Red - Blue (not specified)		Underlying cognitive structure: beliefs (i.e. likelihood and evaluation of 8 salient consequences of buying the product) (cf. Fishbein and Ajzen, 1975)	Attitude towards buying the product (6-item semantic differential)	 No significant effects of colour on attitudes or beliefs for perfume and mineral water Respondents exposed to the pen with blue background and ambient colour had a more positive attitude towards buying the product than those exposed to the pen with the red background and ambient colour. As opposed to respondents in the red condition, those in the blue condition believed that buying the pen was more likely to lead to positive outcomes (especially elegance and uniqueness) and less likely to lead to negative ones.

				DEPENDENT VARIABLES			
Empirical Study	Research	Samula	Colour Stimuli	Internal Responses		Approach	Findings
Empirical Study	Design	Sample	Colour Sumun	Emotions	Cognitions	Behaviour	rindings
Bellizzi and Hite (1992) "Environmental colour, consumer feelings and purchase likelihood" <i>Psychology and</i> <i>Marketing</i>	Laboratory experiment: <u>Experiment 1</u> Manipulated background colours of television displays in a simulated store <u>Experiment 2</u> Manipulated background colours of projected furniture store interior (cfr. Bellizzi et al., 1983)	Experiment 1 Convenience sample 70 females age 18+ Experiment 2 Convenience sample 107 graduate marketing students	Two hues: brightness and saturation not controlled <u>Experiment 1</u> - Red (Canson Paper - red #505) - Blue (Canson Paper - blue #595) <u>Experiment 2</u> "Color Vu Corporation Standard Color Chart": - Red (#1) - Blue(#129)	Experiment 2 Emotions: PAD (Pleasure- Arousal- Dominance) emotion scale (Mehrabian and Russell, 1974; Donovan and Rossiter, 1982)		Experiment 1 Shopping behaviour - Purchase behaviour - Shopping time Experiment 2 Approach/avoidance purchase intentions 8 items (cfr Donovan and Rossiter, 1982)	 Experiment 1 The blue displays were found to produce higher purchase rates (lower postponement rates) than the red displays Shoppers in the blue store spent on average significantly more than those in the red store and they were found to select the most expensive TV-set significantly more often The colour treatment was not found to be significantly related to shopping time Experiment 2 As expected, subjects in the blue environment expressed a greater intention to shop, browse and buy in the simulated store (i.e. they exhibited more approach behaviour) Colour effects were observed on the pleasure dimension only, and not on the arousal or dominance dimensions The blue store environment was found to be more pleasurable than the red store environment

				DEPENDENT VARIABLES		LES	
Empirical Study	Research	Sampla	Colour	Internal Responses		onses Approach	Findings
Empirical Study Design	Design	Sample	Stimuli	Emotions	Cognitions	Behaviour	rmunigs
Crowley (1993) "The two- dimensional impact of colour on shopping" <i>Marketing Letters</i>	Manipulated background colours of projected furniture store interior	Convenience sample 100 females age 18-64	Four fully saturated hues: - Blue - Green - Yellow - Red Not specified, brightness and saturation not controlled	Semantic differential (cf. Osgood et al. 1957) 20 items (13 items from the 'Judgements of environmental quality-scale' – Fisher 1974) 2 dimensions - evaluation - activation (cf. Russel and Pratt, 1980)	Perceptions of merchandize - quality - style - price - selection		 The hypothesis that consumer evaluations exhibit an increasingly positive linear trend from longer (warmer) to shorter (cooler) wavelengths was supported. Support was also found for the hypothesis that the activation dimension within consumer's response to colour exhibits a U-shaped pattern across wavelengths. The results of the study also indicate that colour can affect perceptions of the merchandize within the store environment. (In particular merchandize style and price perceptions show an increasingly positive linear trend with increasing wavelength. For merchandize quality no direct relationship with store colour was found, however merchandize quality perceptions were significantly correlated with the evaluation and activation dimensions. With regard to merchandize selection, a quadratic trend was found, with the blue environment being the most effective for enhancing these perceptions)

4.5. THE INFLUENCE OF COLOUR IN THE STORE ENVIRONMENT: RESEARCH MODEL

To investigate the influence of colour in the store environment, an environmental psychology approach will be applied. As the first level of response to an environment is generally accepted to be affective (Ittelson, 1973, p16), we will, more specifically, apply the classic 'emotional' approach of environmental psychology, which is based on the fundamental premise that emotions elicited by the environment determine human behaviour (Mehrabian, 1978, p14).

Mehrabian and Russell (1974) explored such an emotional theory of environmental psychology. According to this theory, an environment evokes primary emotional reactions, which as intervening variables, determine approach or avoidance reactions to that environment. Our research will be based upon this particular Stimulus-Organism-Response model, which was first applied to the study of store atmosphere by Donovan and Rossiter (1982), and has been adopted by many other atmospheric researchers since (e.g. Donovan et al., 1994; Van Kenhove and Desrumaux, 1997; Sherman, Mathur and Smith, 1997; Tai and Fung, 1997; Gröppel-Klein, 1998; McGoldrick and Pieros, 1998; Matilla and Wirtz, 2001; Sweeney and Wyber, 2002; ...). Based on this model, various atmospheric variables have been found to influence a wide variety of consumer evaluations and behaviours (see Turley and Milliman, 2000 for a review).



With regard to the environmental stimulus (S), the specific effects of colour hue, saturation and brightness in the store design will be investigated. In particular the latter two of these colour dimensions have been totally ignored in previous studies on the effects of colour in a retail context. Worse, they have not even been controlled for, thus confounding the effects of the disparate colour attributes. As far as the internal responses in the organism (O) are concerned, we will focus on the feelings of pleasure, arousal and dominance, evoked by the store colour (e.g. Donovan and Rossiter, 1982). The responses (R) we are interested in, involve approach and/or avoidance intentions with regard to the store.

Thus, applying this model (illustrated in figure 4-1), the proposed study aims to assess (1) the direct effect of colour in the store environment on the emotions experienced in the store (notably pleasure, arousal and dominance) and (2) the direct and indirect effect of colour on consumers' appoach/avoidance behaviour through the store-colour evoked emotions.

The specific hypotheses, generated from the literature review, will be presented in the following paragraph.

4.6. THE INFLUENCE OF COLOUR IN THE STORE ENVIRONMENT: HYPOTHESES

4.6.1. STORE COLOUR-EVOKED PLEASURE

With regard to the effect of store colour on feelings of pleasure, three hypotheses are abstracted from our review of the literature on colour effects, one in relation to each of the three colour attributes: hue, brightness and saturation. Whereas the hypothesis concerning the effect of colour-hue could be based upon the initial empirical findings in a retail context, for brightness and saturation effects we had to rely on the available research based on laboratory experiments with colour patches, because these colour-attributes have been ignored in studies on the impact of colour in the store environment.

4.6.1.1. Store Colour Hue

H1: A negative U-shaped relationship is hypothesized between store-colour wavelength and store-elicited pleasure, with the stores with extreme long wavelength colours (red) expected to be judged less pleasant than the stores with extreme short wavelength colours (blue) and the mid-wavelength colours (yellow-green) at the bottom of the U, being judged least pleasant.

Although it is obvious that a lot of the early studies on the effects of hue on pleasure are methodologically flawed -confounding brightness and saturation effects with the effects of colour hue- there appears to be some consistency in the findings. Extreme short wavelength colours (blue, green) seem to be generally preferred, followed by the extreme long wavelength colour red. Yellow-green, yellow and orange appear to be generally disliked (Walton, Guilford and Guilford, 1933; Guilford, 1934; Eysenck, 1941; Wexner, 1954; Granger, 1955; Guilford and Smith, 1959; Helson and Lansford, 1970; Hopson, Cogan and Batson, 1971; Adams and Osgood, 1973; Sivik, 1974a, 1975; Mehrabian and Russell, 1974;

Jacobs and Suess, 1975; Madden et al., 2000). Thus, pleasure-displeasure reactions to spectral colours appear to approximate a U-shaped function of wavelength, with yellows (green-yellow, yellow and red-yellow) at the bottom portion of the U (Valdez and Mehrabian, 1994).

Nevertheless, Osgood et al. (1957), suggest that an acceptable colour is defined by the object with which it is associated. Studying selected colours in the context of objects, they found yellow to be rated as the "most favourable" colour, which is in contrast with the general finding from using colour patches, that yellow is mostly disliked.

In a store context, a simple negative relationship has been revealed between colour wavelength and colour preference. Warm colours (red in particular) have been found to score low on an evaluative factor, in contrast to the less activating cool colours, which appear to be more preferred (Bellizzi et al., 1983). Bellizzi and Hite (1992) found that a blue store environment was found more pleasant than a red store environment and the hypothesis that consumer evaluations exhibit an increasingly positive linear trend from longer (warmer) to shorter (cooler) wavelengths was supported by Crowley (1993) as well. Nevertheless, in these studies only fully saturated hues have been examined, without controlling for brightness or saturation levels. Moreover, the studies have been confined to examining the effects of two to four hues at the most. Therefore, these findings may not be very reliable, confounding hue effects with saturation and brightness effects.

Based on previous research, by Valdez and Mehrabian (1994) in particular, we propose a negative U-shaped relationship between store-colour wavelength and store-elicited pleasure, with the stores with extreme long wavelength colours (red) expected to be judged less pleasant than the stores with extreme short wavelength colours (blue) and the mid-wavelength colours (yellow-green) at the bottom of the U, being judged least pleasant.

4.6.1.2. Store Colour Brightness

H2: Store colour value is expected to have a positive effect on store-elicited pleasure, with higher value coloured or brighter stores expected to be judged more pleasant than the lower value coloured or darker stores.

There appears to be a general consensus in the literature that lighter colours seem to be preferred over darker ones (Washburn, 1911; Guilford, 1934; Guilford and Smith, 1959, Adams and Osgood, 1973; Sivik, 1974a; 1975; Valdez and Mehrabian, 1994). Guilford and Smith (1959) and Sivik (1974a, 1975) demonstrated the complexity of the relationship between the three dimensions of colour as they relate to pleasantness, but they also noted that as brightness increased, perceived pleasantness also increased, which appeared to be a consistent finding, across all hues. In contrast to Guilford and Smith (1959), who found the positive relationship tending to be curvilinear, Valdez and Mehrabian (1994) found no second-order curvilinear relationship between colour brightness and the elicited pleasure. Examining the contribution of colour brightness to perceived pleasantness in greater detail, Smets (1982) found brightness, as compared to saturation, to account for only a small amount of induced pleasure variance, whereas Valdez and Mehrabian (1994) found the effect of brightness to be considerably stronger than the effect of saturation in determining pleasure responses to colour. In contrast to these findings on general colour effects, Gorn et al. (1997) did not find support for the hypothesis that lower value, darker colours generate more unpleasant feelings when manipulating the colour of a big "swoosh" in the centre of an ad (confining their study to red and blue hues).

Nevertheless, based on our review of the literature on general colour effects, we hypothesize store colour value to have a positive effect on store-elicited pleasure, with higher value coloured or brighter stores expected to be judged more pleasant than the lower value coloured or darker stores.

4.6.1.3. Store Colour Saturation

H3: Store colour saturation is expected to have a positive effect on store-elicited pleasure, with more saturated coloured stores expected to be judged more pleasant than the less saturated coloured stores.

With regard to the contribution of colour saturation to evoked pleasure, the literature is rather straight forward : saturated colours have systematically been found to be preferred over unsaturated ones (Washburn, 1911; Guilford, 1934; Granger, 1955; Guilford and Smith, 1959; Valdez and Mehrabian, 1994). Granger (1955) noted that as saturation increased, so did preference, but only up to a certain point. He found colours seen as too vivid to be less preferred. Guilford and Smith (1959) also found that the greater the saturation within a given hue, the greater its perceived pleasantness appeared to be, with the relationship tending to be curvilinear. In contrast, Valdez and Mehrabian (1994) did not find support for a second-order curvilinear relationship between colour saturation and the elicited pleasure. Examining the contribution of colour saturation to perceived pleasantness in greater detail, Smets (1982) found saturation, as compared to brightness, to account for most of the induced pleasure variance. In contrast, Valdez and Mehrabian (1994) found the effect of saturation to be considerably weaker than the effect of brightness in determining pleasure responses to colour. Conform to these findings on general colour effects, Gorn et al. (1997) found that respondents exposed to ads containing higher chroma colours reported a greater liking for the ad, and that this effect was mediated by greater feelings of excitement experienced, which besides high arousal also involves pleasant feelings.

Thus, based on our review of the literature, we hypothesize store colour saturation to have a positive effect on store-elicited pleasure, with more saturated coloured stores expected to be judged more pleasant than the less saturated coloured stores.

4.6.2. STORE COLOUR-EVOKED AROUSAL

It has been widely accepted that people's feelings are affected by colour. In particular, colour is believed to affect the degree of felt arousal (e.g., Walters et al., 1982; Mikellides, 1990). From the results of early studies, various authors cite support for the premise that certain colours have the capacity to excite and arouse an individual, while others have a calming and relaxing effect. In an attempt to better understand the effect of colour on human behaviour, prior research has examined both arousal perception (i.e., how active a stimulus is perceived to be), as well as the actual physiological arousal engendered by exposure to various colours. In the consumer behaviour literature, colour arousal effects have only recently been addressed (Crowley, 1993; Gorn, 1997). With regard to the effect of store colour on arousal, three hypotheses are abstracted from our review of the literature on colour effects, one in relation to each of the three colour attributes: hue, brightness and saturation. Whereas the hypothesis concerning the effect of colour-hue could be based upon the initial empirical findings in a retail context, for brightness and saturation effects we had to rely on the available research based on laboratory experiments with colour patches, because these colour-attributes have been ignored in studies on the impact of colour in the store environment.

4.6.2.1. Store Colour Hue

H4: A positive U-shaped relationship is hypothesized between store-colour wavelength and store-induced arousal with the stores with extreme long wavelength colours (red) expected to be judged more arousing than the stores with extreme short wavelength colours (blue) and the mid-wavelength colours (green-yellow) at the bottom of the U, being judged least arousing.

With regard to the effect of colour hue on saturation, a great deal of inconsistency exists in the literature, arising from incomplete operational definitions of the stimulus conditions and the multifaceted nature of response measures defined as arousal (Kaiser, 1984; Beach et al., 1988; Valdez and Mehrabian, 1994).

The hypothesis that long-wavelength colours (e.g., red and yellow) are more arousing than short wavelength colours (e.g., blue and green) (Goldstein, 1942; Gerard, 1958, 1959; Ali, 1972) has been very persistent in the literature, despite the fact that numerous studies involving physiological reactions to arousal could not confirm this contention (Erwin, Lerner, Wilson and Wilson, 1961; Jacobs and Hustmyer, 1974; Caldwell and Jones, 1985; Kuller, 1985; Etnier and Hardy, 1997). Studies using semantic differential scales to capture "perceptions of arousal", have more systematically found excitement and activity to be related to the hue dimension, with red and yellow generally seen as active, stimulating and exciting, while green, violet and blue are perceived to be passive, relaxing, soothing and calm (Wexner, 1954; Osgood et al., 1957 Wright and Rainwater, 1962; Aaronson, 1970; Adams and Osgood, 1973; Sharpe, 1974; Jacobs and Suess, 1975; Porter and Mikellides, 1976a; Birren, 1978; Walters et al., 1982; Madden et al., 2000).

Following a Darwinian logic, on the other hand, Wilson (1966) posited a U-shaped relationship between colour wavelength and arousal effects, predicting that the colours with more extreme wavelengths are the most activating, a contention that also has received some support (Erwin, Lerner, Wilson and Wilson, 1961; Wilson, 1966; Nourse and Welch, 1971).

However, contradicting the long-hold stereotype that warm colours are exciting, while cool colours are calming, Sivik (1974a, 1975) concluded on the basis of his comprehensive study, that no difference is exhibited between hues with equal chromaticness on the excitement factor. Sivik hypothesizes that the reason why red and yellow have generally been labelled as exciting, might be attributed to the fact that pure, saturated reds and yellows are more prevalent than pure and highly saturated greens and blues. Likewise, Valdez and Mehrabian (1994, Study 2) could not support the hypothesis that long-wavelength hues are more arousing than short wavelength hues either. Conducting a very systematic and comprehensive study, they found the mean arousal level for green-yellow to be significantly greater than the mean arousal level for purple-blue, yellow-red and red-purple and the mean arousal level for blue-green to be significantly greater than the mean arousal level for purple-blue. A regression analysis to test for a possible significant parabolic inverted-U shaped relationship of arousal to wavelength (note that this is the opposite of Wilson's 1966 contention), revealed no significant effect either. The obtained results, relating hue and arousal, were generally weak and non-significant. The only noteworthy generalization is that, unexpectedly, the green hues
(green-yellow, blue-green, and green) seem to elicit the highest arousal reactions from subjects. Gorn et al. (1997) conclude from their study with regard to the effects of colour in advertising, that when used as an executional cue in an ad, the hue (red vs. blue) of a colour at best has modest effects on feelings of arousal.

Thus, studies using both approaches (physiological responses and perceptions of arousal) have found that certain colours, are indeed more activating than other colours (e.g. Nakshian, 1964; Clynes and Kohn, 1968; Sallis and Buckalew, 1984; Jacobs and Nordan, 1979, see Crowley, 1993). However there is a lot of inconsistency in the findings as to what colours are more and less arousing and whether there is indeed a relationship between colour wavelength and its arousal eliciting quality.

In a retail context, Bellizzi et al. (1983) found that warm colours (red in particular) appeared to score high on an activity factor, in contrast to cool colors. Confining their study to red and blue hues, Bellizzi and Hite (1992) however, did not observe a colour effect on the arousal dimension. Crowley (1993) eventually, found support for the hypothesis that the activation dimension within consumer's response to colour exhibits a U-shaped pattern across wavelengths. It has to be noted though, that in all of these studies fully saturated hues were used as stimuli, which were not controlled for brightness or saturation.

Based on this review of the literature, it is not entirely clear what can be expected with regard to the effect of colour hue on arousal. In accordance with Crowley (1993) (see also Erwin, Lerner, Wilson and Wilson, 1961; Wilson, 1966; Nourse and Welch, 1971), a positive U-shaped relationship is hypothesized between store-colour wavelength and store-induced arousal with the stores with extreme long wavelength colours (red) expected to be judged more arousing than the stores with extreme short wavelength colours (blue) and the mid-wavelength colours (yellow-green) at the bottom of the U, being judged least arousing.

4.6.2.2. Store Colour Brightness

H5: Store colour value is expected to have a negative effect on store-induced arousal, with higher value coloured or brighter stores expected to be judged less arousing than the lower value coloured or darker stores.

The effect of colour brightness on arousal was originally unclear. None of the studies dealing with physiological reactions to colour have investigated these reactions in relation to colour brightness levels. Earlier studies, based on semantic differential ratings, seemed to indicate that arousal is a positive correlate of colour brightness (Wright and Rainwater, 1962; Profusek and Rainey, 1987). More recent studies, however, found exactly the opposite to be true (Valdez and Mehrabian, 1994 - Study 1; Gorn et al., 1997). Indeed, conducting a very comprehensive and systematic study, Valdez and Mehrabian (1994) revealed a negative relationship between colour brightness and induced arousal, indicating that less bright (i.e. darker) colours induce greater feelings of arousal in viewers. From this study, the effect of brightness appears to be considerably weaker than the effect of saturation in determining arousal responses to colour. Furthermore, Valdez and Mehrabian (1994) also found a secondorder curvilinear relationship between colour brightness and the elicited arousal. Arousal declined steeply and monotonically with increasing brightness up to a brightness value of 43, beyond which arousal reversed and increased slightly for the highest brightness values. In other words, arousal appears to decrease as colours range from dark to light, but this trend is slightly reversed for the lightest colours. In agreement with the results obtained by Valdez and Mehrabian (1994), Gorn et al. (1997) also found support for the hypothesis that lighter, higher value colours are more relaxing than lower value, darker colours. Using a two-dimensional view on arousal, they found that those exposed to ads containing higher value colours (for red and blue hues) reported experiencing greater feelings of relaxation, but no difference in the level of felt excitement.

Based on our review of the literature and more in particular in accordance to the more recent findings by Valdez and Mehrabian (1994) and Gorn et al. (1997), we hypothesize store colour value to have a negative effect on store-induced arousal, with higher value coloured or

brighter stores expected to be judged less arousing than the lower value coloured or darker stores.

4.6.2.3. Store Colour Saturation

H6: Store colour saturation is expected to have a positive effect on store-induced arousal, with more saturated coloured stores expected to be judged more arousing than the less saturated coloured stores.

Although none of the studies dealing with physiological reactions to colour have investigated these reactions in relation to colour saturation levels, studies based on semantic differential ratings generally suggest that arousal is a positive correlate of colour saturation, with more saturated (more vivid and purer) colours evoking greater feelings of arousal (Wright and Rainwater, 1962; Hogg, 1969; Adams and Osgood, 1973; Sivik, 1974a, 1975; Valdez and Mehrabian, 1994). Valdez and Mehrabian (1994, Study 1) found a colour's saturation level to explain a substantial portion of the variance in the feelings of arousal it induces, in viewers and noticed the effect of saturation to be considerably stronger than the effect of brightness in determining arousal responses to colour (almost twice the magnitude). Consistent with these findings, Gorn et al. (1997) also found support for the hypothesis that ads with higher chroma colours induce greater feelings of excitement.

Based on a thorough review of the literature with regard to colour saturation effects, we hypothesize this contention to hold also in a store environment, with store colour saturation to have a positive effect on store-induced arousal, with more saturated coloured stores expected to be judged more arousing than the less saturated coloured stores.

4.6.3. STORE COLOUR-EVOKED DOMINANCE

With regard to the effect of store colour on feelings of dominance, three hypotheses are abstracted from our review of the literature on colour effects, one in relation to each of the three colour attributes: hue, brightness and saturation. We had to rely on the available research based on laboratory experiments with colour patches, because dominance effects have been largely ignored in studies on the impact of colour in the store environment. Dominance is a dimension of emotional response that has often been ignored by colour researchers as well. Nevertheless, colour researchers have found colour to elicit emotional responses that cannot be captured by pleasure or arousal dimensions, but seem to infer a dimension of dominance or potency (e.g. Osgood et al., 1957; Adams and Osgood, 1973; Sivik, 1974a; Valdez and Mehrabian, 1994), which is in line with the Pleasure-Arousal-Dominance Model of Emotions proposed by Mehrabian and Russell (1974) (see also Mehrabian, 1995; 1996).

4.6.3.1. Store Colour Hue

H7: An inverted U-shaped relationship is hypothesized between store-colour wavelength and store-elicited dominance, with the stores with mid-wavelength colours (Yellow-Green) expected to elicit most dominance and extreme short or long wavelength coloured stores eliciting less dominance.

Previous studies with regard to the effect of colour-hue on the dominance induced in viewers appear to demonstrate that empirical findings are inconsistent. Not using actual colour stimuli but colour names, Adams and Osgood (1973) found yellow to be rated as weak. Birren (1978) and Sharpe (1974) found green to be associated with withdrawal. In contrast they both found red to be rated as strong and associated with aggression (Adams and Osgood, 1973; Birren, 1978 and Sharpe, 1974). Although the results concerning the impact of colour-hue on dominance, obtained by Valdez and Mehrabian (1994), were generally weak and non-significant, they suggest the exact opposite. According to their findings, middle-wavelength

hues appear to elicit more feelings of dominance in viewers than either the extreme short or long wavelength hues. A plausible explanation for this inconsistency in the findings may be the failure of the earlier colour-studies to provide adequate specifications or controls of colour stimuli (i.e., absence of controls for saturation and brightness, while investigating effects of hue). Another possible explanation for these contrasting findings could also be that colours that are rated as weak elicit feelings of dominance in viewers and vice versa (colours that are rated as strong might elicit feelings of submissiveness). However, further research is needed to find out if this actually is the case.

In a store-context, Bellizzi and Hite (1992) observed no colour effects on the dominance dimension. This could, however, be due to the fact that only two fully saturated hues (red and blue) were investigated.

Nevertheless, based on our review of the literature on general colour effects, and on the research by Valdez and Mehrabian (1994) in particular, we hypothesize an inverted U-shaped relationship between store-colour wavelength and store-elicited dominance, with the stores with mid-wavelength colours (Yellow-Green) expected to elicit most dominance and extreme short or long wavelength coloured stores eliciting less dominance.

4.6.3.2. Store Colour Brightness

H8: Store colour value is expected to have a negative effect on store-induced feelings of dominance, with higher value coloured or brighter stores expected to be judged less dominant than the lower value coloured or darker stores.

On the effects of colour brightness on feelings of dominance, there is general consensus in the literature: dominance appears to be a negative correlate of brightness (Adams and Osgood, 1973; Sivik, 1974a, 1975; Damhorst and Read, 1986; Valdez and Mehrabian, 1994). Indeed, less bright (i.e. darker) colours have been found to induce greater feelings of dominance.

Nevertheless, Sivik (1974a, 1975) notes that the extent of its effect seems to vary across hues, revealing a complex interaction among the colour attributes. Valdez and Mehrabian (1994) found the effect of brightness to be considerably stronger than the effect of saturation in determining dominance responses to colour. Furthermore, Valdez and Mehrabian (1994) found a second-order curvilinear relationship between colour brightness and the elicited dominance. Dominance appeared to decline steeply and monotonically with increasing brightness up to a brightness value of 43, beyond which dominance levels off. In other words, dominance appears to decrease as colours range from dark to light, but levels off for the lightest colours.

Based on our review of the available literature, we hypothesize store colour value to have a negative effect on store-induced feelings of dominance, with higher value coloured or brighter stores expected to be judged less dominant than the lower value coloured or darker stores.

4.6.3.3. Store Colour Saturation

H9: Store colour saturation is expected to have a positive effect on store-elicited dominance, with more saturated coloured stores expected to be judged more dominant than the less saturated coloured stores.

Concerning the effect of colour saturation on dominance, empirical findings reveal consistent results. Potency was found to be directly dependent upon a colour's saturation (Osgood et al., 1957; Hogg, 1969). Although Sivik (1974a, 1975) also found that as chromaticness increased, so did perceived potency, he found this not to be true for yellow, revealing a complex interaction between the colour attributes for the potency dimension as well. Valdez and Mehrabian (1994) also found support for the hypothesis that dominance is a positive correlate of saturation. Apparently, more saturated (more vivid and purer) colours appear to elicit greater feelings of dominance, strength or boldness in viewers. Although they found that a colour's brightness and saturation levels appear to explain a substantial portion of the

variance in the feelings of dominance it induces, a regression analyses indicates the effect of saturation to be considerably weaker than the effect of brightness.

Based on our review of the available literature, we hypothesize store colour saturation to have a positive effect on store-elicited dominance, with more saturated coloured stores expected to be judged more dominant than the less saturated coloured stores.

4.6.4. THE IMPACT OF STORE-COLOUR-EVOKED EMOTIONS ON APPROACH/AVOIDANCE BEHAVIOURS

- H10: Approach intentions towards a store are positively related with the pleasure elicited by the store.
- H11: Avoidance intentions towards a store are negatively related with the pleasure elicited by the store.

In line with previous findings in a retailing context (Donovan and Rossiter, 1982; Baker, Levy and Grewal, 1992; Sherman et al., 1997; Obermiller and Bitner, 1984; Donovan et al., 1994; Hui and Bateson, 1991; Chebat et al., 1995; Van Kenhove and Desrumaux, 1997), we hypothesize approach intentions towards the store to be positively related with the pleasure elicited by the store. With regard to avoidance, we expect the relationship to be reversed (cfr. Foxall, 1997).

- H12: Approach intentions towards a store are positively related with the arousal induced by the store (for pleasant stores).
- H13: Avoidance intentions towards a store are not (or negatively) related with the arousal induced by the store.

In line with Mehrabian and Russell (1974) and Russell and Mehrabian (1976), we expect approach intentions towards a store to be positively related with the arousal induced by the store (for pleasant stores). Support for this proposition in a retailing context has been found in earlier studies (Donovan and Rossiter, 1982; Holbrook and Gardner, 1993; Sherman et al., 1997; Van Kenhove and Desrumaux, 1997). According to the presumed pleasure-arousal interaction effect, we should expect the inverse in the case of unpleasant stores. However, few support for this part of the relationship has been found (Holbrook and Gardner, 1993; Donovan et al., 1994)

With regard to the effect of store-evoked arousal on avoidance behaviour, we expect the inverse, however, considering Foxall's (1997) finding that approach and avoidance are not simply opposites and that arousal exerts an independent influence on both, we could argue to find no relationship between arousal and avoidance, as he did.

- H14: Approach intentions towards a store are positively related with the dominance elicited by the store.
- H15: Avoidance intentions towards a store are not (or negatively) related with the dominance elicited by the store.

Although Donovan and Rossiter (1982), Yalch and Spangenberg (1988) and Greenland and McGoldrick (1994) could not find an important role for dominance in shaping in-store behaviour, Foxall (1997), Foxall and Greenley (1999) and Turley and Milliman (2000) argue that the findings with regard to dominance may be context specific and depending on the independent (store) environmental variables investigated. Babin and Darden (1995) suggest that the lack of findings associated with dominance in previous studies involving retail store environments may also be due to the moderating impact of personality. Based on their findings, Babin and Darden (1995, p61) emphasize that "*rather that dismissing dominance, as some studies have done, future theory development should include dominance as an important emotion influencing shopping behaviour*". Also Gröppel-Klein (1998) points to the possible importance of the dominance factor.

As feelings of dominance have been demonstrated to be evoked by colour (Valdez and Mehrabian, 1994; Osgood, et al., 1957; Adams and Osgood, 1973; Sivik, 1974), we assume that store-colour-elicited dominance can potentially influence approach/avoidance behaviours towards the store. As originally suggested by Mehrabian and Russell (1974) and Proshansky, Ittelson and Rivlin (1974), we hypothesize approach intentions towards a store to be positively related with the dominance elicited by the store.

With regard to the effect of store-evoked feelings of dominance on avoidance behaviour, we expect the inverse, however, considering Foxall's (1997) finding that approach and avoidance are not simply opposites and that dominance exerts an independent influence on both, we could argue to find no relationship between dominance and avoidance, as he did.

4.7. THE MODERATING ROLE OF DEMOGRAPHIC CHARACTERISTICS

It has been generally acknowledged that the physical environment interacts with characteristics of individuals to determine their response (Turley and Milliman, 2000). Therefore, an atmosphere that produces a certain response in one individual or group at a given point in time may produce an entirely different response in another individual or group (Bitner, 1992; Turley and Milliman, 2000). Indeed, different people can have different emotional reactions to the same event or circumstance (Bagozzi, Gopinath and Nyer, 1999). Bitner (1992, p64) included personal characteristics in her conceptual 'servicescape'¹⁹ framework as 'response moderators', which may alter the strength and direction of relationships among environmental response variables. Nevertheless, relatively little attention has been given to the effect of personal characteristics on the relationships between storeevoked emotions and their outcomes (Dawson et al., 1990; Babin and Darden, 1995). Although personal characteristics may be important moderators of colour effects as well, up until now, most studies have examined colour effects also only at the aggregate level (Crowley, 1993). Furthermore, the available empirical findings in this regard remain inconclusive. While some studies have proposed a universal order of colour preferences (Eysenck, 1941; Guilford, 1934; Child et al., 1968), others have suggested that preferences are dependent on such variables as gender and age (e.g., Dorcus, 1926; Guilford and Smith, 1959; Granger, 1955; Jastrow, 1897; Warner, 1949; Meerum Terwocht and Hoeksma, 1995).

In order to extend our research with regard to the impact of colour in the store environment, the moderating effects of individual differences on the hypothesized relationships will also be investigated. More specifically, the moderating impact of two demographic characteristics, age and gender, will be assessed. We intend to investigate whether these demographic characteristics affect (1) the emotions elicited by store-colours and (2) the effects of store-colour induced emotions on consumers' approach-avoidance intentions.

¹⁹ The 'servicescape' framework, proposed by Bitner (1992) offers an explanation of behavioural responses to a retail environment

4.7.1. AGE

Colour preferences have been shown to be age dependent (Norman and Scott, 1952; Birren, 1978; Meerum Terwocht and Hoeksma, 1995). According to Norman and Scott (1952), several studies report a preference for yellow and red in young children, and a preference for blue and green in children in later years of elementary school (Ellis, 1900; 1906; Gale, 1933; Staples and Walton, 1933). The preference for yellow has been found to diminish rather early and is very often found in the least preferred position for older children and adults, whereas the preference of blue appears to build up gradually through the grades, to become the most preferred colour of all (Norman and Scott, 1952). Investigating the preferences for six colours (the primary colours red, blue, yellow and green + black and white - not further specified) among three age groups (72 respondents), Meerum Terwocht and Hoeksma (1995) also found the preference for yellow to decrease with age, whereas the preference for green increased with age. Choungourian (1969) observed a similar trend. The results of a study among children by Boyatzis and Varghese (1994) also suggest an age-related trend in emotional responses to colour: in their study older children tend to show an increasing affinity with bright colours (in this study bright colours referred to the colours pink, red, yellow, green, purple and blue, all taken together for the analyses, whereas black, grey and brown denoted dark colours). Silver and Ferrante (1995) could reveal differences in colour preferences between undergraduate college students and older adults aged 60 to 83 as well.

Moschis (1987) indicates that with increasing age, consumers are less sensitive to external stimuli such as light and colours. Their information-processing abilities appear to decline in general. Moreover, studies with a developmental perspective suggest that there are some age-related changes in emotion preferences (Russell and Bullock, 1985; Russell and Ridgeway, 1983; Meerum Terwocht and Hoeksma, 1995).

It is evident that an atmosphere that produces a positive response in teenagers may produce a negative response in older shoppers. d'Astous (2000), for instance, found the degree of perceived irritation in the shopping environment to be significantly affected by shoppers' age. Playing loud music in the store caused more irritation among older shoppers, while the store being unclean or having a bad smell caused more irritation among younger shoppers. Age therefore is a factor that should not be ignored when studying the impact of store colour on consumers. Nevertheless, because previous research on the effects of colour in a store

environment has not included this demographic variable, it is not clear how exactly it will affect the proposed relationships. For this reason no specific hypotheses will be proposed. We will assess its moderating impact in our analyses.

4.7.2. **Gender**

Although the findings are very ambiguous, many investigations have indicated that there are significant differences between gender in preference for colour (Khouw, 1999a,b). Jastrow (1897) found men to prefer blue to red and women to prefer red to blue. Dorcus (1926) found yellow to have a higher affective value for men than for women. St George (1938) claimed that blue stands out for men far more than for women. Marston (1927) argues that blue is the colour of dominance, preferred primarily by males, while yellow represents submission and is supposedly preferred by females. Eysenck (1941) found no gender difference in colour preferences, except for the colours yellow and orange. He found yellow to be preferred to orange by women and orange to yellow by men, a finding that was confirmed later by Birren (1952) who also found men to prefer orange to yellow, while women tended to rank orange the lowest. McInnis and Shearer (1964) found blue-green to be more preferred among women than men; according to their findings, 56% of men and 76% of women favoured cool colours. Helson and Lansford (1970), on the other hand, reported men to find cool colours most pleasant, while women found warm colours more pleasant. In a study among children, Boyatzis and Varghese (1994) found that when asked to name their favourite colour, boys (26%) cited blue, followed by red, and girls (50%) preferred pink, followed by purple. They found boys more likely than girls to have positive emotional reactions to dark colours (in this study dark colours referred to the colours black, grey and brown, which are actually low in saturation as compared to the colours referred to as bright in the study). Guilford and Smith (1959) found men generally more tolerant towards achromatic colours than women. Analysing the colour-emotion associations among 40 undergraduate students, Hemphill (1996) found women to respond more positively than men to bright colours and to respond more negatively than men to dark colours (in this study bright colours referred to the colours pink, red, yellow, green, purple and blue, wereas black, grey and brown denoted dark colours). Whereas blue was cited as the favourite colour by an equal number of men and women, for other colours some gender differences in colour preferences could be discerned. For women other favourite colours were orange, yellow, purple, green and red. Other favourite colours for men were black, brown, green and red. However, responses to individual colours were very similar for the men and women. In a study among elderly, Silver and Ferrante (1995) found women to prefer black and purple more often than men.

With regard to saturation level, Plater (1967) found that men tended to prefer more saturated colours than women. Likewise, Radeloff (1990) found women to prefer soft, unsaturated colours, whereas men preferred bright, saturated ones. McInnis and Shearer (1964) also found women to prefer lighter colours more than darker ones. Nevertheless, Radeloff (1990) found no significant differences between men and women in the preference for light versus dark colours.

In an exploratory study with regard to colour in interior environments, Khouw (1999a,b) found reinforcing evidence that colour responses are influenced by gender differences. Men were found to rate interiors with highly saturated colours more favorable than women. In contrast, investigating environmental effects on workers' moods, Kwallek et al. (1996) found women to associate depression, confusion and anger with working in low-saturated coloured offices (white, grey and beige), whereas men associated depression, confusion and anger with working in high-saturated coloured offices (green, blue, purple, red, yellow and orange). Despite these psychological responses, both sexes were found to prefer to work in white offices.

Other reviews on sex differences in colour preferences have suggested general similarities in male and female colour preferences, while noting sex differences in the strenghts of those preferences (Norman and Scott, 1952; Whitfield and Wiltshire, 1990). Norman and Scott (1952, p189) note that: "while the *order* of preference is apparently not a function of sex, there is a good reason to suppose a difference in *strength* of the preferences". Valdez and Mehrabian (1994) also found men and women to react with highly similar emotional responses to brightness and saturation levels of colour samples, with results for women being slightly stronger. (e.g. colour saturation appeared to relate significantly to pleasure for women, but not for men). These results suggest that women are slightly more sensitive as compared to men, in terms of their emotional reactions to brightness and saturation levels of colours. As far as hue is concerned, however, Valdez and Mehrabian (1994) found no significant difference in the emotional reactions between men and women.

From his comparison of various studies, Guilford (1934) did observe some sex differences with regard to the affective reactions to colour: the three dimensions of colour accounted for 71% of the affective value judgements of women, but only for 26% of men's affective value judgements. Moreover, Radeloff (1990) found women to be more likely to have a favourite colour than men.

In general, the term "*emotional*" is also considered more typical of women than of men (Antill, 1987; Shields, 1987; Spence et al., 1975). In general women are thought to display more emotion than men and to feel emotion more intensely (Grossman and Wood, 1993; Johnson and Shullman, 1988; Shields and Koster, 1989). Women also appear to report emotions more intensely (Diener, Sandvik and Larsen, 1985; Fujita, Diener and Sandvik, 1991, as reported in Robinson, Johnson and Shields, 2001, p158). Women have also been found to process information in more detail as compared to men, making them more sensitive to environmental factors (Meyers-Levy and Maheswaran, 1991; Meyers-Levy and Sternthal, 1991).

Shopping has been demonstrated to be a gendered activity (Dholakia, 1999). More specifically, d'Astous (2000) found the degree of perceived irritation in the shopping environment to be significantly affected by shoppers' gender. Women appeared to be generally more irritated than men by displeasing aspects of the shopping environment. It is also conceivable that an atmosphere that produces a positive response in females may produce a less positive response in male shoppers. Gender, therefore, is another demographic factor that should not be ignored when studying the impact of store colour on consumers. For this reason, we will assess its moderating impact in our analyses. Nevertheless, because investigations in a retailing context have focussed almost exclusively on females, it is not clear how exactly gender will affect the proposed relationships. Based on the findings with regard to gender differences in emotional reactions to colours in general, by Norman and Scott (1952), Whitfield and Wiltshire (1990) and Valdez and Mehrabian (1994), we hypothesize that for females the effects of colour in the store environment will be stronger than for males. Thus, we do not expect to find any differences in the direction of the proposed relationships between males and females; however, we do expect a difference in strength of the relationships, with the relationships being stronger for women.

Thus far the presentation of the proposed hypotheses. These will be tested in the empirical study, which will be elaborated in the second part of this dissertation.

Part II Empirical Research

Chapter 5

Research Methodology



CHAPTER STRUCTURE

Chapter 5: Research Methodology

- 5.1. Introduction
- 5.2. Experimental Design
- 5.3. Stimuli
- 5.4. Subjects & Procedures
- 5.5. Measures

5.1. INTRODUCTION

As indicated in the previous chapter, this study aims to assess the impact of one store design element, namely colour, on consumers' feelings and subsequent shopping intentions. Since the prevalent research on the impact of colour in the store environment is scarce and methodologically flawed, the employed methodology of the current study will be elaborately presented here. The development of the experimental design, outlined in section 5.2., was guided by our objective to investigate the specific effects of the three attributes of store-interior-colour: hue, saturation and brightness. In section 5.3., the actual store-environment, which was especially developed in computer aided design to serve as a stimulus for this study, is presented. The selection of the specific colour tones for the manipulations of the stimuli is also elucidated here. Section 5.4. overviews the procedures followed for sampling and questioning the subjects. A brief account of the participants is given here as well. Finally the measurement instrument, used to capture their reactions to the manipulated stores, is discussed in section 5.5.

"Experimentation involves the manipulation of one or more variables by the experimenter in such a way that its effects on one or more other variables

can be measured"

TULL AND HAWKINS

1993, p211

5.2. EXPERIMENTAL DESIGN

Experimentation allows us to establish and measure the causal relationships between store colour and consumers' affective and behavioural responses (Tull and Hawkins, 1993, p212). The proposed experimental study can be classified as a 'judgment study' (Aronson, Brewer and Carlsmith, 1985; Judd, Smith and Kidder, 1991), in that the participants are asked to evaluate a stimulus, presented by the experimenter. Effects of variations in the characteristics of the stimulus materials are the key focus of interest. In this case store colour will function as the independent variable and is therefore manipulated.

In contrast to previous studies on the subject, in this study, special attention is devoted towards establishing an experimental design that allows us to investigate causal relationships with respect to each of the three colour attributes: hue, brightness and saturation. Accordingly, the resulting experimental design is a three-way factorial design (colour hue by colour saturation by colour brightness). In fact an 8 by 2 by 2 factorial design was opted for, as will be detailed in the next section.

In order to achieve adequate power, about 25 observations per cell are called for, especially since the colour-effects are expected to be subtle. The selected experimental design is a between-subjects design, in the sense that each respondent is exposed to only one colour condition. This procedure is deemed necessary in order not to reveal the specific nature of the study. Due to the monadic nature of the experiment, the 32 (i.e. 8x2x2) treatment groups will each have to consist of about 25 subjects. Subjects will be matched as much as possible on the basis of gender and three age categories. The experimental procedures, as well as those followed for sampling and questioning the subjects, are described in section 5.4.

5.3. STIMULI

A kind of laboratory experiment, using photographic environmental simulations, was opted for in this study, since such an experiment provides a level of control that would be hard to achieve in a field study (Bateson and Hui, 1992). Moreover, laboratory experiments tend to cost substantially less in terms of resources and time than field experiments (Tull and Hawkins, 1993, p246). Numerous techniques are available for simulating retail environments, ranging in realism from verbal descriptions to actual store interiors (see Eroglu, Scholder and Machleit, 1992 and Bateson and Hui, 1992 for an overview). Photographic slides and pictures, representing static, perceptual simulations (McKechnie, 1977), have been demonstrated to be able to adequately represent the environment (e.g. Hershberger and Cass, 1974; Bitner, 1990; Eroglu and Machleit, 1990; Hui and Bateson, 1991; Areni, Sparks and Dunne, 1996). Hui and Bateson (1991), for example, successfully operationalized consumer density by means of coloured slides portraying different numbers of consumers in a setting. Previous studies on the impact of store colour have also utilized projections of store interiors (e.g. Bellizzi, Crowley and Hasty, 1983; Bellizzi and Hite, 1992; Crowley, 1993).

A major strength of laboratory experiments is the minimization of the effects of 'history', by the removal of a majority of possibly confounding extraneous variables, which provides a high degree of internal validity or replicability (Kerlinger, 1973, p398; Tull and Hawkins, 1993, p246). Unfortunately, as a direct consequence, laboratory experiments are generally rather weak in external validity or generalizability. Nevertheless, the use of photographic slides as environmental simulations has been proven to be a valid method for studying the effects of environments on evaluations, affective responses and behaviour (Hershberger and Cass, 1974; Russell and Mehrabian, 1976; Bateson and Hui, 1992). Russell and Mehrabian (1976) have provided empirical evidence for the generalizability of data based on slides representing physical settings. The use of photographic slides to simulate service settings has also been demonstrated to have 'ecological validity' (McKechnie, 1977, p169), in the sense that (a) they can represent realistic service settings and (b) they are able to elicit the same experience as the actual service setting (Bateson and Hui, 1992). These arguments justify our choice to utilize manipulated pictures to simulate the store interior in the current study.

More specifically, the picture of a store in design furniture and accessories was chosen as stimulus. This choice was made on the basis of two considerations. First of all, to ensure that emotional state and store environment would be relevant to the shopper's buying experience, a store-type carrying high-involvement goods should be selected for this kind of study (Flicker & Speer, 1990, Sherman et al., 1997). Moreover, in the selected store format, the consumer would not be suspicious of the perhaps unusual colours used in the manipulations. This may also be an important factor, especially since Yoo, Park and MacInnis (1998) surmise that consumers' frames of reference and expectations regarding "typical" store characteristics may play a more central role in affecting consumers' emotional responses than the actual store characteristics themselves. Indeed, examining the potential role that store atmosphere expectations play in influencing emotional responses to the shopping experience, Machleit and Eroglu (2000), revealed that as the store atmosphere differed more from what the shopper had expected, then he or she is less likely to feel some of the positively valenced emotions and more likely to feel negatively valenced emotions. Specifically, they found that as the store atmosphere deviated from expectations, the shoppers felt less pleasure, interest and joy and more sadness and disgust. Not unexpectedly, surprise was also significantly higher when the environment was not as anticipated. Moreover, they found that the significant correlations varied by store type and note that it is possible that in a recreational shopping environment, such as a mall, some deviations from expectations could be consistent with the stimulation sought by recreational shoppers. A store in design furniture and accessories involves a store-type carrying high-involvement goods, offered in a recreational shopping environment in which the use of extraordinary colours should not be suspicious. Thus the requirements posed to the stimulus used in this study are fulfilled.

The store in design furniture and accessories was developed in Computer Aided Design (ARKEY), allowing us to realistically manipulate colour in the store environment. As colours can be defined according to three dimensions, appropriate colour-stimuli have to be selected according to these distinct colour attributes. Accordingly, the experimental design is an 8 x 2 x 2 factorial design (colour hue by colour saturation by colour brightness). For the experimental manipulation, initially only four hues from the Munsell Colour System were selected for their divergent pleasure and arousal eliciting properties: blue-green (high pleasantness, high arousal), purple-blue (high pleasantness, low arousal), green-yellow (low pleasantness, high arousal) and yellow-red (low pleasantness, low arousal) (see also Ziems and Christman, 1998; based on Valdez and Mehrabian's 1994 results on affective responses

regarding colour patches). However, since a manipulation check performed on a first set of data revealed that these hues did not elicit pleasure and arousal in respondents as intended, four more hues from the Munsell Colour System were included in the study. For each hue, four specific tones were selected (as indicated in figure 5-1), with low versus high saturation levels (Munsell chroma 6 versus 14) and low versus high brightness levels (Munsell value 5 versus 8, for all hues, except for those for which the former values exceed the existing range, which was the case for blue and red, in which case Munsell value 4 versus 7 was used; purple-blue in which case Munsell value 3 versus 6 was applied and yellow, in which case Munsell value 6 versus 9 was selected).



²⁰ To view displays like this of all the different Munsell hues, we refer to the Gretag Macbeth website (<u>http://standards.gretagmacbeth.com/cmc/index.htm</u>, last updated March 11, 2002), from which Munsell Conversion Software - V6.13 (Win 95/98/NT/2000/XP) Conversion of Munsell > XYZ > L*a*b* > RGB > CYMK > Color on Screen - can be downloaded for free.

The selection of particular colour-tones to use in the experimental design was certainly not obvious, as the three dimensional colour space is highly irregular, which is clearly illustrated in figure 5-2, as discussed earlier, in chapter 3 (paragraph 3.3.4.). In fact, not all brightness-saturation combinations appear to exist for all hues, as different hues reach their maximum chroma at different values. Rather than limiting this study by excluding hues because of their dissimilar chroma-value patterns (as Gorn et al., 1997 did), we have attempted to select a combination of brightness and saturation levels existent in a majority of hues. Only when there was no other possibility because the selected values were out of the exceeding range for a particular hue, brightness levels were slightly adapted.





Source: Handprint Media, 2001 (http://www.handprint.com/HP/WCL/color6.html)

Accordingly the following 32 colours were used in the experiment (Munsell notation, 1905: hue value/chroma, we refer to appendix 1 for a table with the corresponding RGB values):

- blue-green: (5.00 BG 5/6), (5.00 BG 5/14), (5.00 BG 8/6), (5.00 BG 8/14)
- purple-blue: (5.00 PB 3/6), (5.00 PB 3/14), (5.00 PB 6/6), (5.00 PB 6/14)
- green-yellow: (5.00 GY 5/6), (5.00 GY 5/14), (5.00 GY 8/6), (5.00 GY 8/14)
- yellow-red: (5.00 YR 5/6), (5.00 YR 5/14), (5.00 YR 8/6), (5.00 YR 8/14)
- green: (5.00 G 5/6), (5.00 G 5/14), (5.00 G 8/6), (5.00 G 8/14)
- blue: (5.00 B 4/6), (5.00 B 4/14), (5.00 B 7/6), (5.00 B 7/14)
- red: (5.00 R 4/6), (5.00 R 4/14), (5.00 R 7/6), (5.00 R 7/14)
- yellow: (5.00 Y 6/6), (5.00 Y 6/14), (5.00 Y 9/6), (5.00 Y 9/14)

This way 32 colour manipulations of the store are obtained, some of which are presented in the following figures.







Figure 5-3: The four stimuli selected to represent the Munsell 5 Yellow-Red Hue region, with respectively

High Brightness – Low Saturation High Brightness – High Saturation Low Brightness – Low Saturation Low Brightness – High Saturation (upper left corner) (upper right corner) (lower left corner) (lower right corner)

Source: Store environments developed especially for this study by Wim Goris, in ARKEY (Computer Aided Design), BAT NV.



Figure 5-4: The four stimuli selected to represent the Munsell 5 Green-Yellow Hue region, with respectively

High Brightness – Low Saturation High Brightness – High Saturation Low Brightness – Low Saturation Low Brightness – High Saturation (upper left corner) (upper right corner) (lower left corner) (lower right corner)

Source: Store environments developed especially for this study by Wim Goris, in ARKEY (Computer Aided Design), BAT NV.

5.4. SUBJECTS & PROCEDURES

The experiment was carried out in Belgium. Subjects were intercepted by means of a random walk sampling procedure, with predetermined quota for age and gender. About 25 respondents were selected per manipulation. They were matched as much as possible on the basis of gender and three age categories: 18-30, 30-45 and 45-60 year olds. Tables 5-1 and 5-2 give an overview of the demographic and socio-economic characteristics of all the 874 participants.

		Age		
Gender	18-30	30-45	45-60	Total
Male	159	136	131	426
	18.2%	15.6%	15.0%	48.9%
Female	162	151	133	446
	18.6%	17.3%	15.3%	51.1%
Total	321	287	264	872
	36.8%	32.9%	30.3%	100.0%

Table 5-1: Respondents according to age and gender (n=874)

In order to conceal the particular focus and goal of the research, the interviewers presented themselves as working for a market-research company, investigating the market-potential of an Italian store in design furniture and accessories, which is interested in opening up an outlet in Belgium. The presumed objective of the study was to determine whether people would be interested in this store. An incentive was proposed in order to stimulate people to participate to the study. Participants were eligible for a 25 Euro prize of choice (either movie-theatre tickets or book-value-coupons), five of which were raffled at the end of the study.

Household Size:	%
Mean: 3.23	
Std. Dev.: 1.44	
1	11.4
2	23.5
3	19.6
4	29.4
5+	16.1
Number of Kids in Household:	%
Mean: 1.43	
Std. Dev.: 1.33	
0	32.1
1	20.1
2	29.9
3+	17.8
Occurrentian	0/
Occupation:	70
Full-Time Employed	62.1
Part-Time Employed	11.0
Unemployed / Housewife / Retired	6.6
Student	20.3
	-0.0
Level of Education:	%
Junior High School Diploma or less	6.2
Technical or Professional Education	15.7
General High School Diploma	23.7
Short Type Higher Education	18.8
Long Type Higher Education	6.6 20.0
University	29.0
Net Average Monthly Household Income:	%
< 30.000 BEF	7.4
30.000 - 45.000 BEF	10.4
45.000 - 60.000 BEF	19.9
60.000 - 75.000 BEF	15.2
75.000 - 95.000 BEF	10.4
95.000 - 115.000 BEF	14.8
115.000 - 135.000 BEF	9.5
> 135.000 BEF	12.5

Table 5-2: Demographic and socio-economic characteristics of respondents (n=874)

Respondents were asked to fill out a short mood inventory before the actual stimulus – a manipulated picture of the store (16 by 16 cm)– was shown. Each participant was randomly allocated to one of the 32 experimental conditions and shown the corresponding stimulus. As the experiment is monadic in nature, each subject was presented only one store.

The participants were asked to take a look at the picture of the store and fill out the remainder of the questionnaire, including questions pertaining to the emotions elicited by the store and approach/avoidance intentions towards the store. Socio-demographic characteristics of respondents were tapped as well.

Since inherited defects of colour vision affect about 8% of men and 0.5% of women, it is important to screen respondents for colour blindness. However, respondents were not screened for colour vision deficiency before selection, in order not to reveal the purpose of the study. This was done at the end of the experiment.

Finally, respondents were thanked for their participation.
5.5. MEASURES

Each of the 874 respondents, participating in this study, was administered a lengthy questionnaire. The survey instrument contained questions pertaining to participants' general pre-existing mood state, the emotional reactions elicited by the experimental store and their approach/avoidance intentions towards the store. All scales, included in the questionnaire, were carefully translated, with a forward and backward check (Brislin, 1980).

First of all, pre-existing mood was measured, before exposure to the stimulus, by a short 4item mood inventory (Peterson and Sauber, 1983). This Mood Short Form (MSF) has been generated to reflect the 'mood' content domain and was developed, based on a large pool of items, drawn amongst others from Mehrabian's (1972) nonverbal communication scale and Nowlis' (1962) Mood Adjective List. The purified uni-dimensional four-item scale was derived after factor analysis, detailed item analyses and reliability checks. Coefficient alpha for the scale was reported to be .78, .74 and .77 for samples of respectively 1,434, 713 and 248 respondents (Baerden and Netemeyer, 1999). That the transient and varying nature of mood is captured by the MSF, was demonstrated by a test-retest reliability over a 30-day period (n=177) of only .18. Items were measured on a 7-point Likert-type scale (from -3: strongly disagree to +3: strongly agree). Two out of the four items were reversed. The scale is illustrated in table 5-3. Item scores are supposed to be summed to acquire a uni-dimensional MSF index.

Table 5-3: Mood Short Form - MSF (Peterson and Sauber, 1983)

Mood Shor	t Form - MSF (Peterson and Sauber, 1983)
(measured of	on a 7-point Likert-type scale - strongly disagree to strongly agree):
MOOD1	Currently, I am in a good mood. Ik ben momenteel in een goede bui.
MOOD2	As I answer these questions I feel cheerful. Bij het invullen van deze vragenlijst voel ik me opgewekt.
MOOD3	For some reason I am not very comfortable right now. (-) Door omstandigheden voel ik me op dit moment niet erg op mijn gemak.(-)
MOOD4	At this moment I feel edgy or irritable. (-) <i>Ik voel me voor het ogenblik gespannen en prikkelbaar.</i> (-)

Subsequently, three six-item semantic differential scales, measuring the PAD-dimensions of emotions elicited by the experimental store (Mehrabian and Russell, 1974; Russell and Mehrabian, 1977), were incorporated in the questionnaire. The background and validity of these scales (illustrated in table 5-4) have been discussed previously in chapter 2 (paragraph 2.5.). Items of the respective scales were intermixed within an 18-item inventory to minimize respondent awareness of the scale dimensions, thereby enhancing validity (see Mehrabian, 1998). The direction of about half of the items in the scales was reversed.

Table 5-4: PAD measurement scale (Mehrabian and Russell, 1974)

Pleasur	Pleasure items (measured on a 7-point semantic differential scale):						
In this store I feel Deze winkel geeft mij							
P1 P2 P3 P4 P5 P6	depressed unhappy unsatisfied annoyed bored despairing	 	contented happy satisfied pleased relaxed hopeful	een bedrukt gevoel een ongelukkig gevoel een ontevreden gevoel een geërgerd gevoel een verveeld gevoel een wanhopig gevoel	_ _ _ _	een aangenaam gevoel een gelukkig gevoel een tevreden gevoel een behaaglijk gevoel een ontspannen gevoel een hoopvol gevoel	

Arousal items (measured on a 7-point semantic differential scale):

In this store I feel ...

Deze winkel geeft mij ...

A1	relaxed	_	stimulated	een ontspannen gevoel	_	een gestimuleerd gevoel
A2	calm	_	excited	een rustig gevoel	_	een opgewonden gevoel
A3	sluggish	—	frenzied	een loom gevoel	_	een uitzinnig gevoel
A4	dull	_	jittery	een futloos gevoel	_	een zenuwachtig gevoel
A5	sleepy	—	wide-awake	een slaperig gevoel	_	een wakker gevoel
A6	unaroused	-	aroused	een niet geprikkeld gevoel	_	een geprikkeld gevoel

Dominance items (measured on a 7-point semantic differential scale):

In this store I feel ...

Deze winkel geeft mij ...

D1 D2	controlled –	controlling influential	een geremd gevoel	_	een vrij gevoel een ongedwongen gevoel
D2 D3	cared for _	in control	een gehalnen gevoel	_	een autonoom gevoel
D3 D4	awed –	important	een minderwaardig gevoel	_	een superieur gevoel
D5	submissive -	dominant	een onderdanig gevoel	_	een dominant gevoel
D6	guided –	autonomous	een begeleid gevoel	_	een zelfstandig gevoel

Next, the questionnaire contained an eight-item scale measuring approach/avoidance behaviour intentions towards the store (Donovan and Rossiter, 1982). The approach/ avoidance items (illustrated in table 5-5) were measured on 7-point Likert scales (from -3: strongly disagree to +3: strongly agree). In order to avoid response bias, the order of the items was mixed and the direction of three of the items in the scale was reversed. The background and validity of this scale has been discussed previously in chapter 2 (paragraph 2.6.).

Table 5-5: Approach/avoidance measurement scale (Donovan and Rossiter, 1982)

Approach	/avoidance items
(measured	on a 7-point Likert-type scale - strongly disagree to strongly agree):
	Time dimension:
APAV1	I like to spend much time browsing in this store Ik zou in deze winkel graag veel tijd willen doorbrengen om rustig rond te kijken
APAV2	I want to avoid looking around or explore this store (reversed) In deze winkel zou ik vermijden rond te kijken en de winkel te verkennen
	Affiliation:
APAV3	This is a place where I try to avoid other people, and avoid to talk with them (reversed) In deze winkel zou ik trachten mensen te mijden of vermijden ermee te moeten praten
APAV4	This is a place in which I feel friendly and talkative to store personnel who happens to be near me In deze winkel zou ik mij goed gezind voelen en zou ik open staan voor een praatje
	Affect:
APAV5	I like this store environment Ik hou van deze winkelomgeving
APAV6	I enjoy shopping in this store Ik zou er van genieten in deze winkel te komen winkelen
	Spending:
APAV7	I would avoid ever having to return to this store (reversed) Ik zou zoveel mogelijk vermijden naar deze winkel te moeten komen
APAV8	This is the sort of place where I might end up spending more money than I originally set out to spend Ik zou in deze winkel meer geld kunnen uitgeven dan oorspronkelijk voorzien

Finally, the questionnaire also contained questions pertaining to the socio-demographic characteristics of participants, including age, gender, household size, number of kids in household, the respondent's occupation and level of education and the net average monthly household income.

At the very end of the questionnaire, respondents were asked, directly, whether they knew to suffer from any form of colour vision deficiency. To conclude the experiment, respondents' colour vision was also tested by the experimenter by means of three pseudo-isochromatic plates from the Ishihara Colour Vision test (Ishihara, 1969, 1980), for which partially concealed figures have to be identified from a mass of different coloured dots.

Chapter 6

Empirical Results



CHAPTER STRUCTURE

Chapter 6: Empirical Results

- 6.1. Introduction
- 6.2. Screening for colour blindness or colour vision deficiency
- 6.3. Data screening
- 6.4. Respondents
- 6.5. Construct validation
 - 6.5.1. Construct validation of the PAD-emotion scale
 - 6.5.1.1. Exploratory Factor Analysis
 - 6.5.1.2. Uni-Dimensionality
 - 6.5.1.3. Within-Method Convergent Validity
 - 6.5.1.4. Reliability
 - 6.5.1.5. Discussion
 - 6.5.2. Construct validation of the approach-avoidance scale
 - 6.5.2.1. Exploratory Factor Analysis
 - 6.5.2.2. Uni-Dimensionality
 - 6.5.2.3. Within-Method Convergent Validity
 - 6.5.2.4. Reliability
 - 6.5.2.5. Discussion
- 6.6. Store-colour-evoked emotions
 - 6.6.1. Introducing 'mood' as a covariate
 - 6.6.2. Hypotheses testing : Multivariate analysis of variance (MANCOVA)
 - 6.6.3. Store-colour-evoked pleasure
 - 6.6.3.1. Store colour hue
 - 6.6.3.2. Store colour brightness
 - 6.6.3.3. Store colour saturation
 - 6.6.4. Store-colour-evoked arousal
 - 6.6.4.1. Tension and Excitement: A two-dimensional View of Arousal 6.6.4.2. Store colour hue

- 6.6.4.3. Store colour brightness
- 6.6.4.4. Store colour saturation
- 6.6.5. Store-colour-evoked dominance
- 6.6.6. Moderating effects
 - 6.6.6.1. Testing for the moderating effect of gender
 - 6.6.6.2. Testing for the moderating effect of age

6.7. Store-colour-evoked emotions as determinants of approach/avoidance behaviours

- 6.7.1. Hypotheses Testing: Structural Equation Modeling
- 6.7.2. Pleasure as a determinant of approach/avoidance behaviours
- 6.7.3. Arousal as a determinant of approach/avoidance behaviours
- 6.7.4. Overview
- 6.7.5. Moderating effects
 - 6.7.5.1. Testing for the moderating effect of gender
 - 6.7.5.2. Testing for the moderating effect of age
- 6.8. Testing Mediation

6.1. INTRODUCTION

The research model, proposed in chapter 4, was tested by an experimental design, as reviewed in chapter 5. In the current chapter, the research findings are presented, starting of by discussing in section 6.2. how screening for colour vision deficiency among respondents was carried out. In section 6.3. the data screening procedure, which was executed prior to the analyses, is discussed. An account of the respondents in the final sample is given in section 6.4. Subsequently, section 6.5. overviews the construct validation process of both the PADemotion scale and the approach-avoidance scale: the uni-dimensionality, within-method convergent validity and reliability of the respective scales are conscientiously assessed. The actual hypotheses testing falls apart in two main parts: In section 6.6. the hypotheses concerning store-colour-evoked emotions are tested by means of multivariate analysis of variance (MANCOVA). Here pre-existing mood is introduced as a covariate and the different emotions elicited by store-colour hue, brightness and saturation are examined in detail. Moreover, the moderating effects of age and gender on the emotions elicited by store-interiorcolours are assessed. In section 6.7. a second set of hypotheses, concerning the impact of store-colour-evoked emotions on approach/avoidance intentions, is tested by means of structural equation modelling (SEM). The moderating effects of age and gender are also investigated with regard to these relationships. Finally, the preceding findings are linked together, in section 6.8., by verifying whether it is reasonable to assume that store-colourevoked emotions act as a mediator in the relationship between store colour and approachavoidance responses towards the store.

6.2. SCREENING FOR COLOUR BLINDNESS OR COLOUR VISION DEFICIENCY

As mentioned in chapter 3 (paragraph 3.2.3), colour blindness is a visual defect resulting in the inability to distinguish colours (Medical Post, 1997; Neitz, 1998). About 8% of men and 0.5% of women experience some difficulty in colour perception

Due to the specific nature of this study, respondents suffering from colour vision deficiency, or colour blindness have to be removed from the sample, in order to eliminate any possible bias that may be induced this way.

To determine the extent of colour-deficiencies, many researchers have used Ishihara plates (illustrated in figure 6-1) and other colour-discrimination methods (Benjamin, Press, Maoz & Belmaker, 1993; Corsino, 1985; Miyao et al., 1991; Shepard & Cooper, 1992). Others have relied on individuals' self-identification (Kaufman-Scarborough, 2000). However, as the colour vision of perfectly colour-normal persons can be affected over time by age, progressive illness or the use of certain medications and can change subtly, either temporary or permanently (Braus, 1995; Wilson, 1998), an individual who has had normal colour vision, may not realize a gradual decline in perceptual ability for some time.



Figure 6-1: Illustration of pseudo-isochromatic plates from the Ishihara Colour Vision Test Source: Ishihara, 1969, 1980

In the current study colour vision deficiency was measured in two ways, at the end of the questionnaire, in order not to reveal the particular purpose of the study. First of all respondents were asked whether they suffered from any form of colour vision deficiency (i.e. stated colour vision deficiency). We then also tested our respondents' colour vision, using three pseudo-isochromatic plates (identifying partially concealed figures from a mass of different coloured dots) from the Ishihara Colour Vision test (i.e. revealed colour vision deficiency) (Ishihara, 1969, 1980).

Crosstabulations of stated and revealed colour vision deficiency (see table 6-1), demonstrate that the majority of respondents reporting to have good colour vision, scored good on the Ishihara colour vision test (98.6%). Of the 14 participants who were unsure about their colour vision 11 (78.6%) appeared to have good colour vision after all. 59 or 6.8% of all the respondents indicated to be aware of problems with their colour vision. Nevertheless of these participants 32 (54.2%) had good results on the Ishihara colour vision test.

	Stated Colour Vision Deficiency			
Revealed Colour Vision Deficiency (Ishihara Colour Test)	Good Colour Vision	Unsure	Deficient Colour Vision	Total
Good Colour Vision	790	11	32	833
Within Stated	98.6%	78.6%	54.2%	95.3%
Deficient Colour Vision	11	3	27	41
Within Stated	1.4%	21.4%	45.8%	4.7%
Total	801	14	59	874
	91.6%	1.6%	6.8%	100%

Table 6-1: Colour blindness among respondents (n=874)

In order to eliminate any possible bias that may be induced by colour blindness, all respondents that indicated having deficient colour vision (stated colour vision deficiency), as well as all respondents who failed the Ishihara Colour Vision Test (revealed colour vision deficiency) were eliminated from the sample. This totals up to 73 participants who could not be retained for further analyses. When we take a look at the demographic characteristics of these respondents (Table 6-2) it is clear that especially males (76.7%) suffer from this condition of inadequate colour vision. With regard to age the group of 45-60 year-olds seems to be a little stronger represented among the colour blind in our sample.

		Age		
Gender	18-30	30-45	45-60	Total
Male	18	17	21	56
	24.7%	23.3%	28.8%	76.7%
Female	5	5	7	17
	6.8%	6.8%	9.6%	23.3%
Total	23	22	28	73
	31.5%	30.1%	38.4%	100.0%

Table 6-2: Demographic characteristics of colour-blind respondents (n=73)

That colour vision deficiency affects more males than females is clear from the following table (Table 6-3). In our sample 13.1% of males suffer from deficient colour vision as compared to only 3.8% of females.

Table 6-3: The impact of gender on a	colour vision deficiency
--------------------------------------	--------------------------

	Gen	der	
Colour Vision Deficiency	Male	Female	Total
Good Colour Vision	370	431	801
	86.9%	96.2%	91.6%
Deficient Colour Vision	56	17	73
	13.1%	3.8%	8.4%
Total	426	448	874
	100.0%	100.0%	100.0%

 $\chi^2 = .000$

Only 801 respondents with good colour vision are retained in our sample for further analyses.

6.3. DATA SCREENING

As the first step of data analysis should always be a detailed examination of the data (Norusis, 1993, p181), the raw data were inspected before starting with the actual data analyses. Using descriptive statistics, the data set was first checked for typing and coding errors (e.g. out-of-range values). The original questionnaires were consulted in these cases where such errors appeared to be present (Baumgartner and Homburg, 1996). Then, negatively stated items were reverse scored. For the cases where missing values were found, it was double-checked whether the missing data were not simply omitted during data-input. 24 surveys had to be eliminated because of rather incomplete responses.

Since cases that have unusual values for the independent variables can have a substantial impact on the results of analyses, these should be identified (Norusis, 1993, p331). A multivariate procedure for identifying outliers was used, based on the Mahalanobis D^2 measure (Hair et al., 1998). Mahalanobis D^2 is a measure of the distance in the multidimensional space of each observation from the mean centre (i.e. centroid) of the observations (see Tacq, 1997, p251). Thus, in order to locate outliers, for each observation the Mahalanobis Distance was calculated based on all items used in the independent constructs²¹. The value obtained by dividing Mahalanobis D^2 by its degrees of freedom (i.e. the number of independent variables in the analysis) is approximately t-distributed. By means of this statistic, observations can be identified as potential outliers. In our sample only one case was observed with a D^2/df -value of 3.75, slightly exceeding the threshold value (i.e. critical t-value (p=.0005, df=30) = 3.646). Nevertheless, this case was not considered to be a real outlier, because it appeared not to be influential (Cook's Distance <.002)²². Therefore, no real multivariate outliers were found with respect to the items used in the analyses.

²¹ The analysis was based on a total of 30 items: 18 PAD-items, 8 approach/avoidance items and 4 mood items.

²² Cook's distance considers changes in all residuals when the case is omitted (Cook, 1977; Norusis, 1993, p333).

Subsequently, concerning the distribution of the variables, both univariate and multivariate normality were examined. With regard to the individual variables, histograms, normal probability plots and detrended normal plots²³ were visually checked for departures from normality. From the histograms, the distributions of all items appear to be roughly bellshaped. With regard to the normal probability plots, some slight departures from the diagonal can be identified. However, the detrended normal plots did not reveal any striking patterns. Both skewness and kurtosis approximately ranged between -1.5 and +1.5 (Muhtén and Kaplan, 1985). Although some variables exhibit a statistically significant departure from univariate normality, these violations appear to be relatively minor and should not present any serious problems in the course of the data analysis. A subsequent normality check, performed in AMOS 3.62 (Arbuckle, 1997, p238), revealed a significant departure from multivariate normality²⁴. In general, however, large sample sizes, tend to diminish the detrimental effects of non-normality (Hair et al., 1998, p71). Maximum likelihood parameter estimates have been found to be rather robust against this kind of violation, provided that sample size exceeds about 100 observations (Steenkamp and van Trijp, 1991; Boomsma, 1982; Gerbing and Anderson, 1985). Also with regard to MANOVA analyses, violations of the assumption of multivariate normality have little impact with larger sample sizes (Hair et al., 1998, p349).

Thus, a total of 777 cases were retained. The characteristics of the respondents in the final sample are presented in the following paragraph.

²³ A normal probability plot compares the cumulative distribution of actual data values with the cumulative distribution of a normal distribution. The line representing the actual data distribution should follow the diagonal closely in order to be able to assume a normal distribution (Hair et al., 1998, p71). In a detrended normal plot the deviations from this straight diagonal line are plotted. In case of a normal distribution, the points should cluster around a horizontal line, through 0, and there should be no pattern (Norusis, 1993, p189).

²⁴ Multivariate normality assumes that the joint effect of multiple variables is normally distributed.

6.4. **RESPONDENTS**

Tables 6-4 and 6-5 give an overview of the demographic and socio-economic characteristics of the retained 777 participants. The final sample includes a little bit less males than females (respectively 46.5% versus 53.5%). Note that considerably more males suffered from colour vision deficiency. Among the three preconceived age categories, for both men and women fairly equal numbers of respondents were obtained.

		Age		
Gender	18-30	30-45	45-60	Total
Male	137	116	107	360
	17.7%	15.0%	13.8%	46.5%
Female	153	140	122	415
	19.7%	18.1%	15.7%	53.5%
Total	290	256	229	775
	37.4%	33.0%	29.5%	100.0%

Table 6-4: Respondents according to age and gender (n=777)

Moreover, a fairly representative sample was obtained with regard to household size, the number of kids in the households and respondents' occupation, education level and household income.

Household Size:		%
M	ean: 3.26	, 0
Std. I	Dev.: 1.44	
1		10.8
2		23.1
3		19.2
4		30.8
5+		16.1
Number of Kids in Household:		%
Μ	ean: 1.45	
Std. I	Dev.: 1.35	
0		32.0
1		19.4
2		30.7
3+		17.9
Occupation:		%
occupation		/0
Full-Time Employed		61.2
Part-Time Employed		11.5
Unemployed / Housewife / Retired		6.3
Student		21.0
Level of Education:		%
Inniar High School Diploma or loss		57
Tashnical or Professional Education		5.7 15.6
General High School Dinloma		24.0
Short Type Higher Education		183
Long Type Higher Education		6.1
		20.5
LINIVERSILV		795
University		29.5
Net Average Monthly Household In	ncome:	29.5
Net Average Monthly Household In	ncome:	29.5 %
Net Average Monthly Household In < 30.000 BEF	ncome:	29.5 %
Net Average Monthly Household In < 30.000 BEF	ncome:	6.9 11.2
Net Average Monthly Household In < 30.000 BEF	ncome:	6.9 11.2 20.3
Net Average Monthly Household In < 30.000 BEF	icome:	6.9 11.2 20.3 14.7
Net Average Monthly Household In < 30.000 BEF	icome:	29.5 % 6.9 11.2 20.3 14.7 11.0
Net Average Monthly Household In < 30.000 BEF	icome:	29.5 % 6.9 11.2 20.3 14.7 11.0 14.7
Net Average Monthly Household In < 30.000 BEF	icome:	29.5 % 6.9 11.2 20.3 14.7 11.0 14.7 9.3

Table 6-5: Demographic and socio-economic characteristics of respondents (n=777)

Table 6-6 gives an overview of the number of respondents obtained in each condition.

	-	Satur	Total	
Hue	Value	low saturation	high saturation	777
Purple-Blue	dark	24	28	52
	light	27	29	56
	8	51	57	108
Blue	dark	24	24	48
	light	21	15	36
	C C	45	39	84
Blue-Green	dark	27	26	53
	light	27	29	56
	C	54	55	109
Green	dark	22	18	40
	light	29	22	51
	-	51	40	91
Green-Yellow	dark	25	24	49
	light	25	27	52
	-	50	51	101
Yellow	dark	23	21	44
	light	22	24	46
	-	45	45	90
Yellow-Red	dark	25	30	55
	light	28	23	51
	5	53	53	106
Red	dark	25	21	46
	light	20	22	42
		45	43	88

Table 6-6: Number of respondents in each condition (n=777)

6.5. CONSTRUCT VALIDATION

As the validity of constructs is a necessary condition for theory development and testing, it forms a foundation for scientific progress in marketing (Steenkamp & van Trijp, 1991).

Evaluating the <u>unidimensionality</u> of a construct ensures that the questionnaire items cover the same content domain. It can be defined as the 'existence of one construct underlying a set of items' (Steenkamp and van Trijp, 1991, p286) and has been recognized as 'one of the most critical and basic assumptions of measurement theory' (Hattie 1985, p135; Steenkamp and van Trijp, 1991). <u>Within-method convergent validity</u> or internal consistency involves 'the extent to which multiple applications of the same method are in agreement' and <u>reliability</u> refers to 'the degree to which measures are free from random error' (Steenkamp and Van Trijp, 1991, p289).

The dimensionality and validity of the PAD-emotion scale and the approach-avoidance scale will be investigated conscientiously in the following paragraphs. Covariance structure modelling provides substantial advantages over traditional methods, such as coefficient alpha²⁵, exploratory factor analysis and bivariate correlations, to assess the criteria for construct validity (Steenkamp and Van Trijp, 1991). For this reason AMOS²⁶ 3.62 (Arbuckle) will be used to validate the different constructs used in this study and in order to purify the measures. Not only the uni-dimensionality of the constructs will be tested this way, but also their within-method convergent validity or internal consistency and reliability...

²⁵ Cronbach's alpha, which is based on the internal consistency of the items in a scale, is one of the most commonly used reliability coefficients (Norusis, 1994, p147)

²⁶ AMOS is an SPSS program for structural equation modelling.

6.5.1. CONSTRUCT VALIDATION OF THE PAD EMOTION SCALE

6.5.1.1. Exploratory Factor Analysis

According to previous research (Mehrabian and Russell, 1974; Russell and Mehrabian, 1977; Valdez and Mehrabian, 1994) the six pleasure items, the six arousal items and the six dominance items are expected to indicate three underlying dimensions. In order to test the theoretical structure of the measurement instrument, a maximum likelihood confirmatory factor analysis is conducted using AMOS 3.62 (Arbuckle, 1997) on the 18-item, three-factor PAD scale proposed by Mehrabian and Russell (1974). This yielded however unacceptable goodness of fit scores (χ^2 =1046.513, df=132, p<.001, χ^2 /df=7.928)²⁷.

In order to explore the uni-dimensionality of the pleasure, arousal and dominance constructs, Principal Component Factor Analyses with Varimax rotations were conducted on the items measuring the respective constructs. This suggested that in fact two separate dimensions underlie the arousal construct, which may explain the ambiguous results regarding this construct reached by previous researchers (Donovan et al., 1994; Van Kenhove and Desrumaux, 1997; see Gröppel-Klein and Baun, 2001). With regard to the dominance construct, results are rather ambiguous. The pleasure construct, on the contrary, appears to be measured very well. The six items clearly underlie one pleasure construct, explaining 66.7% of the variance (Cronbach's $\alpha = .8970$). The factor scores are presented in table 6-7.

²⁷ The overall fit of the model is an indication of the fit between the calculated variance-covariance matrix and the real variance-covariance matrix. A good model shows no significant difference (p-value>0.05) or for larger samples has a chi-square/d.f. <5.</p>

Table 6-7: Principle Component Factor loadings with Varimax rotated solution

	PLEASURE		
Items	Labels	(Cronbach's $\alpha = .8970$)	
P1 P2 P3 P4 P5 P6	depressed – contented unhappy - happy unsatisfied - satisfied annoyed - pleased bored - relaxed despairing - hopeful	.776 .808 .889 .815 .826 .778	
		66.623 % of the variance explained	
	AROUSAL		
	2-factor solution	(Cronbach's $\alpha = .5902$)	(Pearson corr. = .457, p<.001)
A1 A2 A4	relaxed - stimulated calm - excited dull - jittery	.611 .740 .673	
A6 A3 A5	unaroused - aroused sluggish - frenzied sleepy – wide-awake	.651	.851 .845
		30.029 % of the variance explained	24.362 % of the variance explained
	DOMINANCE		
	1-factor solution	(Cronbach's $\alpha = .5834$)	
D1 D2 D3 D4 D5 D6	controlled – controlling influenced – influential cared for – in control awed – important submissive - dominant guided – autonomous	.756 .385 .493 .608 .440 .675	
		33.029 % of the variance explained	
	2-factor solution	(Cronbach's α = .5602)	(Cronbach's α = .4238)
D1 D2 D6 D3 D4 D5	controlled – controlling influenced – influential guided – autonomous cared for – in control awed – important submissive - dominant	.763 .789 .551	.600 .578 .758
		26.940 % of the variance explained	26.438 % of the variance explained

With regard to the arousal construct, not one, but two factors are extracted with Eigenvalues exceeding unity, explaining together 54% of the variance (or respectively 30% and 24%). Also the Scree plot (presented in figure 6-2), representing the total variance associated with each factor (Cattell, 1966), suggests at least two factors²⁸.



Figure 6-2: Scree plot arousal construct

With regard to this two-factor solution, all items met the cut-off values and could clearly be ascribed to one of the factors. Each item had a minimum factor-loading of .60 and no loading higher than .15 on the other factor. However, the corrected item-total correlations were all unacceptably low (ranging from .33 to .45). The cut-off level for this criterion, opted for by several researchers (Green, Tull and Albaum, 1988; Tabachnik and Fidell, 1989) is .50. Nevertheless, the reliability of the respective dimensions appears adequate (Cronbach's α = .59 for the first factor and Pearson correlation = .457, p<.000 for the second factor²⁹). The factor scores are presented in table 6-7. The presence of two dimensions underlying the arousal construct has been demonstrated and rationalized in previous research (Walters, Apter and Svebak, 1982; Thayer, 1986; Gorn et al., 1997).

²⁸ Experimental evidence demonstrates that the true number of factors, is indicated by the component number at which the Scree (i.e. the gradual trailing off of the total variance associated with each subsequent factor) begins (Norusis, 1994).

²⁹ Because the second factor is composed of only two items, the correlation between the items is ascertained.

Reversal theory, a phenomenological theory originally proposed by Smith and Apter, 1975 (Apter, 1979, 1982), suggests that instead of one single curve (optimal arousal theory) there are two hypothetical curves relating arousal to hedonic tone, as presented in figure 6-3.



In contrast to optimal arousal theorists (Hebb & Thompson, 1954; Hebb, 1955; Fiske and Maddi, 1961), who assume that there is only one arousal system with a single optimal point (i.e. "homeostasis"), the theory of psychological reversals (Apter, 1981) argues that there are two systems, each with its own optimal point (i.e. "bistability"). According to this theory, there are consequently two optimal points or "preferred levels" of arousal positioned towards opposite ends of the arousal dimension (i.e. "reversals"). Thus, one system is an "arousal avoiding system" and the other is an "arousal seeking system". Only one is preferred at a given moment "being the preferred level of the arousal system that is operative at that moment" (Walters, Apter and Svebak, 1982). According to reversal theory, it is possible at any level of arousal to switch from the curve representing one of these systems to the other curve. Low arousal preference has been found to be associated with seriousness and planning orientation (characterizing a "telic state") and high arousal preference is associated with playfulness and spontaneity (characterizing a "paratelic state").

The system of emotions proposed by Wundt (1896) actually incorporates three bipolar dimensions: 'unpleasant-pleasant', 'quiet-excited' and 'relaxed-tense'.

In a retailing context, the PAD-arousal construct has been shown to be rather ambiguous (e.g. Donovan et al., 1994; Van Kenhove and Desrumaux, 1997; see Gröppel-Klein and Baun, 2001). Claiming that 'excitement' represents a pervasively positive reaction in a retail setting, Babin and Attaway (2000) see no reason to include a separate arousal dimension in their study. Also Gröppel-Klein (1998) argues that 'arousal' and 'pleasure' seem to converge in a retail setting, as 'positive activation' appears to be implicitly registered in the arousal construct. Nevertheless, empirical studies by Bost (1987) and Gröppel (1991) reveal that consumers in a retail setting can also experience a pleasant state of low arousal (i.e. relaxation). Thus, too much arousal can, also at the point-of-sale, be experienced as hectic and unpleasant. For this reason, Gröppel-Klein (1998) recommends not to neglect the emotional dimension of "relaxation" experienced in the shopping environment in further studies.

Consistent with such operationalization of arousal, the first dimension (relaxation-tension) appears to be negatively related with pleasure (Pearson correlation = -.732, p<.001), while the second dimension (boredom-excitement) appears to be positively related with pleasure (Pearson correlation = .578, p<.001). The scatterplot in figure 6-4 demonstrates the relation between the two distinct arousal dimensions, 'tension' and 'excitement' and the pleasure dimension. Note that the two arousal dimensions do not seem to be correlated at all (Pearson correlation = .001).³⁰

³⁰ These correlations were based on the in Lisrel 8.5 (Jöreskog and Sörbom, 1993; du Toit et al., 1999, p250) calculated latent variable scores of the purified factors (see further).

Pleasure		
	Arousal-1/Tension	
		Arousal-2/Excitement

Figure 6-4: Matrix scatterplot revealing the correlations between the two distinct arousal factors and pleasure

With regard to the dominance construct, also two factors seem to underlie the measurement instrument, instead of only one. By means of Principle Component Factor Analysis with Varimax Rotation, two factors are extracted with Eigenvalues exceeding unity, explaining together 53% of the variance (or respectively 27% and 26%). The Scree plot (presented in figure 6-5), representing the total variance associated with each factor (Cattell, 1966), also suggests a two-factor solution.



In fact, a one factor solution can explain only 33 % of the total variance. Moreover, in this case, three items appear to have factor-loadings below .50. Also for this construct the corrected item-total correlations were almost all unacceptably³¹ low (ranging from .19 to .51). The reliability is not high either (Cronbach's $\alpha = .58$).

With regard to the possible two-factor solution, all items could rather clearly be ascribed to one of the two factors. Each item had a minimum factor-loading of .55 and no loading higher than .40 on the other factor. The corrected item-total correlations were all unacceptably low (ranging from .20 to .47). The reliability of the respective dimensions also appears to be very low (Cronbach's $\alpha = .56$ and .42 respectively).

³¹ The cut-off level for this criterion, opted for by several researchers is .50 (Green, Tull and Albaum, 1988; Tabachnik and Fidell, 1989).

The factor scores of both the one- and two-factor solutions are presented in table 6-7.

Because of the low reliability of the two-factor solution and the lack of a theoretical framework to support such a multi-dimensional dominance construct, this two-factor solution will be disregarded and a one-factor solution will be pursued. Nevertheless, in that case, deleting item d2 appears to be called for, because of the low factor loading. This also slightly improves the reliability. Cronbach's α for the five remaining items is .60, which is, however, still rather low.

6.5.1.2. Uni-Dimensionality

A maximum likelihood confirmatory factor analysis using AMOS 3.62 (Arbuckle), conducted on each of the constructs separately indeed reveals that there appear to be two separate non-related arousal dimensions within the arousal construct. The pleasure construct on the other hand clearly is uni-dimensional. As far as the dominance construct is concerned, the one-dimensional solution performs best and appears to be highly reliable³².

In order to test the proposed theoretical structure of the measurement instrument, a maximum likelihood confirmatory factor analysis is conducted on the 17 remaining items, using AMOS 3.62 (Arbuckle, 1997). On the basis of the exploratory factor analyses four correlated underlying dimensions are assumed: pleasure, tension, excitement and dominance.

First of all, three items (a6, d3 and d5) obviously have to be eliminated because of extremely low standardized regression weights (<.40). This way a satisfactory overall fit of the model is

³² Correlating three pairs of error-terms (error d3–d6; error d4–d5 and error d3–d5) resulted in excellent goodness of fit scores (χ^2 =.22, df=2, p=0.896, χ^2 /df=.11) . Error-terms can be correlated when several observed variables, which are supposed to be indicators of one latent construct have some systematic error (i.e. they measure something in common, somewhat different from the latent construct they are supposed to measure). Detailed discussions on correlating errors are given by Cote, Netemeyer and Bentler in a recent methodological issue of the Journal of Consumer Psychology (2001, p87-89). In fact, it seems justified to correlate error-terms in this particular case, because we assume that the dominance 'state' emotion scale also captures some dominance 'trait', a personality characteristic (see Mehrabian, 1996). The standardized regression weights for the 5 remaining items (remember that item d2 was removed on the basis of the exploratory factor analysis) are all significant (critical ratio > 1.96).

achieved ($\chi^2=278.767$, df=71, $\chi^2/df=3.926$, p=0.000)³³. Nevertheless, because large standardized residuals³⁴ point to possible multidimensionality (Steenkamp and Van Trijp, 1991), these were carefully examined. 5 standardized residuals were detected exceeding the |2.58| norm. Dropping the corresponding items (p6 and a1) (Jöreskog and Sörbom, 1989), a very satisfactory four-factor model was obtained. Overall goodness of fit measures are presented in table 6-8. Acceptable values for a good fit (Marsh and Hovecar, 1985; Bentler, 1990, Bagozzi and Baumgartner, 1994; Sharma, 1996; Baumgartner and Homburg, 1996), are provided in the first column of this table. With this solution no standardized residuals exceed the |2.58| norm. Overall, the fit indices show a very good fit for the resulting four-factor model. The obtained model is shown in figure 6-6.

³³ It has to be added that the chi square test is highly dependent on the sample size, and that its hypothesis of an exact reproduction of the sample covariance matrix by the implied covariance matrix of the model is often considered to be overly rigid. Models that fit the data well, often have to be rejected on the basis of the chi square value (Bollen and Long, 1993; Bagozzi and Baumgartner, 1994).

³⁴ Standardized residuals are the residuals from the observed and reproduced covariance matrix divided by their asymptotic standard errors, and values exceeding |2.58| indicate misspecification (Jöreskog and Sörbom, 1988).



Figure 6-6: Path coefficients of the four-correlated factor PAD emotion construct in our sample³⁵

Goodness-of-fit indices	Acceptable values	4-correlated factor model	
u^2 (df)	Small	165 811 (48)	
χ, (dl)	Siliali	105.811, (48)	
p-value	>.05	0.000	
χ^2/df	<5	3.454	
Goodness-of-fit index (GFI)	>.90	0.965	
Adjusted Goodness-of-fit index (AGFI)	>.80	0.943	
Tucker and Lewis non-normed fit index (TLI)	>.90	0.957	
Benler's normed Comparative Fit Index (CFI)	>.90	0.969	
Root mean squared error of approximation (RMSEA)	<.08	0.056	
Standardized residuals > 2.58	No	No	
Factor regression coefficients > .50	Yes	10 out of 12	

 Table 6-8: Goodness of fit measures for the four correlated factor structure in the PAD scale (several items were dropped)

 $^{^{35}}$ The exact items referred to, can be retrieved on p 300 where the full scale is presented.

6.5.1.3. Within-Method Convergent Validity

According to Steenkamp and Van Trijp (1991, p289) 'within-method convergent validity' deals with 'the extent to which multiple applications of the same method are in agreement' (Steenkamp and Van Trijp, 1991, p289). They claim that this should be tested before reliability of the measurement instrument is assessed. Provided that the overall fit of the model is acceptable, they argue that a condition for within-method convergent validity is that the factor regression coefficients on a particular item are statistically significant (weak condition) and substantial (stronger condition). With respect to the latter criterion, it has been suggested by Hildebrandt (1987) that the correlation between the item and the construct should exceed .50. In case of the adapted four-factor PAD-emotion scale in this study, almost all of the factor regression coefficients exceed the .50 norm. However, the standardized regression weights of two dominance items score slightly lower. Nevertheless, all factor regression coefficients were highly significant (critical ratio's >11)³⁶.

6.5.1.4. Reliability

Reliability refers to 'the degree to which measures are free from random error' (Steenkamp and Van Trijp, 1991, p289) and should be estimated only provided that within-method convergent validity is achieved (Steenkamp and Van Trijp, 1991; see also Bagozzi, 1981b; Bagozzi et al., 1990). With regard to the reliability of the individual items, Bagozzi and Baumgartner (1994) suggest that the lower acceptable bound of the squared correlation between the item and the construct is 0.4. Two dominance items (d4 and d6) and one tension item (a4) score somewhat lower on this criterion. To assess the reliability of the respective dimensions, the composite reliabilities were ascertained, as well as the average variances extracted³⁷ (Bagozzi, 1980; Bagozzi and Yi, 1988; Fornell and Larcker, 1981; Steenkamp and van Trijp, 1991; Baumgartner and Homburg, 1996; Bagozzi and Baumgartner, 1994). The 'pleasure' factor proved to be very reliable with a composite reliability of .89 and an average variance extracted of .62. 'Tension' scores considerably lower, with a composite reliability of

³⁶ In order for a regression weight to reach statistical significance at the .05 level, its critical ratio should exceed [1.96] (see Arbuckle, 1997, p292).

³⁷ For a scale to possess good reliability, its composite reliability should be between 0.60 and 0.80 and the average variance extracted at least 0.50 (Bagozzi and Yi, 1988).

only .55. The average variance extracted reached only .38. Because only two items were retained to assess this factor, also Pearson's correlation coefficient is checked, which, nevertheless, appears to be highly significant (r=.368, p<.001). The composite reliability of 'excitement' reaches .62, with an average variance extracted of .45. We verified Pearson's correlation coefficient in this case too, as only two items are included here as well. This proved also to be highly significant (r=.452, p<.001). Finally the composite reliability for the 'dominance' construct amounts to .58, which is also rather low. The average variance extracted reached only .35. We can conclude that some random error is captured by our measures, in addition to what we intended to measure. Whereas the pleasure factor scores extremely well with regard to its reliability, the other factors score somewhat lower, but still almost meet the minimum criteria of .60 (Bagozzi and Yi, 1988), which is rather satisfactory considering the low number of items retained to capture the underlying constructs.

6.5.1.5. Discussion

Overall, the goodness of fit indices show a good fit for the resulting four-factor model. The uni-dimensionality, within-method convergent validity and reliability of the underlying pleasure, tension, excitement and dominance constructs prove to be adequate. However, we note that dominance does not score very well with regard to reliability and that the standardized regression weights of two dominance items do not score high either. Moreover, the 'dominance' dimension appears to be extremely highly correlated with the 'pleasure' dimension (r = .91). In order to test whether 'pleasure' and 'dominance' do not involve in fact one and the same dimension, we will investigate the discriminant validity of the dominance construct. According to Fornell and Larcker (1981) a scale possesses discriminant validity if the average variance extracted by the underlying construct is larger than the shared variance (i.e. the squared intercorrelation) with other latent constructs. Since the squared correlation among pleasure and dominance (.83) exceeds the variance extracted in both the pleasure and dominance scales (respectively .62 and .35), the discriminant validity of the dominance construct is rather questionable. An additional, less stringent, analysis of discriminant validity (Steenkamp and van Trijp, 1991) consists of examining whether applying an additional restriction of perfect correlation between the two dimensions does not result in a better fit. This can be examined by determining the difference between the χ^2 of the nested model (i.e. the more restricted model with the constraint of perfect correlation between

pleasure and dominance) and the χ^2 of the initial model (allowing for free correlation) and dividing this result by the difference in the degrees of freedom between the two models (Sharma, 1996). This test³⁸ [(180.56 – 165.81) / (49-48) = 14.75 > 3.84 ($\chi^2_{0.95}$, 1df)] reveals that the additional constraint of perfect correlation between the pleasure and dominance dimensions is not justified. The model assuming free correlation between the two factors fitted the data significantly better than the model constraining the correlation to one. Therefore, pleasure and dominance prove to be two highly correlated but separate factors³⁹.

Still, because of the low within-method convergent validity and reliability, the question remains whether or not the 'dominance' dimension should be dropped altogether. In line with Russell and Pratt's (1980) conceptualisation that the 'dominance' dimension is not applicable in environments calling for an affective response, we could consider ignoring the 'dominance' dimension, as several researchers have done in the past (e.g., Van Kenhove and Desrumaux, 1997, Sherman et al., 1997). However, several other researchers in the retailing context, have referred to the importance of the all too often ignored 'dominance' dimension (Gröppel-Klein, 1998; Foxall, 2000). Foxall (1997) proposes that the role of 'dominance' may depend on the type of consumer setting investigated. This may be responsible for the failure of past studies to find a role for 'dominance'. In the same vain, Turley and Milliman (2000) also note that the importance of the 'dominance' dimension may be context specific, depending on the independent store environmental variables investigated. Since the specific effects of colour on the 'dominance' dimension have been demonstrated (Valdez and Mehrabian, 1994), it seems appropriate to retain the construct. For these theoretical reasons and because the adapted 4factor model appears to have an adequate overall fit, it was decided to retain the 'dominance' dimension for further analyses. Although it has to be noted that results concerning the 'dominance' dimension should be interpreted with caution.

³⁸ Discriminant validity among these constructs was tested in Lisrel 8.5 (Jöreskog and Sörbom, 2001).

³⁹ This procedure for identifying discriminant validity was also followed by Babin, Darden and Babin, 1998.

6.5.2. CONSTRUCT VALIDATION OF THE APPROACH – AVOIDANCE SCALE

6.5.2.1. Exploratory Factor Analysis

The original approach–avoidance intentions scale, introduced to the study of environmental psychology by Mehrabian and Russell (1974), was adapted to the retailing context by Donovan and Rossiter (1982). Since there appears to be some inconsistency in the literature with regard to the conceptualisation of the approach-avoidance construct (see chapter 2), both exploratory and confirmatory factor analyses will be conducted on the 8-item retail-specific approach-avoidance scale proposed by Donovan and Rossiter (1982).

In order to explore the underlying factor structure of the eight items, measuring approachavoidance intentions, first a Principal Component Factor Analysis with Varimax rotation was performed. This way, two factors are extracted with Eigenvalues exceeding unity, explaining 67.57% of the variance (40.35% and 26.97% respectively for F1 (approach) and F2 (avoidance).

The Scree plot (presented in figure 6-7), representing the total variance associated with each factor (Cattell, 1966), also clearly suggests a two-factor solution.



Figure 6-7: Scree plot approach-avoidance construct

All items could clearly be ascribed to one of these two factors. Each item had a minimum factor-loading of .65 and no loading higher than .4 on the other factor. The corrected item-total correlations were all high, except for the reversed 'affiliation' item (Apav3) and the retail-specific 'money spending' item (Apav8), for which the item-total correlations amounted respectively to .43 and .49. Although the cut-off level for this criterion, opted for by several researchers (Green, Tull and Albaum, 1988; Tabachnik and Fidell, 1989) is .50, both items were provisionally retained. The factor scores are presented in table 6-9.

The resulting factors clearly reveal two separate approach and avoidance dimensions with high coefficient alphas, pointing to a high internal validity of the factors. Recently, Foxall (1990, 1997; Foxall and Greenley, 2000) also suggested that approach and avoidance make up two separate underlying dimensions of the approach-avoidance construct. Referring to Alhadeff (1982), Foxall argues that consumer behaviour in any given situation can be explained as the outcome of two opposing learning histories, with the strength of approach being a function of the individual's learning history with respect to prior approach behaviour and their consequences, while the strength of avoidance/escape constitutes a function of the individual's learning history escape and their consequences (Foxall, 1997, p514, Foxall and Greenley, 2000). Therefore, he resumes that approach and avoidance to approach. Moreover, pleasure, arousal and dominance have been found to exert independent influences upon approach and avoidance responses (Foxall, 1997; Foxall and Greenley, 2000).

Table 6-9: Principle Component Factor loadings with Varimax rotated solution

APPROACH-AVOIDANCE

		Approach	Avoidance
Items	Factorloading mean Labels	0.772 Cronbach's $\alpha = 0.874$	0.748 Cronbach's $\alpha = 0.707$
Apav 1	I like to spend much time browsing in this store (Time)	.843	
Apav4	This is a place in which I feel friendly and talkative to store personnel who happens to be near me (Affiliation)	.654	
Apav5	I like this store environment (Affect)	.792	
Apav6	I enjoy shopping in this store (Affect)	.854	
Apav8	This is the sort of place where I might end up spending more money than I originally set out to spend (Spending)	.716	
Apav2	I want to avoid looking around or explore this store (Time - Reversed)		.730
Apav3	This is a place where I try to avoid other people, and avoid to talk with them (Affiliation - Reversed)		.805
Apav7	I would avoid ever having to return to this store (Affect - Reversed)		.710
	Explained variance: 67.32%	40.35%	26.97%

6.5.2.2. Uni-Dimensionality

Because traditional approach–avoidance studies (Mehrabian and Russell, 1974; Russell and Mehrabian, 1976, 1978; Donovan and Rossiter, 1982; Bellizzi and Hite, 1992; Van Kenhove and Desrumaux, 1997, Matilla and Wirtz, 2001) have always considered approach and avoidance to be the opposite poles in a uni-dimensional approach-avoidance construct, we will submit the 8-item approach-avoidance scale to a maximum likelihood confirmatory factor analysis, performed in AMOS 3.62 (Arbuckle, 1997). This yielded however
unacceptable goodness of fit scores (χ^2 =332.528, df=20, p<.001, χ^2 /df=16.626). Five standardized residuals, with values exceeding |2.58|, were detected. Because large standardized residuals point to possible multidimensionality (Steenkamp and Van Trijp, 1991), these were carefully examined. The pattern of standardized residuals was informative for re-specification⁴⁰. Clearly three items could be identified, which are responsible for the unacceptable fit of the initial one-factor model. The originally reversed-scored items Apav3, Apav7 and Apav2 appear to constitute a separate 'avoidance' dimension, conform to the second factor extracted by the exploratory Principal Component factor analysis with Varimax rotation.

Thus, two dimensions appear to underlie approach and avoidance responses in our study. A separate approach dimension, containing 5 high loading items, and an avoidance dimension, consisting of 3 high loading items, could be revealed. In order to determine whether this two-factor solution is tenable, a maximum likelihood confirmatory factor analysis was undertaken on the re-specified model. This yielded however still unacceptable goodness of fit scores (χ^2 =134.834, df=19, p<.001, χ^2 /df=7.097). One standardized residual was detected exceeding |2.58|. Dropping an item (Apav4) pertaining to this 'offending' residual⁴¹ (Jöreskog and Sörbom, 1989) resulted in a satisfactory two-factor model. Overall goodness of fit measures are provided in table 6-10. With this solution no standardized residuals exceed |2.58|. Overall, the goodness of fit indices show a good fit for the resulting two-factor model. The obtained model is shown in figure 6-8.

⁴⁰ A subset of items could be detected with large positive residuals among each other (representing underfitting) and negative, although small, standardized residuals with most of the other items pertaining to the original factor, which may suggest that the subset constitutes a separate factor (see Steenkamp and van Trijp, 1991, p287).

⁴¹ If no clear pattern indicating misspecification emerges, as in this case, Steenkamp and van Trijp (1991) suggest it is best to delete the defective item.



Figure 6-8: Path coefficients in the two-correlated-factor model underlying the Approach-Avoidance construct in our sample

Goodness-of-fit indices	Acceptable values	2-correlated factor model
χ^2 , (df)	Small	61.348, (13)
p-value	>.05	0.000
χ^2/df	<5	4.719
Goodness-of-fit index (GFI)	>.90	0.978
Adjusted Goodness-of-fit index (AGFI)	>.80	0.953
Tucker and Lewis non-normed fit index (TLI)	>.90	0.971
Benler's normed Comparative Fit Index (CFI)	>.90	0.982
Root mean squared error of approximation (RMSEA)	<.08	0.069
Standardized residuals > 2.58	No	No
Factor regression coefficients > .50	Yes	5 out of 7

 Table 6-10:
 Goodness of fit measures for the two correlated factor structure of the 7-item approach-avoidance scale (item apav 4 was dropped)

6.5.2.3. Within-Method Convergent Validity

Considering that the overall fit of the model is satisfactory, we verify whether the factor regression coefficients are statistically significant and substantial (Steenkamp and van Trijp, 1991; Hildebrandt, 1987). In case of the two-dimensional approach-avoidance scale in this study, standardized regression weights appear to be high and almost all exceed the .50 norm. Only two standardized regression weights score slightly lower, but almost achieve the norm: .47 and .49. Moreover, all factor regression coefficients were highly significant (critical ratio's >11.45)⁴².

6.5.2.4. Reliability

As within-method convergent validity appears to be adequate, reliability can be ascertained (Steenkamp and Van Trijp, 1991; see also Bagozzi, 1981b; Bagozzi et al., 1990). As far as the individual items are concerned, one approach item (apav8) and one avoidance item (apav3) do not appear to be very reliable. The squared correlations, between these items and the constructs they are supposed to capture (respectively approach and avoidance), do not meet the minimum criterion of 0.4 proposed by Bagozzi and Baumgartner (1994). To assess the reliability of the respective dimensions, the composite reliabilities were ascertained, as well as the average variances extracted⁴³ (Bagozzi, 1980; Bagozzi and Yi, 1988; Fornell and Larcker, 1981; Steenkamp and van Trijp, 1991; Baumgartner and Homburg, 1996; Bagozzi and Baumgartner, 1994). The approach dimension proved to be very reliable with a composite reliability of .88 and an average variance extracted of .65. Also for the avoidance dimension an adequate composite reliability was obtained: .71. The average variance extracted reached .46. We can conclude that the approach/avoidance measure was highly reliable, with the composite reliabilities of the approach and avoidance factors both well exceeding the minimum criteria of .60 (Bagozzi and Yi, 1988). The average variance extracted of the avoidance construct was slightly lower than the .50 criterion proposed by Bagozzi and Yi (1988).

⁴² In order for a regression weight to reach statistical significance at the .05 level, its critical ratio should exceed |1.96| (see Arbuckle, 1997, p292).

⁴³ For a scale to possess good reliability, its composite reliability should be between 0.60 and 0.80 and the average variance extracted at least 0.50 (Bagozzi and Yi, 1988).

6.5.2.5. Discussion

Overall, the goodness of fit indices show a good fit for the resulting two-factor model. The uni-dimensionality, within-method convergent validity and reliability of the underlying approach and avoidance constructs prove to be satisfactory. However, the 'avoidance' dimension appears to be rather highly correlated with the 'approach' dimension (r = -.69). In order to test whether 'approach' and 'avoidance' do not involve in fact two opposite poles of one and the same dimension, as traditionally suggested, their discriminant validity was investigated (cfr. Babin, Darden and Babin, 1998). The squared correlation among approach and avoidance (.48) slightly exceeds the variance extracted in the avoidance scale (.46), but not the average variance extracted in the approach scale (.65)⁴⁴. An additional, less stringent, analysis of discriminant validity⁴⁵ (Steenkamp and van Trijp, 1991) [(267.02 - 61.35) / (14-13) = 205.67 > 3.84 ($\chi^2_{0.95}$, 1df)] reveals that the additional constraint of perfect correlation between the 'approach' and 'avoidance' dimensions is not justified. The model assuming free correlation between the two factors fitted the data significantly better than the model constraining the correlation to one. Therefore, 'approach' and 'avoidance' prove to be two correlated but separate factors, as suggested by Foxall (1990, 1997; Foxall and Greenley, 2000).

Discriminant validity of 'approach' and 'avoidance' was also examined with regard to the 'pleasure', 'tension', 'excitement' and 'dominance' constructs⁴⁶. Correlations between each of the constructs proved all to be significantly different from unity.

⁴⁴ A scale is presumed to possess discriminant validity if the average variance extracted by the underlying construct is larger than the shared variance (i.e. the squared intercorrelation) with other latent constructs (Fornell and Larcker, 1981).

⁴⁵ Examined was whether applying an additional restriction of perfect correlation between the two dimensions does not result in a better fit. This was done by determining the difference between the χ^2 of the nested model (i.e. the more restricted model with the constraint of perfect correlation between 'approach' and 'avoidance') and the χ^2 of the initial model (allowing for free correlation) and dividing this result by the difference in the degrees of freedom between the two models (Sharma, 1996). This test was performed in Lisrel 8.5 (Jöreskog and Sörbom, 2001).

⁴⁶ Discriminant validity among these constructs was also tested in Lisrel 8.5 (Jöreskog and Sörbom, 2001).

6.6.1. INTRODUCING MOOD AS A COVARIATE

It is clear that assessments of feeling states within a store environment will reflect feeling states brought to the environment, as well as those induced by the environment (Dawson et al., 1990; Swinyard, 1993; Mano, 1999). No doubt, emotions measured in the store will include some combination of both. Therefore Donovan et al. (1994) suggests that future studies should measure emotions prior to entering the store, as well as some time after entering the store. For this reason, we would like to take into account the pre-existing mood states of the participants in the current empirical study. We acknowledge that respondents' general pre-existing moods at the time of questioning, can significantly affect how they will respond to the store colour (Walters, Apter and Svebak, 1982; Ziems and Christman, 1998). Indeed, Ziems and Christman (1998) found mood to have an effect on affective reactions to colour. They induced participants' moods by having them listen to happy versus sad music and instructing them to think happy versus sad thoughts. Thus, induced mood was shown to affect the speed with which colours, differing on the arousal dimension, could be discriminated. Nevertheless, as consumers' moods are difficult to manipulate and as they are not the main focus of our study, we will try to control for them by including general mood as a covariate⁴⁷ in the analyses.

Pre-existing mood was measured by a short mood inventory (Peterson and Sauber, 1983) before exposure to the stimulus. The uni-dimensionality of the mood-scale was assessed conducting a maximum likelihood confirmatory factor analysis in AMOS 3.62 (Arbuckle, 1997). A satisfactory fit was obtained when the error-terms of the reversed-scored items are allowed to correlate⁴⁸ (χ^2 =0.401, df=1, χ^2 /df=0.401, p=0.527). One regression weight (mood

⁴⁷ Metric covariates are typically included in an experimental design to remove extraneous influences from the dependent variable, thus increasing the within-group variance (Hair et al., 1998, p346).

⁴⁸ It seems justified to correlate error-terms in this particular case, because we suspect a potential method effect. In fact, a major controversy actually concerns the bi-polarity of the mood construct (Babin et al., 1998). Several researchers have however imputed the independence of a positive and a negative mood dimension to an artifact of method (Russell and Carroll, 1999). As the scale to measure pre-existing mood, used in this study, was intended to be bi-polar (Peterson and Sauber, 1983), we assume it is justified to correlate the error terms of the originally reversed scored items. Nevertheless, we acknowledge that our decision may be subject to debate (detailed discussions on the bi-polarity controversy, which appears to be central to the psychology of affect, can be found in Russell and Carroll, 1999 and Babin et al., 1998). Detailed discussions on correlating

3) scored slightly lower (.39), but remained significant (critical ratio=9.025). As with this solution none of the standardised residuals exceed the |2.58| norm and a satisfactory composite reliability of .73 was obtained, it was decided to retain the four items of the original scale. Scores for the latent mood variable are computed in LISREL 8.5 and used as a covariate in the further analyses. Thus, by measuring the potentially confounding pre-existing mood variable, we can control statistically its effect on the dependent variables (Cook and Campbell, 1979).

6.6.2. Hypotheses Testing: Multivariate Analysis of Variance (MANCOVA)

In order to test a first set of hypotheses (H1-H9), concerning the emotions elicited by store colour, a three-way (8 x 2 x 2) multivariate analysis of variance⁴⁹ (MANOVA) was performed in SPSS 10, with pleasure, tension, excitement and dominance as the dependent variables and *hue* (eight categories: Purple-Blue, Blue, Blue-Green, Green, Green-Yellow, Yellow, Yellow-Red and Red), *saturation level* (high/low) and *value* (high/low) as fixed factors. Pre-existing mood was included in the analysis as a covariate (i.e. MANCOVA) (see Tacq, 1997, p346).

Thus, this analysis will test more specifically whether there are significant differences between the 'centroids' or vectors of multiple means of the emotions (i.e. pleasure, tension, excitement and dominance) elicited by (1) store colour hue, (2) store colour brightness and (3) store colour saturation.

But first let us take a look at the basic assumptions for this analysis: the dependent variables are required to have a multivariate normal distribution with the same variance-covariance matrix in each group.

errors are given by Cote, Netemeyer and Bentler in a recent methodological issue of the Journal of Consumer Psychology (2001, p87-89).

⁴⁹ Multivariate analysis of variance (MANOVA) is an extension of univariate analysis of variance (ANOVA), which can accommodate more than one dependent variable and is particularly useful when used in conjunction with an experimental design, in which the independent variable is manipulated in order to examine its effect on the dependent variables (Hair et al., 1998, p326-327). Although ANOVA tests can be computed separately for each of the dependent variables, this approach ignores the interrelation among them, and thus the possibility that some composite (linear combination) of the dependent variables may provide evidence of an overall group difference, that may not be detected otherwise. Moreover, conducting a series of separate ANOVA tests would fail to provide control of our effective overall experiment-wide Type 1 error rate (Hair et al., 1998, p339).

With regard to the normality assumption, we refer to section 6.3., where this supposition has already been examined. While it is apparent that multivariate normality is not achieved, it has been demonstrated that such a violation has little impact for multivariate analyses of variance when larger sample sizes are analysed (Hair et al., 1998, p349). Because we have a sufficiently large sample, this violation should not pose any problems for our analyses.

Secondly, the assumption of homoscedasticity (an equal dispersion in each cell) must be examined before starting with the analyses. Moreover, since we are dealing with dependent variables, which are mutually associated, the co-variations among them should not differ too much per cell either (Tacq, 1997, p354). Box's M test is used to examine this homogeneity of within-variations and co-variations. It tests the null hypothesis that the observed covariance matrices of the dependent variables are equal across groups. Box's M reaches a significance level of .049 for the eight hue-categories; for the two brightness levels Box's M attained a p-value of .794 and for the two saturation levels, eventually, a significance level of .683 is obtained. For saturation and brightness levels these p-values are well above the postulated α of .05. With regard to the hue categories, the significance level also almost reaches the required level⁵⁰. Because the observed covariance matrices of the dependent variables do not appear to be very different across the groups, we can accept the assumption of homoscedasticity.

As the violation of one of the basic assumptions (i.e. multivariate normality) is not expected to cause any real problems because of the sufficient sample size, we can go ahead and conduct the MANCOVA analysis. The results of this analysis are presented next.

Wilks' lambda⁵¹ reveals significant main effects of store colour hue (p=.028) and store colour value (p=.017) on the emotions elicited, but not of store colour saturation (p=.101). Nevertheless, a significant two-way (hue x saturation) interaction effect seems to exist (p=.026), as well as a significant three-way (hue x value x saturation) interaction effect (p=.031).

⁵⁰ Note that Box's M is very sensitive to departures from normality (Norusis, 1994, p72).

⁵¹ Wilks' lambda is a rather robust criterion to assess multivariate differences across groups, which is relatively immune to violations of the assumptions underlying MANOVA (especially for larger sample sizes) and yet maintains high power (Hair et al., 1998, p351).

As expected, the respondents' pre-existing mood (the covariate introduced to the model) also proved to have a highly significant effect on the emotions experienced after exposure to the stimulus (Wilks' lambda, p<.001), thus revealing clearly the necessity to control for this intervening variable. Although a participant's mood did not seem to significantly affect his/her perceived tension (Pearson's r = -.037, p=.357), mood did appear to be significantly related to experienced pleasure (Pearson's r = .100, p=.010), excitement (Pearson's r = .142, p<.001) and dominance (Pearson's r = .142, p<.001). As a consequence, in all subsequent analyses, respondents' pre-existing general state of mind or mood will be controlled for.

With regard to the power⁵² of the experiment, we assumed that the large sample size would be adequate to detect rather small true differences. Retrospectively, satisfactory average observed power-levels were obtained, in general well exceeding the suggested .80 norm⁵³.

In the following paragraphs we will take a closer look at the individual hypotheses with regard to the effects of store colour hue, saturation and brightness on experienced feelings of pleasure, tension, excitement and dominance. Table 6-11 gives a comprehensive overview of the findings.

⁵² The power of an experiment refers to the probability of rejecting the null hypothesis when it is false and depends on the magnitude of the true differences and the sample size (Norusis, 1994, p41; Cohen, 1977).

⁵³ With regard to the effects of colour saturation and the value x saturation interaction, smaller power-levels (.61 and .10 respectively) were observed. However, the latter probably only reflects the low actual effect-size (partial-Eta-squared for the value x saturation interaction amounted to a mere .001). For clear discussions on observed power and effect sizes, we refer to Trochim, 2002, Becker, 1999 and the following anonymous website: http://www.linguistics.ucla.edu/faciliti/facilities/statistics/power.htm).

	MAIN EFFECTS										INTERACTION EFFECTS								
	Colour Wavelength						Colour Value		Colour Saturation		Hue x Value	Hue x Saturation	Value x Saturation	Hue x Saturation x Value					
	РВ	В	BG	G	GY	Y	YR	R	р	Low	High	р	Low	High	р	р	р	р	р
EMOTIONS																			
Pleasure	5	1 ^a	6	4	8 ^a	2	3	7	.037	074	.075	.037	.034	033	ns	ns	ns	ns	ns
Tension	5	8 ^b	6	7	1 ^b	4	3	2	.047	.081	081	.024	076	.077	.033	ns	ns	ns	ns
Excitement	7	5	4	6	3	2	1	8	ns	026	.010	ns	010	006	ns	ns	.010	ns	.019
Dominance	6	1	4	5	8	3	2	7	ns	040	.034	ns	.035	040	ns	ns	ns	ns	ns
													Γ				[[
WILK'S LAMBDA									.028			.017			ns	ns	.026	ns	.031

Table 6-11: Main and interaction effects of store colour wavelength, saturation and value on elicited emotions

Results from a MANCOVA analysis, including pre-existing mood as a covariate (Wilk's lambda, p<.001). With regard to colour wavelength average rankings are given, from 1 (highest scoring) to 8 (lowest scoring). ^a Means scores are significantly different (post-hoc Bonferroni, p=.028) ^b Means scores are significantly different (post-hoc Bonferroni, p=.012)

6.6.3. STORE COLOUR-EVOKED PLEASURE

6.6.3.1. Store Colour Hue

H1: A negative U-shaped relationship was hypothesized between store-colour wavelength and store-elicited pleasure, with the stores with extreme long wavelength colours (red) expected to be judged less pleasant than the stores with extreme short wavelength colours (blue) and the mid-wavelength colours (green-yellow) at the bottom of the U, being judged least pleasant.

Since, with regard to elicited pleasure, independence⁵⁴ was reached in the factorial design among hue, value and saturation, the main effects could be interpreted directly (Hair et al., 1998, p344). Store-colour-evoked pleasure was in fact found to differ significantly among the different hue categories (p=.037). The average pleasure elicited by the various hues is represented in figure 6-9. The hues are arranged according to increasing wavelengths. In contrast to expectations, however, no clear pattern seems to emerge in the pleasure evoked by short to long wavelength hues. Post-hoc comparisons among the hues showed only the difference in elicited pleasure between blue and green-yellow to be significant (Bonferroni, p=.028). This difference was in the hypothesized direction, with blue being the store-colour eliciting most pleasure and green-yellow evoking least pleasure (respective standardized average pleasure ratings were .219 versus -.263). The extreme long wavelength-colour red also scored rather low with regard to elicited pleasure (the standardized pleasure rating for red averaged -.142). In contrast, other long-wavelength hues, yellow and yellow-red, were judged to be very pleasant. According to increasing pleasure, hues can be ranged as follows:

Green-Yellow / Red / Blue-Green / Purple-Blue / Green / Yellow-Red / Yellow / Blue.

Average pleasure-scores are given in table 6-12.

⁵⁴ Independence in factorial design means that the effect of one treatment is the same for each level of the other treatment(s) (Hair et al., 1998, p344).



Figure 6-9: Average pleasure elicited by the various store-colour-hues

Pleasure						
Hue	Hue Average score					
B Y YR G PB BG R GY	.219 .117 .070 .045 003 039 142 263					

Table 6-12: Average pleasure elicited by the various store-colour-hues⁵⁵

⁵⁵ Vertical lines indicate homogeneous sub-groups obtained post-hoc with Duncan's multiple-range test. Note that Duncan's multiple range test is less conservative than Bonferonni's post-hoc comparisons with respect to type I errors and has therefore more power, consequently identifying more group differences among the hues (Hair et al., 1998, p356).

6.6.3.2. Store Colour Brightness

H2: Store colour value is expected to have a positive effect on store-elicited pleasure, with higher value coloured or brighter stores expected to be judged more pleasant than the lower value coloured or darker stores.

As hypothesized, the value or brightness of the colour in the store interior does appear to have a significant impact on pleasurable feelings. Our findings indeed confirm that brighter coloured stores evoke more pleasant feelings than darker coloured stores (respective average pleasure scores of .075 versus -.074; p=.037), as demonstrated in figure 6-10.



346

Although this hypothesis generally appears to be true, a small (insignificant) interaction can be revealed between store colour hue and store-colour brightness, as demonstrated in figure 6-11. Whereas generally brighter coloured stores are rated more pleasant than darker coloured stores, the opposite seems to be true for yellow and for purple-blue. Note however, that this interaction effect was not significant.



Figure 6-11: Average pleasure elicited by light and dark tints of the various store-colour-hues

6.6.3.3. Store Colour Saturation

H3: Store colour saturation is expected to have a positive effect on store-elicited pleasure, with more saturated coloured stores expected to be judged more pleasant than the less saturated coloured stores.

From our findings, saturation appears, on the contrary, to have a slightly negative impact on feelings of pleasure (figure 6-12), with more saturated coloured store interiors being judged a little bit less pleasant (-.033 for highly saturated coloured stores versus .034 for less saturated coloured stores). Although the effect is not in the hypothesized direction, it is not significant either.



Figure 6-12: Average pleasure elicited by saturated and unsaturated store-colours

When we take a closer look at the matter (figure 6-13), we see that actually there is a huesaturation interaction, which is however not significant either. In general more saturated colours seem to elicit less intensive feelings of pleasure, although there are some exceptions. For green-yellow, for example, the opposite applies.



Figure 6-13: Average pleasure elicited by saturated and unsaturated tints of the various store-colour-hues

6.6.4. STORE COLOUR-EVOKED AROUSAL

6.6.4.1. Tension and Excitement: A Two-dimensional View of Arousal

As discussed in paragraph 6.5.1., the arousal construct was found to be composed of two separate un-correlated arousal dimensions. A first dimension (relaxation-tension) appeared to be negatively related with pleasure, while a second dimension (boredom-excitement) was found to be positively related with pleasure. Such a two-dimensional view of arousal has also been demonstrated and rationalized in previous research (Walters, Apter and Svebak, 1982; Thayer, 1986; Gorn et al., 1997). As no separate hypotheses have been formulated with regard to these distinct arousal dimensions, the general hypotheses on the arousal elicited by store colour hue, brightness and saturation were tested with regard to each arousal dimension. For each hypothesis we will first present the findings for the negatively valenced tension dimension. Subsequently the results with regard to the positively valenced excitement dimension will be discussed.

6.6.4.2. Store Colour Hue

H4: A positive U-shaped relationship is hypothesized between store-colour wavelength and store-induced arousal, with the stores with extreme long wavelength colours (red) expected to be judged more arousing than the stores with extreme short wavelength colours (blue) and the mid-wavelength colours (green-yellow) at the bottom of the U, being judged least arousing.

Tension

With regard to tension, independence was reached among hue, value and saturation in the factorial design. Therefore the main effects could be interpreted directly. In fact, among the different hue categories a significant difference could be revealed in store-colour-evoked tension (p=.047). The average tension elicited by the various hues is represented in figure 6-14.

The hues are arranged according to increasing wavelengths. In contrast to expectations, however, again no clear pattern seems to emerge in the tension evoked by short to long wavelength hues. Post-hoc comparisons among the hues showed only a significant difference in elicited tension between blue and green-yellow (Bonferroni, p=.012). This difference was, however, not in the hypothesized direction. Contrary to expectations, the blue-coloured store environment appeared to cause less tension than the green-yellow-coloured interior (standardized average tension scores were -.272 versus .240, respectively). The mid-wavelength colour green-yellow actually appeared to be the most tense colour (which is in accordance with findings by Valdez and Mehrabian, 1994), followed by red. In contrast, blue, green, blue-green and purple-blue seemed to be, as hypothesized, more relaxing colours. Yellow does not appear to cause tension either.



Figure 6-14: Average tension elicited by the various store-colour-hues

According to increasing tension, hues can be ranged as follows:

Blue / Green / Blue-Green / Purple-Blue / Yellow / Yellow-Red / Red / Green-Yellow.

Average tension-scores are given in table 6-13.

Tension							
Hue Average score							
GY R YR Y PB BG G B	.240 .106 .076 014 028 045 061 272						
2							

Table 6-13: Average tension elicited by the various store-colour-hues

Excitement

With regard to the excitement elicited by store colours, the interpretation of the results is more complex. Actually, in this case no independence was achieved between hue, brightness and saturation effects. Instead, complex interaction effects were found. None of the main effects appeared to be significant. Nevertheless, a plot of the mean excitement scores for store colour hue is presented here anyway (figure 6-15). However the reader should keep in mind that this is an oversimplification of the facts, since the interaction-effects are not represented here.



Figure 6-15: Average excitement elicited by the various store-colour-hues

According to increasing excitement, hues can be ranged as follows (note however that the differences among the hues are insignificant):

Red / Purple-Blue/ Green/ Blue / Blue-Green / Green-Yellow / Yellow / Yellow-Red

Average excitement-scores are given in table 6-14.

Excitement							
Hue	Average score						
YR	.202						
Y	.083						
GY	.036						
BG	019						
B	052						
G	092						
PB	107						
R	113						

Table 6-14: Average excitement elicited by the various store-colour-hues⁵⁶

Yellow-red turned out to be the most exciting colour. Ignoring the colours green and red, an upward trend can be revealed in elicited excitement, going from short to long-wavelength hues. Thus, with the exception of green and red, the warmer the store-colour, the more excitement it appears to elicit.

In figure 6-16 the joint effects of store-colour-hue on the two orthogonal dimensions of arousal, excitement and tension, are comprehensively illustrated. Note, for instance, for red that, although it is conceived as a rather tense colour, on the other hand it is perceived to be rather boring. Yellow-red, in contrast, appears to evoke excitement as well as tension. Blue is found to be very relaxing, but a little bit boring. Whereas yellow and purple-blue evoke approximately the same small amount of relaxation, yellow is still perceived as rather exciting, whereas purple-blue appears to elicit feelings of boredom.

⁵⁶ Vertical lines indicate homogeneous sub-groups obtained post-hoc with Duncan's less conservative multiplerange test.



Figure 6-16: Store-colour-hue evoked tension and excitement

In fact, in this diagram, the bottom-left to top-right diagonal can be considered as a main arousal axis, onto which, for each colour, a perpendicular line can be dropped, in order to determine the colour's general arousal score. In the same vain, the bottom-right to top-left diagonal approximates a valence or pleasure axis. Thus, disregarding the valence or feelings of pleasure involved, yellow and red can be conceived as approximately equally arousing. Nevertheless, the feelings of arousal elicited by a yellow-coloured store interior appear to be more pleasant than those evoked by a red-coloured retail environment.

6.6.4.3. Store Colour Brightness

H5: Store colour value is expected to have a negative effect on store-induced arousal, with higher value coloured or brighter stores expected to be judged less arousing than the lower value coloured or darker stores.

Tension

As hypothesized, the value or brightness of the colour in the store interior appears to have a significant impact on experienced tension. Our findings indeed confirm that brighter coloured stores evoke less tension than darker coloured stores (respective average tension scores of - .081 versus 0.81; p=.024), as demonstrated in figure 6-17.



Figure 6-17: Average tension elicited by dark and light store-colours

Although this hypothesis generally appears to hold, a small (insignificant) interaction can be revealed between store colour hue and store-colour brightness, as demonstrated in figure 6-18.



Figure 6-18: Average tension elicited by light and dark tints of the various store-colour-hues

Whereas generally darker coloured stores appear to be rated more tense than brighter coloured stores, the opposite seems to be true for purple-blue. With regard to yellow no difference can be found in elicited tension between bright and dark tones. The finding, that darker coloured store interiors are experienced as more tense than brighter coloured stores is apparently most pronounced for the colours green, yellow-red and red. For blue, blue-green and green-yellow, the difference in elicited tension among bright coloured and dark coloured store environments is much smaller. Note however, that this 'hue x value' interaction effect was not found to be significant.

Excitement

Although brighter coloured stores appear to elicit less tension (or more relaxation), they also appear to generate a little more excitement (figure 6-19), which was not hypothesized. However, this effect appeared to be insignificant.



Estimated Marginal Means of Excitement

Figure 6-19: Average excitement elicited by dark and light store-colours

Although the 'hue x saturation' interaction was also found to be insignificant, the plot in figure 6-20 reveals some interesting contrasts. Whereas, in general, brighter coloured stores seemed to elicit a little more excitement (remember that this effect turned out not to be significant), the opposite appeared to be true for yellow. Moreover, for blue, green and red, no such effect could be discovered.



Figure 6-20: Average excitement elicited by light and dark tints of the various store-colour-hues

In fact a complex 'hue x saturation x value' interaction effect appeared to be significant, as will be demonstrated later on.

Discussion

Thus far, the results obtained, regarding the effects of store colour value on elicited feelings of tension and excitement, are conform to the findings by Gorn et al. (1997) in an advertising context. Although their study was confined to only two hues (red and blue), they also found that brighter colours, when used as a peripheral cue in advertising, elicited more feelings of relaxation (and thus less tension) than darker colours on the one hand, but found, on the other hand, no effect of value on excitement. We found generally the same to be true for a wider range of colours applied in a store interior.

6.6.4.4. Store Colour Saturation

H6: Store colour saturation is expected to have a positive effect on store-induced arousal, with more saturated coloured stores expected to be judged more arousing than the less saturated coloured stores.

Tension

From our findings, saturation appears indeed to have a significant impact on experienced tension, with more saturated coloured store interiors being judged more tense (average experienced tension of .077 for highly saturated coloured stores versus -.076 for less saturated coloured stores, p=.033) as demonstrated in figure 6-21.



Estimated Marginal Means of Tension

Figure 6-21: Average tension elicited by saturated and unsaturated store-colours

However, when we take a closer look at the matter (figure 6-22), we see that actually there is a 'hue x saturation' interaction, which is, nevertheless, not significant. In general, more saturated colours seem to elicit more tension, although there are some exceptions. Greenyellow is again a notable exception, as well as blue-green and red. For blue, no effect of saturation is apparent, while for green, yellow and yellow-red, in contrast, it is very pronounced.



Figure 6-22: Average tension elicited by saturated and unsaturated tones of the various store-colour-hues

Excitement

With regard to store-colour-evoked excitement, none of the main effects were found to be significant. Instead, complex significant interaction effects could be revealed. In fact a significant 'hue x saturation' interaction-effect (p=.010), as well as a significant 'hue x saturation x value' interaction-effect (p=.019), appeared to be present. In figures 6-23, 6-24 and 6-25 these complex interactions are demonstrated. As in this case no general interpretation of the effects can be given, the effects for each hue will be discussed separately.







Figure 6-23: Average excitement elicited by saturated and unsaturated tones of the various store-colour-hues, respectively for dark tints (top) and for bright tints (bottom), demonstrating a complex 'hue x saturation x value' interaction.

Estimated Marginal Means of Excitement



Estimated Marginal Means of Excitement



Figure 6-24: Average excitement elicited by light and dark tints of the various store-colourhues, respectively for unsaturated tones (top) and for saturated tones (bottom), demonstrating a complex 'hue x saturation x value' interaction.



Figure 6-25: Comprehensive illustration of the complex 'hue x saturation x value' interactions.

Purple-Blue:

The darker and more saturated Purple-Blue coloured store appeared to elicit a little more boredom than the other Purple-Blue tints, which seemed to score pretty similar with regard to evoked excitement.

Blue:

With regard to Blue, there seems to be no difference in elicited excitement between the saturated and unsaturated tones when dark tints are involved. The unsaturated light coloured blue store interior elicited more excitement, whereas the saturated light coloured blue environment evoked less excitement than the dark coloured designs.

Blue-Green:

With regard to blue-green, there seems to be no difference in elicited excitement between saturated and unsaturated tones or dark and light tints.

Green:

For Green, low-saturated/low-value and high-saturated/high-value tones elicit the same amount of excitement. The same is true for low-saturated/high-value and high-saturated/low-value greens, which can both be considered as a little more boring.

Green-Yellow:

Highly saturated, bright (high value) Green-Yellow was found to elicit the most excitement of all the tints. The level of excitement elicited by this particular Green-Yellow tone was also considerably higher than for the other Green-Yellows, which were generally perceived as more boring.

Yellow:

With regard to Yellow, there seems to be no difference in elicited excitement between the dark and bright tints for the unsaturated tones. In contrast, the dark, highly saturated yellow store interior elicited more excitement, whereas the bright (high value), highly saturated yellow environment appeared to evoke less excitement than the unsaturated yellow coloured store designs.

Yellow-Red:

Dark, unsaturated yellow-red, applied to the store interior, appeared to be rather boring, while the following tones were able to elicit more excitement (in order of increasing evoked excitement): high-value/low-saturation yellow-red, high-value/high-saturation yellow-red and, ultimately, dark, highly saturated yellow-red, which elicited a considerable amount of excitement.

Red:

With regard to Red, there seems to be no difference in elicited excitement between the saturated and unsaturated tones when bright (high value) tints are concerned. The less saturated dark coloured red store interior elicited more excitement, whereas the more saturated dark coloured red environment evoked less excitement than the bright coloured red store designs.

From these hue-by-hue results, it should be clear that no generalizations can be drawn with regard to the excitement elicited by store-colour saturation and brightness.

Discussion

The results obtained, regarding the effects of store colour saturation on elicited feelings of tension and excitement, are not conform to the findings by Gorn et al. (1997) in an advertising context. For red and blue hues, used as a peripheral cue in advertising, they found that more saturation elicited more feelings of excitement, but not less relaxation. In contrast, in a retailing context, for the eight hues investigated, we found in general more saturation to cause more tension (and thus in fact less relaxation). Nevertheless, when we consider only blue and red hues, we cannot find a significant effect of colour saturation on elicited tension, as illustrated in figure 6-26. From the interaction results discussed earlier, we cannot say much in general with regard to the excitement elicited by store-colour saturation. In fact, when we consider only red and blue coloured store interiors, we could say that more saturation actually appears to evoke less excitement (see figure 6-26), which is exactly opposite to Gorn's (1997) findings.



Figure 6-26: Store-colour-saturation-evoked tension and excitement (for red and blue hues)

These results reveal the importance of including many colour-hues in experimental designs testing saturation and brightness effects. Otherwise very misleading results can be obtained. Generalizations on the basis of findings with regard to only two hues (which are common in colour research) are odious.

6.6.5. STORE COLOUR-EVOKED DOMINANCE

From the results of the MANCOVA analysis, we could not reveal any significant effects of store-colour hue, brightness nor saturation on feelings of dominance. Consequently, none of the hypothesized effects with regard to store-colour-evoked dominance could be supported. As our results turned out not to be significant, we will not pursue to present our findings with regard to the individual hypotheses on the specific effects of store colour hue, saturation and brightness on feelings of dominance in further detail.

In subsequent analyses, the latent dominance variable will be omitted, as it does not seem to be impacted significantly by store-colour⁵⁷.

⁵⁷ Note that the dominance construct did not turn out to be very reliably measured, as demonstrated in the paragraph on construct validation. Moreover, because of the high correlation (r = .91) with the pleasure construct, the discriminant validity of the dominance construct could also be questioned (Fornell and Larcker, 1981) and multicolliniarity generally decreases statistical efficiency (Hair et al., 1998).

6.6.6. MODERATING EFFECTS

In this section we will investigate whether there are any moderating effects of age and gender regarding the relationships between store-colour and the emotions elicited. Our aim is to determine, more specifically, whether age or gender affect these relationships or, in other words, whether store colours elicit different emotions in males than in females or in younger than in middle-aged or older consumers.

Before starting with the analyses, the assumption of homoscedasticity is examined. Box's M reaches a satisfactory significance level of .147 for the three age-categories, demonstrating homogeneity of within-variations and co-variations among the age groups. Nevertheless, for the gender groups a significance level of only .013 is obtained. Fortunately, a violation of the assumption of homoscedasticity has minimal impact if the groups are of approximately equal size⁵⁸ (Hair et al., 1998, p348). Moreover, Wilk's lambda is a rather robust criterion to assess multivariate differences across groups and is relatively immune for violations of the assumptions underlying MANOVA, especially for larger sample sizes (Hair et al., 1998, p351). Thus, although the observed covariance matrices of the gender groups appear to differ to some degree, we feel we can go ahead and proceed with the analyses conscientiously.

6.6.6.1. Testing for the moderating effect of gender

First, it was verified whether males and females react differently to store colour. To investigate whether there is such a moderating effect of gender on the emotions elicited by store colours, a MANOVA analysis was conducted on the full-factorial model, including the gender-factor. Wilks' lambda reveals neither a significant (p<.05) main nor significant interaction effects of gender on store-colour-evoked emotions in general. Nevertheless, the univariate statistics reveal a significant hue by gender interaction (p=.026) with respect to store-colour-evoked tension, suggesting that different store-colour-hues appear to elicit a different amount of tension in males than in females, which is demonstrated in figure 6-27.

⁵⁸ The sampling design was actually set up to attain equal sample sizes among age and gender groups. Nevertheless, a significant proportion of the male participants turned out to suffer to some degree from colour vision deficiency. Although this should not be surprising, it was not anticipated when setting up the sampling-quota for each condition. As the female group size (N=417) divided by the some smaller male group size (N=360) still amounts to only 1.16, which is smaller than the suggested 1.5 norm, we should not expect any problems in this regard.



Figure 6-27: Store-colour-hue-evoked tension for males and females

Whereas purple-blue, blue and red seem to elicit about the same amount of tension in males and females, for the other hues considerable differences can be discerned. In fact, applied to a store-environment, the hues blue-green and green-yellow appear to elicit more tension in females than in males. On the other hand, green, yellow and yellow-red seem to evoke a little more tension in males than in females. Average tension-scores for males and females are given in table 6-15. These results suggest that the gender of the target audience should not be ignored in store-colour decision-making.

Tension									
Averag	e scores	Male	Female						
GY	.240	YR .178	GY .534						
R	.106	G .095	BG .138						
YR	.076	Y .065	R .134						
Y	014	R .060	PB016						
PB	028	PB056	YR018						
BG	045	GY073	Y120						
G	061	BG211	G171						
B	272	B254	B302						

Table 6-15: Average tension elicited by the various store-colour-hues among males and females

6.6.6.2. Testing for the moderating effect of age

To determine whether store colours elicit different emotions in younger, middle-aged or older consumers, a MANOVA analysis is conducted on the full-factorial model, including the age factor. Wilks' lambda reveals no significant main or interaction effects of age on store-colour-evoked emotions either. Nevertheless, taking a look at the univariate statistics, we find that age does affect experienced excitement significantly, as demonstrated in figure 6-28.



Figure 6-28: Store-colour-evoked excitement among different age-groups

Younger respondents appear to experience more excitement than middle-aged and older respondents (respective excitement scores of .123 versus -.086 and -.088, p=.020). More specifically, we can identify a significant hue by age interaction with regard to elicited excitement (p=.038). Figure 6-29 clearly illustrates this rather complex interaction effect. Although, generally store-colours elicited higher amounts of excitement in younger respondents, the opposite appears to be true for blue, blue-green and green-yellow, which actually seem to elicit most excitement in the oldest age-category. Average excitement-scores are given per age-category in table 6-16. In fact, no obvious pattern seems to emerge from the obtained results, suggesting that for each hue different conclusions should be drawn. This points to the importance of considering the target-age-group when making store-colour decisions.



Figure 6-29: Store-colour-hue-evoked excitement among different age-groups

Excitement											
Average scores		18-	29	30-	44	45-46					
YR	.202	YR	.558	R	.144	GY	.257				
Y	.083	Y	.416	YR	.140	В	.129				
GY	.036	GY	.154	В	055	BG	.095				
BG	019	PB	.144	BG	090	Y	094				
В	052	G	.072	GY	138	G	151				
G	092	BG	057	Y	158	YR	199				
PB	107	R	080	G	205	PB	245				
R	113	В	218	PB	326	R	497				

Table 6-16: Average excitement elicited by the various store-colour-hues by age-group
6.7. STORE-COLOUR-EVOKED EMOTIONS AS DETERMINANTS OF APPROACH/AVOIDANCE BEHAVIOURS

6.7.1. HYPOTHESIS TESTING: STRUCTURAL EQUATION MODELING

Structural Equation Modelling was deemed ideally suited to analyse our data with regard to the next set of hypotheses, because this multivariate data analysis technique enables the researcher (1) to deal with multiple relationships simultaneously in a statistically efficient way and (2) to assess these relationships comprehensively, transitioning from exploratory to confirmatory analysis (Hair et al., 1998, p578). More specifically, we will be using the statistical package 'AMOS 3.62' (Arbuckle) to test the second part of relationships in the hypothesized model, proposed earlier.

Nevertheless, because of the rather high correlations among the pleasure construct on the one hand, and the arousal constructs (tension and excitement) on the other hand, the analyses will be performed in two parts. First the hypotheses will be tested with regard to the pleasure construct and subsequently the analyses will be performed with regard to the two arousal constructs (which are not correlated among themselves). As dominance did not appear to be significantly impacted by store colour, relationships regarding this dimension were not examined any further.

The inter-correlations among the latent constructs⁵⁹ are tabulated in table 6-17. Note that preexisting mood was controlled for in this and subsequent analyses according to procedures recommended by Draper and Smith (1980) and Green (1978). By regressing the mean composite mood scores on the composite as well as on the individual emotion and approachavoidance items and using the resulting standardized residuals as respective indicators, the effects of pre-existing mood could be completely partialled out.

⁵⁹ Intercorrelations among latent constructs (Φ matrix from confirmatory factor analysis) are corrected for attenuation due to measurement error. Latent construct scores were calculated in Lisrel 8.5 (Jöreskog and Sörbom, 1993; du Toit et al., 1999, p250).

	STORE-COLOUR-EVOKED EMOTIONS			BEHAVIOUR	
	Pleasure	Tension	Excitement	Approach	Avoidance
EMOTIONS Pleasure	1.000			.771 **** p=.000	644 *** p=.000
Tension	732 **** p=.000	1.000		391 *** p=.000	.376 *** p=.000
Excitement	.573 *** p=.000	n.s.	1.000	.431 *** p=.000	323 **** p=.000
BEHAVIOUR Approach	771***	301***	431***	1 000	
Арргоасп	p=.000	391 p=.000	.431 p=.000	1.000	
Avoidance	644 **** p=.000	.376 *** p=.000	323 **** p=.000	695***	1.000

Table 6-17 : Construct Correlations

Pearson's correlation coefficients.

Pre-existing mood was controlled for.

** Correlation is significant at the 0.01 level (2-tailed)

6.7.2. PLEASURE AS A DETERMINANT OF APPROACH/AVOIDANCE BEHAVIOURS

H10: Approach intentions towards a store are positively related with the pleasure elicited by the store.

H11: Avoidance intentions towards a store are negatively related with the pleasure elicited by the store.

Table 6-18 displays the overall fit statistics resulting from testing the pleasure-part of the model. Overall, the fit indices show that the model fits the data quite well. The individual standardized path estimates are shown in figure 6-30. The results suggest a significantly positive path between store-colour-evoked pleasure and approach intentions towards the store (.79) and a significantly negative path between elicited in-store pleasure and avoidance responses (-.68). Thus our findings provide strong support for the proposed hypotheses.



Figure 6-30: Empirical test of hypothesized model (pleasure)

Goodness-of-fit indices	Acceptable values	Model
χ^2 , (df)	Small	237.836, (52)
p-value	>.05	0.000
χ^2/df	<5	4.574
Goodness-of-fit index (GFI)	>.90	0.953
Adjusted Goodness-of-fit index (AGFI)	>.80	0.929
Tucker and Lewis non-normed fit index (TLI)	>.90	0.956
Benler's normed Comparative Fit Index (CFI)	>.90	0.966
Root mean squared error of approximation (RMSEA)	<.08	0.068
-		

Table 6-18: Goodness of fit measures for the hypothesized model (pleasure)

6.7.3. AROUSAL AS A DETERMINANT OF APPROACH/AVOIDANCE BEHAVIOURS

- H12: Approach intentions towards a store are positively related with the arousal induced by the store (for pleasant stores).
- H13: Avoidance intentions towards a store are not (or negatively) related with the arousal induced by the store.

As discussed in paragraph 6.5.1., in this study, two separate, un-correlated dimensions were found to underlie the arousal construct⁶⁰. The first dimension (relaxation-tension) appeared to be negatively related with pleasure, while the second dimension (boredom-excitement) was found to be positively related with pleasure. Nevertheless, as we had anticipated a uni-dimensional arousal construct, no separate hypotheses have been formulated with regard to these distinct arousal dimensions. Yet, when an environmental psychology approach is applied in a retailing context, sometimes a pleasure-arousal interaction effect is hypothesized, in the sense that for pleasant environments a positive relation is expected between store-elicited arousal and approach, whereas in unpleasant environments this relation is supposed to be negative. A test of this interaction effect can be accommodated in our hypotheses, when we expect the tension dimension to have inverse effects to the excitement dimension. Therefore, we propose that approach intentions towards a store are positively related with the excitement induced by the store, but negatively with the store-elicited tension. With regard to avoidance, we expect no or inverse effects.

Table 6-19 displays the overall fit statistics, resulting from testing this arousal-part of the model. Overall, the fit indices show that the model fits the data quite well. The individual standardized path estimates are shown in figure 6-31. The results suggest a significantly positive path between store-colour-evoked excitement and approach intentions towards the store (.47) and a significantly negative path between elicited tension and approach (-.69). With regard to avoidance, a negative effect was detected for excitement (-.38) and a positive one for tension (.72). Thus, our findings provide strong support for the proposed interaction effect. Not only with regard to approach intentions significant paths could be identified from

⁶⁰ Such a two-dimensional view of arousal has also been demonstrated and rationalized in previous research (Walters, Apter and Svebak, 1982; Thayer, 1986; Gorn et al., 1997).

store-elicited feelings of tension and excitement, but also for avoidance reverse effects could be revealed.



Figure 6-31: Empirical test of hypothesized model (arousal)

Goodness-of-fit indices	Acceptable values	Model
r^2 (df)	Small	164 287 (40)
χ, (u)	Siliali	104.287, (40)
p-value	>.05	0.000
χ^2/df	<5	4.107
Goodness-of-fit index (GFI)	>.90	0.962
Adjusted Goodness-of-fit index (AGFI)	>.80	0.938
Tucker and Lewis non-normed fit index (TLI)	>.90	0.946
Benler's normed Comparative Fit Index (CFI)	>.90	0.961
Root mean squared error of approximation (RMSEA)	<.08	0.063

 Table 6-19:
 Goodness of fit measures for the hypothesized model (arousal)

6.7.4. OVERVIEW

Our findings support the presumption that in-store induced emotions can induce approach and avoidance responses. Approach intentions towards the store appear to be positively affected by store-colour-evoked pleasure and excitement and negatively by feelings of tension. Store-colour-evoked tension, on the other hand, appears to induce avoidance responses, which also seem to increase when less pleasure and less excitement are experienced. An overview of all individual standardized path estimates is given in table 6-20. Note that the relatively high standardized regression weights may be due to some common-method variance.

Parameter Estimates: Total Sample				
	Path from:			
Pleasure	Tension	Excitement		
.79	69	.47		
68	.72	.38		
	Parame Pleasure .79 68	Parameter Estimates: TotalPath from:PleasureTension.796968.72		

Table 6-20: Store-colour-evoked emotions: parameter estimates

As the path –coefficients with regard to approach and avoidance intentions appear to be very similar, only pointing in opposite directions, critical ratio's for differences in both parameters were verified. As these proved to exceed the critical ratio of 1.96, (crdiff_{approach-avoidance}=19.801 for pleasure, 8.856 for tension and 8.554 for excitement), it is deemed worthwhile holding on to both related, but separate constructs. A comprehensive overview of the respective hypotheses tested and results obtained is rendered in table 6-21.

	Path	Hypothesis	Coefficient	Conclusion
H10	Pleasure \rightarrow Approach	+	.79	Supported
H11	Pleasure \rightarrow Avoidance	-	68	Supported
H12a	Tension \rightarrow Approach	-	69	Supported
H13a	Tension \rightarrow Avoidance	+	.72	Supported
H12b	Excitement \rightarrow Approach	+	.47	Supported
H13b	Excitement \rightarrow Avoidance	-	38	Supported

Table 6-21: Store-colour-evoked emotions: Summary of hypotheses and results

6.7.5. MODERATING EFFECTS

In this section we will investigate whether there are any moderating effects of age and gender. Our purpose is to determine, more specifically, whether the relationships found, between the emotions elicited by store-interior-colour and approach and avoidance responses towards the store, are affected by consumers' age and/or gender. This will be tested using the multigroupprocedure in AMOS 3.62 (Arbuckle, 1997).

Up until now, we have always reported standardized coefficients, because these closely approximate effect sizes, which are particularly useful for interpretation. Nevertheless, whereas standardized coefficients are useful for determining relative importance, they are sample specific and not comparable across samples (Hair et al., 1998, p614). As in the current analyses, we are not interested in the actual size of the parameters, but in group-comparisons, we will convert to the un-standardized solutions in this section.

6.7.5.1. Testing for the moderating effect of gender

The research question of interest at this point is whether a consumer's gender moderates the relationship between store-colour-evoked emotions and subsequent approach and avoidance intentions. Thus far, the path coefficients in the structural models have been calculated for both genders together. We now run the structural equation model for both groups separately, in order to be able to determine whether there are any significant differences in the path coefficients between both groups. Overall, the fit indices show that the models fit the data from the two groups quite well. Goodness of fit measures for both groups are presented in table 6-22 and the results are illustrated in figure 6-32. Table 6-23 gives the parameter estimates by gender.

We can subsequently compare a new model, in which all paths are constrained to be equal across male and female sub-samples, with the original model for which these constraints are set free (i.e. in this case the paths for males and females are allowed to vary). Thus, this new model is actually nested in the original one (i.e. it simply is a more restricted version of the initial model).





Figure 6-32: Path coefficients for males versus females (male/female)

Goodness-of-fit indices	Acceptable	Pleasure Model		Arousal Model	
	values	Males	Females	Males	Females
	~ 4				
χ^2 , (df)	Small	300.53	6, (104)	225.29	91, (80)
p-value	>.05	0.	000	.0	00
χ^2/df	<5	2.	890	2.8	816
Goodness-of-fit index (GFI)	>.90	.9	941	.9	50
Adjusted Goodness-of-fit index (AGFI)	>.80	.9	912	.9	17
Tucker and Lewis non-normed fit index (TLI)	>.90	.9	954	.9	38
Benler's normed Comparative Fit Index (CFI)	>.90	.9	964	.9	55
Root mean squared error of approximation	<.08	0.)49	.0	48
(RMSEA)					

Table 6-22: Goodness of fit measures for the hypothesized models for male and female sub-samples

Male N=360	Parameter Estimates by Gender			
	Path from:			
Path to:	Pleasure	Tension	Excitement	
Approach Avoidance	1.00 76	-1.72 1.31	.55 45	
Female				
N=417	Path from:			
Path to:	Pleasure	Tension	Excitement	
Approach Avoidance	.94 59	-1.32 1.19	.67 31	

Table 6-23: Unstandardized regression weights for gender

In order to test whether it is justified to hold on to this constrained model or whether we should rather use separate models for males and females, we have to compare both models, which can be done by determining the difference between the χ^2 of the nested model (i.e. the more restricted model with the constraint of equal paths for males and females) and the χ^2 of the initial model (allowing for discrepancies in path coefficients) and dividing this result by the difference in the degrees of freedom between the two models (Sharma, 1996).

With regard to the impact of store-colour-evoked pleasure, this test $[(303.476 - 300.536) / (102-104) = 2.94 < 5.99 (\chi^2_{0.95}, 2df)]$ reveals that the constraint of equal path coefficients for males and females is justified. The model assuming equal effects for gender did not fit the data significantly worse, than the model allowing for differences. Therefore, this simpler model is preferred and we can conclude that there are no significant differences for men and women in the relationships between store-colour-evoked pleasure and approach or avoidance intentions.

For the effect of store-colour-evoked arousal (i.e. tension and excitement) on approach and avoidance intentions, no significant difference can be detected between the sexes either $[(232.006 - 225.291) / (84-80) = 6.715 < 9.49 (\chi^2_{0.95}, 4df)].$

Therefore, we can conclude that no significant differences can be revealed between males and females in the relationships between store-colour-elicited emotions and subsequent approach and avoidance intentions.

6.7.5.2. Testing for a moderating effect of age

In order to examine whether the relationships between store-colour-evoked emotions and subsequent approach and avoidance intentions are moderated by the consumer's age, we run the structural equation models again, this time for the three age-categories (18-29, 30-44 and 45-60) separately.

Goodness-of-fit indices	Acceptable	Pleasure Model	Arousal Model
	values	Young Middle Old	Young Middle Old
\mathbf{x}^2 (df)	Small	415 626 (156)	253 670 (120)
p-value	>.05	.000	.000
χ^2/df	<5	2.664	2.114
Goodness-of-fit index (GFI)	>.90	.920	.945
Adjusted Goodness-of-fit index (AGFI)	>.80	.880	.909
Tucker and Lewis non-normed fit index (TLI)	>.90	.940	.943
Benler's normed Comparative Fit Index	>.90	.953	.959
(CFI)			
Root mean squared error of approximation	<.08	.046	.038
(RMSEA)			

Table 6-24: Goodness of fit measures for the hypothesized models for the different age sub-samples







Overall, the fit indices show that the models fit the data from the three age groups quite well. Goodness of fit measures for the three groups are presented in table 6-24 and the results are illustrated in figure 6-33. Table 6-25 gives the parameter estimates by age-group.

Younger (18-29) N=290	Parameter Estimates by Age group			
		Path from:		
Path to:	Pleasure	Tension	Excitement	
	06	2.428	(7	
Approach	.96	-2.43"	.6/	
Avoidance	69	1.97°	42	
Middle (30-44)				
N=256				
		Path from:		
Path to:	Pleasure	Tension	Excitement	
Approach	1.09	-1.36	.63	
Avoidance	64	1.08	28 ^c	
Older (45-60)				
N=229				
		Path from:		
Path to:	Pleasure	Tension	Excitement	
Annroach	86	87 ^a	60	
Approach	.00	0/ 76 ^b	.00 62 °	
Avoidance	03	./0	03	

^a Critical ratio for differences between parameters = 2.323

^b Critical ratio for differences between parameters = -2.163

^c Critical ratio for differences between parameters = -2.032

To determine whether there are any significant differences in the path coefficients between the age-groups, we can again compare the more restricted model, in which all paths are constrained to be equal across age-group sub-samples, with our initial model for which these constraints are set free (i.e. in this case the paths are allowed to vary between the age-groups).

With regard to the impact of store-colour-evoked pleasure, this test [(419.980 – 415.626) / (160-156) = $4.354 < 9.49 (\chi^2_{0.95}, 4df)$] reveals that the constraint of equal path coefficients for the three age-groups is also justified. Because the model assuming equal effects among the

Table 6-25: Unstandardized regression weights for age-groups

three age-groups, did not fit the data significantly worse than the model allowing for differences, this simpler model is preferred over the more complex one. We can conclude that there are no significant differences in the relationships between store-colour-evoked pleasure and approach or avoidance intentions among younger, middle-aged and older respondents.

For the effect of store-colour-evoked arousal (i.e. tension and excitement) on approach and avoidance intentions, no significant differences can be detected between the age-groups either $[(268.963 - 253.670) / (128-120) = 15.293 < 15.51 (\chi^2_{0.95}, 8df)]$. However, as in this case significance was almost reached, we also closely examined the individual parameter estimates. By a rough inspection, the regression paths appear relatively similar among the age categories, with the exception of the tension-approach and the tension-avoidance paths, which seem to differ considerably between young and older respondents. In fact, these paths appear to be stronger among the young sub-sample. When calculating the critical ratio's for differences between parameters, we discover in fact three parameters that differ significantly between particular subgroups. Besides the two already mentioned differences, the regressionpath between excitement and avoidance also appears to differ significantly among older and middle-aged respondents (-.63 versus -.28, critical ratio=-2.032). Comparing the fully constrained model, with another model in which these three (out of the twelve) path coefficients are released, suggests that the latter model is significantly better [(268.963 -253.670 / (128-125) = 9.552 > 7.81 ($\chi^2_{0.95}$, 3df)]. Thus, for older respondents (aged between 45-60), the relations between store-colour evoked tension and subsequent approach and avoidance intentions seem weaker than those for the younger age-group, while the relation between store-colour-evoked boredom (i.e. the opposite pole of excitement) and avoidance appears to be stronger, especially with regard to the middle-age-group.

Nevertheless, in general we can conclude that the differences, among the different age-groups, in the relationships between store-colour-elicited emotions and subsequent approach and avoidance intentions are not very significant.

6.8. TESTING MEDIATION

In the previous sections we demonstrated that store interior colour can affect feelings of pleasure, excitement, but also tension. Moreover, we showed that the emotions elicited by store colour are significantly related to approach and avoidance intentions towards the store. Tables 6-26 and 6-27 provide a comprehensive summary of these research findings:

	Path	Hypothesis	р	Conclusion
H1	Hue \rightarrow Pleasure	- U	.037*	Partially Supported [*]
H2	Brightness \rightarrow Pleasure	+	.037	Supported
Н3	Saturation \rightarrow Pleasure	+	n.s.	Not Supported
H4a	Hue \rightarrow Tension	+ U	.047*	Partially Supported*
H5a	Brightness \rightarrow Tension	-	.024	Supported
H6a	Saturation \rightarrow Tension	+	.033	Supported
H4b	Hue \rightarrow Excitement	+U	n.s.**	Partially Supported**
H5b	Brightness \rightarrow Excitement	-	n.s.**	Partially Supported ^{**}
H6b	Saturation \rightarrow Excitement	+	n.s.**	Partially Supported**
H7	Hue \rightarrow Dominance	U^{-1}	n.s.	Not Supported
H8	Brightness \rightarrow Dominance	-	n.s.	Not Supported
H9	Saturation \rightarrow Dominance	+	n.s.	Not Supported

* Significant hue effects could be discerned, yet, no apparent pattern seemed to emerge.

** Significant complex hue x saturation and hue x saturation x value interaction effects could be revealed (p=.01 and p=.019 respectively), although no clear pattern was apparent.

Table 6-26: Store-colour-evoked emotions: Summary of hypotheses and results

 Table 6-27:
 Effects of Store-colour-evoked emotions on approach/avoidance intentions:

 Summary of hypotheses and results

What we are interested in at this point is whether it is reasonable to assume that there is mediation. We want to establish whether store-colour-elicited emotions act as a 'mediator', representing the general mechanism through which store-colour (the focal independent variable) is able to influence approach/avoidance intentions towards the store (the dependent variable of interest) (see Baron and Kenny, 1986).

According to Baron and Kenny (1986), "*in general, a given variable may be said to function as a mediator to the extent that it accounts for the relation between the predictor and the criterion*". The procedure for testing mediation, outlined by Baron and Kenny (1986) will be followed next to determine whether there is mediation in our model. This procedure will be explained on the basis of the illustration in figure 6-34, which diagrams the basic causal chain involved in mediation In this model two causal paths lead to the outcome variable, one direct from the independent variable (path c) and one from the mediator (path b), which is in turn affected by the independent variable (path a).



Figure 6-34: Path-diagram of mediational model

To test for mediation, three regression equations should be estimated and separate coefficients for each equation should be estimated and tested (Baron and Kenny, 1986):

- (1) Path a: the mediator should be regressed on the independent variable
- (2) Path c: the dependent variable should be regressed on the independent variable
- (3) Path b and path c': the dependent variable should be regressed on both the independent variable and the mediator.

Mediation is established when:

- (1) Path a is significant, i.e. the independent variable has impact on the mediator in the first equation.
- (2) Path c is significant, i.e. the independent variable has impact on the outcome variable in the second equation.
- (3) Path b is significant, i.e. the mediator has impact on the outcome variable in the third equation.
- (4) If the first three conditions hold, path coefficient c' from the third equation should be less than path coefficient c from the second equation, or zero in case of perfect mediation.

Baron and Kenny (1986) suggest to use the multiple indicator approach and estimate mediation paths by latent variable structural modeling. By means of this technique all the relevant paths can be directly tested and measurement error is taken into account. They argue that although structural modeling techniques were developed for the analysis of non-experimental data, the experimental context actually strengthens their use.

Nevertheless, due to the nature of this study, the independent variable we are interested in, store colour, is composed of three dimensions: store colour hue, brightness and saturation, which were manipulated according to a three-way (8x2x2) factorial design. Thus, the colour variable cannot be incorporated in a testable mediational model. As a consequence, instead we chose to include a single item, capturing *'store-interior colour appreciation'*, which was included in the questionnaire in a series of questions pertaining to judgements of several aspects of the store interior. The item was measured on a 7-point Likert-type scale, with scores ranging from -3 "very ugly colours" to +3 "very pretty colours". The use of a single item is deemed acceptable in this case, since in mediator-oriented research, we are more interested in the mechanism than in the exogeneous variable itself.

In figures 6-35, 6-36 and 6-37 the direct and respective pleasure-, tension- and excitementmediated effects of store-colour-appreciation on approach-avoidance intentions are represented.





Figure 6-36: Direct and tension-mediated effects of store-colour-appreciation on approach/avoidance intentions

.58 / .51



Figure 6-37: Direct and excitement-mediated effects of store-colour-appreciation on approach/avoidance intentions

-.41 / .16 (n.s.)

A first investigation concerned whether the necessary conditions for mediation were fulfilled. This was checked respectively for pleasure, tension and excitement. To begin with, store-colour-appreciation was found to affect these potential mediating emotions significantly. Moreover, store-colour-appreciation also had a significant impact on both, approach and avoidance responses. Finally, the store-colour-evoked emotions of pleasure, tension and excitement were each found to be significantly related to approach and avoidance intentions, as was proved earlier. As these necessary conditions appear to hold, the path coefficients can be examined further.

From a rough inspection of these coefficients, it is obvious that the direct paths from storecolour-appreciation to approach and avoidance responses strongly diminish when respectively pleasure and excitement are controlled (compare the blue with the red coefficients). Running the mediated and un-mediated models together in AMOS, gives us the possibility to test whether this decrease in path coefficients is significant, which appeared to be true in both cases of pleasure and excitement (crdiff>5). The direct effects of store-colour-appreciation become insignificant when the emotions of pleasure and excitement are introduced in the model, which clearly illustrates their mediational role. Only in the case of pleasure, the direct colour-appreciation – approach path remained significant, but was nevertheless also strongly reduced. Here, no full mediation was reached, but a significant direct effect remains.

With regard to tension, we could not reveal a significant mediating effect, in that, although the path-coefficients seem to diminish slightly, the differences are not significant (crdiff<1.96). However, according to Lehman (2001, p91), treating mediation as a yes-no question leaves out a lot of information. He suggests a more continuous measure, which concentrates on the size of the effects. Lehman's (2001) simple index of mediation [(c-c')/c] amounts to [(.58-.51)/.58 = .12] for the colour-appreciation – approach relationship and [(.41-.29)/.41 = .29] for the colour-appreciation – avoidance relationship, indicating some small amount of mediation anyway.

Therefore, we can conclude that it is reasonable to assume that approach-avoidance responses to store-colours are, at least partially, mediated by the emotions they evoke. This is particularly true for store-colour-evoked feelings of pleasure and excitement.

Chapter 7

Discussion, Limitations and Implications



CHAPTER STRUCTURE

Chapter 7: Discussion, Limitations and Implications

- 7.1. Introduction
- 7.2. Overview of the Results
- 7.3. Implications
 - 7.3.1. Theoretical Implications
 - 7.3.2. Managerial Implications
 - 7.3.3. Consumer Implications
- 7.4. Limitations
- 7.5. Directions for Further Research

7.1. INTRODUCTION

In this final chapter the results of the current study and its limitations will be discussed and directions for further research will be suggested. First, an overview of the results is presented in section 7.2. Then, in section 7.3., the implications of the findings are assessed. In this section, theoretical, managerial as well as consumer implications are provided. Subsequently, section 7.4. outlines the limitations of this study and finally in section 7.5. suggestions for further research are proposed.

7.2. OVERVIEW OF THE RESULTS

After data screening and screening for colour blindness, responses of 777 participants were retained for further analyses.

Before testing the actual hypotheses, strong efforts were made to enhance the validity and reliability of the constructs included in this study. A careful validation of the underlying constructs of both the PAD-emotion scale and the approach-avoidance scale was performed, assessing conscientiously the uni-dimensionality, within-method convergent validity and reliability of the respective scales.

Although the six pleasure items, the six arousal items and the six dominance items were expected to indicate three underlying dimensions in the PAD emotion scale, the scale turned out to be composed of four underlying constructs. In fact two separate unrelated arousal dimensions emerged: 'tension', which appeared to be negatively related to pleasure, and 'excitement', which appeared to be positively related to pleasure. After the deletion of several items, maximum likelihood confirmatory factor analysis of the resulting four-factor model attained very satisfactory goodness-of-fit scores. Whereas the pleasure factor appeared to score extremely well with regard to its reliability, the other factors scored somewhat lower. Although dominance scored actually rather weak with regard to within-method convergent validity and reliability, the construct was originally retained, because the potential role of dominance has been noted to be context specific (Turley and Milliman, 2000) and colour has been demonstrated to affect this dimension (e.g. Valdez and Mehrabian, 1994; Osgood et al., 1957; Adams and Osgood, 1973; Sivik, 1974a). The pleasure, tension, excitement and dominance factors also appeared to be highly correlated, but still, discriminant validity among them was achieved. Mehrabian (1998, p6) has indicated high correlations between the respective PAD dimensions not to be unusual, when stimuli represent only "limited realms of affective experience", which apparently seems to be the case when only colour is manipulated (see also Valdez and Mehrabian, 1994).

Traditionally, in a retailing context approach and avoidance have always been considered to be the opposite poles of a uni-dimensional approach-avoidance construct (Donovan and Rossiter, 1982; Bellizzi and Hite, 1992; Van Kenhove and Desrumaux, 1997; Matilla and Wirtz, 2001). Nevertheless, in this study, both exploratory as well as confirmatory factor analyses clearly revealed two related but separate underlying approach and avoidance

dimensions, each demonstrating high internal validity. This finding is in line with Foxall's (1990, 1997, Foxall and Greenley, 2000) conceptualization that approach and avoidance make up two distinct dimensions, which can be explained as the outcome of two opposing learning histories, and can, therefore, not be represented psychometrically by a single continuum from escape/avoidance to approach. After the deletion of one 'affiliation' item, a satisfactory two-factor approach-avoidance model was obtained, with adequate goodness-of-fit scores.

The actual hypothesis testing was split up into two main parts. First of all, the hypotheses concerning store-colour-evoked emotions were tested by means of multivariate analysis of variance (MANCOVA). Here pre-existing mood was introduced as a covariate, to control for its potentially confounding effect, and the different emotions elicited by store-colour hue, brightness and saturation were examined in detail. A three-way full factorial multivariate analysis of variance, with mood as a covariate, revealed significant main effects for store-colour hue and value on the emotions elicited, but not for store-colour saturation. Nevertheless, a significant two-way (hue x saturation) as well as three-way (hue x value x saturation) interaction effect also emerged. The respondents' pre-existing mood (the covariate introduced to the model) also proved, as expected, to have a highly significant effect on the emotions experienced after exposure to the stimulus, thus revealing clearly the necessity to control for this intervening variable. We refer to paragraph 6.8. for a comprehensive table, summarizing the proposed hypotheses and obtained results. We present a short overview:

Pleasure

As expected (H1), store-colour-evoked pleasure was found to differ significantly among the different hue categories. In contrast to expectations however, no clear pattern seemed to emerge in the pleasure evoked by short to long wavelength hues. Post-hoc comparisons among the hues showed only the difference in elicited pleasure between blue and green-yellow to be significant, with blue eliciting the most pleasure and green-yellow the least. According to increasing pleasure, hues can be ranged as follows: Green-Yellow / Red / Blue-Green / Purple-Blue / Green / Yellow-Red / Yellow / Blue. The value or brightness of the colour in the store interior also appears to have a significant impact on pleasurable feelings, as hypothesized (H2), with brighter coloured stores evoking more pleasant feelings than darker coloured stores. Our expectation (H3), that colour saturation would also have a positive effect on store-elicited pleasure could, however, not be confirmed.

Tension

As hypothesized (H4a), significant differences could be revealed in store-colour-evoked tension among the different hue categories. Still, no clear pattern appeared to emerge in the tension elicited by short to long wavelength hues. Post-hoc comparisons among the hues showed only the difference in elicited tension between blue and green-yellow to be significant. The blue-coloured store environment appeared to cause less tension (i.e. more relaxation) than the green-yellow coloured interior. This finding was contrary to expectations, which were based on findings in a retail context by Crowley (1993), but is in line with findings by Valdez and Mehrabian (1994) concerning colour-patches. The mid-wavelength colour green-yellow actually appeared to be the most tense colour, followed by red. According to increasing tension, hues can be arranged as follows: Blue / Green / Blue-Green / Purple-Blue / Yellow / Yellow-Red / Red / Green-Yellow. The value or brightness of the store-interior, as hypothesized (H5a), also appears to have a significant impact on experienced tension. The obtained results confirm that brighter coloured stores, in general, evoke less tension than darker coloured stores. Our findings were also supportive of the expectation (H6a) that, in general, more saturated coloured stores are judged more tense (i.e. less relaxing) than less saturated coloured stores.

Excitement

With regard to the excitement elicited by store colours, the interpretation of the results is more complex. In this case no independence was achieved between hue, brightness and saturation effects, but instead, complex two-way (hue x saturation) and three-way (hue x value x saturation) interaction effects were found, which will not be discussed here in further detail. We refer to section 6.6.4. for a more detailed interpretation.

Dominance

From the results of the MANCOVA analysis, we could not reveal any significant effects for store-colour hue, brightness, or saturation on feelings of dominance. Consequently, none of the hypothesized effects with regard to store-colour-evoked dominance (H7-H8-H9) could be supported. For this reason, the dominance construct was omitted in subsequent analyses.

Testing for *the moderating effects of gender* on the emotions elicited by store-interior-colours revealed no significant main or interaction effects with regard to store-colour-evoked emotions in general. Nevertheless, the univariate statistics reveal a significant hue by gender

interaction with regard to store-colour-evoked tension, suggesting that different store-colourhues appear to elicit a different amount of tension in males and females. Applied to a storeenvironment, the hues blue-green, and green-yellow appear to elicit more tension in females than in males. On the other hand, green, yellow and yellow-red seem to evoke a little more tension in males than in females. These results suggest that the gender of the target audience should not be ignored in store-colour decision-making.

Assessing *the moderating effect of age*, we also could not reveal significant main or interaction effects of age-category with regard to store-colour-evoked emotions in general. Nevertheless, the univariate statistics reveal that younger respondents appear to experience more excitement than middle-aged and older respondents. We can also identify a significant 'hue by age' interaction with regard to elicited excitement. Although, generally, store colours elicited higher amounts of excitement in younger respondents, the opposite appears to be true for blue, blue-green and green-yellow, which actually seem to elicit most excitement in the oldest age category. However, no obvious pattern seems to emerge from the obtained results, suggesting that for each hue different conclusions should be drawn. This result points to the importance of considering the target-age-group when making store colour decisions.

The next set of hypotheses was tested by means of Structural Equation Modelling (SEM). Providing support for the proposed hypotheses (H10-H11), the results reveal a significant positive path between store-colour-evoked pleasure and approach intentions towards the store (.79) and a significant negative path between elicited in-store pleasure and avoidance responses (-.68). In light of the fact that two separate arousal factors, tension and excitement, emerged from our data, the original hypotheses concerning arousal were adapted to take on a two-dimensional view of arousal. This allowed us to integrate the presumed pleasure-arousal interaction in our hypotheses. As expected (H12a-H12b), the results suggest a significant positive path between store-colour-evoked excitement and approach intentions towards the store (.47) and a significant negative path between elicited tension and approach (-.69). With regard to avoidance, as hypothesized (H13a-H13b), a negative effect was detected for excitement (-.38) and a positive one for tension (.72). Thus, our findings provide strong support for the proposed interaction effect. Not only with regard to approach intentions, significant paths could be identified from store-elicited feelings of pleasure, tension and excitement, but also for avoidance reverse effects could be revealed. Thus our findings support the presumption that in-store induced emotions can induce approach and avoidance

responses. Approach intentions towards the store appear to be positively affected by storecolour-evoked pleasure and excitement and negatively by feelings of tension. Store-colourevoked tension, on the other hand, appears to induce avoidance responses, which also seem to increase when less pleasure and less excitement are experienced.

Although the path-coefficients with regard to approach and avoidance intentions appear to be very similar, only pointing at opposite directions, they were found to differ significantly from each other, emphasizing the importance of holding on to both related, but separate constructs.

The relationships between the emotions elicited by store-interior-colour and approach and avoidance responses towards the store, were in general not found to be significantly affected by a consumer's gender or age. Nevertheless, some minor discrepancies between the different age-categories could be revealed. Actually, for older respondents (aged between 45-60), the relations between store-colour evoked tension and subsequent approach and avoidance intentions seemed to be weaker than those for the younger age-group (aged between 18-30), while the relation between store-colour-evoked boredom (i.e. the opposite pole of excitement) and avoidance appeared to be stronger, especially with regard to the middle-age-group (aged between 30-45). However, in general, we can conclude that the differences among the different age-groups, in the relationships between store-colour-elicited emotions and subsequent approach and avoidance intentions, are not very significant.

In conclusion, store interior colour was demonstrated to affect feelings of pleasure and excitement, but also tension. Moreover, such store-colour-evoked emotions were found to be significantly related to approach and avoidance intentions towards the store. A comprehensive overview of these findings can be found in section 6.8. Furthermore, we established that it is reasonable to assume that store-colour-elicited emotions act, at least partially, as a 'mediator', representing the general mechanism through which store-colour is able to influence approach/avoidance intentions towards the store. This appeared to be particularly the case for store-colour-evoked feelings of pleasure and excitement.

7.3. IMPLICATIONS

7.3.1. THEORETICAL IMPLICATIONS

In this study we examined the impact of colour in the store environment on affective responses and subsequent approach and avoidance behaviours. As the prevalent research on the impact of store-interior colour is scarce and methodologically flawed, we contributed with our study to marketing scholars' understanding of this subject. Based on the literature, we first supplied a detailed review of general colour effects and of the effects of colour applied to the store environment. We demonstrated the shortcomings in these research streams and attempted to address them in our own empirical study.

In a methodological sense, this study distinguishes itself from other studies on the impact of colour in the store environment in several respects. First of all, in this study a full range of store-colours has been examined. A total of 32 different colours, have been included as stimuli (8 hues by 2 brightness levels by 2 saturation levels), in contrast to the existing studies, which have focused on a maximum of only four hues. Moreover, this study clearly illustrates the necessity of investigating the effects of several individual hues, as findings on the basis of two or four hues are by no means to be generalized to other warm or cold colours. Secondly, in the current study hue-effects have been examined while controlling for saturation and brightness levels. The few studies on the impact of store-interior-colour carried out so far, have mostly used fully saturated colours and have neglected to control for the potentially confounding colour attributes of brightness and saturation when investigating hue effects. Moreover, they have neglected to fully specify the utilized colour samples according to a standard colour system. Thirdly, besides hue effects, the specified research design, allowed us to examine the effects of store-colour-saturation and brightness, as well as interaction effects between the three colour dimensions. Our findings showed that saturation and brightness have a significant impact on colour-evoked emotions in their own right, while also significant interaction effects appeared to emerge between store-colour hue, brightness and saturation. These results clearly emphasize the importance of considering the three colour attributes together when studying environmental colour effects.

Although the effects of colour brightness and saturation on elicited emotions have been examined in an advertising context and with regard to plain colour patches, they have, thus far, never been examined in a retail context. However, this study clearly demonstrates the necessity to study colour in context, as previous findings on colour effects in other contexts, could not always be confirmed in a store context.

Finally, in this study special attention was devoted to construct validation of the PADemotion scale and the approach-avoidance scale.

Construct validation of the PAD-emotion scale revealed two separate 'tension' and 'excitement' dimensions within the arousal scale. These uncorrelated factors actually appeared very useful to examine the pleasure-arousal interaction effect on approach responses, as 'excitement' is positively related to pleasure, whereas 'tension' is negatively related to pleasure. Distinguishing these two separate arousal dimensions can possibly resolve some of the ambiguity with regard to the effects of arousal in a retail context. In line with previous empirical studies in a retailing context, dominance was found to be a rather weak construct, which appeared in this study highly correlated with pleasure. Although significant correlations among the PAD scales are not unusual when used to assess stimuli that represent 'limited realms of affective experience' (cfr. Valdez and Mehrabian, 1994; see Mehrabian, 1998), no significant differences could be detected in dominance responses elicited by different store colours. Therefore, this dimension appears to be a redundant factor, which is in line with the contention by Russell and Pratt (1980) and findings by Donovan and Rossiter (1982) and Greenland and McGoldrick (1994).

Concerning the approach-avoidance scale, the results obtained in this study through confirmatoy factor analysis provide strong empirical evidence for the existence of two separate approach and avoidance dimensions within the approach/avoidance construct (cfr. Foxall 1990, 1997, Foxall and Greenley, 2000), revealing that also in a retailing context these are two related but separate dimensions. As a matter of fact, the path coefficients with regard to the emotional determinants regarding approach and avoidance behaviours, have been proven to be not only reversed, but actually significantly different from each other. Thus, instore elicited pleasure, tension and excitement were found to exert independent influences upon approach and avoidance responses. This demonstrates the necessity to consider both behavioural intentions of approach as well as avoidance, as these are not evoked to the same extent by different emotions.

7.3.2. MANAGERIAL IMPLICATIONS

The recent finding that almost two thirds of all purchase decisions are made in the store (POPAI, 1998) has turned the attention to the point-of-purchase and to store atmospherics. As other marketing tools become neutralized in the competitive battle, 'atmospherics' play, as contended by Kotler (1973), a growing role in the unending search of firms for differential advantage. According to Bitner (1990), atmospheric planning can make the difference between a business success or failure. As the effect of the store environment on retailer performance has been demonstrated (Kumar and Karande, 2000), 'atmospherics' has become a highly relevant marketing instrument for retailers.

Particularly because store-interior-colour has been proven to be a potentially influential storedesign element, which can, in practice, easily be adapted at minimal costs, the booming retailing industry appears very interested in this specific research topic (Brengman, 2002; Retail-Update, 2002; POPAI-Benelux; Cahan, 2002). Knowing the effects of specific colours applied to the store environment can provide retailers with powerful guidelines for making store-interior colour decisions, enabling them to design shopping environments that induce specific emotions in shoppers, which enhance their purchase probability. We feel that this study can provide guidance in making appropriate store interior colour decisions. Based on insights gained from this study, store-managers can make better-informed decisions in the selection or avoidance of specific store colours.

Our results suggest to avoid dark and saturated colours, as they appear to evoke tension in shoppers, which, in turn, leads to avoidance behaviours. The application of light store interior colours can, on the other hand, be advocated because they seem to elicit pleasurable and relaxed feelings, which appear to prompt approach intentions, such as spending time browsing around in the store, enjoying shopping and spending more money than planned. With regard to particular hues, basic findings suggest that, in general, the use of blue, green, yellow and yellow-red in the store interior could elicit favourable effects, while green-yellow and red should clearly be avoided. However, a careful examination of the detailed results is recommended because of the complex 'hue x brightness x saturation' interaction effects.

Our findings also specifically suggest that the age and gender of the target store-audience should not be ignored in store-colour decision-making. In fact, a significant 'hue by gender' interaction with regard to store-colour-evoked tension could be revealed, suggesting that different store-colour-hues appear to elicit a different amount of tension in males and females. Our results also point to the importance of considering the target-age-group when making store colour decisions. Whereas, in general, younger respondents appear to experience more excitement than middle-aged and older respondents, a significant 'hue by age' interaction reveals that blue, blue-green and green-yellow actually seem to elicit most excitement in the oldest age category. Therefore in order to be able to select an appropriate store-colour, the main target's age and gender should be taken into account.

7.3.3. CONSUMER IMPLICATIONS

Also for consumers it is nicer to shop in a pleasant and attractive store environment. Nevertheless, it may be important for them to realize that the store itself can be a hidden persuader. Even when a consumer is not planning to buy anything, an attractive store atmophere may impel him/her to do so anyway. Although this may happen unconsciously, it is important for consumers to realize that they may be affected by such subtleties as store colour. At times when consumers cannot afford any excess spending, the only way to protect themselves from these temptations is by avoiding such attractive store environments.

7.4. LIMITATIONS

It has to be acknowledged that this study has some limitations. Several biases might have occurred in collecting our data and interpreting our results.

One important limitation to this study concerns the '*external validity*' of the findings. Unfortunately, as a direct consequence of the main concern for internal validity, laboratory experiments are generally rather weak in generalizability. Therefore, an extension of this study to field experiments is called for to determine the ecological validity and generalizability of our findings. Actually 'being' in an environment where the wall colours are manipulated, would probably evoke stronger emotional reactions than 'viewing a picture' in which wall colours are manipulated.

Furthermore, the findings of this study cannot be generalized to other hues and other value and saturation levels than those investigated. More empirical research is necessary to determine to what extent our findings can be generalized, with regard to other store colours. Also with regard to other store formats more research is called for. It is obvious that colours that evoke pleasure in a furniture store may differ from the ones that do so in a grocery store.

As this study is based on a brief exposure to the stimulus, only short-term affective responses to store-interior-colour were assessed. It would be interesting to assess emotional reactions over a longer time period, in order to determine whether some adaptation to the environmental colour occurs or not.

A potential source of bias in this study, may result from the fact that 'the external environment' of the respondent, while viewing the experimental picture, was not controlled for. The influence of lighting and other colours in the environment, for example, could represent potentially confounding factors. However, controlling for these factors would have called for a very stringent laboratory setting, which would have been very costly and might have introduced other biases of its own.
The '*omission of potentially important variables*' might have introduced another bias. We did, for example, not check for the respondents' involvement with the product category of 'design furniture and accessories', which might possibly have a moderating role in the relation between store-colour-evoked emotions and subsequent approach/avoidance responses.

Another limitation concerns the measurement of the 'arousal construct'. In this study, emotional responses to the store environment were measured by means of self-report questionnaires. Gröppel-Klein and Baun (2001) however, point out that verbal scales may not be appropriate to capture arousal. Construct validation of the PAD-emotion scale revealed two separate 'tension' and 'excitement' dimensions within the arousal scale. However, purification of the scale resulted in only two remaining items capturing tension and two other items measuring excitement. For future research, we recommend the use of better 'tension' and 'excitement' scales and suggest that perhaps also electrodermal activity could be measured as an arousal indicator (cfr. Gröppel-Klein and Baun, 2001). In retrospect, some verbal protocol procedures could also have been used to capture tension and excitement (cfr. Gorn et al., 1997). However, such procedures do not always provide usable responses (e.g. Geuens, 1998).

'*Common method variance*' may be another possible threat to validity. As we used one single questionnaire to measure all constructs included in the study, the strength of the relationships between these constructs may be somewhat inflated. This is related to the potential existence of *halo effects*, in that respondents might be worried to provide inconsistent answers. However, mixing the items and reversing some of them should have reduced this potential bias.

Another potential bias that may have confounded our findings might result from *'experimenter bias'*. However, intensive experimenter briefing and training were provided, prior to data collection, aiming to overcome this potential weakness. The use of a structured questionnaire, including detailed instructions for the respondents, automatically diminished this risk. Moreover, experimenters were not aware of the actual hypotheses investigated.

A myriad of unresolved research issues still needs to be addressed. Some of the limitations mentioned above, provide interesting directions for further research. A major avenue for such research encompasses the replication and extension of our study across different store types and involving different colour-samples. Such replications might shed light on the generalizability of our findings in order to establish external validity of our results. In the current study the impact of store colour was investigated with respect to a design furniture and accessories store. Future research should study the effects of interior colours for different store to what extent the obtained results can be generalized. Also more colour-samples should be tested. Future studies should especially focus on more hues: as no clear patterns seem to emerge with regard to the emotions elicited with respect to colour-wavelength, results are prone to be hue-specific. Also the impact of more value and saturation-levels should be examined.

Moreover, an extension of this study to field experiments is called for in order to determine the ecological validity and generalizability of our findings. As a matter of fact, POPAI Benelux, the Point-of-Purchase-Advertising-Institute has already shown a profound interest in this study and appears to be willing to carry this experiment one step further, into the actual store environment.

A cross-cultural replication could also be extremely valuable to determine the extent to which store-colour-effects are generalizable across cultures.

In this study the moderating effects of the demographic characteristics age and gender have been assessed. It may also be interesting to examine the moderating effects of personality variables such as arousal seeking tendency or optimum stimulation level.

Longitudinal research could examine the potential moderating influence of colour trends on store colour effects.

Another potential avenue for further research involves the possible interactions of the effects of store colour with the effects of other store characteristics such as music or crowding.

As Gorn et al. (1997) already indicated, there is little systematic empirical research in marketing on the effects of colour. Although some initial work has been carried out by Gorn et al. (1997) with regard to the use of colour in advertising, there are still a lot of colour-applications in a marketing context, which remain unexplored. For instance the effects of colour applied to packaging, coupons, in-store displays, shopping carts, salespersons' dress, web-sites, etc... all still have to be investigated.

In fact, a lot of further research is required in order to increase our understanding of the effects of colour in consumer behaviour...

References

- Aaronson, B., (1970). Some Affective Stereotypes of Color. International Journal of Symbology, vol. 2, n° 1, 15-27.
- Abramov, I., & Gordon, J., (1994). Color Appearance: on seeing Red or Yellow, or Green or Blue. Annual Revue of Psychology, vol. 45, 451-485.
- Adams, F. M., & Osgood, C. E., (1973). A Cross-Cultural Study of the Affective Meanings of Color. *Journal of Cross-Cultural Psychology*, vol. 4, 135-156.
- Agoston, G. A., (1979). Color Theory and its Application in Art and Design. New York: Springer-Verlag.
- Ajzen, I., & Fishbein, M., (1980). Understanding Attitudes and Predicting Social Behavior. Englewood Cliffs, NJ: Prentice-Hall.
- Alhadeff, D. A., (1982). *Microeconomics and Human Behaviour*. Los Angeles, CA: University of California Press.
- Ali, M. R., (1972). Clinical Note: Pattern of EEG Recovery under Photic Stimulation by Light of Different Colors. *Electroencephalography and Clinical Neurophysiology*, vol. 33, 332-335.
- Antill, J. K., (1987). Parents' Beliefs and Values about Sex Roles, Sex Differences, and Sexuality: Their Sources and Implications. In: P. Shaver & C. Hendrick (Eds.), *Review of Personality and Social Psychology*, vol. 7, 294-328. Newbury Park. CA: Sage.
- Apter, M. J., (1976). Some Data Inconsistent with the Optimal Arousal Theory of Motivation. *Perceptual and Motor Skills*, vol. 43, 1209-1210.
- Apter, M. J., (1979). Human Action and the Theory of Psychological Reversals. In: G. Underwood & R. Stevens (Eds.), *Aspects of Consciousness*, vol. 1. London: Academic Press.
- Apter, M. J., (1981). On the Concept of Bistability. International Journal of General Systems, vol. 6, 225-232.
- Apter, M. J., (1982). The Experience of Motivation: The Theory of Psychological Reversals. New York: Academic Press.
- Arbuckle, J. L., (1997). Amos Users' Guide, Version 3-6. U.S.: Small Waters Corporation.
- Areni, C. S., Sparks, J. R., & Dunne P., (1996). Assessing Consumers' Affective Responses to Retail Environments: A Table of Two Simulation Techniques. *Advances in Consumer Research*, vol. 23, 504-509.
- Argue, J., (1991). Color Counts. The Vancouver Sun, June 10th, B7.
- Armore, S. J., (1973). Elementary Statistics and Decision Making. Columbus, OH. Merrill.
- Aronson, E., Brewer, M., & Carlsmith, J. M., (1985). Experimentation in Social Psychology. In: G. Lindzey & E. Aronson (Eds.), *Handbook of Social Psychology*, vol. 1, 3rd ed. New York: Random House.
- Averill, J. R., (1973). Personal Control over Aversive Stimuli and Its Relationship to Stress. Psychological Bulletin, vol. 80 (4), 286-303.

- Babin, B. J., & Attaway, J. S., (2000). Atmospheric Affect as a Tool for Creating Value and Gaining Share of Customer. *Journal of Business Research*, vol. 49, 91-99.
- Babin, B. J., & Babin, L., (2001). Seeking Something Different? A Model of Schema Typicality, Consumer Affect, Purchase Intentions and Perceived Shopping Value. *Journal of Business Research*, vol. 54, 89-96.
- Babin, B. J., Darden W. R., & Griffin, M., (1994). Work and/or Fun: Measuring Hedonic and Utilitarian Shopping Value. *Journal of Consumer Research*, March, 644-657.
- Babin, B. J., & Darden. W. R., (1995). Consumer Self-Regulation in a Retail Environment. *Journal of Retailing*, vol. 71, n° 1, 47-70.
- Babin, B. J., Darden, R., & Babin, A., (1998). Negative Emotions in Marketing Research: Affect or Artifact? Journal of Business Research, vol. 42, 271-285.
- Baerden, W. O., & Netemeyer, R. G., (1999). Handbook of Marketing Scales: Multi-Item Measures for Marketing and Consumer Behavior Research, 2nd ed. Thousand Oaks, California: Association for Consumer Research Series, Safe Publications.
- Bagozzi, R. P., (1980). Causal Models in Marketing. New York: Wiley.
- Bagozzi, R. P., (1981b). An Examination of the Validity of Two Models of Attitude. *Multivariate Behavioral Research*, vol. 16, 323-359.
- Bagozzi, R. P., & Baumgartner, H. (1994). The Evaluation of Structural Equation Models and Hypothesis Testing. In: Bagozzi, R.P. (Ed.), *Principles of Marketing Research*, 386-422. Cambridge, MA: Blackwell.
- Bagozzi, R.P., Gopinath, M., & Nyer, P. U., (1999). The Role of Emotions in Marketing. *Journal of the Academy of Marketing Science*, vol. 27, n° 2, 184-206.
- Bagozzi, R. P., & Moore, D. J., (1994). Public Service Advertisements: Emotions and Empathy Guide Prosocial Behavior. *Journal of Marketing*, vol. 58, January, 56-70.
- Bagozzi, R. P., & Yi, Y., (1988). On the Evaluation of Structural Equation Models, *Journal of the Academy of Marketing Science*, vol. 16, Spring, 7-94.
- Bagozzi, R. P., & Yi, Y., (1990). Assessing Method Variance in Multitrait-Multimethod Matrices: The Case of Self-reported Effect and Perceptions at Work. *Journal of Applied Psychology*, vol. 75, 547-560.
- Baker, J., (1986). The Role of Environment in Marketing Services: The Consumer Perspective. In: J.A. Cepeil et al. (Eds.) *The Services Challenge: Integrating for Competitive Advantage*, 79-84. Chicago, IL: American Marketing Association.
- Baker, J., Grewal, D., & Parasuraman, A., (1994). The Influence of Store Environment on Quality Inferences and Store Image. *Journal of the Academy of Marketing Science*, vol. 22, (Fall), 328-339.
- Baker, J., Levy, M., & Grewal, D., (1993). An Experimental Approach to Making Retail Store Environmental Decisions. *Journal of Retailing*, (Winter 1992), vol. 68, n° 4, 445-460.
- Ball, V. K., (1965). The Aesthetics of Colour: A Review of Fifty Years of Experimentation. *Journal of Aesthetic Art Criticism*, vol. 23, 441-452.
- Baron, R. M., & Kenny, D. A., (1986). The Moderator-Mediator Variable Distinction in Social Psychological Research Conceptual, Strategic, and Statistical Considerations. *Journal of Personality and Social Psychology*, vol. 51, n° 6, 1173-1182.

- Bateson, J. E. G., & Hui, M. K. M., (1987). A Model of Crowding in the Service Experience: Empirical Findings. In: J. Czepiel, C. Congram, & J. Shanahan (Eds.), *The Services Challenge: Integrating for Competitive Advantage*, 85-89. Chicago, IL.: American Marketing Association.
- Bateson, J. E. G., & Hui, M. K. M., (1992). The Ecological Validity of Photographic Slides and Videotapes in Simulating the Service Setting. *Journal of Consumer Research*, vol. 19, September, 271-281.
- Batra, R., & Holbrook, M. B., (1990). Developing a Typology of Affective Responses to Advertising. *Psychology & Marketing*, vol. 7, Spring, 11-25.
- Batra, R., & Ray, M. L., (1986). Affective Responses Mediating Acceptance of Advertising. *Journal of Consumer Research*, vol. 13, September, 234-249.
- Baumgartner, H., & Homburg, C., (1996). Applications of Structural Equation Modelling in Marketing and Consumer Research: a Review. *International Journal of Research in Marketing*, vol. 13, 139-161.
- Beach L. R., Wise B. K. & Wise J. A., (1988). *The Human Factors of Color in Environmental Design : A Critical Review*. Nasa Report n° NASA CR-177498 Washington, DC: National Aeronautics and Space Administration.
- Beard, J., (1995). Color for the Color Blind. Popular Science, vol. 247 nº 1, 44.
- Becker, L. A., (1999). *Measures of Effect Size* (*Strength of Association*). Available from: <u>http://web.uccs.edu/lbecker/SPSS/glm_effectsize.htm</u>.
- Bellizzi, J. A., Crowley, A. E., & Hasty, R. W., (1983). The Effects of Color in Store Design. *Journal of Retailing*, vol. 59, n° 1, 21-45.
- Bellizzi, J. A., & Hite, R. E., (1992). Environmental Color, Consumer Feelings, and Purchase Likelihood. *Psychology and Marketing*, vol. 9, 347-363.
- Benjamin, J., Press, J., Maoz, B., & Belmaker, R. H., (1993). Linkage of a Normal Personality Trait to the Color-Blindness Gene: Preliminary Evidence. *Biological Psychiatry*, vol. 34, 581-583.
- Bennet, C. P., Haque, A., & Perkins, C., (1991). The Use of Baker-Miller Pink in Police Operational and University Experimental Situations in Britain. *International Journal of Biosocial and Medical Research*, vol. 13, n° 1, 118-128.
- Bentler, P. M., (1990). Comparative Fit Indexes in Structural Models. *Psychological Bulletin*, vol. 107 n° 2, 238-246.
- Bentler, P. M., (2001). Structural Equations Modelling: VI.C. Improving Model Fit by Correlating Errors. In: D. Iacobucci (Ed.), *Journal of Consumer Psychology*, Special Issue: Methodological and Statistical Concerns of the Experimental Behavioral Researcher, vol. 10, (1-2), 87-89.
- Berlin, B., & Kay, P., (1969). *Basic Color Terms, their Universality Evolution*. Berkeley, CA: University of California Press.
- Berlyne, D. E., (1960). Conflict, Arousal and Curiosity. New York: McGrow-Hill.
- Berlyne, D. E., (1967). Arousal and Reinforcement. In: D. Levine, (Ed.), *Nebraska Symposium on Motivation*, 1-110. Lincoln, Nebraska: University of Nebraska Press.
- Berman, B., & Evans, J. R., (2001). *Retail Management: A Strategic Approach*, 8th ed. Upper Saddle River, NJ: Prentice Hall.
- Billmeyer, F. W. Jr., (1987). Survey of Color Order Systems, Color Research and Application, vol. 12, 173-186.

- Birren, F., (1952). The Emotional Significance of Color Preference. *American Journal of Occupational Therapy*, vol. 79, 61-63.
- Birren, F., (1978). Color and Human Response. New York: Van Nostrand Reinhold.
- Birren, F., (1983). Color and Human Response. Color Research and Application, vol. 8 n° 2, 75-81.
- Bitner, M. J., (1990). Evaluating Service Encounters: The Effects of Physical Surroundings and Employee Responses. *Journal of Marketing*, vol. 54, April, 69-82.
- Bitner, M. J., (1992). Servicescapes: The Impact of Psychical Surroundings on Customers and Employees. Journal of Marketing, vol. 56, n° 2, 57-70.
- Bjerstedt, A., (1960). Warm-Cool Color Preferences as Potential Personality Indicators: Preliminary Note. *Perceptual and Motor Skills*, vol. 10, February, 31-34.
- Boedeker, M., (1995). Optimum Stimulation Level and Recreational Shopping Tendency. *European Advances in Consumer Research*, vol. 2, 372-380.
- Bollen, K. A., & Long, J. S., (1993). Testing Structural Equation Models. Newbury Park: Sage Publications.
- Boomsma, A., (1982). The Robustness of LISREL against Small Samples in Factor Analysis Models. In: Jöreskog, K. G., & Wold, H. (Eds.), Systems under Indirect Observation, Part I, 149-173. Amsterdam: North Holland.
- Bost, E., (1987). Ladenatmosphäre und Konsumentenverhalten, Heidelberg.
- Boyatzis, C. J., & Varghese, R. (1994). Children's Emotional Associations with Colors. *The Journal of Genetic Psychology*, vol. 155, n° 1, 77-85.
- Boynton, R. M., (1979). Human Color Vision. New York: Holt, Rinehart and Winston.
- Braus, P., (1995). Vision in an Aging America. American Demographics, vol. 17, n° 6, 34-38.
- Brengman, M., (2002). De Invloed van Kleur in de Winkelomgeving op de Consument (*The Influence of Colour in the Store Environment on the Consumer*), Presentation at the Trade Conference: Winnen op het Winkelpunt... (*Winning at the Point-of-Sale*), organized by the Institute for International Research, Dorint Hotel, Brussels, May, 23-24.
- Brislin, R. W., (1980). Translation and Content Analysis of Oral and Written Material. In: H.C. Triandis and J.W. Bercy (Eds.), *Handbook of Cross-Cultural Psychology*, vol. 2, 389-444. Boston: Allyn and Bacon.
- Brown, P. K., & Wald, G., (1964). Visual Pigment in Single Rods and Cones of the Human Retina. *Science*, vol. 144, 45-52.
- Buehler, S., (1996). Monitor testing: Subjective, Objective, Comprehensive. PC Magazine Online, July 8.
- Bullough, E., (1908). The Perceptive Problem in the Aesthetic Appreciation of Single Colours. *British Journal of Psychology*, vol. 2, 406-463.
- Bush, L. E. II., (1973). Individual Differences Multidimensional Scaling of Adjectives Denoting Feelings. Journal of Personality and Social Psychology, vol. 25, 50-57.

Byrnes, D. A., (1983). Color Associations of Children. The Journal of Psychology, vol. 113, 247-250.

- Cahan, L., (2002). *Inspiration: The Power of Color*. Presentation at the VM+SD, International Retail Design Conference, September 18-20, Passadina, California.
- Caldwell, J. A., & Jones, G. E., (1985). The Effects of Exposure to Red and Blue Light on Physiological Indices and Time Estimation. *Perception*, vol. 14, 19-29.
- Calkins, D. J., Tsukamoto, Y. & Sterling, P., (1998). Micro-circuitry and Mosaic of a Blue-Yellow Ganglion Cell in the Primate Retina. *Journal of Neuroscience*, vol. 18, 3373-3385.
- Campbell, D. T., (1963). Social Attitudes and Other Acquired Behavioral Dispositions. In: S. Koch (Ed.), *Psychology: A Study of a Science*, vol. 6, 94-172. New York: McGraw-Hill.
- Campbell, D. T., & Fiske, D. W., (1959). Convergent and Discrimant Validation by the Multimethod Matrix. *Psychological Bulletin*, vol. 56, 81-105.
- Campo, K., (1993). Product-Related Differences in Variety Seeking Behavior. 2nd EMAC Proceedings. Barcelona, Spain, vol. 1, 311-330.
- Carver, C. S., Sutton, S. K., & Scheier, M. F., (2000). Action, Emotion, and Personality: Emerging Conceptual Integration. *Personality and Social Psychology Bulletin*, vol. 26, n° 6, June, 741-751.
- Cattell, R. B., (1966). The Scree Test for the Number of Factors. *Multivariate Behavioral Research*, vol. 1, April, 245-276.
- Cattell, R. B., (1966). The Meaning and Strategic Use of Factor Analysis. In: R.B. Cattall, (Ed.), *Handbook of Multivariate Experimental Psychology*. Chicago: Rand McNally.
- Chebat, J.-C., & Dubé, L., (2000). Evolution and Challenges Facing Retail Atmospherics: The Apprentice Sorcerer is Dying. *Journal of Business Research*, vol. 49, 89-90.
- Chebat, J.-C., Filiatrault, P., Gelinas-Ghebat, P., & Vaninsky, A., (1995). Impact of Waiting Attribution and Consumer's Mood on Perceived Quality. *Journal of Business Research*, vol. 34, 191-196.
- Chebat, J.-C., Gelinas-Chebat, C., Vaninsky, A., & Filiatrault, P., (1995). The Impact of Mood on Time Perception, Memorization, and Acceptance of Waiting. *Genetic, Social, and General Psychological Monographs*, vol. 121, 413-424.
- Child, I. L., Hansen, J. A., & Hornbeck F. W., (1968). Age and Sex Difference in Children's Color Preferences. *Child Development*, vol. 39, 237-247.
- Choungourian, A., (1969). Color Preferences: A Cross-cultural and Cross-sectional Study. *Perception and Motor Skills*, vol. 28, 801-803.
- Clynes, M., & Kohn, M., (1968). Recognition of Visual Stimuli from the Electric Response of the Brain. In: N.S. Kline & F. Laska (Eds.). *Computers and Electronic Devices in Psychiatry*. New York: Grune & Stratton.
- Cohen, J. B., (1977). Statistical Power Analysis for the Behavioral Sciences. New York: Academic Press.
- Cohen, J. B., (1990). Attitude, Affect and Consumer Behavior. In: B. S. Moore & A. M. Isen, (Eds.), Affect and Social Behavior, 152-206. New York: Cambridge University Press.
- Cohn, J., (1894). Experimentelle Undersuchungen über die Gefühlsbetonung der Farben. Helligkeiten und ihrer Combinationen (Experimental Investigation of the Emotional Effect on Color, Lightness and their Combination). *Philosophische Studiën*, vol. 10, 562-603.

Colorsystem.com, (2002), http://www.colorsystem.com.

Consumer Reports, (1976). Bread: You Can't Judge a Loaf by its Color., vol. 41, May, 256-260.

- Cook, R.D., (1977). Detection of Influential Observations in Linear Regression. Technometrics, vol. 19, 15-18.
- Cook, T. D., & Campbell, D. T., (1979). Quasi-experimentation: *Design and Analysis Issues for Field Settings*. Chicago: Rand McNally.
- Corsino, B. V., (1985). Color Blindness and Rorschach Color Responsivity. *Journal of Personality Assessment*, vol. 49, 533-534.
- Costa, P. T., Jr., & McCrae, R. R., (1992). Revised NEO Personality Inventory and NEO Five-factor Inventory Professional Manual. Odessa, FL: Psychological Assessment Resources.
- Costa, P. T., Jr., & McCrae, R. R., (1995). Domains and Facets: Hierarchical Personality Assessment Using the Revised NEO Personality Inventory. *Journal of Personality Assessment*, vol. 64, 21-50.
- Cote, J., (2001). Structural Equations Modelling: VI.C. Improving Model Fit by Correlating Errors, In: D. Iacobucci (Ed.), *Journal of Consumer Psychology*, Special Issue: Methodological and Statistical Concerns of the Experimental Behavioral Researcher, vol. 10, (1-2), 87-89.
- Craik, K. H., (1970a). Environmental Psychology. In: K. H. Craik, et al., (Eds.), *New Directions in Psychology*, vol. 4, 1-121. New York: Holt Rinehart & Winston.
- Craik, K. H., (1970b). The Environmental Dispositions of Environmental Decision-Makers, Annals of the American Academy of Political and Social Sciences, vol. 389, May, 87-94.
- Craik, K. H., (1976). The Personality Research Paradigm in Environmental Psychology. In: S. Wapner, S. B. Cohen, & B. Kaplan, (Eds.), *Experiencing the Environment*, 55-79. New York: Plenum Press.
- Craik, K. H., & McKechnie, G. E., (1977). Editor's Introduction: Personality and the Environment. *Environment and Behavior*, vol. 9, June, 155-168.
- Crowley, A. E., (1993). The Two-Dimensional Impact of Color on Shopping. *Marketing Letters*, vol. 4, n° 1, 59-69.

D

- Dacey, D. M., (1996). Circuitry for Color Coding in the Primate Retina. *Proceedings of National Academy of Science*, vol. 93, 582-588.
- Damhorst, M. L., & Reed, J. A. P., (1986). Clothing Color Value and Facial Expression: Effects on Evaluations of Female Job Applicants. Social Behavior and Personality, vol. 14, 89-98.
- Darden, W. R., & Babin, B. J., (1994). Exploring the Concept of Affective Quality: Expanding the Concept of Retail Personality. *Journal of Business Research*, vol. 29, February, 101-110.
- Darley, J. M., & Gilbert, D. T., (1985). Social Psychological Aspects of Environmental Psychology. In: G. Lindzey & E. Aronson, (Eds.), *Handbook of Social Psychology*, 3rd ed., vol. II, 949-991. New York: Random House, Inc.
- Dashiell, J. F., (1917). Children's Sense of Harmonies in Colors and Tones. Journal of Experimental Psychology, vol. 11, 466-475.

- D'Astous, A., (2000). Irritating Aspects of the Shopping Environment. Journal of Business Research, vol. 49, 149-156.
- Dawson, S., Bloch, P. H., & Ridgway N. M., (1990). Shopping Motives, Emotional States and Retail Outcomes. Journal of Retailing, vol. 66, Winter, 408-427.
- Dellaert, B. G. C., Arentze, T. A., Bierlaire, M., Borgers, A. W. J., & Timmermans, H. I. P., (1998). Investigating Consumers' Tendency to Combine Multiple Shopping Purposes and Destinations. *Journal of Marketing Research*, vol. 35, May, 177-188.
- Dember, W. N., & Earl, R. W., (1957). Analysis of Exploratory, Manipulatory and Curiosity Behaviors. *Psychological Review*, vol. 64, 91-96.
- DeMonasterio, F. M., Gouras, P., & Tolhurst, D. J., (1975). Trichromatic Colour Opponency in Ganglion Cells of the Rhesus Monkey Retina. *Journal of Physiology*, vol. 251, 197-216.
- Dériberé, M., (1968). La Couleur dans les Activités Humaines. Paris: Dunod, 3e ed.
- DeValois, R. L., Abramov, I., & Jacobs, G. H., (1966 and 1974). Analysis of Response Patterns of LGN Cells. *Journal of the Optical Society of America.*, vol. 56, 966-977.
- Dholakia, R. R., (1999). Going Shopping: Key Determinants of Shopping Behaviors and Motivations. International Journal of Retail & Distribution Management, vol. 27, 154-165.
- Diener, E., (1999). Introduction to Special Section on the Structure of Emotion. *Journal of Personality and Social Psychology*, vol. 76, 803-804.
- Diener, E., Sandvik, E., & Larsen, R. J., (1985). Age and Sex Effects for Emotional Intensity. *Developmental Psychology*, vol. 21, 542-546.
- Donovan, R. J., & Rossiter J. R., (1982). Store Atmosphere: An Environmental Psychology Approach. *Journal* of *Retailing*, vol. 58, Spring, 34-57.
- Donovan, R. J., Rossister, J. R., Marcoolyn, G., Nesdale, A., (1994). Store Atmosphere and Purchasing Behavior. *Journal of Retailing*, vol. 70, n° 3, 283-294.
- Dorcus, R. M., (1926). Color Preference and Color Association. Pedagogical Seminary, vol. 33, 432.
- Draper, N. R., & Smith, H., (1980). Applied Regression Analysis. New York: Wiley.
- Dubé, L., & Morin, S., (2001). Background Music Pleasure and Store Evaluation Intensity Effects and Psychological Mechanisms. *Journal of Business Research*, vol. 54, 107-113.
- Du Toit, S., du Toit, M., Jöreskog, K., & Sörbom, D., (1999). Interactive Lisrel, User's Guide, SSI Scientific Software International.

Ε

- Edell, J. A., & Burke, M. C., (1987). The Power of Feelings in Understanding Advertising Effects. *Journal of Consumer Research*, vol. 14, December, 421-433.
- Edell, J. A., & Staelin, R., (1983). The Information Processing of Pictures in Print Advertisements. *Journal of Consumer Research*, vol. 10, 45-61.
- Ellis, H., (1900). The Psychology of Red, Popular Science, vol. 57, 356-375.

Ellis, H., (1906). The Psychology of Yellow. Popular Science, vol. 68, 456-463.

- Eroglu, S. A., & Machleit, K. A., & Davis, L. M., (2000). Online Retail Atmospherics: Empirical Tests of a Cue Typology. In: Evans J. R. and Berman B. (Eds.), *Retailing 2000: Launching the New Millennium*, *Proceedings of the 6th Triennial National Retailing Conference*, vol. 9, 144-150. Hofstra University, Hemstead, NY: Academy of Marketing Science and the American Collegiate Retailing Association.
- Eroglu, S. A., & Machleit, K. A., (1990). An Empirical Study of Retail Crowding: Antecedents and Consequences. *Journal of Retailing*, vol. 66 (2), 201-221.
- Eroglu, S. A., Machleit, K. A., & Davis, L. M., (2001). Atmospheric Qualities of Online Retailing: A Conceptual Model and Implications. *Journal of Business Research, Special Issue on Retail Strategy and Consumer Decision Research*, vol. 54, 177-184.
- Eroglu, S. A., Scholder P. E. & Machleit K. A., (1992). Environmental Cues in Retailing: Suggestions for a Research Agenda. Proceedings of the 1991 Symposium on Patronage Behavior and Retail Strategy, 51-60. Cutting Edge II.
- Erwin, C. W., Lerner, M., Wilson, N. J., & Wilson, W. P., (1961). Observations on the Photically Elicited Arousal Response. *Electroencephalography Clinical Neurophysiology*, vol. 13, 391-394.
- Etnier, J. L., & Hardy, C. J., (1997). The Effects of Environmental Color. *Journal of Sport Behavior*, vol. 3, August, p 299, 14, 1 chart. 5 graphs.
- Evans, D. R., & Day, H. I., (1971). The factorial Structure of Responses to Perceptual Complexity. *Psychonomic Science*, vol. 22, 357-359.
- Everett, P. B., Pieters, R. G. M., & Titus P. A., (1994). The Consumer-Environment Interaction. *International Journal of Research in Marketing*, vol. 11, 97-105.
- Eysenck, H. J., (1941). A Critical and Experimental Study of Color Preferences. American Journal of Psychology, vol. 54, 385-391.

\mathbf{F}

- Farrell, R. J. & Booth, J. M., (1984). Design Handbook for Imagery Interpretation Equipment. Boeing, February.
- Fishbein, M. M., & Ajzen, I., (1975). Belief, Attitude, Intention, and Behavior: An introduction to Theory and Research. Reading, MA: Addison-Wesley.
- Fisher, J. D., (1974). Situation-Specific Variables as Determinants of Perceived Environmental Aesthetic Quality and Perceived Crowdedness, *Journal of Research in Personality*, vol. 8, August, 177-188.
- Fiske, D. W., & Maddi, S. R., (1961). Functions of Varied Experience. Homewood, Illinois: Dorsey.
- Flicker, M. H., & Speer, W. C., (1990). Emotional Responses to Store Layout and Design: An Experimental Approach. In: Parasuraman et al. (Eds.). AMA Educators' Proceedings: Enhancing Knowledge Development in Marketing (1-5). Chicago, IL: American Marketing Association.
- Fonberg, E. A., (1986). Emotions, Motivation and Depressive States in Emotion. In: R. Plutchik et al., (Eds.), *Research and Experience*, 300-365. New York: Kluwer Press.
- Fornell, C., & Larcker, F. D., (1981). Evaluating Structural Equation Models with Unobservable Variables and Measurement Errors. *Journal of Marketing Research*, vol. 18, 39-50.

- Foxall, G. R., (1990). *Consumer Psychology in Behavioural Perspective*. London and New York, NY: Routledge /International Thomson Press.
- Foxall, G. R., (1992). The Behavioral Perspective Model of Purchase and Consumption: From Consumer Theory to Marketing Practice. *Journal of the Academy of Marketing Science*, vol. 20, Summer, 189-198.
- Foxall, G. R., (1993). Situated Consumer Behavioural Interpretation of Purchase and Consumption, *Research in Consumer Behaviour*, vol. 6, 113-152.
- Foxall, G. R., (1997). Affective Responses to Consumer Situations. *The International Review of Retail, Distribution and Consumer Research*, vol. 7, Issue 3, July, 192-225.
- Foxall, G. R., (1999). The Behavioural Perspective Model: Consensibility and Consensuality. *European Journal* of Marketing, vol. 33, (5-6), 570-596.
- Foxall, G. R., (2000). The Contextual Stance in Consumer Research. *European Journal of Marketing*, vol. 34, n° 7, 768-779.
- Foxall, G. R., & Greenley, G. E., (1999). Consumers' Emotional Responses to Service Environments. Journal of Business Research, vol. 46, 149-158.
- Foxall, G. R., & Greenley, G. E., (2000). Predicting and Explaining Responses to Consumer Environments: An Empirical Test and Theoretical Extension of the Behavioural Perspective Model. *The Service Industries Journal*, vol. 20, n° 2, April, 39-63.
- Frank, O. M. G., & Gilovich, T., (1988). The Dark Side of Self- and Social Perception: Black Uniforms and Aggression in Professional Sports. *Journal of Personality and Social Psychology*, vol. 54, 74-85.
- Frijda, N. H., (1987). The Emotions. Cambridge: University Press.
- Frijda, N. H., (1993). Moods, Emotion Episodes, and Emotions. In: M. Lewis & J. M. Haviland, (Eds.), Handbook of Emotions, 381-403, New York: Guilford.
- Frijda, N. H., Kuipers, P., & ter Schure, E., (1989). Relations Among Emotion, Appraisal, and Emotional Action Readiness. *Journal of Personality and Social Psychology*, vol. 57, 212-228.
- Fujita, F., Diener, E., & Sandvic, E., (1991). Gender Differences in Negative Affect and Well-being: The Case for Emotional Intensity. *Journal of Personality and Social Psychology*, vol. 61, 427-434.
- Funder, D. C., (1991). Global Traits: A Neo-Allportian Approach to Personality. *Psychological Science*, vol. 2, 31-39.
- Funderburk, L., (2001). A Palette of Colorful Expression, 2/03/2001. Available at: <u>http://www.insteam.com/LauraFunderburk/mycolor.htm</u>.

G

- Gale, A. V., (1933). Children's Preferences for Colors, Color Combinations, and Color Arrangements. Chicago: Univ. Chicago Press.
- Gardner, M. P., (1985). Mood States and Consumer Behavior: A Critical Review. Journal of Consumer Research, vol. 12, December, 281-300.
- Garrett, J. C., & Brooks, C. I., (1987). Effect of Ballot Color, Sex of Candidate, and Sex of College Students of Voting Age on their Voting Behavior. *Psychological Reports*, vol. 60, 39-44.

- Gelineau, E. P., (1981). A Psychometric Approach to the Measurement of Color Preference. *Perceptual and Motor Skills*, vol. 53, 163-174.
- Gerard, R. M., (1957-1958). Differential Effects of Colored Lights on Psychophysiological Functions. Unpublished Ph.D. Dissertation, UCLA.
- Gerard, R. M., (1958). Color and Emotional Arousal. The American Psychologist, July, 340.
- Gerbing, D. W., & Anderson, J. C., (1985). The Effects of Sampling Error and Model Characteristics on Parameter Estimation for Maximum Likelihood Confirmatory Factor Analysis. *Multivariate Behavioral Research*, vol. 20, 255-271.
- Geuens, M., (1998). *De Invloed van Persoonlijkheid op de Effectiviteit van Emotionele Reclame*, (The Influence of Personality on the Effectiveness of Emotional Advertising). Unpublished Dissertation, Antwerp, RUCA.
- Goldstein, K., (1939). The Organism. New York: American Book Co.
- Goldstein, K., (1942). Some Experimental Observations Concerning the Influence of Color on the Function of the Organism. *Occupational Therapy*, vol. 21, 147-151.
- Goodfellow, R. A. II., & Smith, P. C., (1973). Effects of Environmental Color on Two Psychomotor Tasks. *Perceptual and Motor Skills*, vol. 37, 296-298.
- Gorn, G. J., (1982). The Effects of Music in Advertising on Choice Behavior: A Classical Conditioning Approach. *Journal of Marketing*, vol. 46, Winter, 94-101.
- Gorn, G. J., Chattopadhyay, A., Yi, T., & Dahl D. W., (1997). Effects of Color as an Executional Cue in Advertising: They're in the Shade. *Management Science*, vol. 43, n° 10, October, 1387-1400.
- Gotz, K., & Gotz, K., (1974). Color Preferences of Art Students: Surface Colors: I. *Perceptual and Motor Skills*, vol. 39, n° 3, 1103-1109.
- Gotz, K., & Gotz, K., (1975). Color Preferences of Art Students: Surface Colors: II. *Perceptual and Motor Skills*, vol. 41, 271-278.
- Gouras, P., (1968). Identification of Cone Mechanisms in Monkey Ganglion Cells. *Journal of Physiology*, vol. 199, 533-547.
- Granger, G. W., (1955). An Experimental Study of Colour Preferences. *Journal of General Psychology*, vol. 52, 3-20.
- Green, P., (1978). Analyzing Mutivariate Data. Hinsdale, IL.: The Dryden Press.
- Green, P. E., Tull, D. S., & Albaum, G. (1988). Research for Marketing Decisions. New Jersey: Englewood Cliffs.
- Green, R. R., & Goldfried, M. R., (1965). On the Bipolarity of Sematic Space. *Psychological Monographs*, vol. 79, n° 599.
- Green, W. K., Hanson, S. M., Mohammes, S. J., Phillips, C. L., Richards, P. E., Smith, M. E., & White, A., (1982). Effects of Viewing Selected Colors on Performance of Gross and Fine Motor Tasks. *Perceptual* and Motor Skills, vol. 54, 778.
- Greenland, S. J., & McGoldrick P. J., (1994). Atmospherics, Attitudes and Behaviour: Modelling the Impact of Designed Space. *The International Review of Retail, Distribution and Consumer Research*, 1-16.

Gretag Macbeth, Munsell Utilities, (2002). (http://standards.gretagmacbeth.com/cmc/index.htm)

- Grewal, D., & Baker, J., (1994). Do Retail Store Environmental Factors Affect Consumers' Price Acceptability? An Empirical Examination. *International Journal of Research in Marketing*, vol. 11, 107-115.
- Griffitt, W., (1970). Environmental Effects on Interpersonal Affective Behavior: Ambient Effective Temperature and Attraction. *Journal of Personality and Social Psychology*, vol.15, 250-244.
- Griffitt, W., & Veitch, R., (1971). Hot and Crowded: Influences of Population Density and Temperature on Interpersonal Affective Behavior. *Journal of Personality and Social Psychology*, vol. 17, 92-98.
- Gröppel, A., (1991). Erlebnisstrategien im EinzelhandelBAnalyse der Zielgruppen, der Ladengestaltung und der Warenpräsentation zur Vermittlung von Einkaufserlebnissen. Heidelberg.
- Gröppel-Klein, A., (1997). The Influence of the Dominance Perceived at the Point-of-Sale on the Price-Assessment. In: Englis, B. G. & A. Olofsson, (Eds.), *European Advances in Consumer Research*, vol. 3, 304-311. Provo, UT: Association for Consumer Research.
- Gröppel-Klein, A., (1998). The Influence of the Dominance Perceived at the Point-of-Sale on the Price-Assessment. *European Advances in Consumer Research*, vol. 3, 304-311.
- Gröppel-Klein, A., & Baun, D., (2001). The Role of Consumers' Arousal for Retail Stores Results from an Experimental Pilot Study Using Electrodermal Activity as Indicator. In: M. C. Gilly & J. Meyers-Levy (Eds.), Advances in Consumer Research, vol. 28.
- Grossbart, S. L., Hampton, R., Rammohan, B., & Lapidus, R. S., (1990). Environmental Dispositions and Customer Response to Store Atmospherics. *Journal of Business Research*, vol..31, November, 225-241.
- Grossbart, S. L., & Mittelstaedt, R. A., Curtis, W. W., & Rogers, R. D., (1975). Environmental Sensitivity and Shopping Behavior. *Journal of Business Research*, vol. 3, October, 281-294.
- Grossman, M., & Wood, (1993). Sex Differences in Intensity of Emotional Expression: A Social Role Interpretation. *Journal of Personality and Social Psychology*, vol. 65, 1010-1022.
- Guilford, J. P., (1934). The Affective Value of Color as a Function of Hue, Tint and Chroma. *Journal of Experimental Psychology*, vol. 17, 342-370.
- Guilford, J. P., & Smith, P. C., (1959). A System of Color Preferences. *American Journal of Psychology*, vol. 62, 487-502.
- Gulas, C. S., & Schewe, C. D., (1994). Atmospheric Segmentation: Managing Store Image with Background Music. In: Ravi Acrol and A. Mitchell, (Eds.), *Enhancing Knowledge Development in Marketing*, 325-330. Chicago, IL.: American Marketing Association.

Η

- Hair, Anderson, Tatham, & Black, (1998). *Multivariate Data Analysis*, Chapter 11: Structural Equation Modelling.
- Hall, E. L., (1979). Computer Image Processing and Recognition. New York: Academic Press. Available at : (http://kiptron.psyc.virginia.edu/steve boker/ColorVision2/node16.html)
- Hamid, P. N., & Newport, A. G., (1989). Effect of Color on Physical Strength and Mood in Children. *Perceptual and Motor Skills*, vol. 69, 179-185.
- Hammes, J. A., & Wiggens, S. L., (1962). Perceptual Motor Steadiness, Manifest Anxiety and Color Illumination. *Perceptual and Motor Skills*, vol. 14, 59-61.

Hand Media, (2001). http://www.handprint.com/HP/WCL/color1.html).

Handprint Media, (2001). http://www.handprint.com/HP/WCL/color2.html).

Handprint Media, (2001). http://www.handprint.com/HP/WCL/color6.html).

- Hard, A, (1975). NCS: A Descriptive Color Order and Scaling System With Application for Environmental Design. *Man-Environment Systems*, vol. 5, 161-167.
- Hard, A. & Sivik, L., (1979). Outlines of a Theory of Colors in Combination. *Man-Environment Systems*, vol. 9, 217-228.
- Hattie, J. A., (1985). Methodology Review: Assessing Unidimensionality of Tests and Items. Applied Psychological Measurement, vol. 9, 139-164.
- Havlena, W. J., & Holbrook, M. B., (1986). The Varieties of Consumption Experience: Comparing Two Typologies of Emotion in Consumer Behavior. *Journal of Consumer Research*, vol. 13, December, 97-112.
- Heath, R. G., (1964a). Pleasure Response of Human Subjects to Direct Stimulation of the Brain: Physiologic and Psychodynamic Considerations. In: R. G. Heath, (Ed.): *The Role of Pleasure in Behavior*, 219-243. New York: Harper & Row, Publishers.
- Heath, R. G., (1964b). R. G. Heath, (Ed.): *The Role of Pleasure in Behavior*. New York: Harper & Row, Publishers.
- Hebb, D. O., (1955). Drives and the C.N.S. (Conceptual Nervous System). *Psychological Review*, vol. 62, 243-254.
- Hebb, D. O., & Thompson, W. R., (1954). The Social Significance of Animal Studies. In: G. Lindzey (Ed.), *Handbook of Social Psychology*. Cambridge, Massachusetts: Addison-Wesley.
- Helson, H., & Lansford, T., (1970). The Role of Spectral Energy of Source and Background Pleasantness of Object Colors. *Applied Optics*, vol. 9, 1513-1562.
- Hemphill, M., (1996). Color Psychological Aspects. *Journal of Genetic Psychology*, vol. 157, Issue 3, September 275, 6p, 1 chart. 1 graph.
- Henthorne, M., (2000). Colour Blindness What is it? Available at: http://home-pages.enterprise.net/markhenthorne/colour vision/redgreen.html
- Hering, E., (1878). Zur Lehre vom Lichtsinne. Wien: Gerold & Sohn. (Cited from Mueller & Rudolph, 1969).
- Hering, E., (1920). Grundzüge der Lehre vom Lichtsinn. Berlin.
- Herman Miller, (2001). Experience of Color. 8 February 2001. Available at: <u>http://www.HermanMiller.com/us/indes.bbk/5519.htm</u>.
- Hershberger, R. G., & Cass, R. C., (1974). Predicting User Responses to Building. In: D. H. Carson (Ed.), Man-Environment Interactions: Evaluations and Applications, Pt. 2, 117-134. Stroudsbury, PA: Dowden, Hutchinson & Ross.
- Hildebrandt, L., (1987). Consumer Retail Satisfaction in Rural Areas: A Reanalysis of Survey Data. *Journal of Economic Psychology*, vol. 8, 19-42.
- Hogg, J., (1969). A Principal Component Analysis of Semantic Differential Judgements of Single Colors and Color Pairs. *Journal of General Psychology*, vol. 80, 129-140.

Holahan, C., (1986). Environmental Psychology. Annual Review of Psychology, 381-407.

- Holbrook, M. B., & Batra, R., (1987). Assessing the Role of Emotions as Mediators of Consumer Responses to Advertising. *Journal of Consumer Research*, vol. 14, December, 404-420.
- Holbrook, M. B., Chestnut, R. W., Oliva, T. A., & Greenleaf, E. A., (1984). The Roles of Emotions, Performance, and Personality in the Enjoyment of Games. *Journal of Consumer Research*, vol. 11, September, 728-730.
- Holbrook, M. B., & Gardner, M., (1993). An Approach to Investigating the Emotional Determinants of Consumption Durations: Why do People Consume What They Consume for as Long as They Consume It? *Journal of Consumer Psychology*, vol. 2, 123-142.
- Holland, S, (2001). 312. Testing for Color Blindness. 29 November 2001. http://www3.telus.net/softwaves/hiddentalents/brain/vision/312-color-tests.html.
- Honnolulu Star Bulletin, (1999). The Status of Pink Football Locker Rooms Today, October, 24th.
- Hopson, J., Cogan, R., & Batson, C., (1971). Color Preference as a Function of Background and Illumination *Perceptual and Motor Skills*, vol. 33, n° 2, part 3, 1083-1088.
- Howett, G. L., (1985). *Linear Opponent-Color Model Optimized for Brightness Prediction*. Gaithersburg, Md.: U.S. Department of Commerce, NBS.
- Hoyer, W. D., & MacInnis, D. J., (2001). Consumer Behavior, 2nd ed. Boston: Houghton Miffin Company.
- Huffman, C., & Houston, M. J., (1993). Goal-Oriented Experiences and the Development of Knowledge. Journal of Consumer Research, vol. 20, September, 190-207.
- Hui, M. K., & Bateson, J. E. G., (1991). Perceived Control and the Effects of Crowding and Consumer Choice on the Service Experience. *Journal of Consumer Research*, vol. 18, September, 174-184.
- Hui, M. K., Dubé, L., & Chebat, J-C., (1997). The Impact of Music on Consumers' Reactions to Waiting for Services. *Journal of Retailing*, vol. 73, Spring, 87-104.
- Humphrey, N. K., (1971). Colour and Brightness Preferences in Monkeys. Nature, vol. 229, 615-617.
- Humphrey, N. K., & Keeble, G. R., (1977). Do Monkeys' Subjective Clocks Run Faster in Red Light than in Blue? *Perception*, vol. 6, 7-14.
- Humphrey, N. K. (1976). The Colour Currency of Nature. In: T. Porter and B. Mikellides, (Eds.), *Color For Architecture*, 95-98. New York: Van Nostrand Reinholt, Co.
- Hurlbert, A., (1991). Deciphering the Colour Code. Nature, vol. 349, 191-193.
- Hurlbert, A., (1997). Colour Vision. Current Biolology, vol. 7, 400-402.
- Hurvich, L. M. & Jameson, D., (1957). An Opponent-Process Theory of Color Vision. *Psychological. Review*, vol. 64, 384-404.
- Hurvich, L. M. (1981). Color Vision. Massachusetts: Sinauer Associates, Inc.

Ι

Ilsen, A. M., Shalker, T. E., Clark, M., & Karp. L., (1978). Affect, Accessibility of Material in Memory, and Behavior: A Cognitive Loop? *Journal of Personality and Social Psychology*, vol. 36, n° 1, 1-12.

- Ishihara, S., (1960). Test for Color-Blindness: The Series of Plates Designed as a Test for Color-Blindness. Kyoto Tokyo: Kanchara Shuppan.
- Ishihara, S., (1969). Tests for Colour-Blindness. Tokyo: Kanchara Shuppan.

Ishihara, S., (1980). Ishihara Tests for Color Blindness. Tokyo: Kanehara & Co. Ltd.

- Ishihara, S., (2001). *Ishihara Tests for Color Blindness*. Available at: http://www.toledo-bend.com/colorblind/Ishihara.html.
- Ittelson, W. H., (1973). Environment Perception and Contemporary Perceptual Theory. In: W. H. Ittelson, (Ed.), *Environment and Cognition*. New York: Seminar.
- Izard, C. E., (1977). Human Emotion. New York: Plenum Press.
- Izard, C. E., (1994). Die Emotionen des Menschen. Eine Einführung in die Grundlagen der Emotionspsychologie., 3ed. Weinheim: Beltz.

J

- Jacobs, K. W., & Hustmyer, F. E., (1974). Effects of Four Psychological Primary Colors on GSR, Heart Rate and Respiration Rate. *Perceptual and Motor Skills*, vol. 38, 763-766.
- Jacobs, K. W., & Nordan, F. M., (1979). Classification of Placebo Drugs. Effect of Color. Perceptual and Motor Skills, vol. 49, 367-372.
- Jacobs, K. W., & Suess, J. F., (1975). Effects of Four Psychological Colors on Anxiety States. Perceptual and Motor Skills, vol. 41, 201-210.
- James, W. T., & Domingos, W. R., (1953). The Effect of Color Shock on Motor Performance and Tremor. Journal of General Psychology, vol. 48, 187-193.
- Janis, I. L., Kaye, D., & Kirschner, P., (1965). Facilitating Effects of 'Eating-while-Reading' on Responsiveness to Persuasive Communications. *Journal of Personality and Social Psychology*, vol. 1, 181-186.
- Janssens, J., & Mikellides, B., (1998). Color Research in Architectural Education: A Cross-Cultural Explorative Study. *Color Research and Application*, vol. 23, n° 5, October, 328-333.
- Jastrow, J., (1897). The Popular Aesthetics of Color. Popular Science, vol. 50, 361-368.
- Johnson, I. T., & Shulman, G. A., (1988). More Alike Than Meets The Eye. Perceived Gender Differences in Subjective Experience and its Display. *Sex Roles*, vol. 19, 67-79.
- Jokl, M. V., (1982). The Effect of the Environment on Human Performance. *Applied Ergonomics*, vol. 13, 229-280.
- Jöreskog, K. G., & Sörbom, D., (1988). *LISREL 7: A Guide to the Program and Applications*. Chicago, IL: SPSS Inc.
- Jöreskog, K. G., & Sörbom, D., (1989). *LISREL 7: A Guide to the Program and Applications*. Chicago: Jöreskog & Sörbom/SPSS Inc.
- Jöreskog, K. G., & Sörbom, D., (1993). *LISREL 8: Structural Equation Modelling with the SIMPLIS Command Language*. Chicago: Scientific Software International, Inc.
- Jöreskog, K. G., & Sörbom, D., (2001). LISREL 8.5.

- Jud, D. B., (1971). Choosing Pleasant Color Combinations. Lighting Design and Application, August, 31-41.
- Judd, C. M., Smith, E. R., & Kidder, L. H., (1991). *Research Methods in Social Relations*, 6th ed. U.S.: Harcourt Brace Jovanovich College Publishers.
- Judd, D. D. & Nickerson, D., (1975). Relation Between Munsell and Swedish Natural Color System Scales. *Journal of the Optical Society of America*, vol. 65, n° 1, 85-90.

Κ

- Kaiser, P. K., (1984). Physiological Response to Color: A Critical Review. Color *Research and Application*, vol. 9 n° 1, 29-36.
- Kaplan, S., (1987). Aesthetics, Affect, and Cognition. Environment and Behavior, vol. 19, January, 3-32.
- Kassarjian, H. H., (1971). Content Analysis in Consumer Research, *Journal of Consumer Research*, vol. 4, June, 8-18.
- Kassarjian, H. H., & Sheffet, M. J., (1991). Personality and Consumer Behavior: An Update. In: H. H. Kassarjian & T. S. Robertson (Eds.), *Perspectives in Consumer Behavior*. 281-303. Englewood Cliffs, NJ: Prentice-Hall.
- Kaufman-Scarborough, C., (2000). Seeing Through The Eyes Of The Color-Deficient Shopper: Consumer Issues for Public Policy. *Journal of Consumer Policy*, vol. 23, 461-492.
- Kelly, G. A., (1955). The Psychology of Personal Constructs. New York: Norton.
- Kelly, K. L., & Judd, D. B., (1976). Color: Universal Language and Dictionary of Names. National Bureau of Standards, Spec. Publ., 440.
- Kerlinger, F. N., (1973). Foundations of Behavioral Research. New York: Holt, Rinehart and Winston, Inc.
- Khouw, N., (1999a). Gender and the Meaning of Color in Interior Environments, available at: <u>http://www.colormatters.com/khouw.html</u>, accessed (2/03/2001).
- Khouw, N., (1999b). The Meaning of Color for Gender. Available at: http://www.colormattrs.com/khouw.html, accessed (2/03/2001).
- Klemz, B. R., & Boshoff, C. (2001). Environmental and Emotional Influences on Willingness-to-buy in Small and Large Retailers. *European Journal of Marketing*, vol. 35, n° 1/2, 70-91.
- Konorski, J., (1948). Conditioned Reflexes and Neuron Organization. Cambridge, UK: Cambridge University Press.
- Kotler, P., (1973). Atmosphere as a Marketing Tool. Journal of Retailing, vol. 49, Winter, 48-64.
- Kreitler, H., & Kreitler, S., (1972). Psychology of Arts, Durham, N.C.: Duke University Press.
- Kuehni, R. G., (2002). The Early Development of the Munsell System. *Color Research and Application*, vol. 27, n° 1, February.
- Kuhl, J., (1986). Motivation and Information Processing: A New Look at Decision Making, Dynamic Change, and Action Control. In: *Handbook of Motivation and Cognition: Foundations of Social Behavior*, 404, 434. New York: The Guilford Press.

- Kuller, R., (1981). Non-Visual Effects of Light and Colour: Annotated Bibliography. Stockholm: The Swedish Council for Building Research.
- Kuller, R., (1985). Seeing Red. Working Environments.
- Kumar, V., & Karande, K., (2000). The Effect of Retail Environment on Retailer Performance. Journal of Business Research, vol. 49 (2), 167-181.
- Kwallek, N., & Lewis, C. M., (1990). Effects of Environmental Color on Males and Females: A Red or White or Green Office. *Applied Ergonomics*, vol. 21, 275-278.

L

Lane, R., (1991). Does Orange Mean Cheap? Forbes, vol. 23, December, 144-147.

- Lang, P. J., Bradley, M. M., & Cuthbert, B. N., (1990). *Emotion, Attention, and the Startle* Psychological Review, vol. 97, 377-395.
- Lans, M. S., (1995). Supreme Court to Rule on Colors as Trademarks. Marketing News, January, 28th.
- Larsen, R.J., & Diener E., ((1992). Promises and Problems with the Circumplex Model of Emotion. In: M. S.. Clark, (Ed.), *Review of Personality and Social Psychology: Emotional and Social Behavior*. vol. 14, 25-59. Newbury Park, CA: Sage.
- Lawler, C. O., & Lawler, E. E., (1965). Color-Mood Associations in Young Children. The Journal of Genetic Psychology, vol. 107, 29-32.
- Lazarus, R. S., (1991). Emotion and Adaptation. New York: Oxford University Press.
- Lean, E., (1984). Color Me Training. Training and Development Journal, vol. 38, 42-51.
- LeDoux, J., (1996). The Emotional Brain: The Mysterious Underpinnings of Emotional Life. New York: Simon & Schuster.
- Lee, S., & Barnes, J. H. Jr., (1990). Using Color Preferences in Magazine Advertising. *Journal of Advertising Research*, vol. 29, n° 6, 21-30.
- Lehman, D., (2001). Structural Equations Modelling: VI.E. Mediators and Moderators. In: D. Iacobucci (Ed.), Journal of Consumer Psychology, Special Issue: Methodological and Statistical Concerns of the Experimental Behavioral Researcher, vol. 10, (1-2), 90-91.
- Levine, M. W., (2000). Levine and Shefner's Fundamentals of Sensation and Perception., 3rd ed., Oxford University Press.
- Lewin, K., (1935). A Dynamic Theory of Personality. New York: McGraw-Hill.
- Lewin, K., (1936). Principles of Topological Psychology. New York: McGraw-Hill.
- Lindsay, P. H., & Norman, D. A., (1973). *Human Information Processing: An Introduction to Psychology*. New York: Academic Press.
- Lindsay, P. H. & Norman, D. A., (1977). *Human Information Processing: An Introduction to Psychology*. New York: Academic Press.
- Luria, S. M., & Kobus, D. A., (1985). Red Light, White Light. Proceedings of U.S. Naval Institute, August, 123-126.

- Lutz, R. J., & Kakkar, P., (1975a). Situational Influence on Interpersonal Persuasion. Advances in Consumer Research, vol. 3, 370-378.
- Lutz, R. J., & Kakkar, P., (1975b). The Psychological Situation as a Determinant of Consumer Behaviour. Advances in Consumer Research, vol. 3, 439-453.

Μ

- Machleit, K. A., & Eroglu, S. A., (2000). Describing and Measuring Emotional Response to Shopping Experience. *Journal of Business Research*, vol. 49, 101-111.
- Madden, T. J., Hewett K., & Roth M. S., (2000). Managing Images in Different Cultures: A Cross-National Study of Color Meanings and Preferences. *Journal of International Marketing*, vol. 4, 90-107.
- Mahnke, F., (1981). Color in Medical Facilities. Interior Design, vol. 52, 256-263.
- Mano, H., (1997). Affect and Persuasion: The Influence of Pleasantness and Arousal on Attitude Formation and Message Elaboration. *Psychology and Marketing*, vol. 14, n° 4, 315-335.
- Mano, H., (1999). The Influence of Pre-Existing Negative Affect on Store Purchase Intentions. *Journal of Retailing*, vol. 75, n° 2, 149-172.
- Mano, H., & Oliver, R. L., (1993). Assessing the Dimensionality and Structure of the Consumption Experience: Evaluation, Feeling and Satisfaction. *Journal of Consumer Research*, vol. 20, December, 451-466.
- Marans, R. W., & Spreckelmeyer, K. F., (1982). Measuring Overall Architectural Quality. *Environment and Behavior*, vol. 14, November, 652-670.
- Markin, R. J., Lilis, C. M., & Narayana, C. L., (1976). Social-Psychological Significance of Store Space, *Journal* of *Retailing*, vol. 52, Spring, 43-54.
- Marks, W. B., Dobelle, W. H., & MacNichol, E. F., (1964). Visual Pigments of Single Primate Cones. *Science*, vol. 143, 1181-1183.
- Marsh, H. W., & Hovecar, D., (1985). Application of Confirmatory Factor Analysis to the Study of Self-Concept: First and Higher Order Factor Models and their Invariance across Groups. *Psychological Bulletin*, vol. 97, Issue 3, 562-582.
- Marston, W. M., (1927). Primary Colors and Primary Emotions. Psyche, vol. 30, 4-33.
- Martineau, P., (1958). The Personality of the Retail Store, *Harvard Business Review*, vol. 36, January-February, 47-55.
- Maslow, A. H., & Mintz, N. L., (1956). Effects of Esthetic Surroundings: I. Initial Effects if Three Esthetic Conditions Upon Perceiving 'Energy' and 'Well-Being' in Faces. *Journal of Psychology*, vol. 41, 247-254.
- Matilla A. S., & Wirtz, J., (2001). Congruency of Scent and Music as a Driver of In-Store Evaluations and Behavior. *Journal of Retailing*, vol. 77, 273-289.
- McCloy, C. H., (1942). Blocks Tests of Multiple Response. Psychometrika, vol. 7, 165-169.
- McCracken, Grant, (1988). Culture and Consumption. Bloomington IN: Indiana University Press.
- McGoderick, P. J., & Pieros, C. P., (1998). Atmospherics, Pleasure and Arousal, the Influence of Response Moderators. *Journal of Marketing Management*, vol. 14, 173-197.

- McInnis, J. H., & Shearer, J. K., (1964). Relationship between Color Choices and Selected Preferences for the Individual. *Journal op Home Economics*, vol. 56, 181-187.
- McKechnie, G. E., (1974). *ERI Manual: Environmental Response Inventory*. Pablo Alto, CA: Consulting Psychologists Press.
- McKechnie, G. E., (1977). Simulation Techniques in Environmental Psychology. In: D. Stokols (Ed.), *Perspectives on Environment and Behavior: Theory, Research and Applications*, 169-189. New York: Plenum.
- Medical Post, (1997). New Tests bring to Light Severity of Color Blindness (Pencil and Paper Test, and Genetic Test). Author non Specified, October, 28th.
- Meerum Terwocht, M. & Hoeksma, J. B., (1995). *Colors and Emotions: Preferences and Combinations*. Provincetown: The Journal of General Psychology, January.
- Mehrabian, A., (1972). Nonverbal Communication. In: J. K. Cole (Ed.), *Nebraska Symposium on Motivation*, 1971, vol. 19, 107-161. Lincoln, NE: University of Nebraska Press.
- Mehrabian, A., (1978). Measures of Individual Differences in Temperament. Educational and Psychological Measurement, vol. 38, 1105-1117.
- Mehrabian, A., (1979). Effect of Emotional State on Alcohol Consumption. *Psychological Reports*, vol. 44, 272-282.
- Mehabrian, A., (1995). Framework for a Comprehensive Description and Measurement of Emotional States. *Genetic, Social and General Psychology Monographs*, vol. 121, 133-144.
- Mehrabian, A., (1996). Pleasure-Arousal-Dominance: A General Framework for Describing and Measuring Individual Differences in Temperament. *Current Psychology*, vol. 14, Winter, Issue 4, 261-293.
- Mehrabian, A., (1997). Comparison of the PAD and PANAS as Models for Describing Emotions and for Differentiating Anxiety from Depression. *Journal of Psychopathology and Behavioral Assessment*, vol. 19, 331-357.
- Mehrabian, A., (1998). Manual for a Comprehensive System of Measures of Emotional States: The Pad Model. (Available from Albert Mehrabian, 1130 Alta Mesa Road, Monterey, CA, USA 93940). Los Angeles: University of California.
- Mehrabian, A., & Blum, J. S., (1997). Physical Appearance, Attractiveness, and the Mediating Role of Emotions. *Current Psychology: Developmental, Learning, Personality, Social*, vol. 16, 20-42.
- Mehrabian, A., & de Wetter, R., (1987). Experimental Test of an Emotion-based Approach to Fitting Brand Names to Products. *Journal of Applied Psychology*, vol. 72, 125-130.
- Mehrabian, A., & Ksionzky, S., (1974). A Theory of Affiliation. Lexington, MA: D.C. Heath.
- Mehrabian, A., & Riccioni, M., (1986). Measures of Eating-Related Characteristics for the General Population: Relationships with Temperament. *Journal of Personality Assessment*, vol. 50, 610-629.
- Mehrabian, A., & Russell, J. A., (1974a). An Approach to Environmental Psychology. Cambridge MA: MIT Press.
- Mehrabian, A., & Russell, J. A., (1974b). A Verbal Measure of Information Rate for Studies in Environmental Psychology. *Environment and Behavior*, vol. 6, 233-252.
- Mehrabian, A., & Russell, J. A., (1975). Environmental Effects on Affiliation Among Strangers. *Humanitas*, vol. 11, 219-230.

- Mehrabian, A., Wihardja, C., & Ljunggren, E., (1997). Emotional Correlates of Preferences for Situation-Activity Combinations in Everyday Life. *Genetic, Social, and General Psychology Monographs*, vol. 123, 461-477.
- Melara, R. D., Marks, L. E., & Potts, B. C., (1993). Primacy of Dimensions in Color Perception. *Journal of Experimental Psychology: Human Perception and Performance*. vol. 19. n° 5, 1082-1104.
- Melgosa, M., Rivas, M. J., Hita, E., & Viénot, F., (2000). Are We Able to Distinguish Color Attributes? *Color Research and Application*, vol. 25, n° 5, 356-367.
- Menon, K., & Dubé, L., (2000). Ensuring Greater Satisfaction by Engineering Salesperson Response to Customer Emotions. *Journal of Retailing*, vol. 76, Fall, Issue 3, 285-308.
- Merbs, S. L., & Nathans, J., (1992). Absorption Spectra of the Hybrid Pigments Responsible for Anomalous Color Vision. *Science*, vol. 258, 464-466.
- Meyers-Levy, J., & Maheswaran, D., (1991). Expoloring Differences in Males' and Females' Processing Strategies. *Journal of Consumer Research*, vol. 18, June, 63-70.
- Meyers-Levy, J., & Sternthal, B. (1991). Gender Differences in the Use of Message Cues and Judgments. *Journal of Marketing Research*, vol. 28, February, 84-96.
- Meyers-Levy, J., & Peracchio, L. A., (1995). Understanding the Effects of Color: How the Correspondence between Available and Required Resources Affects Attitudes. *Journal of Consumer Research*, vol. 22, 121-138.
- Middlestadt, S. E., (1990). The Effect of Background and Ambient Color on Product Attitudes and Beliefs. *Advances in Consumer Research.*, vol. 17, 244-249.
- Mikellides, B., (1979). Conflicting Experiences of Color Space. In: J. G. Simon, (Ed.), *Conflicting Experiences of Space*. Proceedings of the 4th I.A.P.C., vol. II, 679-703. (Louvain-la-Neuve).
- Miller, N. E., (1944). Experimental Studies of Conflict. In: J. McV. Hunt (Ed.) *Personality and the Behavior Disorders*, vol. 1, 431-465. New York: Ronald Press,
- Miller, N. E., & Dollard, J., (1941). Social Learning and Imitation. New Haven, CT: Yale University Press.
- Milliman, R. E., (1982). Using Background Music to Affect the Behavior of Supermarket Shoppers. *Journal of Marketing*, vol. 46, Summer, 86-91.
- Milliman, R. E., (1986). The Influence of Background Music on the Behavior of Restaurant Patrons. *Journal of Consumer Research*, vol. 13, September, 286-289.
- Miyao, M., Ishihara, S., Sakakibara, H., Kondo, T., Yamada, S., Furuta, M., Yamanaka, K., Yasuma, T., & Toda, Y., (1991). Effects of Color CRT Display on Pupil Size in Color-blind Subjects. *Journal of Human Ergonomics*, vol. 20, 241-247.
- Mooradian, T. A., & Olver J. M., (1996). Shopping Motives and the Five Factor Model: An Integration and Preliminary Study. *Psychological Reports*, vol. 78, 579-592.
- Morton, J., (1999). *Color Voodoo # 6 Color Voodoo for the Office*. Color Voodoo Publications. Available at: <u>http://www.colorvoodoo.com/cvoodoo6.html</u>.
- Moschis, G. P., (1987). Consumer Socialization: A Life-Cycle Perspective. Lexington, MA: Lexington Books.
- Moye, L. N., & Kincade, D. H., (2002). Influence of Usage Situations and Consumer Shopping Orientations on the Importance of the Retail Store Environment. *International Review of Retail, Distribution and Consumer Research*, vol. 12, n° 1, January, 59-79.

Munsell, A. H., (1899-1918). *Color Diary*. Handwritten Notes typed and Figures and Sketches traced from Munsell Color Company Orginals from Early 1920. Copy of typed Version made Available by Luke J. T.

Munsell, A. H., (1905). A Color Notation. Boston: GEO.H.Ellis.

Munsell, A. H., (1915). The Atlas of the Munsell Color System. Boston.

Munsell, A. H., (1929). The Munsell Book of Colours.

Munsell, A. H., (1966). A Munsell Color Product. New York: Kollmorgen Corporation.

Munsell.com, 2002. The Munsell System of Colour Notation. http://www.munsell.com/munsell1.htm.

Muthén, B., & Kaplan, D., (1985). A Comparison of Some Methodologies for the Factor Analysis of Non-Normal Likert Variables. *British Journal of Mathematical and Statistical Psychology*, vol. 38, 171-189.

Ν

- Naka, K. I., & Rushton, W. A. H., (1966). S-Potentials from Colour Units in the Retina of Fish (Cyprinidae). Journal of Physiology, vol.185, 536-555.
- Nakahara, H., (2000). WhatColor. This is a PC Software Program that is Available to Assist Color Blind Persons in Identifying Colors on their Computer Screens. It is found at: http://hp.vector.co.jp/authors/VA011243/wcolor e.htm.
- Nakashima, T., (1909). Contributions to the Study of the Affective Processes. *American Journal of Psychology*, vol. 20, 157-193.
- Nakshian, J. S., (1964). The Effect of Red and Green Surroundings on Behavior. *Journal of General Psychology*, vol. 70, 143-161.
- Nash Graphic Design Group, (1994). Color Blindness: Color Vision Deficiency (Color Vision Guide). Menlo Park CA: Nash Graphic Design Group.
- Nassau, K., (1983). *The Physics and Chemistry of Color: The Fifteen Causes of Color*. New York: John Wiley and Sons.
- Nathans, J., Piantanida, T. P., Eddy, R. L., Shows, T. B., & Hogness, D. S., (1986a). Molecular Genetics of Inherited Variation in Human Color Vision. *Science*, vol. 232, 203-210.
- Nathans, J., Thomas, D., & Hogness, D. S., (1986b). Molecular Genetics of Human Color Vision: The Genes Encoding Blue, Green, and Red Pigments. *Science*, vol. 232, 193-202.

National Aeronautics and Space Administration, (1976). Johnson Spacecraft Centre.

- Neitz, J., Neitz, M., & Kainz, P. M., (1996). Visual Pigment Gene Structure and the Severity of Color Vision Defects. *Science*, vol. 274, 801-804.
- Neitz, M., (1998). Society and Colorblindness: a View in Need of Correction. *Research to Prevent Blindness Scientific Writers Seminar*, 64-66.
- Neri, D. F., Luria, S. M., & Kobus, D. A., (1986). The Detection of Various Color Combination Under Different Chromatic Ambient Illuminations. Aviation, Space, and Environmental Medicine, June, 555-560.

- Netemeyer, R., (2001). Structural Equations Modelling: VI.C. Improving Model Fit by Correlating Errors. In: D. Iacobucci (Ed.), *Journal of Consumer Psychology*, Special Issue: Methodological and Statistical Concerns of the Experimental Behavioral Researcher, vol. 10, (1-2), 87-89.
- Norman, R. D., & Scott, W. A., (1952). Color and Affect: A Review and Sematic Evaluation. *Journal of General Psychology*, vol. 46, 185-223.
- Norusis, M.J., (1993). SPSS for Windows: Base System User's Guide, Release 6.0. Chicago IL: SPSS Inc.
- Norusis, M. J., (1994). SPSS Advanced Statistics 6.1. Chicago IL: SPSS Inc.
- Nourse, E. W., & Welch, R. B., (1971). Emotional Attributes of Colour: A Comparison of Violet and Green. *Perceptual and Motor Skills*, vol. 32, 403-406.
- Nowlis, V., (1962). Research with the Mood Adjective Check List. In: S. S. Tomkins & C. E. Izard (Eds.), *Affect, Cognition and Personality.* New York: Springer.

0

- Oatley, K., (1992). Best Laid Schemes: The Psychology of Emotions. Cambridge, UK: Cambridge University Press.
- Obermiller, C., & Bitner, M. J., (1984). Store Atmosphere: A Peripheral Cue for Product Evaluation. In: D. C. Stewart, (Ed.), American Psychological Association Annual Conference Proceedings, Consumer Psychology Division, 52-53. Washington, DC: American Psychological Association.
- Olds, J., (1959). Pleasure Centres in the Brain. Scientific American, vol. 195, nº 4, 105-116.
- Oliver, R., (1993). Cognitive, Affective, and Attribute Bases of the Satisfaction Response. *Journal of Consumer Research*, vol. 20, December, 418-430.
- Olney, T., Holbrook, M. B., Batra, R., (1991). Consumer Responses to Advertising: The Effects of Ad Content, Emotions, and Attitude Toward the Ad on Viewing Time. *Journal of Consumer Research*, vol. 17, March, 440-453.
- Osgood, C. E., (1953). Method and Theory in Experimental Psychology. New York: Oxford University Press.
- Osgood, C. E., Suci, G. J., & Tannenbaum, P. H., (1957). *The Measurement of Meaning*. Urbana, IL: University of Illinois Press.
- Ott, J. N., (1976). Influence of Fluorescent Lights on Hyperactivity and Learning Disabilities. Journal of Learning Disabilities, vol. 9, n° 1, 417-422.

P

- Park, C. W., & Smith, D. C., (1989). Product Level Choice: A Top-Down of Bottom-Up Process. Journal of Consumer Research, vol. 16, December, 289-295.
- Percy, L., Ketchum, MacLeod, & Grove, (1974). Determining the Influence of Color on a Product Cognitive Structure: a Multidimensional Scaling Application. *Advances in Consumer Research*, vol. 1, 218-227.

- Peterson, R. A., (1977). Consumer Perceptions as a Function of Product Color, Price and Nutrition Labelling. Advances in Consumer Research, vol. 4, 61-63.
- Peterson, R. A., & Sauber, M., (1983). A Mood Scale for Survey Research. In: P. Murphy et al. (Eds.), *American Marketing Association Educator's Proceedings*, 409-414. Chicago: American Marketing Association.
- Piantanida, T., (1988). The Molecular Genetics of Color Vision and Color Blindness. *Trends in Genetics*, vol. 4, 319-323.
- Plack, J. J., & Shick, J., (1974). The Effects of Color on Human Behavior. *Journal of the Association for the Study of Perception*, vol. 9, n° 1, 4-16.
- Plater, G., (1967). Adolescent Preferences for Fabric, Color, and Design on Usual Task. Unpublished Master's Thesis. Terre Haute, Indiana: Indiana State College.
- Plutchik, R., (1980). Emotion: A Psycho-evolutionary Synthesis. New York: Harper & Row.
- POPAI Europe, (1998). *The POPAI Europe Consumer Buying Habits Study*. Point-of-Purchase Advertising Institute. Co-ordination by Retail Marketing In-Store Services Limited, Watford, Herts : POPAI Europe.
- Porter, T., & Mikellides, B., (1976). Color for Architecture. New York: Van Nostrand Reinhold Co.
- Pressey, S. L., (1921). The Influence of Color upon Mental and Motor Efficiency. American Journal of Psychology, vol. 32, 326-356.
- Pressly, P. K., & Heesacker, M., (2001). The Physical Environment and Counseling: A Review of Theory and Research. *Journal of Counseling and Development*, vol. 79, Spring, 148-160.
- Pribram, K. H., (1962). Interrelations of Psychology and the Neurological Disciplines. In: S. Koch (Ed.) *Psychology : A study of Science*, vol New York: McGraw-Hill.
- Profusek, P. J., & Rainey, D. W., (1987). Effects of Baker-Miller Pink and Red on State Anxiety, Grip Strength, and Motor Precision. *Perceptual and Motor Skills*, vol. 65, 941-942.
- Proshansky, H. M., Ittelson W. H., & Rivlin L. G., (1974). Freedom of Choice and Behavior in a Physical Setting. In: J. M. Proshansky et al. (Ed.), *Environmental Psychology*, 170-181. New York : Holt, Rinehart & Winston.

R

- Radeloff, D. J., (1990). Role of Color in Perception of Attractiveness. *Perceptual and Motor Skills*, vol. 71, 151-160.
- Raju, P. S., (1980). Optimum Stimulation Level: Its Relationship to Personality, Demographics and Exploratory Behavior. *Journal of Consumer Research*, vol. 7, 272-282.
- Raju, P. S., (1981). Theories of Exploratory Behavior: Review and Consumer *Research* Implications. *Research in Marketing*, vol. 4, 223-249.
- Ratneshwar, S., & Shocker, A. D., (1991). Substitution-in-Use and the Cognitive Structure of Product Categories. *Journal of Marketing Research*, vol. 28, August, 281-295.
- Razran, G. H. S., (1938). Conditioning Away Social Bias by the Luncheon Technique. *Psychological Bulletin*, vol. 35, 693.
- Razran, G. H. S., (1940). Conditional Response Changes in Rating and Appraising Sociopolitical Slogans. Psychological Bulletin, vol 37, 481.

- Read, M. A., Sugawara A. I., & Brandt, J., (1999). Impact of Space and Color in the Physical Environment on Preschool Children's Cooperative Behavior. *Environment and Behavior*, vol. 31 n° 3, May, 413-428.
- Reddy, T. V., & Bennet, C. A., (1985). Cultural Differences in Color Preferences. Proceedings of the Human Factors Society – 29th Annual Meeting, 590-593.
- Retail-Update, (2002). Interview with Brengman, M., De Invloed van Kleur in de Winkelomgeving op de Consument (*The Influence of Colour in the Store Environment on the Consumer*), by J.H. Verbank, Retail-Update, n° 39, 4, July, 5th.
- Rigden. C., (2000). The Eye of the Beholder Designing for Colour-blind Users. Retrieved from the World Wide Web at: <u>http://www.labs.bt.com:people/rigdence/coulours</u>.
- Robinson, M. D., Johnson, J. T., & Shields, S. A., (2001). The Gender Heuristic and the Database: Factors Affecting the Perception of Gender-Related Differences in the Experience and Display of Emotions. In: W. G. Parrott (Ed.), *Emotions in Social Psychology, Key Readings in Social Psychology*, Reading 8. Philadelphia, PA, US: Taylor & Francis Group Psychology Press.
- Rogers, R. D., (1979). Commentary on "The Neglected Variety Drive". *Journal of Consumer Research*, vol. 6, June, 88-91.
- Rolls, B. J., (1985). Experimental Analyses of the Effects of Variety in a Meal on Human Feeding. *American Journal of Clinical Nutrition*, vol. 42, 932-939.
- Rood, N. O., (1879). Modern Chromatics. New York: Appleton, 218.
- Rosenstein; L. D., (1985). Effect of Color of the Environment on Task Performance and Mood of Males and Females with High or Low Scores on the Scholastic Aptitude Test. *Perceptual and Motor Skills*, vol. 60, 550.
- Rosenthal, O., & Philips, R. H. (1997). *Coping with Color-Blindness*. Garde, City Park, New York: Avery Publishing Group.
- Rotter, J. B., (1954). Social Learning and Clinical Psychology. Englewood Cliffs, NJ: Prenice-Hall.
- Russell, J. A., (1980). A Circumplex Model of Affect. *Journal of Personality and Social Psychology*, vol. 39, December, 1161-1178.
- Russell, J.A., & Bullock, M., (1985). Multidimensional Scaling of Emotional Facial Expressions: Similarity from Pre-schoolers to Adults. *Journal of Personality and Social Psychology*, vol. 48, 1290-1298.
- Russell, J. A., & Carroll, J. M., (1999). On the Bipolarity of Positive and Negative Affect. *Psychological Bulletin*, vol. 125, 3-30.
- Russell, J. A., & Feldman Barrett, L., (1999). Core Affect, Prototypical Emotional Episodes and Other Things Called Emotion: Dissecting the Elephant. *Journal of Personality and Social Psychology*, vol. 76, 805-819.
- Russell, J. A., & Lemay, G., (2000). Emotion Concepts. In: M. Lewis & J. M. Haviland-Jones (Eds.) *Handbook* of *Emotions*, 2nd ed. chapter 13, 491-503. New York/London: The Guilford Press.
- Russell, J. A., & Mehrabian, A., (1976). Some Behavioral Effects of the Physical Environment. In: S. Wagner, S. Cohen, & B. Kaplan (Eds.), *Experiencing the Environment*, 5-18. New York: Plenum.
- Russell, J. A., & Mehrabian, A., (1976). Environmental Variables in Consumer Research. Journal of Consumer Research, vol. 3, 62-63.
- Russell, J. A., & Mehrabian, A., (1977). Evidence for a Three-Factor Theory of Emotions. *Journal of Research in Personality*, vol. 11, 273-274.

- Russell, J. A., & Mehrabian, A., (1978). Approach-avoidance and Affiliation as Functions of the Emotion-Eliciting Quality of an Environment. *Environment and Behaviour*, vol. 10, 355-387.
- Russell, J. A., & Pratt, G., (1980). A Description of the Affective Quality Attributed to Environments. *Journal of Personality and Social Psychology*, vol. 38, n° 2, 311-322.
- Russell, J. A., & Ridgeway, O., (1983). Dimensions Underlying Children's Emotion Concepts. Development Psychology, vol. 19, 795-804.
- Russell, J. A., & Snodgrass J., (1987). Emotion and the Environment. In: Daniel Stokols and Irwin Altman, (Eds.), *Handbook of Environmental Psychology*, vol. 1, 245-281. New York: John Wiley & Sons, Inc.
- Russell, J. A., & Ward, L. M., (1982). Environmental Psychology. Annual Review of Psychology, 651-688.

S

- Sallis, R. F., & Buckalew, I. W., (1984). Relation of Capsule Color and Perceived Potency. *Perceptual Motor Skills*, vol. 58, 897-898.
- Schaie, K. W., (1961). Scaling the Associations Between Colors and Mood Tones. American Journal of Psychology, vol. 74, 266-273.
- Schaie, K. W., (1962). A Q-Sort Study of Color-Mood Association. *Journal of Projective Techniques*, vol. 25, 341-346.
- Schauss, A. G., (1979). Tranquilizing Effects of Color Reduces Aggressive Behavior and Potential Violence. Orthomolecular Psychiatry, vol. 8, n° 4.
- Schauss, A. G., (1981). Application of Behavioral Photobiology to Human Aggression: Baker-Miller Pink. International Journal of Biosocial Research, vol. 2, 1-9.
- Schauss, A. G., (1985). The Physiological Effect of Color on the Suppression of Human Aggression: Research on Baker-Miller Pink. *International Journal of Biosocial Research*, vol. 7, n° 2, 55-64. http://www.predesign.org/baker-miller.html.
- Schmitt, B. H., & Simonson, A., (1997). Marketing Aesthetics: The Strategic Management of Brands, Identity and Image. New York: The Free Press.
- Schutz, W. C., (1966). The Interpersonal Underworld. Palo Alto, CA: Science & Behavior.
- Schneirla, T. C., (1959). An Evolutionary and Developmental Theory of Biphasic Processes Underlying Approach and Withdrawal. In: M. R., Jones, (Ed.) *Nebraska Symposium on Motivation*, vol. 7, 1-42. Lincoln: University of Nebraska Press.
- Scitovsky, T., (1992). *The Joyless Economy. The Psychology of Human Satisfaction*, Revised Edition. New York: Oxford University Press.
- Sewell, J. H., (1983). Color Counts too! Academic Therapy. January, 329-336.
- Sharma, S., (1996). Applied Multivariate Techniques. New York: John Wiley & Sons.
- Sharpe, D. T., (1974). The Psychology of Color and Design. Chicago, IL: Nelson-Hall.
- Sharpe, D. T., (1981). The Psychology of Color and Design. New Jersey: Littlefield, Adams & Co.

- Shaver, P., Schwartz, J., Kirson, D., & O'Connor, C., (1987). Emotion Knowledge: Further Exploration of a Prototype Approach. *Journal of Personality and Social Psychology*, vol. 52, 1061-1086.
- Shaver, P., Schwartz, J., Kirson, D., & O'Connor, C., (2001). Emotion Knowledge: Further Exploration of a Prototype Approach. In: W. Gerrod Parrott, (Ed.) *Emotions in Psychology: Essential Readings*, 1st Reading, 26-55. US: Psychology Press-Taylor & Francis.
- Shepard, R. N., & Cooper, L. A., (1992). Representation of Colors in the Blind, Color-blind, and Normallysighted. *Psychological Science*, vol. 3, 97-104.
- Sherman, E., Mathur, A., & Smith, R. B., (1997). Store Environment and Consumer Purchase Behavior: Mediating Role of Consumer Emotions. *Psychology & Marketing*, vol. 14, n° 4, 361-378.
- Sherman, E., & Smith, R.B., (1986). Mood States of Shoppers and Store Image: Promising Interactions and Possible Behavioral Effects. In: P. Anderson (Ed.), *Advances on Consumer Research*, vol. 14. Provo, UT: Association for Consumer Research.
- Shields, S. A., (1987). Women, Men, and the Dilemma of Emotions. In: P. Shaver & C. Hendric (Eds.), Sex and Gender: Review of Personality and Social Psychology, vol. 7, 229-250. Newbury Park, CA: Sage.
- Shield, S. A., & Koster, B. A., (1989). Emotional Stereotyping of Parents in Child Rearing Manuals, 1915-1980. Social Psychology Quarterly, vol. 52, 44-55.
- Silver, N. C., & Ferrante, R., (1995). Sex Differences in Color Preferences among Elderly Sample. *Perceptual and Motor Skills*, vol. 80, 920-922.
- Silver, N. C., & McCulley, W. I., (1988). Sex and Race Differences in Color and Number Preferences. *Perceptual and Motor Skills*, vol. 66, 295-299.
- Sivik, L., (1974a). Color Meaning and Perceptual Color Dimensions: A Study of Color Samples. *Goteborg Psychological Reports*, vol. 4, n° 1.
- Sivik, L., (1974c). Measuring the Meaning of Colors: Problems of Sematic Biopolarity. *Goteborg Psychological Reports*, vol. 4, n° 13.
- Sivik, L., (1974d). Measuring the Meaning of Colors: Reliability and Stability. *Goteborg Psychological Reports*, vol. 4, n° 12.
- Sivik, L., (1975). Studies of Color Meaning. Man-Environment Systems, vol. 5, 155-160.
- Sjoberg, L., & Svensson, E., (1976). The Polarity and Dimensionality of Mood. Goteborg Psychological Reports, vol. 6, n° 12.
- Smets, G., (1969). Time Expression of Red and Blue. Perceptual and Motor Skills, vol. 29, 511-514.
- Smets, G., (1982). A Tool for Measuring Relative Effects of Hue, Brightness and Saturation in Color Pleasantness. *Perceptual and Motor Skills*, 1159-1164.
- Smith, K. C. P., & Apter, M. J., (1975). A Theory of Psychological Reversal. Chippenham, England: Pleton.
- Smith, C., & Ellsworth, P., (1985). Patterns of Cognitive Appraisal in Emotions. *Journal of Personality and Social Psychology*, vol. 48, 813-838.
- Smith, V. C., & Pokorny, J., (1972). Spectral Sensitivity of Color Blind Observers and the Cone Photopigments. *Vision Research*, vol. 12, 2059-2071.
- Smith, V. C., Pokorny, J., & Starr, S. J. (1976). Variability of Color Mixture Data I. Interobserver Variability in the Unit Coordinates. *Vision Research*, vol. 16, 1087-1094

- Sokolov, E. N., & Boucsein, W., (2000). A Psychophysiological Model of Emotion Space. *Integrative Physiological and Behavioral Science*, vol. 35, n° 2, April-June, 81-119.
- Solomon, M. R., (1995). Consumer Behavior, 3rd. ed. New York: Simon & Schuster.
- Spence, J. T., Helmreich, R., & Stapp, J., (1975). Ratings of Self and Peers on Sex Role Attributes and their Relation to Self-Esteem and Conceptions of Masculinity and Feminity. *Journal of Personality and Social Psychology*, vol. 32, 29-39.
- Spielberger, C. D., Gorsuch, R. L., & Lushene, R. E., (1970). *Manual for the State-Trait Anxiety Inventory*. Palo Alto, CA: Consulting Psychologists Press.
- Srivastava, R. J., & Peel, T. S., (1968). *Human Movement as a Function of Color Stimulation*. Topeka, Kansas: Environmental Research Foundation.
- Staples, R., & Walton, W. E. (1933). A Study of Pleasurable Experience as a Factor in Color Preference. Journal of Genetic Psychology, vol. 43, 217-223.
- Stark, G. M., Saunders, D. M. & Wookey, P. E., (1982). Differential Effects of Red and Blue Coloured Lighting on Gambling Behavior. *Current Psychological Research*, vol. 2, 95-100.
- Steenkamp, J-B. E. M., & Baumgartner, H., (1992). The Role of Optimum Stimulation Level in Exploratory Consumer Behavior. *Journal of Consumer Research*, vol. 19, December, 434-448.
- Steenkamp, J-B. E. M. & Baumgartner, H., (1995). Development and Cross-Cultural Validation of a Short Form of CSI as a Measure of Optimum Stimulation Level. *International Journal of Research in Marketing*, vol. 12, Issue 2, 97-105
- Steenkamp, J-B. E. M., Baumgartner, H., & van der Wulp, E., (1994). The Relationships among Arousal Potential, Arousal and Stimulus Attractiveness, and the Moderating Role of Need for Stimulation. Unpublished Paper, presented in the 23rd EMAC Conference.
- Steenkamp, J-B. E. M., & van Trijp, H. C. M., (1991). The Use of LISREL in Validating Marketing Constructs. *International Journal of Research in Marketing*, vol. 8, 283-299.
- Stevens, S. S., (1946). On the Theory of Scales of Measurement, Science, vol. 105, June 7, 677-680.
- St.George, M. W., (1938). Color Preferences of College Students with Reference to Chromatic Pull, Learning, and Association. *American Journal of Psychology*, vol. 13, 268-271.
- Stokols, D., & Altman, I., (1987). Handbook of Environmental Psychology. New York: John Wiley & Sons, Inc.
- Sucov, E. W., (1973). A Review of Research on the Psychological Influence of Color on People. Pittsburgh, Pa.: Westinghouse Research Laboratories.
- Sweeney, J. C., & Wyber, F., (2002). The Role of Cognitions and Emotions in the Music-Approach-Avoidance Behavior Relationship. *Journal of Services Marketing*, vol. 16, n° 1, 51-67.
- Swinyard, W. R., (1993). The Effects of Mood, Involvement, and Quality of Store Experience on Shopping Intentions. *Journal of Consumer Research*, vol. 20, September, 271-280.
- Swirnoff, L., (1989). Dimension Color. Cambridge, MA,: Birkhauser Boston .

T

Tabachnik, B. G., & Fidell, L. S., (1989). Using Multivariate Statistics. New York: Harper and Row Publishers.

- Tacq, J., (1997). Multivariate Analysis Techniques in Social Science Research: From Problem to Analysis. (English ed.). London: Sage Publications.
- Taft, C., (1997). Color Meaning and Context: Comparisons of Sematic Ratings of Colors on Samples and Objects. *Color Research and Application*, vol. 22, n° 1, February, 41-50.
- Tauber, E. M., (1972). Why Do People Shop? Journal of Marketing, vol. 36, October, 46-49.
- Thayer, R. E., (1967). Measurement of Activation through Self-Report. Psychological Reports, vol. 20, 663-678.
- Thayer, R. E., (1070). Activation States as Assessed by Verbal Report and Four Psychophysiological Variables. *Psychophysiology*, vol. 7, 86-94.
- Thayer, R. E., (1986). Activation and Deactivation: Current Overview and Structural Analysis. *Psychological Reports*, vol. 58, 607-614.
- Thayer, R. E., (1989). The Biopsycholgy of Mood and Activation. New York: Oxford.
- Thompson, C. J., & Hirschman. E. C., (1995). Understanding the Socialized Body: A Poststructuralist Analysis of Consumers' Self-Conceptions, Body Images, and Self-Care Practices. *Journal of Consumer Research*, vol. 22, 139-153.
- Tom, G., Barnett, T., Lew, W., & Selmants, J., (1987). Cueing the Consumer. The Role of Salient Cues in Consumer Perception, *Journal of Consumer Marketing*, vol. 4, n° 2, 23-27.
- Trochim, W. M. K., (2002). *The Research Methods Knowledge Base*. Available from: <u>http://trochim.human.cornell.edu/kb/power.htm</u>.
- Tull, D. S., & Hawkins, D. I., (1993). Marketing Research: Measurement & Method. 6th ed. Upper Saddle River, NJ: Prentice Hall.
- Turley, L. W., & Milliman R. E., (2000). Atmospheric Effects on Shopping Behavior: A Review of the Experimental Evidence. *Journal of Business Research, vol.* 49, 193-211.
- Turley, L. W., & Bolton, D. L., (1999). Measuring the Affective Evaluations of Retail Service Environments. Journal of Professional Services Marketing, vol. 19, n° 1, 31-37.

V

- Valdez, P., (1993). *Emotion Responses to Color*. Unpublished Doctoral Dissertation. Los Angeles: University of California.
- Valdez, P. & Mehrabian, J., (1994). Effect of Color on Emotions. *Journal of Experimental Psychology: General*, vol. 123, Issue 4, 394-409.
- Vandewalle, H., Peres, G., & Monod, H., (1987). Standard Anaerobic Exercise Tests. Sports Medicine, vol. 4, 268-289.
- Van Kenhove, P., & Desrumaux, P., (1997). The Relationship between Emotional States and Approach or Avoidance Responses in a Retail Environment. *The International Review of Retail, Distribution and Consumer Research*, vol. 7, Issue 3, July, 351-368.
- Van Kenhove, P., De Wulf, K., & Van Waterschoot, W., (1999). The Impact of Task Definition on Store-Attribute Saliences and Store Choice. *Journal of Retailing*, vol. 75, n° 1, 125-137.

- Van Kenhove, P., & De Wulf, K., (2000). Income and Time Pressure: a Person-Situation Grocery Retail Typology. *International Review of Retail, Distribution and Consumer Research*, vol. 10, n° 2, 149-166.
- Venolia, C., (1988). Healing Environments. Berkeley, CA: Celestial Arts.
- Vischeck, (2000). *The World's first Online Color-blindness Simulator* (2000). Menlo Park, California: Tiny Eyes. Retrieved at: <u>http://www.vischeck.com/what.shtml</u>.
- Voke, J., (1998). Color Vision . . . Dangers at Work. The Safety & Health Pratitioner, vol. 16, nº 1, 28-30.
- Von Helmholtz, H. C. F., (1924). *Physiological Optics.*, vol. (trans. J. Southall). Rochester, N: Optical Society of America.

W

- Wahlers, R. G., & Etzell, M., (1985). A Consumer Response to Incongruity between Optimal Stimulation and Life Style Satisfaction. Advances in Consumer Research, vol. 12, 97-101.
- Wald, M., (1997). Vision Tests for Rail Engineers Vary, and May Deceive. New York Times, March 27th, 6.
- Walker, M., (1991). The Power of Color. New York: Avery Publishing Group, 50-52.
- Walsh, L. M., & Toma, R. B., (1990). Color Preference and Food Choice among Children. Journal of Psychology Interdisciplinary & Applied, vol. 124, November, Issue 6, 645, 9p, 4 charts.
- Walters, J., Apter, M. J., & Svebak, S., (1982). Color Preference, Arousal, and Theory of Psychological Reversals. *Motivation and Emotion*, vol. 6, n° 3, 193-215.
- Walton, W. E., Guilford, R. B., & Guilford, J. P., (1933). Color Preferences of 1279 University Students. *American Journal of Psychology*, vol. 45, 322-328.
- Ward, G., (1995). Colors and Employee Stress Reduction. Supervision, vol. 56, February, 3-5.
- Ward, J. C., Bitner, M. J. & Barnes, J., (1992). Measuring the Prototypicality and Meaning of Store Environments. *Journal of Retailing*, vol. 68, n° 2, 194-220.
- Ward, L. M., & Russell, J. A., (1981). Cognitive Set and the Perception of Place. *Environment & Behavior*, vol. 13, September, 610-632.
- Warner, S. J., (1949). The Color Preferences of Psychiatric Groups. Psychological. Monographs, vol. 63, n° 6.
- Washburn, M. F., (1911). A Note on Affective Value of Color. American Journal of Psychology, vol. 22, 114-115.
- Wasburn, M. F., & Grose, S. L., (1921). Voluntary Control of Likes and Dislikes: The Effects of an Attempt Voluntarily to Change the Affective Value of Colors. *American Journal of Psychology*, vol. 32, 284-289.
- Washburn, M. F., McLean, K. G., & Dodge, A., (1934). The Effect of Area on the Pleasantness and Unpleasantness of Colors. *American Journal of Psychology*, vol. 46, 638-640.
- Watson, D., & Tellegen, A., (1985). Toward a Consensual Structure of Mood. *Psychological Bulletin*, vol. 98, September, 219-235.
- Watson, D., & Tellegen, A., (1999). Issues in the Dimensional Structure of Affect: Effects of Descriptors, Measurement Error, and Response Formats. *Psychological Bulletin*, vol. 125, 601-610.

- Weller, L., & Livingston, R., (1988). Effects of Color of Questionnaire on Emotional Responses. Journal of General Psychology, vol. 115, 433-440.
- Westbrook, R. A., (1980). Intrapersonal Affective Influences on Consumer Satisfaction with Products. Journal of Consumer Research, vol. 7, June, 49-54.
- Westbrook, R. A., (1987). Product/Consumption-Based Affective Responses and Postpurchase Processes. *Journal of Marketing Research*, vol. 24, August, 258-270.
- Westbrook, R. A., & Black, W. C., (1985). A Motivation Based Shopper Typology. *Journal of Retailing*, vol. 61, 78-103.
- Wexner, L. B., (1954). The Degree to which Colors (Hues) are Associated with Mood Tones. *Journal of Applied Psychology*, vol. 38, 432-435.
- White, R. W., (1984). An Introduction to Ecological Psychology. Monterey, CA: Brooks/Cole.
- Whitfield, A., (1984). Individual Differences in Evaluation of Architectural Colour: Categorization Effects. *Perceptual and Motor Skills*, vol. 59, 183-186.
- Whitfield, T. W. A., & Wiltshire, T. J., (1990). Color Psychology : A Critical Review. Genetic, Social and General Psychology Monographs, vol. 116, 387-405.
- Wiegersma, Sjoerd, & Van der Elst, (1988). Blue Phenomenon: Spontaneity or Preference. *Perceptual and Motor Skills*, vol. 66 n° 2, 308-310.
- Wiesel, T.N., & Hubel, D.H., (1966). Spacial and Chromatic Interactions in the Lateral Geniculate Body of the Rhesus Monkey. *Journal of Neurophysiology*, vol. 29, 1115-1156.
- Wilson, G. D., (1966). Arousal Proprieties of Red vs. Green. Perceptual and Motor Skills, vol. 23, 947-949.
- Wilson, G. D. (1998). Retrieved from http://www.lava.net/~dewilson/web/color.html.
- Wilson, N. J., & Wilson, W. P., (1959). The Duration of Human Electroencephaloghic Arousal Responses Elicited by Photic Stimulation. *Electroencephalography and Clinical Neurophysiology*, vol. 11, 85-91.
- Wineman, J. D., (1979). Color in Environmental Design: Its Impact on Human Behavior. In Environmental Design: *Research Theory and Applications*, EDRA 10, 436-439.
- Wineman, J. D., (1982). Office Design and Evaluation: An Overview. *Environment & Behavior*, vol. 14, n° 3, May, 271-298.
- Wohlfarth., H., (1986). Color and Light Effects on Students' Achievement. *Behavior and Physiology*. Edmonton, Alberta: Funded by Alberta Education.
- Wolfmaier, T. G. (1999). Designing for the Color-challenged: A Challenge. Inter-net-working, 2(1), retrieved at the following web address: http://www.sandia.gov/itg/newsletter/mar99/accessibility_color_challenged.html.
- Wolfson, S., Case, G., (2000) . The Effects of Sound and Colour on Responses to a Computer Game. *Interacting with Computers*, vol. 13, 182-192.
- Woodson, W. E., (1981). Human Factors Design Handbook: Information and Guidelines for the Design of Systems, Facilities, Equipment, and Products of Human Use. New York: McGraw-Hill.
- Wright, B., & Rainwater, L., (1962). The Meanings of Color. Journal of General Psychology, vol. 67, 89-99.
- Wright, D., (1984). The Perception of Light and Colour. In: W. W. Dawson and J. M. Enoch (Ed.) *Foundations* of Sensory Science, 229-258, Berlin: Springer-Verlag.

Wundt, W., (1896, 1905). Grundriss der Psychologic. (Fundamentals of Psychology.) Leipzig: Wilhelm Engelmann.

Y

- Yalch, R., & Spangenberg, E., (1988). An Environmental Psychological Study of Foreground and Background Music as Retail Atmospheric Factors. In: A. W. Walle, (Ed.), AMA Educators' Conference Proceedings, vol. 54, 106-110. Chicago, IL: American Marketing Association.
- Yalch, R., & Spangenberg, E., (1990). Effects of Store Music on Shopping Behavior, Journal of Consumer Marketing, vol. 7, n° 2, 55-63.
- Yalch, R., & Spangenberg, E., (1993). Using Store Music for Retail Zoning: A Field Experiment. In: Leigh McAlister and Michael L. Rothschild, (Eds.), Advances in Consumer Research, vol. 20, 632-636. Provo, UT: Association for Consumer Research.
- Yik, M. S. M., Russell J. A., & Feldman Barrett, L., (1999). Structure of Self-Reported Current Affect: Integration and Beyond. *Journal of Personality and Social Psychology*, vol. 77, n° 3, 600-619.
- Yoo, C., Park, J., & MacInnis, D. J., (1998). Effects of Store Characteristics and In-Store Emotional Experiences on Store Attitude. *Journal of Business Research*, vol. 42, 253-263.
- Young, T., (1807). On the Theory of Light and Colors. Lectures in Natural Philosophy. London (cited form Mueller & Rudolph, 1969).

Ζ

- Ziems, D., & Christman, S., (1998). Effects of Mood on Color Perception as a Function of Dimensions of Valence and Arousal. *Perceptual and Motor Skills*, vol. 87, 531-535.
- Zuckerman, M., (1977). Development of a Situation-Specific Trait-State Test for the Predition and Measurement of Affective Responses. *Journal of Consulting and Clinical Psychology*, vol. 45, 513-523.
- Zurcher, J., (2001). Sportspeak. Available at: http://www.aceweekly.com/Backissues_ACEWeekly/01004/cb011004.html.
Appendix

5.00 GY				
Chroma	14	14	6	6
Value	5	8	5	8
R	110	188	125	202
G	146	223	142	217
В	0	0	72	139

Selected colours according to the Munsell notation and their corresponding RGB values

5.00PB				
Chroma	14	14	6	6
Value	3	6	3	6
R	0	54	59	138
G	87	163	88	161
В	177	255	128	199

5.00G				
Chroma	14	14	6	6
Value	5	8	5	8
R	0	0	80	152
G	158	237	147	224
В	102	167	118	187

5.00B				
Chroma	14	14	6	6
Value	4	7	4	7
R	0	0	22	123
G	128	205	119	194
В	187	255	142	215

5.00YR				
Chroma	14	14	6	6
Value	5	8	5	8
R	201	255	176	254
G	113	186	124	199
В	0	63	85	157

5.00BG				
Chroma	14	14	6	6
Value	5	8	5	8
R	0	0	60	140
G	158	237	147	224
В	154	221	143	213

5.00R				
Chroma	14	14	6	6
Value	4	7	4	7
R	197	255	155	231
G	56	142	94	170
В	67	135	92	165

5.00Y				
Chroma	14	14	6	6
Value	6	9	6	9
R	190	255	180	255
G	157	234	159	235
В	0	0	85	154